Original Page 1

Effective: September 1, 2019

This pricing and service guide includes the service descriptions, rates, charges, terms and conditions of service for the provision of switched interstate, international and intrastate telecommunications services by BellSouth Long Distance, Inc. d/b/a AT&T Long Distance Service between locations within the United States and between the United States and those countries listed herein.

For Business and Residential Customers in Alabama, Florida, Georgia, Kentucky, Louisiana, Mississippi, North Carolina, South Carolina and Tennessee whose services are offered on a detariffed basis and are governed by the AT&T Business Services Agreement or the AT&T Residential Service Agreement, the service descriptions, terms and conditions and pricing included in this pricing and service guide also apply to intrastate services.

Service is provided by any means of wire, terrestrial communications systems, satellite, microwave, and other transmission systems, or any combination thereof.

©2019 AT&T Intellectual Property. All rights reserved. AT&T, the AT&T logo and all other AT&T marks contained herein are trademarks of AT&T Intellectual Property and/or AT&T affiliated companies.

Original Page 1

Effective: September 1, 2019

# **EXPLANATION OF SYMBOLS**

Changes to this pricing and service guide shall be identified on the revised page(s) using symbols. The following are the only symbols used for the purposes indicated below:

- (C) Changed term, or change in text
- (D) Deleted text, discontinued rate, or discontinued term
- (I) Increased rate
- (N) New text
- (R) Rate reduction
- /n/ Moved text (where n is the next available footnote number)

Original Page 1

Effective: September 1, 2019

### SECTION 1 – TERMS AND ABBREVIATIONS

Access Line – A facility arrangement which connects the Customer's or Authorized User's location to the Company's network switching center.

Additional Period – The rate element used to bill chargeable time when a call continues beyond the Initial Period. The Additional Period starts when the Initial Period ends. Additional Period rates apply to any fraction of the time period for chargeable time beyond the Initial Period. Additional Periods vary by rate schedule and are specified in each individual rate table contained in later sections of this pricing and service guide.

Affiliated ILEC – An incumbent local exchange company (ILEC) with which the Company has any of the following relationships: (1) owns or controls it; (2) is owned or controlled by it; or (3) is under common ownership with it.

Aggregation – The combining of Customer's total usage across multiple BTNs into a group for the purpose of determining a common usage rate for call(s) associated with individual BTNs.

ANI – A calling telephone number identification which is forwarded to a interexchange carrier by a LEC as a call is placed.

Authorization Code – A numerical code, one or more of which are available to Customers to enable them to access the Company's network, and which are used by the Company both to prevent unauthorized access to its facilities and to identify Customers for billing purposes.

Authorized User – A natural person or legal entity which is authorized by the Customer to use the Company's Service under the terms and regulations of this pricing and service guide.

BellSouth – BellSouth Corporation and its affiliates.

Business Customer – For the purposes of this pricing and service guide, a Business Customer is a Customer of the Company whose primary use of the Company's Service is for business purposes. A Business Customer is also a Customer who accesses the Company's Service using a presubscribed Access Line that has been assigned a business class of service by the local service provider.

BTN – Billed Telephone Number. May consist of one or more WTNs.

Carrier – BellSouth Long Distance, Inc. d/b/a AT&T Long Distance Service

Casual Calling – Access to the Company's network and the subsequent use of Service by an End User Customer who does not have a current account with the Company.

CIC – An interexchange carrier identification code.

Commercial Credit Card – A billing arrangement whereby the originating caller may bill the charges for a call or service to an approved commercial credit card. The terms and conditions of the company issuing the credit card apply to payment arrangements.

Commission – The Federal Communications Commission.

Original Page 2

Effective: September 1, 2019

SECTION 1 – TERMS AND ABBREVIATIONS

Company – BellSouth Long Distance, Inc. d/b/a AT&T Long Distance Service

Confinement Institution – Used throughout this pricing and service guide to refer to prisons, jails, penal facilities or other institutions used for penalty purposes which contract with BellSouth Long Distance, Inc. for the provision of service for use by their Inmate population.

Customer – A person, firm, partnership, corporation or other entity which arranges for the Company to provide, discontinue or rearrange telecommunications services on behalf of itself or others; uses the Carrier's telecommunications services; and is responsible for payment of charges, all under the provisions and terms of this pricing and service guide. Customer also includes any natural person or legal entity 1) which resells the services of the Company to End Users; or 2) places Casual Calls using the Company's service.

Dedicated Access - See Special Access

Domestic – Relating and limited to the United States and the United States territories of Puerto Rico, U.S. Virgin Islands, Guam, and Commonwealth of Northern Mariana Islands and American Samoa.

End User – Any person, firm, corporation, partnership or other entity which uses the services of the Company under the provisions and regulations of this pricing and service guide. The End User does not have an account, service agreement or other relationship with the Company. The End User is responsible for payment unless the charges for the services utilized are accepted and paid by another Customer.

Equal Access – A form of dialed access provided by local exchange companies whereby telephone calls dialed by the Customer are automatically routed to the Company's network. Customers may also route calls to the Company's network by dialing an access code provided by the Company.

Initial Period – The initial period is the length of a call for minimum billing purposes. The initial periods vary by rate schedule and are specified in each individual rate table contained in other sections of this pricing and service guide.

Inmates – The confined population of Confinement Institutions.

LATA – A geographic area existing on February 8, 1996, as previously established by the U.S. District Court for the District of Columbia in Civil Action No. 82-0192, or established by a Bell operating company after February 8, 1996, and approved by the FCC.

LEC – Local Exchange Carrier.

IXC - Interexchange Carrier.

Original Page 3

Effective: September 1, 2019

### SECTION 1 – TERMS AND ABBREVIATIONS

Mainland United States – The forty-eight (48) states within the continental United States of America and the District of Columbia.

MOU - Minutes of Use

MMC - Minimum Monthly Commitment

MRC - Monthly Recurring Charge

Pay Telephone – Telephone instruments provided by the Company, Customer, Confinement Institution or other third party for use by the transient general public or Inmates. Pay Telephones permit the user to place calls to other parties and bill such calls on a non sent-paid or sent paid-basis. To facilitate sent-paid calling, Pay Telephones can be equipped with a credit card reader, coin box, or similar device that allows charges to be collected for each call at the instrument.

PIC – Primary interexchange carrier.

PIC Authorization – A Customer's or End User's selection of a PIC that meets the requirements of federal and state law.

Primary Carrier – The IXC designated by Customer or End User as its first routing choice and primary overflow carrier.

Primary Carrier Service – Outbound long distance service provided to a customer when the Company is selected as the Customer's Primary Carrier.

Premises – A building or buildings on contiguous property.

Private Line – A dedicated path between two locations.

Rate Periods – A collective reference to the Day Rate Period, Evening Rate Period, Night/Weekend Rate Period, Peak Rate Period, Off-Peak Rate Period, Standard Rate Period, Discount Rate Period, and Economy Rate Period.

Residential Customer – For the purposes of this pricing and service guide, a Residential Customer is a Customer of the Company whose primary use of the Company's Service is for personal use in a house, apartment or other residential dwelling unit. A Residential Customer is also a Customer who accesses the Company's Service using a presubscribed Access Line that has been assigned a residential class of service by the local service provider.

Service – Any or all services by the Company provided pursuant to this pricing and service guide.

Special Access – Where originating or terminating access between the Customer and the Company is provided on dedicated circuits. A method of reaching the Company's Services whereby the Customer is connected directly to the Company's access point without utilizing the services of the local switched network.

Station-to-Station - A service whereby the person originating the call uses the assistance of a live or mechanized operator to place a call to a particular destination number.

Original Page 4

Effective: September 1, 2019

### SECTION 1 – TERMS AND ABBREVIATIONS

Switched Access – Where access between the Customer and the Company is provided on local exchange company Feature Group circuits and the connection to the Customer is a LEC-provided business or residential access line. The cost of switched Feature Group access is billed to the Company.

TFS - AT&T Long Distance Toll Free<sup>SM</sup> Service

Telecommunications Device for the Deaf (TDD) – machine that uses the transmission of coded signals instead of verbal communications to enable hearing impaired users to communicate with each other and with non-hearing impaired individuals.

United States – For purposes of this pricing and service guide the term "United States" includes the Mainland United States, Alaska, Hawaii, Puerto Rico, U.S. Virgin Islands, Guam, American Samoa and the Commonwealth of the Northern Mariana Islands (CNMI).

WTN - Working Telephone Number

Original Page 1

Effective: September 1, 2019

### **SECTION 2 – REGULATIONS**

The services provided hereunder are subject to these regulations and any written agreement. These regulations apply to all interexchange interstate and international long distance services provided by the Company. These regulations also apply to intrastate services in the following states: Alabama, Florida, Georgia, Kentucky, Louisiana, Mississippi, North Carolina, South Carolina and Tennessee, excepted as noted below.

For Residential Customers in Alabama, Florida, Georgia, Kentucky, Louisiana, Mississippi, North Carolina, South Carolina and Tennessee (whose services are governed by the AT&T Residential Service Agreement available at www.att.com/servicepublications), only the following sections apply:

SECTION 2.1 SECTION 2.2.8 SECTION 2.7.2 SECTION 2.7.4 SECTION 2.7.8 (B) SECTION 2.7.9 SECTION 2.7.11 and 2.7.12 SECTION 2.8.3 SECTION 2.14.5 SECTION 2.16.1

### 2.1 Undertaking of the Company

Service is offered to Customers of the Company to provide outbound direct dialed, inbound toll-free number and operator assisted long distance calls between locations within the United States and between the United States and those countries listed in Section 5.0. The Company provides switched long distance network services for voice grade and low speed dial-up data transmission services.

The Company does not undertake to transmit messages but furnishes the use of its facilities to its Customers for communications. All Services are provided subject to the terms and conditions set forth in this pricing and service guide. In the event of a conflict between a contract entered into by the Company and this pricing and service guide, the terms of this pricing and service guide shall prevail.

The Company provides for the installation, operation, and maintenance of the telecommunications services provided herein in accordance with the terms and conditions set forth under this pricing and service guide. The Company may act as the Customer's agent for ordering access connection facilities provided by other carriers or entities, when authorized by the Customer, to allow connection of a Customer's location to the Company services. The Customer shall be responsible for all charges due for such service arrangement.

The Company's services are provided on a monthly basis unless otherwise provided and are available twenty-four (24) hours per day, seven (7) days per week.

### **SECTION 2 – REGULATIONS**

### 2.2 Limitations on Service

2.2.1 Service is offered subject to the availability of the necessary facilities, including established billing arrangements, and subject to the provisions of this pricing and service guide.

Original Page 2

- 2.2.2 The Company reserves the right to discontinue furnishing Service, or to limit the use of Service, when necessitated by conditions beyond its control, when the Customer or its Authorized User or End User is using Service in violation of the law or in violation of the provisions of this pricing and service guide, or for non-payment by the Customer.
- 2.2.3 Service provided under this pricing and service guide is directly controlled by the Company, and the Customer may not transfer or assign the use of Service, except with the prior written consent of the Company. Such transfer or assignment shall only apply where there is no interruption in the use or location of Service, and all regulations and conditions contained in this pricing and service guide, as well as all conditions for Service, shall apply to all such permitted assignees or transferees.
- 2.2.4 The Customer may, where applicable, request the Company to assign one or more sub-accounts for billing purposes and to direct sub-account invoices to affiliates of the Customer or other designated entities for payment purposes. Such requests shall not affect the liability of the Customer, who shall remain solely liable to the Company for payment of all invoices for Service requested and obtained by the Customer, whether invoiced by the Company to the Customer, its affiliates, or other designated entities.
- 2.2.5 Customers may use the services provided under this pricing and service guide for any lawful purpose consistent with the transmission and switching parameters of the telecommunications network and may resell its use (or the use of any part thereof) to a third party in the normal course of the Customer's business, except as specifically prohibited or limited herein. In any instance in which any of the services provided hereunder are being used in an unlawful manner, the Company may, immediately and upon written notice to the Customer, and without liability, restrict, suspend or discontinue providing such service.
- 2.2.6 Service may be limited or discontinued by the Company, without notice to the Customer, by blocking traffic to or from certain countries, cities, or NXX exchanges, or by blocking calls using certain Authorization Codes, when the Company deems it necessary to take such action to prevent unlawful use of its service. The Company will restore service as soon as it can be provided without undue risk, and will, upon request by the Customer affected, assign a new Authorization Code to replace the one that has been deactivated.
- 2.2.7 Service provided at Confinement Institutions for use by Inmates may be restricted or otherwise limited under the direction of authorized personnel or the Confinement Institution's administration at their own discretion.
- 2.2.8 Foreign Telecommunications Administrations may impose upon the portion of the end-to-end international service or facilities which they provide certain limitations, prohibitions or restrictions which may have the effect of limiting the ability of Subscribers or Customers to utilize the services furnished by the Company. It is the responsibility of the Customer to confirm any limitations, prohibitions or restrictions imposed by the Foreign Telecommunications Administration.

SECTION 2 – REGULATIONS

### 2.3 Limitations on Liabilities

2.3.1 Except as otherwise stated in this section, the liability of the Company for damages arising out of either: (1) the furnishing of its Services, including but not limited to mistakes, omissions, interruptions, delays, or errors, or other defects, representations, or use of these Services, or (2) the failure to furnish its Service, whether caused by acts or omission, shall be limited to the extension of allowances to the Customer for interruptions in Service as set forth in Section 2.14.

Original Page 3

- 2.3.2 Except for the extension of allowances to the Customer for interruptions in Service as set forth in this pricing and service guide, the Company shall not be liable to a Customer or any third party for any direct, indirect, special, incidental, reliance, consequential, exemplary or punitive damages, including, but not limited to, loss of revenue or profits, for any reason whatsoever, including, but not limited to, any act or omission, failure to perform, delay, interruption, failure to provide any Service or any failure in or breakdown of facilities associated with the Service.
- 2.3.3 The liability of the Company for errors in billing that result in overpayment by the Customer shall be limited to a credit equal to the dollar amount erroneously billed or, in the event that payment has been made and Service has been discontinued, to a refund of the amount erroneously billed.

### **SECTION 2 – REGULATIONS**

- 2.3 Limitations on Liabilities (continued)
  - 2.3.4 The Company shall not be liable for any claims for loss or damages involving:
    - (A) Any act or omission of: (a) the Customer, (b) any other entity furnishing service, equipment or facilities for use in conjunction with Services provided by the Company; or (c) common carriers or warehousemen;

Original Page 4

- (B) Any delay or failure of performance or equipment due to causes beyond the Company's control, including but not limited to, acts of God, fires, floods, earthquakes, hurricanes, or other catastrophes; acts of government authority, national emergencies, insurrections, acts of terrorism, riots, wars (declared or undeclared) or other civil commotions; strikes, lockouts, work stoppages or other labor difficulties; criminal actions taken against the Company; unavailability, failure or malfunction of equipment or facilities provided by the Customer or third parties; and any law, order, regulation or other action of any governing authority or agency thereof;
- (C) Any unlawful or unauthorized use of the Company's Services;
- (D) Libel, slander, invasion of privacy or infringement of trademarks, patents, trade secrets, or copyrights arising from or in connection with the transmission of communications by means of Company provided facilities or Services; or by means of the combination of Company provided facilities or Services with Customer provided services;
- (E) Breach in the privacy or security of communications transmitted over the Company's Service;
- (F) Changes in any of the facilities, operations or procedures of the Company that render any equipment, facilities or services provided by the Customer obsolete, or require modification or alteration of such equipment, facilities or services, or otherwise affect their use or performance, except where reasonable notice is required by the Company and is not provided to the Customer, in which event the Company's liability is limited as set forth in paragraph 2.3.1 of this Subsection;
- (G) Defacement of or damage to Customer premises resulting from the furnishing of Services or equipment on such premises or the installation or removal thereof;
- (H) Injury to property or injury or death to persons, including claims for payments made under Workers' Compensation law or under any plan for employee disability or death benefits, arising out of, or caused by, any act or omission of the Customer, or the construction, installation, maintenance, presence, use or removal of the Customer's facilities or equipment connected, or to be connected to the Company's facilities.
- 2.3.5 The Company shall be indemnified, defended and held harmless by the Customer from and against any and all claims, loss, demands, suits, expense, or other action or any liability whatsoever, including attorney fees, whether suffered, made, instituted, or asserted by the Customer or by any other party, for any personal injury to or death or any person or persons, and for any loss, damage or destruction of any property, including environmental contamination, whether owned by the Customer or by any other party, caused or claimed to have been caused directly or indirectly by the installation, operation, failure to operate, maintenance, presence, condition, location, use or removal of any Company or Customer equipment or facilities or Service provided by the Company.

Original Page 5

Effective: September 1, 2019

### **SECTION 2 – REGULATIONS**

### 2.3 Limitations on Liabilities (continued)

2.3.6 The Company does not guarantee nor make any warranty with respect to installations provided by it for use in an explosive atmosphere. The Company shall be indemnified, defended and held harmless by the Customer from and against any and all claims, loss, demands, suits, or other action, or any liability whatsoever, including attorney fees, whether suffered, made, instituted or asserted by the Customer or by any other party, for any personal injury to or death of any person or persons, and for any loss, damage or destruction of any property, including environmental contamination, whether owned by the Customer or by any other party, caused or claimed to have been caused directly or indirectly by the installation, operation, failure to operate, maintenance, presence, condition, location, use or removal of any equipment or facilities of the Service.

### 2.4 Cancellation or Discontinuance of Service by the Company

Without incurring any liability, the Company may, under the following conditions, cancel Service prior to commencement and/or discontinue Service that is being furnished for the following reasons; provided that, unless otherwise stated, the Customer shall be given proper notice of such cancellation or discontinuance of Service:

- 2.4.1 Without notice, for noncompliance with or violation of any applicable State, municipal or Federal law, ordinance or regulation or noncompliance with or violation of any Commission regulation, provided that notice may be required by order of such regulatory authorities.
- 2.4.2 For the Customer's or its Authorized User's or End User's refusal to provide reasonable access to the Company or its agents for the purpose of inspection and maintenance of equipment owned by the Company.
- 2.4.3 For noncompliance with any of the provisions of this pricing and service guide governing Service.
- 2.4.4 For nonpayment of any sum due the Company for more than thirty (30) days after delivery of an invoice to the custody of the U.S. Mail or other standard delivery service.
- 2.4.5 Without notice, in the event of the Customer's or its Authorized User's or End User's use of equipment in such a manner as to adversely affect the Company's equipment or Service to others.
- 2.4.6 Without notice, in the event of unauthorized or fraudulent use of Service. Whenever Service is discontinued for unauthorized use of Service, the Company may, before restoring Service, require the Customer to make, at its own expense, all changes in facilities or equipment necessary to eliminate unauthorized use and to pay to the Company an amount reasonably estimated by the Company as the loss in revenues to the Company resulting from such unauthorized use plus claims lodged against the Company by third parties.
- 2.4.7 Without notice, by reason of any order or decision of a court or other government authority having jurisdiction that prohibits the Company from furnishing Service to the Customer or its Authorized Users or End Users.
- 2.4.8 For periods of inactivity over sixty (60) days.

### **SECTION 2 – REGULATIONS**

# 2.5 Cancellation or Termination of Service by Customer

2.5.1 Service shall be canceled by the Company promptly upon receipt of a cancellation request from the Customer. Upon cancellation a final bill will be prepared, per the specifications set forth in this pricing and service guide. The Customer shall be liable for all recurring charges prior to proper notice if a change in presubscribed carrier is initiated by the Customer.

Original Page 6

Effective: September 1, 2019

2.5.2 If the Customer, either on behalf of itself or an Authorized User or End User, orders Service from the Company which requires special construction or facilities, and then cancels its order before Service begins, a charge shall be made to the Customer for the non-recoverable portions of the expenditures or liabilities incurred by the Company on behalf of the Customer or its Authorized User or End User by Company.

### 2.6 Restoration of Service

The use and restoration of Service shall in all cases be in accordance with the priority system specified in Part 64, Subpart D, of the Rules and Regulations of the Federal Communications Commission.

# 2.7 Payment and Billing

- 2.7.1 The Customer is responsible for payment of all charges for services and equipment furnished to the Customer for transmission of calls via the Company. In particular and without limitation to the foregoing, the Customer is responsible for any and all cost(s) incurred as the result of:
  - (A) any delegation of authority resulting in the use of the Customer's or its Authorized User's or End User's communications equipment and/or network services which result in the placement of calls via the Company;
  - (B) any and all use of the services provided by the Company, including calls which the Customer did not individually authorize;
  - (C) any calls placed by or through the Customer's or its Authorized User's or End User's equipment via any remote access feature(s);
  - (D) any and all calls placed to an "800" or "888" or other toll-free service number provided to the Customer by the Company;
  - (E) any calls placed through misused or fraudulently used account codes or authorization codes associated with optional features for 1+ or toll-free service.
- 2.7.2 Non-recurring charges for installations, service connections, moves, and rearrangements, where applicable, are payable upon demand to the Company or its authorized agent. Billing thereafter will include recurring charges and/or actual usage as defined in this pricing and service guide.
- 2.7.3 Service is provided and billed by the Company on a monthly basis. Usage sensitive charges are billed in arrears and fixed monthly recurring charges may be billed one month in advance or in arrears. The Customer shall pay monthly in advance or on demand all monthly recurring charges for Service and shall pay on demand all charges for usage at any agency duly authorized to receive such payments.

Original Page 7

Effective: September 1, 2019

### SECTION 2 – REGULATIONS

### 2.7 Payment and Billing (continued)

- 2.7.4 Bills are due and payable upon receipt. Interest at the lesser of a rate of one and one-half percent (1.5%) per month, or the maximum rate allowed by law, shall be charged on any amount remaining unpaid, including late payment charges, at the time the next bill is prepared. If an AT&T LEC bills you for services on the Company's behalf, the AT&T LEC late payment charges and policies will apply.
- 2.7.5 Should service be suspended for nonpayment of charges, it will be restored when appropriate payments are made. Any monthly service charges will continue to accrue until appropriate payments are made.
- 2.7.6 When service has been disconnected for nonpayment, the service agreement is considered to have been terminated. Reestablishment of service may be made only upon the execution of a new service agreement which is subject to the provisions of this pricing and service guide.
- 2.7.7 In its discretion, the Company may restore or reestablish service which has been suspended or disconnected for nonpayment of charges, prior to payment of all charges due. Such restoration or reestablishment shall not be construed as a waiver of the Company's right to receive full payment for all charges due or any rights to suspend or disconnect service for nonpayment of any such or other charges due and unpaid or for the violation of the provisions of this pricing and service guide; nor shall the failure to suspend or disconnect service for nonpayment of any past due account or accounts operate as a waiver or estoppel to suspend or disconnect service for nonpayment of such account or of any other past due account.
- 2.7.8 The Company may demand immediate payment under the following circumstances:
  - (A) Where Service is terminated or abandoned.
  - (B) Where actual usage is two times greater than the Customer's average usage as reflected on the monthly bills for the three months prior to the current bill or, in the case of a new Customer who has been receiving Service for less than four months, where the actual usage is twice the estimated monthly usage charge.
  - (C) Where the Company has reason to believe that a Business Customer or Reseller Customer is about to go out of business or that bankruptcy is imminent for that Customer.
- 2.7.9 A charge will apply whenever a check, draft, or electronic funds transfer presented for payment for service is not accepted by the institution on which it is written. See Section 7 of this pricing and service guide for the applicable charge.
- 2.7.10 The security of Authorization Codes used by the Customer or its Authorized Users or End Users are the responsibility of the Customer. All calls placed using such Authorization Codes or using facilities owned or controlled by the Customer, its Authorized Users or End Users shall be billed to the Customer and must be paid by the Customer.
- 2.7.11 If notice from the Customer of a dispute as to charges is not received in writing by the Company within thirty (30) days after delivery of an invoice to the custody of the U.S. Mail or other standard delivery service, the billing will be considered correct and binding.
- 2.7.12 The minimum period for which services are provided and for which rates and charges are applicable is one (1) month unless otherwise specified in this pricing and service guide or by mutually agreed upon contract. When a service is discontinued prior to the expiration of the minimum period, charges are applicable, whether the service is used or not.

Original Page 8

Effective: September 1, 2019

### **SECTION 2 – REGULATIONS**

### 2.7 Payment and Billing (continued)

2.7.13 With respect to any Services for which there is a flat rate monthly service charge for a specified number of minutes of long distance usage, the per minute charge for actual minutes of usage that are less than such specified number of minutes in any month shall be equal to the flat rate charge for the specified number of minutes of usage divided by the actual minutes of usage in such month. In the event of no usage in a billing period, the monthly service charge will nonetheless apply.

### 2.8 Deposits

2.8.1 The Company reserves the right to validate the credit worthiness of Customers and billed parties through available verification procedures and to establish a maximum predetermined credit amount. Where a requested billing method cannot be validated, or maximum credit amount established, the Company may refuse to provide service.

Any applicant whose credit has not been duly established and acceptable to the Company may be required to make a deposit to be held as a guarantee of payment of charges. In addition, an existing Customer may be required to make a deposit or increase a deposit presently held.

- 2.8.2 A deposit is not to exceed two month's Service and the estimated usage charges for two (2) month's service plus installation. A Deposit will be returned as follows:
  - (A) When an application for service has been canceled prior to the establishment of Service. The deposit will be applied to any charges applicable in accordance with this pricing and service guide and the excess portion of the deposit, if any, will be returned.
  - (B) Upon termination of Service, the deposit and accrued interest, as described herein, will be applied to any charges applicable in accordance with this pricing and service guide and the excess portion of the deposit, if any, will be returned promptly.
- 2.8.3 The interest rate used will be simple interest at the rate of six percent annually unless a different rate has been established by the appropriate legal authority in the state where the Service offering is located.
- 2.8.4 The fact that a deposit has been made in no way relieves the applicant or Customer from complying with the Company's regulations as to advance payments and the prompt payment of bills on presentation nor does it constitute a waiver or modification of the regular practices of the Company providing for the discontinuance of Service for nonpayment of any sums due for Service rendered.

# 2.9 Advance Payments

In order to protect the Company against revenue loss, an applicant for Service may be required to pay in advance of installation an amount not to exceed applicable service charges or other nonrecurring charges, plus estimated charges for two months of Service. Where special construction charges are applicable the payment thereof may be required in advance of start of construction. The amount of advance payment will be credited to the Customer's account on the first bill rendered for Service, and a new advance payment may be collected each month to be applied to each subsequent bill for Service.

Original Page 9

Effective: September 1, 2019

### SECTION 2 – REGULATIONS

### 2.10 Taxes

Any assessments, franchise fees, privilege, license, occupation, excise, or other similar taxes or fees, whether in lump sum or at a flat rate, or based on receipts, or based on poles, wire or other utility property units, imposed upon the Company by any governmental authority shall be added pro rata, insofar as practical, in amounts which in the aggregate for the Company's Customers of any political entity shall be equal to the amount of any such tax upon the Company. The Company shall, so long as any such tax or fee is in effect, add to the bills of the Customers in such political entity pro rata on the basis of the revenue derived by the Company from each such Customer, an amount sufficient to recover any such tax or fee.

Unless otherwise specified in this pricing and service guide, any such taxes and fees are in addition to rates as quoted in this pricing and service guide and are included as separate line items on the Customer's bill.

# 2.11 Terminal Equipment

Service may be used with or terminated in terminal equipment or communications systems, such as a PBX or key telephone system, provided by the Customer, its Authorized User or End User. Such terminal equipment or communications systems shall be furnished by and maintained at the expense of the Customer, its Authorized User or End User, except as otherwise provided. The Customer and its Authorized User or End User is also responsible for all costs at its premises incurred in the use of Service, including but not limited to equipment, wiring, electrical power, and personnel. When such terminal equipment or communications systems are used, they shall in all respects comply with the generally accepted minimum protective standards of the telecommunications industry as endorsed by the Federal Communications Commission.

### 2.12 Interconnection

Service furnished by the Company to the Customer or its Authorized Users or End Users may be connected with the services or facilities of other carriers. The Customer is responsible for all charges billed by other carriers in connection with the use of Service. Any special equipment or facilities necessary to achieve compatibility between carriers are the sole responsibility of the Customer.

### 2.13 Inspection, Testing and Adjustment

- 2.13.1 The Company may, upon reasonable notice, make such tests and inspections as may be necessary to determine whether pricing and service guide requirements are being complied with in the installation, operation, and maintenance of the Customer's, Authorized User's, End User's or the Company's equipment. The Company may, without notice, interrupt Service at any time, as necessary, because of a departure from any of these requirements and may continue such interruption until its requirements have been satisfied.
- 2.13.2 Upon reasonable notice, access to the facilities provided by the Company shall be made available to the Company by the Customer or its Authorized Users for such tests and adjustments as may be necessary for their maintenance to a condition satisfactory to the Company.
- 2.13.3 The Company shall not be liable to the Customer, its Authorized Users or End Users for any damages for Service interruption pursuant to this SubSection. Neither the Customer nor its Authorized Users and End Users shall be entitled to any credit for interruption of Service pursuant to this Section when the interruption of Service is less than two (2) hours.

Original Page 10

Effective: September 1, 2019

### **SECTION 2 – REGULATIONS**

# 2.14 Interruption of Service

- 2.14.1 The Customer shall be given a credit allowance for any interruption of Service which is not due to (a) the Company's inspection, testing or adjustment, if for a period of two (2) hours or less; (b) mistakes or errors of the Customer, its Authorized Users or its End Users; or (c) the failure of facilities or equipment provided by the Customer or its Authorized Users and End Users.
- 2.14.2 Credit allowances shall be subject to the general liability provisions set forth in Section 2.3 herein. It shall be the obligation of the Customer to notify the Company immediately of any interruption of Service for which a credit allowance is desired. Before giving such notice, the Customer shall ascertain that the trouble is not being caused by action or omission of the Customer, its Authorized Users or End Users, or is not in facilities or equipment, if any, furnished by the Customer, its Authorized User or End User and connected to the Company's Services.
- 2.14.3 For the purposes of credit computation, every month shall be considered to have seven hundred twenty (720) hours. No credit shall be allowed for any interruption of Service of a continuous duration of less than two (2) hours.
- 2.14.4 The Customer shall be credited for an interruption of Service of two (2) hours or more at the rate of 1/720th of the monthly non-usage sensitive charges for the Service affected for each hour or major fraction thereof that the interruption continues. The formula for calculating credit shall be as follows:

Credit = 
$$\frac{A}{720}$$
 x B

"A" = Outage time in hours

"B" = Total monthly fixed, non-usage sensitive charge for affected facility

2.14.5 For usage rated toll services, credits will be limited to, at maximum, the price of the Initial Period for the individual call that was interrupted plus any applicable per call service charges or surcharges required to reconnect the caller.

# 2.15 Use of Recording Devices

Customers, Authorized Users and End Users who use recording devices do so at their own risk. A Customer, Authorized User or End User may only use a recording device if the Customer, Authorized User or End User complies with the requirements of this section and is able to connect or disconnect the recording device, or turn the recording device on or off, at will.

- 2.15.1 Customers, Authorized Users and End Users may record a conversation if the Customer, Authorized User or End User first obtains written or verbal consent to the recording of all parties to the conversation prior to or at the beginning of the conversation.
- 2.15.2 A distinctive recorder tone must be repeated at intervals of approximately fifteen (15) seconds to alert all parties to the conversation that a recording device is being used.
- 2.15.3 The requirements of 2.15.1 and 2.15.2 are waived for Broadcast licensees who use a recording device to record a conversation for broadcast if all parties to the conversation are aware that the conversation will be broadcast.

Original Page 11

Effective: September 1, 2019

### SECTION 2 – REGULATIONS

### 2.16 Toll-Free Numbers

The following additional regulations apply to Inbound Long Distance Services when offered by the Company:

- 2.16.1 The Company will make every effort to reserve toll-free vanity numbers (*e.g.*, 800, 888) on behalf of Customers, but makes no guarantee or warranty that the requested toll-free number(s) will be available or assigned to the Customer requesting the number.
- 2.16.2 If a Customer accumulates undisputed past-due charges, the Company reserves the right not to honor the Customer's request for a change in toll-free service to another carrier (*e.g.*, "porting" of the toll-free number), including a request for a Responsible Organization (Resp Org) change, until such time as all charges are paid in full.
- 2.17 Trade Names, Trademarks, Service Marks and Registered Marks

Neither the Customer nor the Company shall use the other's trade names, trademarks or service marks ("Marks") without the prior written approval of the other party. Neither shall they display or use the other's Marks, nor permit the same to be displayed or used by third parties. Nothing in this pricing and service guide creates in a party rights in the Marks of the other.

### 2.18 Adjustment to Rates and Charges

The Company may adjust its rates and charges or impose additional rates and charges on its Customers to recover amounts it is required by governmental or other authorities to collect from or pay to others in support of statutory or regulatory programs. Examples of such programs include, but are not limited to, the Universal Service Fund, the Primary Interexchange Carrier Charge, and compensation to payphone service providers for the use of their payphones to access BellSouth Long Distance services. Such charges, if applicable, are described in Section 9 of the pricing and service guide.

### SECTION 3 - GENERAL INFORMATION

### 3.1 Interstate Service Descriptions

### 3.1.1 Service Descriptions

(A) The Company offers outbound direct dial (1+), inbound toll-free (*e.g.*, 800 or 888) number and operator assisted (0+) long distance services to its Customers. The Company's long distance service charges may vary based upon call duration, time of day rate period, day of week, mileage, call type, and or calling plan. Other services offered by the Company include, but are not limited to, directory assistance and Private Line services. All Company services are available 24 hours a day, seven days a week.

Original Page 1

Effective: September 1, 2019

- (B) Outbound direct dial services are available from originating locations within the Mainland United States and Hawaii. Calls may be placed to locations within the United States.
- (C) Inbound toll-free service is available to Customers served from locations within the Mainland United States. Toll-free calls may be originated from locations within the United States.
- (D) Access to the Company's Services may be via Switched or Special Access. Outbound direct dial services are offered as Primary Carrier Service from locations served with equal access end offices.
- (E) Special Access Lines, if utilized, are provided and billed to the Customer by the Company or by another special access provider. Charges for Special Access Lines are determined by the special access provider.

# 3.1.2 Timing of Calls

Billing for calls placed over the Company's network is based in part on the duration of the call as follows, unless otherwise specified in this pricing and service guide:

- (A) Timing of each call begins when the called station is answered (*i.e.*, when two-way communications are established.) Answer detection is based on standard industry answer detection methods, including hardware and software answer detection.
- (B) Chargeable time for calls ends when one of the parties disconnects from the call.
- (C) The initial and additional timing periods for billing purposes vary by product and are specified in Sections 5, 6, 7, 8 and 12 of this pricing and service guide.
- (D) The Company will not knowingly bill for unanswered calls. When a Customer indicates that he/she was billed for an incomplete call, the Company will reasonably issue credit for the call.
- (E) Time of day designations are used in this pricing and service guide to indicate rate period boundaries. Rate periods begin at the first time of day designation and continue up to but not including the second time of day designation.
- (F) Calls will be billed at the rate in effect during the call, with two rates applied if the call spans over more than one billing rate period.

### SECTION 3 – GENERAL INFORMATION

# 3.1 Interstate Service Descriptions (continued)

# 3.1.3 Rate Periods and Holidays

(A) Unless otherwise specified in this pricing and service guide, the appropriate rates apply for Day, Evening and Night/Weekend calls based on the following chart.

TIMES	MON	TUES	WED	THURS	FRI	SAT	SUN
8:00 AM to 5:00 PM*	Daytime	Daytime Period					
5:00 PM to 11:00 PM*	Evening Period						
11:00 PM to 8:00 AM*	Night/V	Night/Weekend Period					

Original Page 2

Effective: September 1, 2019

(B) Unless otherwise specified in this pricing and service guide, the appropriate rates apply for Peak and Off-Peak calls based on the following chart.

TIMES	MON	TUES	WED	THURS	FRI	SAT	SUN
7:00 AM to 7:00 PM*	Peak Pe	Peak Period					
7:00 PM to 7:00 AM*	Off-Pea	Off-Peak Period					

<sup>\*</sup>To, but not including

# (C) Holidays

For services subject to holiday discounts, the Night/Weekend or Off-Peak rate will apply to the holidays listed below unless a lower rate normally applies or unless otherwise specified in this pricing and service guide:

New Year's Day	January 1
Martin Luther King Day	Nationally Recognized Day
Presidents' Day	Nationally Recognized Day
Memorial Day	Nationally Recognized Day
Independence Day	July 4
Labor Day	Nationally Recognized Day
Columbus Day	Nationally Recognized Day
Veterans Day	Nationally Recognized Day
Thanksgiving Day	Nationally Recognized Day
Christmas Day	December 25

<sup>\*</sup>To, but not including

### SECTION 3 - GENERAL INFORMATION

- 3.1 Interstate Service Descriptions (continued)
  - 3.1.4 Outbound Long Distance Services

The Company's outbound long distance services allow Customers to place calls to terminating locations on a direct dial basis. Customers must dial "1+" the destination telephone number, including area code to reach the called party. With outbound long distance services, the Customer as caller is billed for services provided.

Original Page 3

Effective: September 1, 2019

Outbound long distance services are available to Customers who: 1) presubscribe their local access lines to the Company's network, or 2) dial the Company's access code (where available) to gain access to the network.

The following optional features and services are available for an additional charge to Business Customers who subscribe to Business MTS or a direct dial outbound Business optional calling plan:

- (A) Range Privileges Allows the Customer to restrict certain types of outbound calling on a line or per account basis. Range Privileges may be standard or custom. Range privileges will not restrict access to dial-around (10-10-XXX) calls. Customer is responsible for such calls.
  - .1 Standard Range Privileges: Provides the Customer the ability to choose from predefined privileges as follows:
    - No calls allowed except operator-assisted, toll-free and local calls
    - No calls allowed except to domestic 50 states
    - No calls allowed except domestic 50 states, Canada, Puerto Rico and/or U.S. Virgin Islands
    - No calls allowed except to entire North American Dialing Plan locations
  - .2 Custom Range Privileges: Allows the Customer to restrict outbound calling by NPA, by NPA-NXX, by state and/or by international country code.
- (B) Override Codes Provides the Customer the ability to override range privileges by using a specific code.

Rates, billing increments, discounts and qualifications, if applicable, vary by service plan and are provided in later sections of this pricing and service guide.

### SECTION 3 - GENERAL INFORMATION

### 3.1 Interstate Service Descriptions (continued)

### 3.1.5 Inbound Long Distance Services

The Company's inbound long distance services are toll-free number (*e.g.*, 800, 888) services. Calls may originate from any valid exchange and terminate to the Customer's location at no charge to the calling party. With inbound long distance services, the Customer is billed for the call rather than the calling party.

Original Page 4

Effective: September 1, 2019

Inbound long distance services may terminate to Customer local exchange access lines using Switched Access termination. Customers with larger call volumes may have inbound calls routed to Special Access facilities connecting the Customer's premises directly with the Company network.

The following standard features are available at no additional charge to Business Customers who subscribe to AT&T Long Distance Toll Free<sup>SM</sup> Service.

- (A) On Demand Reroute Allows the Customer to designate, upon demand, alternate routing of incoming toll-free calls to another location.
- (B) Vanity Numbers Personalized toll-free number requested by the Customer, subject to the availability of the requested number.

The following optional features and services are available for an additional charge to Business Customers who subscribe to AT&T Long Distance Toll Free<sup>SM</sup> Service.

- (A) Authorization Codes Allows the Customer to limit unintentional or unauthorized calls placed on a toll-free number by requiring the caller to enter a specific code before the toll-free call can be completed.
- (B) Call Blocking Allows the Customer to define specific geographic areas where calls cannot originate or allows blocking of calls that originate from a payphone.
- (C) National Toll-Free Directory Assistance Listing Service Beginning September 1, 2003, this service will allow up to one (1) listing per each toll-free number in the National Toll-Free Directory Assistance Listing Service, except that Customer who previously have been allowed up to three (3) listings per each toll-free number in the National Toll-Free Directory Assistance Listing Service will be permitted to retain these listings in accordance with other terms and conditions contained in this pricing and service guide.

Rates, billing increments, discounts and qualifications, if applicable, vary by service plan and are provided in later sections of this pricing and service guide.

SECTION 3 - GENERAL INFORMATION

### 3.1 Interstate Service Descriptions (continued)

### 3.1.6 Operator Services

Operator Services are available on a presubscribed basis to Customers. Operator Services allow the Customer to place a call and arrange for billing to the presubscribed Business or Residential Access Line or to a commercial credit card. Calls are rounded up to the next whole minute for billing purposes.

Original Page 5

Effective: September 1, 2019

Services provided to transient end users from Pay Telephones, Hotels, Motels or other Aggregator location are provided for under the Company's FCC OSP Informational Tariff on file with the FCC.

The following billing arrangement is available to Customers through the Company's Operator Services:

• Operator Station (Operator Assisted 0+-)

This is a Service whereby the Customer places a non-person-to-person call by dialing 0 + area code + station number and chooses to wait for the assistance of an operator (live or automated). When placing an operator station call, the caller is connected to a non-specified individual at the terminating end. Such calls may be billed to a credit card.

### SECTION 3 – GENERAL INFORMATION

# 3.2 International Service Descriptions

### 3.2.1 General

(A) The Company offers international telecommunications services between points within the United States and foreign locations as specified in the following paragraphs. The Company's rates and charges may vary based upon service type, call duration, time of day rate period, day of week, mileage, call type, foreign location, and/or calling plan. All Company services are available 24 hours a day, seven days a week.

Original Page 6

- (B) Outbound direct dial services are available from originating locations within the Mainland United States and Hawaii. Calls may be placed to locations between the United States and those countries listed in Sections 6, 7 and 12.
- (C) Access to the Company's Services may be via Switched or Special Access. Outbound direct dial services are offered as Primary Carrier Service from locations served with equal access end offices.
- (D) Special Access Lines, if utilized, are provided and billed to the Customer by the Company or by another special access provider. Charges for Special Access Lines are determined by the special access provider.

### SECTION 3 - GENERAL INFORMATION

# 3.2 International Service Descriptions (continued)

### 3.2.2 Timing of Calls

Billing for calls placed over the Company's network is based in part on the duration of the call as follows, unless otherwise specified in this pricing and service guide:

Original Page 7

Effective: September 1, 2019

- (A) Timing of each call begins when the called station is answered (*i.e.*, when two-way communications are established.) Answer detection is based on standard industry answer detection methods, including hardware and software answer detection.
- (B) Chargeable time for calls ends when one of the parties disconnects from the call.
- (C) The initial and additional timing periods for billing purposes vary by product and are specified in Section 6, Section 7 and Section 12 of this pricing and service guide.
- (D) The Company will not knowingly bill for unanswered calls. When a Customer indicates that he/she was billed for an incomplete call, the Company will reasonably issue credit for the call.
- (E) Time of day designations are used in this pricing and service guide to indicate rate period boundaries. Rate periods begin at the first time of day designation and continue up to but not including the second time of day designation.
- (F) A call will be billed for its entire duration at the rate in effect when the call began.

# 3.2.3 Rate Periods and Holidays

For services subject to time of day discounts, the following rate periods unless otherwise specified in this pricing and service guide. Time-of-day periods are based on the local time of the originating location within the United States.

### (A) Rate Periods for Service to Canada

Times	Mon	Tues	Wed	Thur	Fri	Sat	Sun
8:00 AM to 6:00 PM*	Standa	rd Perio	d				
6:00 PM to 8:00 AM*	Economy Period						

<sup>\*</sup>To, but not including

### (B) Rate Periods for Service to Mexico

Times	Mon	Tues	Wed	Thur	Fri	Sat	Sun
7:00 AM to 7:00 PM*	Standard Period						
7:00 PM to 7:00 AM*	Economy Period						

<sup>\*</sup>To, but not including

# SECTION 3 – GENERAL INFORMATION

# 3.2 International Service Descriptions (continued)

# 3.2.3 Rate Periods and Holidays (continued)

# (C) Rate Periods for Service to Other Countries

Country	Standard	Economy
Afghanistan	6p-11a	11a-6p
Albania	7а-бр	6p-7a
Algeria	6a-5p	5p-6a
American Samoa	10a-11p	11p-10a
Andorra	7а-бр	6p-7a
Angola	6a-5p	5p-6a
Anguilla	7a-10p	10p-7a
Antarctica	10a-11p	11p-10a
Antarctica-Scott	10a-11p	11p-10a
Antigua	7a-10p	10p-7a
Argentina	8a-12m	12m-8a
Armenia	8a-12m 7a-2a	2a-7a
Aruba	7a-10p	10p-7a
Ascension Island	6a-5p 2p-3a	5p-6a
Australia (includes	2p-3a	3a-2p
Christmas and Cocos		_
Islands)		
Austria	7a-6p	6p-7a
Azerbaijan	7a-2a	2a-7a
Bahamas	8a-11p	11p-8a
Bahrain	9p-3p	3p-9p
Bangladesh	6a-6p	6p-6a
Barbados	7a-10p	10p-7a 2a-7a
Belarus	7a-2a	2a-7a
Belgium	7a-6p	6p-7a
Belize	7a-6p 8a-11p	11p-8a
Benin	6a-5p	5p-6a
Bermuda	8a-11p	11p-8a
Bhutan	6a-6p 7a-12m	6р-6а
Bolivia	7a-12m	12m-7a
Bosnia-Herzegovina	7a-12m	12m-7a
Botswana	6a-5p	5p-6a
Brazil	8a-12m	12m-8a
British Virgin Islands	8a-11p	11p-8a
Brunei	10a-11p 7a-2a	11p-10a 2a-7a
Bulgaria	7a-2a	2a-7a
Burkina Faso	6a-5p	5p-6a
Burundi	6a-5p	5p-6a
Cambodia	5p-11a	11a-5p
Cameroon	6a-5p 6a-5p	5p-6a
Cape Verde Islands	6a-5p	5p-6a
Cayman Islands	8a-11p	11p-8a
Central African Rep.	6a-5p 6a-5p	5p-6a
Chad	6a-5p	5p-6a
Chile	8a-12m	12m-8a
China	5p-11a	11a-5p

Country	Standard	Economy
Colombia	7a-12m	12m-7a
Comoros	6a-5p	5p-6a
Congo	6a-5p	5p-6a
Cook Islands	10a-11p	11p-10a
Costa Rica	8a-11p	11p-8a
Croatia	7a-12m	12m-7a
Cuba	7a-10p	10p-7a 6p-7a
Cyprus Czech Republic	7a-6p	6p-7a
Czech Republic	7a-12a	12a-7a
Denmark	7а-бр	6p-7a
Diego Garcia	6a-6p	6p-6a
Djibouti	6a-5p	5p-6a
Dominica	7a-10p	10p-7a
Dominican Republic	7a-10p	10p-7a
East Timor	10a-11p	10p-7a 11p-10a
	1	•
Easter Island	8a-12m	12m-8a
Ecuador	7a-12m	12m-7a
Egypt	7a-2a	2a-7a
El Salvador	8a-11p	11p-8a
Eq Guinea	6a-5p 7a-2a	5p-6a
Eritrea	7a-2a	2a-7a
Estonia	7a-2a	2a-7a
Ethiopia	7a-2a	2a-7a 6p-7a
Faeroe Islands	7а-бр	6p-7a
Falkland Islands	8a-12m	12m-8a
Fiji Islands	9a-2a	2a-9a
Finland	7а-бр	6p-7a
France	7a-6p	6p-7a
French Antilles	8a-11p	11p-8a
French Guiana	8a-12m	12m-8a
French Polynesia	10a-11p	11p-10a
Gabon	6a-5p 6a-5p	5p-6a 5p-6a
Gambia	6a-5p	5p-6a
Georgia	7a-2a	2a-7a
Germany	7a-6p	6p-7a
Ghana	6a-5p	5p-6a
Gibraltar	7a-6p	6p-7a
Greece	7a-6p	6p-7a
Greenland	7a-6p	6p-7a
Grenada	7a-10p	10p-7a
Guadeloupe	8a-11p	11p-8a
Guatemala	8a-11p	11p-8a
Guinea	6a-5p	5p-6a
Guinea Bissau	6a-5p	5p-6a

Original Page 8

# SECTION 3 – GENERAL INFORMATION

# 3.2 International Service Descriptions (continued)

# 3.2.3 Rate Periods and Holidays (continued)

(C) Rate Periods for Service to Other Countries (continued)

Country	Standard	Economy
Guyana	8a-12m	12m-8a
Haiti	7a-10p	10p-7a
Honduras	8a-11p	11p-8a
Hong Kong	10a-11p	11p-10a
Hungary	7a-6p	6p-7a 8p-7a
Iceland	7a-8p	8p-7a
India	6a-6p	6p-6a
India Indonesia	6a-6p 5p-11a	6p-6a 11a-5p 2a-7a
Iran	7a-2a 7a-2a	2a-7a
Iraq	7a-2a	2a-7a
Ireland	7a-6p	6p-7a
Israel	7a-6p 12m-5p	5p-12m
Italy	7a-6p	6p-7a
Italy Ivory Coast	6a-5p	6p-7a 5p-6a
Jamaica	7a-10p	10p-7a
Japan	2p-3a	3a-2p
Jordan	12m-5p	5p-12m
Kazakhstan	7a-2a	2a-7a
Kenya	7a-1a	1a-7a
Kiribati	10a-11p 2p-3a	11p-10a 3a-2p
Korea, North	2p-3a	3a-2p
Korea, North Korea, South	2p-3a	3a-2p 6p-7a
Kosovo	7a-6p	6p-7a
Kuwait	7a-1a	1a-7a
Laos	5p-11p	11p-10a
Latvia	7a-2a	2a-7a
Lebanon	9p-3p 7a-1a	3p-9p 1a-7a
Lesotho	7a-1a	1a-7a
Liberia	6a-5p	5p-6a
Libya	6a-5p 6a-5p	5p-6a
Liechtenstein	7a-6p	6p-7a
Lithuania	7a-6p 7a-2a	6p-7a 2a-7a
Luxembourg	7a-6p	6p-7a
Macao	7a-6p 5p-11a 7a-12m 6a-5p	11a-5p
Macedonia	7a-12m	12m-7a
Madagascar	6a-5p	5p-6a
Malawi	6a-5p	5p-6a
Malaysia Maldives	5p-11a	11a-5p
Maldives	6p-11a	11а-бр
Mali	6p-11a 6a-5p	5p-6a
Malta	7a-6p	6p-7a
Marshall Islands	l 10a-11n	11p-10a
Mauritania	6a-5p	5p-6a
Mauritius	6a-5p	5p-6a
Laos	5p-11p	11p-10a
Latvia	6a-5p 6a-5p 5p-11p 7a-2a 7a-2a	5p-6a 11p-10a 2a-7a 2a-7a
Kyrgyzstan	7a-2a	2a-7a

Country	Standard	Economy
Laos	5p-11p	11p-10a
Latvia	7a-2a	2a-7a
Lebanon	9p-3p 7a-1a	3p-9p
Lesotho	7a-1a	1a-7a
Liberia	6a-5p	5p-6a
Libya	6a-5p 6a-5p	5p-6a
Liechtenstein	/a-6p	6p-7a 2a-7a 6p-7a
Lithuania	7a-2a	2a-7a
Luxembourg	7a-6p	6p-7a
Macao	5p-11a	11a-5p
Macedonia	7a-12m 6a-5p	12m-7a
Madagascar	6a-5p	5p-6a
Malawi	6a-5p	5p-6a
Malaysia	5p-11a	11a-5p
Maldives	5p-11a 6p-11a 6a-5p	11а-бр
Mali	6a-5p	5p-6a
Malta	7а-бр 10а-11р	6p-7a
Marshall Islands	10a-11p	11p-10a
Mauritania	6a-5p	5p-6a
Mauritius	6a-5p 6a-5p 10a-11p	5p-6a 11p-10a
Micronesia	10a-11p	11p-10a
Moldova	7a-2a	2a-/a
Monaco	7a-6p	6p-7a
Mongolia	10a-11p	11p-10a
Montserrat	4p-10p	10p-4p
Morocco	6a-5p	5p-6a
Mozambique	6a-5p	5p-6a
Myanmar	10a-11p	5p-6a 11p-10a
Namibia	6a-5p 10a-11p	5p-6a
Nauru	10a-11p	11p-10a
Nepal	6а-бр	6p-6a
Netherlands	7а-бр	6p-7a 11p-8a
Netherland Antilles	8a-11p	11p-8a
Nevis	7a-10p	10p-7a
New Caledonia	10a-11p	11p-10a
New Zealand	10a-11p	11p-10a
Nicaragua	8a-11p	11p-8a
Niger	6a-5p	5p-6a
Nigeria	7a-1a	1a-7a
Niue	10a-11p	11p-10a
Norfolk Islands	10a-11p	11p-10a 6p-7a
Norway	7a-6p	6p-7a
Oman	9p-3p	3p-9p
Pakistan	5a-6p 10a-11p	6p-6a
Palau	10a-11p	11p-10a
Palestinian Authority	12a-5p	5p-12a
Panama	8a-11p	11p-8a

Original Page 9

# SECTION 3 – GENERAL INFORMATION

# 3.2 International Service Descriptions (continued)

# 3.2.3 Rate Periods and Holidays (continued)

# (C) Rate Periods for Service to Other Countries (continued)

Country	Standard	Economy
Papua New Guinea	10a-11p	11p-10a
Paraguay	8a-12m	12m-8a
Peru	7a-12m	12m-7a
Philippines	5p-11a	11a-5p
Pitcairn	10a-11p	11p-10a
Poland	7a-12m	12m-7a
Portugal	7a-8p	8p-7a
Qatar	7a-1a	1a-7a
Reunion Island	6a-5p	5p-6a
Papua New Guinea	10a-11p	11p-10a
Paraguay	8a-12m	12m-8a
Peru	7a-12m	12m-7a
Philippines	5p-11a	11a-5p
Pitcairn	10a-11p	11p-10a
Poland	7a-12m	12m-7a
Portugal	7a-8p	8p-7a
Qatar	7a-1a	1a-7a
Reunion Island		
(includes Mayotte	6a-5p	5p-6a
Ísland)	-	-
Romania	7a-2a	2a-7a
Russia	7a-2a	2a-7a
Rwanda	6a-5p	5p-6a
San Marino	7a-6p	6p-7a
Sao Tome	6a-5p	5p-6a
Saudi Arabia	7a-1a	1a-7a
Senegal	6a-5p	5p-6a
Serbia	7a-12m	12m-7a
Seychelles	6a-5p	5p-6a
Sierra Leone	6a-5p	5p-6a
Singapore	10a-11p	11p-10a
Slovakia	7a-12m	12m-7a
Slovenia	7a-12m	12m-7a
Solomon Islands	10a-11p	11p-10a
Somalia	6a-5p	5p-6a
South Africa	6a-5p	5р-6а
South Sudan	7a-2a	2a-7a
Spain	7а-бр	6p-7a
Spanish Sahara	6a-5p	5p-6a
Sri Lanka	ба-бр	6p-6a
St Helena	6a-5p	5p-6a

Country	Standard	Economy
St Kitts	7a-10p	10p-7a
St Lucia	7a-10p	10p-7a
St Pier	7a-10p	10p-7a
St Vincent/Grenada	7a-10p	10p-7a 10p-7a
Sudan	7a-2a	2a-7a
Suriname	8a-12m	12m-8a
Swaziland	6a-5p	5p-6a
Sweden	7a-6p	6p-7a
Switzerland	7a-6p	6p-7a
Syria	7a-2a	2a-7a
Taiwan	10a-11p	11p-10a
Tajikistan	7a-2a	2a-7a
Tanzania	7a-2a	2a-7a
Thailand	5p-11a	11a-5p
Togo	6a-5p	5p-6a
Togo Tokelau	6a-5p	5p-6a
Tonga	9a-2a	2a-9a
Trinidad/Tobago	7a-10p	10p-7a
Tunisia	6a-5p	5p-6a
Turkey	7a-6p 8a-11p	6p-7a
Turks/Caicos	8a-11p	11p-8a
Turkmenistan	7a-2a	11p-8a 2a-7a
Tuvalu	10a-11p	11p-10a
Uganda	7a-2a	2a-7a
Ukraine	7a-2a	2a-7a
United Arab Emir.	9p-3p	3p-9p 6p-7a 12m-7a
United Kingdom	7a-6p 7a-12m	6p-7a
Uruguay	7a-12m	12m-7a
Uzbekistan	7a-2a	2a-7a
Vanuatu	10a-11p	11p-10a
Vatican City	7a-6p	6p-7a
Venezuela	8a-12m	12m-8a
Vietnam	5p-11a	11a-5p
Wallis	10a-11p	11p-10a
West Samoa	10a-11p	11p-10a
Yemen	9p-3p	3p-9p
Zaire	6a-5p	5p-6a
Zambia	6a-5p 6a-5p	5p-6a
Zimbabwe	6a-5p	5p-6a

Original Page 10

Effective: September 1, 2019

### (D) Holidays

For services subject to holiday discounts, the Evening or Economy rate will apply to the holidays listed below unless a lower rate normally applies or unless otherwise specified in this pricing and service guide.

These holiday rates only apply to operator service calls terminating to Canadian locations.

New Year's Day January 1 Christmas Day December 25

SECTION 3 – GENERAL INFORMATION

- 3.2 International Service Descriptions (continued)
  - 3.2.4 International Outbound Long Distance Services

International Outbound Long Distance services are offered to Customers for origination of calls from locations within the Mainland United States and Hawaii to foreign destinations. Calls may be placed on a direct dial basis to foreign locations as specified by service type in Section 4 of this pricing and service guide. Customers must dial the destination telephone number, including country code and/or area code to reach the called party. With outbound long distance services, the Customer as caller is billed for services provided. Outbound long distance services are available to Customers who; 1) presubscribe their local access lines to Company's network.

Original Page 11

Effective: September 1, 2019

3.2.5 International Inbound Long Distance Service

See Sections 4.4, 6, 7 and 12 for a description of Inbound Long Distance Service.

3.2.6 International Operator Services

See Section 8 for a description of Operator Services.

Original Page 12

Effective: September 1, 2019

SECTION 3 – GENERAL INFORMATION

### 3.3 Calculation of Distance

For mileage sensitive services, the distance between originating and terminating points of a call are determined using vertical ("V") and horizontal ("H") coordinates for the serving wire center(s) or the Company access point(s) associated with the call. For purposes of determining the airline mileage of a call the Company references the V and H coordinates as found in Telcordia's V&H Tape and NECA FCC Tariff No. 4. The use of coordinates for wire centers versus access points and the method for calculating actual distances varies based on the type of service and the form of access used to reach the Company network.

For outbound and inbound switched long distance services utilizing Switched Access Origination, mileage measurements are based on the airline distance between serving wire centers associated with the originating and terminating points of the call. The serving wire centers of a call are determined by the area codes and exchanges of the origination and terminating points. Distance measurements are determined using the mileage calculation method shown in this section.

Certain services provided to Mexico utilize Rate Bands associated with terminating locations for determining rates and charges for each call. For purposes of determining Rate Bands for calls to Mexico, the Company references industry standard Rate Bands.

Original Page 1

Effective: September 1, 2019

SECTION 4 – BUSINESS DOMESTIC SERVICE DESCRIPTIONS, RATES AND CHARGES

### 4.1 General

Customers are billed based on their usage of the Company's services. Rates may vary by service type, call duration, time of day, rate period, day of week, mileage, call type and/or calling plan. Fixed recurring charges, not dependent upon usage, are billed in advance. Usage-based charges are billed after each usage cycle.

Customers will be charged for each individual call placed through the Company. Call duration is determined as described in Section 3 of this pricing and service guide. For distance or time of day sensitive offerings, charges are determined based on mileage calculations and applicable rate periods found in Section 3 unless otherwise specified in this pricing and service guide.

Outbound direct dial services are available from originating locations within Alabama, Florida, Georgia, Kentucky, Louisiana, Mississippi, North Carolina, South Carolina and Tennessee. Calls may be placed to locations within the United States.

Original Page 2

Effective: September 1, 2019

SECTION 4 – BUSINESS DOMESTIC SERVICE DESCRIPTIONS, RATES AND CHARGES

- 4.2 Business Message Telecommunications Service (MTS)
  - (A) Business MTS is an outbound only direct dialed long distance service offered to Business Customers from switched access lines. Service is offered on a flat rate basis, twenty-four hours per day, seven days per week. Customers must dial 1 plus the area code, if applicable, and the terminating number to complete a call utilizing this service. All calls are billed in one (1) minute increments after an initial period, for billing purpose, of one (1) minute. Customers who do not designate and qualify for a specific optional calling plan are automatically placed on Business MTS. Holiday discounts do not apply.
  - (B) Business MTS is available to Business Customers who:
    - .1 subscribe to an access line from one of the following Affiliated ILECs: AT&T Alabama, AT&T Florida, AT&T Georgia, AT&T Kentucky, AT&T Louisiana, AT&T Mississippi, AT&T North Carolina, AT&T South Carolina or AT&T Tennessee; and,
    - .2 subscribe to the Company for interLATA (interstate and intrastate) Business MTS service and/or intraLATA (local) toll Business MTS service.
  - (C) Rates and Charges

Monthly Minimum Usage Charge (MUC): \$47.50

If monthly outbound usage charges or outbound and inbound usage charges combined equal or exceed the MUC in a billing period, the MUC will not apply. If the monthly outbound usage charges or outbound and inbound usage charges combined in a billing period are less than the MUC, the charge that will apply will be the difference between that month's usage charges and the MUC.

Per-Minute Usage Rate:

Jurisdiction	Per-Minute Usage Rate
Interstate	\$0.99
Intrastate	
Alabama, Florida, Kentucky, Louisiana, Mississippi,	\$0.99
North Carolina, South Carolina and Tennessee	
Georgia InterLATA and IntraLATA	
Mileage 0-22	\$0.00
Mileage 23 +	\$0.99

SECTION 4 – BUSINESS DOMESTIC SERVICE DESCRIPTIONS, RATES AND CHARGES

# 4.3 Business Optional Calling Plans

- 4.3.1 AT&T Business Unlimited Calling<sup>SM</sup>
  - (A) AT&T Business Unlimited Calling<sup>SM</sup> is a direct dialed domestic outbound long distance voice service offered to business Customers with one to fifteen switched access lines. Service is offered on a flat rate basis, twenty-four hours per day, seven days per week. Customers must dial 1 plus the area code, if applicable, and the terminating number to complete a call utilizing this service. This plan provides the Customer with unlimited minutes of interexchange long distance usage for a flat rate monthly charge (certain restrictions, as outlined below, do apply). This plan does not provide call detail information on the Customer's monthly bill for usage eligible for unlimited calling but does provide call detail information for usage not included in the unlimited calling. Customers may choose to receive the call detail information for unlimited calling for an additional monthly recurring charge. This plan is not offered on an intraLATA or intrastate only basis.

Original Page 3

- (B) Toll free service is available with this plan for Customers who also subscribe to AT&T Long Distance Toll Free<sup>SM</sup> Service. The provisions and rates and charges of the AT&T Long Distance Toll Free<sup>SM</sup> Service will apply as described in Section 4.4 of this pricing and service guide except that usage rates are specified below. Toll-free usage is not included in the unlimited usage.
- (C) In order to be eligible for this plan, the Company must be able to verify that the Customer meets the following eligibility requirements. If the Company determines that the Customer's use of service violates any of these restrictions the Company, at its sole option, may move the Customer to another plan or suspend, restrict or cancel the Customer's service without prior notice. Customers are eligible for this service if they meet the following requirements:
  - .1 Customers must subscribe to BellSouth Long Distance and to the Company's affiliated incumbent local exchange carrier. Customers must subscribe to Complete Choice® For Business, individual line measured rate service for business, and/or flat rate service for business offered by the Company's Affiliated ILEC on all lines on the account.
  - .2 This plan is available to business Customers with one to fifteen business lines at a single location and on a single billing account. All lines at the location must subscribe to this plan.
  - .3 Customers must presubscribe to AT&T Business Unlimited Calling<sup>SM</sup> for both intraLATA and interLATA usage on all lines that are at a single location and on a single billing account.
  - .4 Unlimited plan usage does not include multi-party conference calls (except those calls placed by using the Three-Way Calling feature are included), calls to 900, 976, 700 numbers and other calls to access information services, directory assistance, operator services, international calling and toll-free calling services.
  - .5 This plan is not available for resale.

# SECTION 4 – BUSINESS DOMESTIC SERVICE DESCRIPTIONS, RATES AND CHARGES

- 4.3 Business Optional Calling Plans (continued)
  - 4.3.1 AT&T Business Unlimited Calling<sup>SM</sup> (continued)
    - (C) (continued)
      - .6 The Company reserves the right to exclude certain terminating telephone numbers from this plan. Calls to these numbers will be billed at applicable MTS rates.

Original Page 4

Effective: September 1, 2019

- .7 Service under this plan cannot be used with the following applications or services: call center environment or in connection with any similar such application, auto-dialers or any similar type of device, PBX or PBX-like equipment, data transmission equipment, Centrex service, foreign exchange service, public telephone services, ISDN services, broadcast facsimile services or the equivalents of any such services.
- .8 Customers on this plan may only place as many concurrent calls as they have individual lines subscribed to under this plan.
- .9 Term discounts are not applicable to this service.
- .10 Customer lines associated with education institutions (colleges, Universities, etc.) or other businesses that aggregate end user traffic are not eligible for this plan.
- (E) Rates and Charges
  - .1 Monthly Recurring Charges (MRC)
    - .a Direct Dial Service

Primary Line \$20.00 Each Additional Line after the Primary Line \$20.00

.b Toll-free Inbound Service:

All toll-free inbound calls are billed in six (6) second increments after an initial period, for billing purposes; of thirty (30) seconds. The per minute usage rate is \$0.06. See Section 4.4 of pricing and service guide for the toll-free inbound MRC.

.c Call Detail - \$5.00 per line

SECTION 4 – BUSINESS DOMESTIC SERVICE DESCRIPTIONS, RATES AND CHARGES

- 4.3 Business Optional Calling Plans (continued)
  - 4.3.2 AT&T Business Calling
    - (A) AT&T Business Calling is a combination inbound and outbound optional pricing plan available to business Customers on switched access lines.
    - (B) AT&T Business Calling is available to new or existing Business Customers who:
      - .1 utilize Switched Access to reach the long distance network for outbound calling and to receive calls from the long distance network for AT&T Long Distance Toll Free<sup>SM</sup> Service;

Original Page 5

Effective: September 1, 2019

- .2 subscribe to an access line from one of the following Affiliated ILECs: AT&T Alabama, AT&T Florida, AT&T Georgia, AT&T Kentucky, AT&T Louisiana, AT&T Mississippi, AT&T North Carolina, AT&T South Carolina or AT&T Tennessee;
- .3 subscribe to the Company for interLATA (interstate and intrastate) service, intraLATA (local toll) service, or interLATA/intraLATA service combined; and,
- .4 request to be provisioned under this plan.
- (C) Calls are billed in increments of six (6) seconds subject to a minimum billed connect time (initial period) of sixty (60) seconds.
- (D) To receive toll-free (inbound) service Customers must subscribe to the AT&T Long Distance Toll Free<sup>SM</sup> Service. The provisions and rates and charges of the AT&T Long Distance Toll Free<sup>SM</sup> Service will apply as described in Section 4.4 of this pricing and service guide except that usage rates are specified in Section 4.3.2 (E) herein.
- (E) Rates and Charges

The Monthly Recurring Charge is \$3.00, and the per-minute rate for Direct-Dial Outbound and AT&T Long Distance Toll Free<sup>SM</sup> Service calls is \$0.5550.

SECTION 4 – BUSINESS DOMESTIC SERVICE DESCRIPTIONS, RATES AND CHARGES

- 4.3 Business Optional Calling Plans (continued)
  - 4.3.3 AT&T Business Calling Monthly
    - (A) AT&T Business Calling Monthly is a combination inbound and outbound optional pricing plan available to business Customers on switched access lines.
    - (B) AT&T Business Calling Monthly is available to new or existing Business Customers who:
      - .1 utilize Switched Access to reach the long distance network for outbound calling and to receive calls from the long distance network for AT&T Long Distance Toll Free<sup>SM</sup> Service;

Original Page 6

Effective: September 1, 2019

- .2 subscribe to an access line from one of the following Affiliated ILECs: AT&T Alabama, AT&T Florida, AT&T Georgia, AT&T Kentucky, AT&T Louisiana, AT&T Mississippi, AT&T North Carolina, AT&T South Carolina or AT&T Tennessee;
- .3 subscribe to the Company for interLATA (interstate and intrastate) service, intraLATA (local toll) service, or interLATA/intraLATA service combined; and,
- .4 request to be provisioned under this plan.
- (C) Calls are billed in increments of six (6) seconds subject to a minimum billed connect time (initial period) of sixty (60) seconds.
- (D) To receive toll-free (inbound) service Customers must subscribe to the AT&T Long Distance Toll Free<sup>SM</sup> Service. The provisions and rates and charges of the AT&T Long Distance Toll Free<sup>SM</sup> Service will apply as described in Section 4.4 of this pricing and service guide except that usage rates are specified in Section 4.3.3 (E) herein.
- (E) Rates and Charges

The Monthly Recurring Charge is \$10.00, and the per-minute rate for Direct-Dial Outbound and AT&T Long Distance Toll Free<sup>SM</sup> Service calls is \$0.140.

### SECTION 4 – BUSINESS DOMESTIC SERVICE DESCRIPTIONS, RATES AND CHARGES

- 4.3 Business Optional Calling Plans (continued)
  - 4.3.4 AT&T Business Unlimited Calling<sup>SM</sup> II
    - (A) AT&T Business Unlimited Calling SM II is an outbound calling and optional inbound long distance calling plan. This plan is established at the BTN level; multiple BTN aggregation is not available with this plan. If the Customer or Applicant selects different long distance calling plans for specific WTNs, the Customer is required to establish a separate BTN for each variation.

Original Page 7

Effective: September 1, 2019

(B) This plan is available to Business Customers that:

### OPTION 1 - NEW OR EXISTING

- .1 utilize Switched Access to reach the long distance network for outbound calling and to receive calls from the long distance network (for AT&T Long Distance Toll Free<sup>SM</sup> service);
- .2 subscribe to the Company for the provision of interstate and intrastate (interLATA and/or intraLATA) Service for outbound long distance calling, and may also subscribe to the Company for the provision of toll-free service (AT&T Long Distance Toll Free<sup>SM</sup> service) for inbound long distance calling;
- .3 newly subscribe to or currently subscribe to Complete Choice<sup>®</sup> for Business or AT&T Business Local Calling from the Company's Affiliated ILEC;
- .4 subscribe to and maintain or currently subscribe to and maintain at least one (1) but no more than ten (10) business access lines (*i.e.*, BTNs and WTNs) from the Company's Affiliated ILEC;
- .5 at the time of subscription to this plan, Customers must subscribe to a local service term agreement with the Company's Affiliated ILEC; and
- .6 request to be provisioned under this plan.

Original Page 8

Effective: September 1, 2019

### SECTION 4 – BUSINESS DOMESTIC SERVICE DESCRIPTIONS, RATES AND CHARGES

- 4.3 Business Optional Calling Plans (continued)
  - 4.3.4 AT&T Business Unlimited Calling<sup>SM</sup> II (continued)
    - (B) This plan is available to Business Customers that (continued):

#### **OPTION 2 - RETENTION**

- .1 utilize Switched Access to reach the long distance network for outbound calling and to receive calls from the long distance network (for AT&T Long Distance Toll Free<sup>SM</sup> service);
- .2 currently subscribe to AT&T Business Unlimited Calling<sup>SM</sup> II and state an intention to move long distance service to a competitor of the Company;
- .3 currently or newly subscribe to the Company for the provision of interstate and intrastate (interLATA and/or intraLATA) Service for outbound long distance calling, and may also subscribe to the Company for the provision of toll-free service (AT&T Long Distance Toll Free<sup>SM</sup> service) for inbound long distance calling;
- .4 currently or newly subscribe to Complete Choice® for Business or AT&T Business Local Calling from the Company's Affiliated ILEC;
- .5 currently or newly subscribe to and maintain at least one (1) but no more than ten (10) business access lines (*i.e.*, BTNs and WTNs) from the Company's Affiliated ILEC; and
- at the time of subscription to this plan, Customers must subscribe to a local service term agreement with the Company's Affiliated ILEC.
- (C) A single business entity with more than one BTN at that business entity's physical service location is eligible for this plan on one (1) BTN only, provided the total number of business access lines do not exceed ten (10) at that location.
- (D) The Customer may subscribe to this plan for outbound Service only, or for both outbound Service and inbound toll-free calling. Subject to restrictions, Customers subscribing to the plan receive unlimited domestic 1+ direct-dialed outbound calling. Where a Customer subscribes to AT&T Long Distance Toll Free<sup>SM</sup> service, inbound toll-free calls may originate on any type of access and are terminated via Switched Access to the Customer's location. (See Section 4.4 and Section 4.5 of this pricing and service guide for optional feature, rules and regulations, and general information regarding AT&T Long Distance Toll Free<sup>SM</sup> service).
- (E) Inbound switched toll-free calls and fully automated operator assisted are not included in the unlimited MOUs. Toll-free calls are billed in increments of six (6) seconds subject to a minimum connect time (initial period) of thirty (30) seconds. Toll-free calls per minute rates associated with this plan are subject to change with prior notification to the Customer.

SECTION 4 – BUSINESS DOMESTIC SERVICE DESCRIPTIONS, RATES AND CHARGES

- 4.3 Business Optional Calling Plans (continued)
  - 4.3.4 AT&T Business Unlimited Calling<sup>SM</sup> II (continued)
    - (F) Restrictions
      - .1 All business access lines and/or working lines under the participating BTN must be provisioned on this plan and term discounts do not apply to this plan.

Original Page 9

- This plan is provided for standard voice calling involving live dialog between .2 individuals. This plan is not intended for use as a substitute for dedicated or open circuits, or similar applications. This plan is not available for resale. This plan may not be used for data connections, including but not limited to, modem to modem calls, remote access applications, Internet access, or Intranet access, including access to corporate LANs. Additionally, this plan may not be used for dedicated point-to-point connections between equipment that leaves a circuit connected without a contemporary, continuous voice communication. This plan may not be used for any of the following: auto dialers; PBX trunks; ground start line or trunks; ISDN service, including PRI; foreign exchange services; Remote Call Forwarding/Telebranch Service; public telephone service; public access smartpay phones; analog to digital conversion; digital PBX service, including local access provisioned via T-1 facilities; WATS service; PBX/PABX/EABX services; non-square electronic key telephone systems; hybrid key telephone systems; predictive calling/dialing systems; automatic outbound dialing systems; any type of automatic call distribution system; or the functional equivalent of any such systems listed above. This plan may be used for fax transmissions, excluding broadcast fax applications. If the Company determines that the Customer is in violation of the above listed restrictions, the Customer shall forfeit eligibility for rates under this plan and will be moved to Business MTS unless an alternative plan is selected by the Customer. If, during a billing cycle, the Customer has more than 10 calls to any single number of a duration of more than 3 hours each or any single call of greater than 10 hours of duration, the Customer will be presumed to be in violation of these restrictions; in such case, Customer may be asked to provide reasonable proof to the Company that it is not using the Service for a prohibited purpose, and failure to provide such proof to the Company shall be treated as a violation of the terms and conditions of this plan.
- .3 If the Company determines that the Customer's use of service violates any of these Restrictions, the Company, at its sole option, may move the Customer to another plan, or suspend, restrict or cancel the Customer's service without prior notice.

SECTION 4 – BUSINESS DOMESTIC SERVICE DESCRIPTIONS, RATES AND CHARGES

### 4.3 Business Optional Calling Plans (continued)

## 4.3.4 AT&T Business Unlimited Calling<sup>SM</sup> II (continued)

## (G) Call Detail Suppression

Call Detail is available as an optional feature to this plan. For an additional MRC, Customers who select this option will receive itemized call detail of their zero-rated calls. Activation and deactivation of the feature will begin within the next billing cycle following a Customer's request. Customers may add or remove the Call Detail optional feature from this plan at no charge.

Original Page 10

Effective: September 1, 2019

### (H) Rates and Charges

### .1 Direct Dialed Outbound Service MRC

Number of Lines	MRC
1	\$15
2	\$30
3	\$45
4	\$60
5	\$75
6	\$90
7	\$105
8	\$120
9	\$135
10	\$150

## .2 Toll-Free Service

The per minute usage rate is \$0.0600. All toll-free inbound calls are billed in six (6) second increments after an initial period, for billing purposes, of thirty (30) seconds. See Section 4.4 of this pricing and service guide for the AT&T Long Distance Toll Free<sup>SM</sup> MRC.

## .3 Call Detail MRC

Number of Lines	MRC
1	\$5
2	\$10
3	\$15
4	\$20
5	\$25
6	\$30
7	\$35
8	\$40
9	\$45
10	\$50

SECTION 4 – BUSINESS DOMESTIC SERVICE DESCRIPTIONS, RATES AND CHARGES

- 4.3 Business Optional Calling Plans (continued)
  - 4.3.5 AT&T Business Block of Time<sup>SM</sup> II Plans
    - (A) AT&T Business Block of Time<sup>SM</sup> II plans are bundled intrastate/interstate outbound calling and inbound long distance calling plans.
    - (B) AT&T Business Block of Time<sup>SM</sup> II plans are available to new or existing Business Customers who:
      - .1 utilize Switched Access to reach the long distance network for outbound calling and to receive calls from the long distance network for AT&T Long Distance Toll Free<sup>SM</sup> Service;

Original Page 11

- .2 subscribe to access line from an Affiliated ILEC of the Company that operates in whole or in part as an AT&T ILEC;
- .3 subscribe to the Company for interLATA (interstate and intrastate) and intraLATA (local toll) Service for outbound long distance calling on each access line, and may also subscribe to the Company's AT&T Long Distance Toll Free<sup>SM</sup> Service for inbound long distance calling;
- .4 specify at the time of ordering if the minutes are to be used for outbound calling, inbound toll-free calls, or both; and,
- .5 request to be provisioned under this plan.
- (C) For a specific MRC the Customer receives a specific amount (block) of time for placing (1+) Direct-Dialed domestic outbound long distance calling or for both (1+) Direct-Dialed domestic outbound long distance calling and domestic inbound switched AT&T Long Distance Toll Free<sup>SM</sup> Service calling. All usage in excess of the selected block of time will be billed at a fixed rate per minute. Calls are billed in increments of one (1) second subject to a minimum billed connect time (initial period) of thirty (30) seconds. Any minutes not used in a billing cycle will not be carried over to the next billing cycle. No credits will be given for any unused minutes. Term plan discounts are not applicable to this Service.
- (D) To receive toll-free service (inbound) Customers must subscribe to the AT&T Long Distance Toll Free<sup>SM</sup> Service. The terms and conditions and rates and charges of the AT&T Long Distance Toll Free<sup>SM</sup> Service will apply as described in Section 4.4 of this pricing and service guide except that usage rates are specified in Section 4.3.5 (G) herein.
- (E) The AT&T Business Block of Time<sup>SM</sup> II plan minutes cannot be used for calls to 900 services (or other similar calls to informational services), international calling, directory assistance or operator services.
- (F) AT&T Business Block of Time<sup>SM</sup> II plans will remain in effect until: 1) either cancelled or changed by the Customer, 2) until the Company no longer offers AT&T Business Block of Time<sup>SM</sup> II plans or 3) the Company notifies the Customer this plan is no longer available.

SECTION 4 – BUSINESS DOMESTIC SERVICE DESCRIPTIONS, RATES AND CHARGES

- 4.3 Business Optional Calling Plans (continued)
  - 4.3.5 AT&T Business Block of Time<sup>SM</sup> II Plans (continued)
    - (G) Rate Options

The following rate options provide the Customer with an xx-minute block of time for placing (1+) Direct-Dialed domestic outbound long distance calling or for both (1+) Direct-Dialed domestic outbound long distance calling and inbound toll-free calls. The MRC is shown for each rate option, along with the outbound intrastate/interstate and/or inbound toll-free calling per-minute usage rates for calls completed/or received after the xx-minute block of time has been used.

Original Page 12

AT&T Business Block of Time <sup>SM</sup> II Plans			
			Per-Minute
Plan Name	Minutes	MRC	Usage Rate
250	250	\$15.00	\$0.0600
700	700	\$35.00	\$0.0550
1200	1200	\$55.00	\$0.0470
2500	2500	\$105.00	\$0.0450
5000	5000	\$200.00	\$0.0410
7500	7500	\$275.00	\$0.0370
10,000	10,000	\$350.00	\$0.0360

SECTION 4 – BUSINESS DOMESTIC SERVICE DESCRIPTIONS, RATES AND CHARGES

- 4.3 Business Optional Calling Plans (continued)
  - 4.3.6 AT&T Business Calling Florida
    - (A) AT&T Business Calling Florida is a bundled intrastate/ interstate domestic outbound and domestic inbound long distance plan offered to business Customers on switched access lines.
    - (B) AT&T Business Calling Florida is available to new or existing Business Customers who:
      - .1 utilize Switched Access to reach the long distance network for outbound calling and to receive calls from the long distance network for AT&T Long Distance Toll Free<sup>SM</sup> Service;

Original Page 13

Effective: September 1, 2019

- .2 subscribe to and maintain a minimum of ten (10) access lines or access line equivalents located within the same LATA from an Affiliated ILEC that operates in whole or in part as an AT&T ILEC;
- .3 subscribe to the Company for the provision of interstate, intrastate intraLATA and intrastate interLATA Service;
- .4 have service and be located in Florida; and,
- .5 request to be provisioned under this plan.
- (C) For a specific MRC the Customer receives 10,000 minutes for placing both (1+) direct-dialed domestic outbound intrastate intraLATA long distance calling and domestic inbound intrastate intraLATA switched AT&T Long Distance Toll Free<sup>SM</sup> Service calling. Additional intrastate intraLATA usage over the initial 10,000 minutes, intrastate interLATA and interstate usage will be billed at a fixed rate per minute specified in Section 4.3.6 (F) herein. Calls are billed in increments of (1) second subject to a minimum billed connect time (initial period) of thirty (30) seconds. For partial invoice months the 10,000 minutes are prorated. Any minutes not used in a billing cycle will not be carried over to the next billing cycle. No credits will be given for any unused minutes.
- (D) This plan is not available for resale, broadcast facsimile services, foreign exchange service, public telephone services, for use with auto-dialers or any similar type of device or equivalents of any such services, or Customer lines associated with education institutions (colleges, universities, etc.) or other businesses that aggregate end user traffic. The 10,000 minutes cannot be used for calls to 900 services or other calls to access information services, directory assistance or operator services calls.
- (E) To receive toll-free (inbound) service Customers must subscribe to the AT&T Long Distance Toll Free<sup>SM</sup> Service. The provisions and rates and charges of the AT&T Long Distance Toll Free<sup>SM</sup> plan will apply as described in Section 4.4 of this pricing and service guide except that usage rates are specified in Section 4.3.6 (F) herein.
- (F) Rates and Charges

The Monthly Recurring Charge is \$20.00, and the Direct-Dial Outbound and Inbound Toll-Free per-minute rate for interstate calls, intrastate interLATA calls and intrastate intraLATA calls after the initial 10,000 minutes have been used is \$0.0650.

FC-19-0011

SECTION 4 – BUSINESS DOMESTIC SERVICE DESCRIPTIONS, RATES AND CHARGES

- 4.3 Business Optional Calling Plans (continued)
  - 4.3.7 AT&T Business Block of Time<sup>SM</sup> III Plans
    - (A) AT&T Block of Time<sup>SM</sup> III plans are bundled intrastate/interstate outbound calling and optional inbound Switched TFS long distance calling plans.
    - (B) Unless otherwise specified in the rate options defined in Section 4.3.3 (G), AT&T Block of Time<sup>SM</sup> III plans are available to new or existing Business Customers who:
      - .1 meet one of the following: (a) have previously subscribed to local dial tone Service of an Affiliated ILEC of the Company and have cancelled that Service; or (b) are currently a local telephone customer of a competitor in the local serving territory of one of the Affiliated ILECs<sup>1</sup> of the Company and are now moving dial tone services from the competitor to the Affiliated ILEC<sup>1</sup>;

Original Page 14

- .2 utilize Switched Access to reach the long distance network for outbound calling and to receive calls from the long distance network for TFS;
- .3 subscribe to access line from an Affiliate of the Company that operates in whole or in part as an AT&T ILEC;
- .4 subscribes to the Company for interLATA (interstate and intrastate) and intraLATA (local toll) Service for outbound long distance calling on each access line, and may also subscribe to the Company's TFS for inbound long distance calling;
- specify at the time of ordering if the MOUs are to be used for outbound calling, or both outbound and inbound TFS calling; and,
- .6 request to be provisioned under this plan.
- (C) Customers may subscribe to an AT&T Block of Time<sup>SM</sup> III rate option for outbound Service only or for both outbound and inbound TFS. Where a Customer subscribes to TFS, inbound toll-free calls may originate on any type of access and are terminated via Switched Access to the Customer's location. See Section 4.4 and Section 4.5 of this pricing and service guide for optional features, rules and regulations, and general information regarding TFS.
- (D) For a specific MRC the Customer receives a specific amount (block) of time for placing (1+) Direct-Dialed domestic outbound long distance calling or for both (1+) Direct-Dialed domestic outbound calling and inbound switched TFS calls. All usage in excess of the selected block of time will be billed at a fixed rate per minute. Calls are billed in increments of one (1) second subject to a minimum connect time (initial period) of thirty (30) seconds. Any minutes not used in a billing cycle will not be carried over to the next billing cycle. Credits will not be given for any unused minutes.
- (E) Term plan discounts are not applicable to this Service.
- (F) The AT&T Business Block of Time<sup>SM</sup> plan minutes cannot be used for calls to 900 services (or other similar calls to informational services), international calling, directory assistance or operator services.

<sup>&</sup>lt;sup>1</sup> Not including where that Affiliate is operating as a CLEC.

SECTION 4 – BUSINESS DOMESTIC SERVICE DESCRIPTIONS, RATES AND CHARGES

- 4.3 Business Optional Calling Plans (continued)
  - 4.3.7 AT&T Business Block of Time<sup>SM</sup> III Plans (continued)
    - (G) Rate Options

The following MRC and Additional Per-Minute Rate will apply based on the rate option selected and the number of months that Customer has subscribed to this rate option (Months of Service).

Original Page 15

Effective: September 1, 2019

The following rate options provide the Customer with an *xx-minute* block of time for placing (1+) Direct-Dialed domestic outbound long distance calling or for both (1+) Direct-Dialed domestic outbound calling and inbound switched TFS calls. The MRC is shown for each rate option, along with the per-minute rate that will apply after the *xx-minute* block of time has been used.

AT&T Business Block of Time <sup>SM</sup> III Plans				
Plan Name	Minutes Allotted in MRC	Months in Service	MRC	Additional Per-Minute Rate Over Allotment
700	700	1 to12 Months	\$30.00	\$0.0550
700	700	13 Months or Greater	\$35.00	\$0.0550
1,200	1,200	1 to12 Months	\$50.00	\$0.0470
1,200	1,200	13 Months or Greater	\$55.00	\$0.0470
2,500	2,500	1 to12 Months	\$96.00	\$0.0450
2,300	2,300	13 Months or Greater	\$105.00	\$0.0450
5,000	5,000	1 to12 Months	\$184.00	\$0.0410
5,000	5,000	13 Months or Greater	\$200.00	\$0.0410

SECTION 4 – BUSINESS DOMESTIC SERVICE DESCRIPTIONS, RATES AND CHARGES

4.3 Business Optional Calling Plans (continued)

### 4.3.8 All for Less® Unlimited

(A) This plan is an outbound calling and optional inbound Switched Toll Free Service (TFS) long distance calling plan. This plan is established at the BTN level. Aggregation of multiple BTNs is not available with this plan. If the Customer or Applicant selects a different long distance calling plan for specific WTNs, the Customer is required to establish a separate BTN for each variation. This plan is available to Business Customers in Alabama, Florida, Georgia, Kentucky, Louisiana, Mississippi, North Carolina, South Carolina and Tennessee.

Original Page 16

- (B) This plan is available to Business Customers who:
  - .1 utilize Switched Access to reach the long distance network for outbound calling and optionally, to receive calls from the long distance network for inbound TFS;
  - .2 subscribe to the Company for interLATA (interstate and intrastate) and intraLATA (local toll) Service for outbound long distance calling on each AT&T business access line per service location, and may also subscribe to the Company's TFS for inbound long distance calling;
  - .3 currently or newly subscribe to and maintain a minimum of one (1) but no more than ten (10) business access lines (*i.e.*, BTNs and WTNs) from an Affiliated ILEC of the Company (excluding where that Affiliate is operating as a CLEC) on a single BTN account per service location;
  - .4 subscribe to this plan only on business access lines subscribing to AT&T Business Local Calling<sup>SM</sup> from an Affiliated ILEC of the Company;
  - .5 commit to subscribe to this service for an initial one (1) year term agreement<sup>1</sup>; or under the non-term rate option;
  - .6 request to be provisioned under this plan; and
  - .7 meet one or more of the Qualification Options defined in Section 4.3.8 (C) of this pricing and service guide.

<sup>&</sup>lt;sup>1</sup> Effective September 1, 2017, the 1-Year term option is no longer available to new Customers. Existing Customers at the end of their current 1-Year term may request to renew this plan for a subsequent 1-Year term, or they may continue to subscribe to this plan on the out-of-term month-to-month basis, until they move locations and/or make changes to their service, whichever occurs first.

SECTION 4 – BUSINESS DOMESTIC SERVICE DESCRIPTIONS, RATES AND CHARGES

- 4.3 Business Optional Calling Plans (continued)
  - 4.3.8 All for Less® Unlimited (continued)
    - (C) Qualification Options

#### **OPTION 1**

Original Page 17

Effective: September 1, 2019

newly subscribe to a qualifying Internet service as defined and offered by an Affiliate of the Company

#### **OPTION 2**

- 1. newly subscribe to a qualifying wireless service as defined and offered by an Affiliated Wireless provider of the Company; and
- agree the Company may combine its wireless Affiliate billing with the Affiliated ILEC billing except when Customers are participating in the wireless Affiliate's SBS Wireless Deposit Waiver Program.

### **OPTION 3**

currently subscribe to Internet Service from an Affiliate of the Company and agree to upgrade current internet service to any 6Mbps or higher, Internet speed.

#### **OPTION 4**

- currently subscribe to: AT&T Business Unlimited Calling<sup>SM</sup>, AT&T Business Unlimited
   Calling<sup>SM</sup> III, AT&T Business Unlimited Calling<sup>SM</sup> IV, AT&T Business Unlimited Calling<sup>SM</sup> V,
   Business Unlimited Plan or All for Less Unlimited;
- 2. currently subscribe to AT&T Business Local Calling<sup>SM</sup> from an Affiliated ILEC of the Company;
- currently subscribe to Internet service as defined and offered by an Affiliated of the Company and/or wireless service as defined and offered by an Affiliated Wireless provider of the Company and/or TV service as defined and offered by an Affiliate of the Company.

### **OPTION 5**

- currently subscribe to and maintain at least one (1) but no more than ten (10) business access
  lines (i.e., BTNs and WTNs) from an Affiliated ILEC of the Company (excluding where that
  Affiliate is operating as a CLEC) on a single BTN account per service location and state an
  intention to change local carriers for economic reasons and/or in response to a competitive offer;
- currently subscribe to Internet service as defined and offered by an Affiliate of the Company and/or wireless service as defined and offered by an Affiliated Wireless provider of the Company and/or TV service as defined and offered by an Affiliate of the Company.

### **OPTION 6**

newly subscribe to qualifying TV service as defined and offered by an Affiliate of the Company.

SECTION 4 – BUSINESS DOMESTIC SERVICE DESCRIPTIONS, RATES AND CHARGES

- 4.3 Business Optional Calling Plans (continued)
  - 4.3.8 All for Less® Unlimited (continued)
    - (D) A single business entity with more than one BTN at the business entity's physical service location is eligible for this plan on one (1) BTN only, provided the total number of AT&T business access lines do not exceed ten (10) at that location. Additionally, service under this plan is limited to one BTN with one (1) to ten (10) business access lines per legal business entity.

Original Page 18

Effective: September 1, 2019

- (E) The Customer may subscribe to this plan for outbound Service only, or for both outbound service and inbound TFS. Customer should specify at the time of ordering if the plan is to be used for outbound calling, or both outbound and inbound TFS calling. Customers subscribing to the plan receive unlimited domestic 1+ direct-dialed outbound calling. Where a Customer subscribes to TFS, inbound toll-free calls may originate on any type of access and are terminated via Switched Access to the Customer's location. (See Section 4.4 and Section 4.5 of this pricing and service guide for optional features, rules, regulations, TFS MRC and general information regarding TFS).
- (F) TFS calls are billed in increments of six (6) seconds subject to a minimum connect time (initial period) of thirty (30) seconds. TFS per minute rates associated with this plan are subject to change with prior notification to the customer.
- (G) TFS calls, multi-party conference calls (except those calls placed by using the Three-Way Calling feature are included), calls to 900, 976, 700 numbers or other calls to access information services, directory assistance, operator services and international calling are not included in the unlimited minutes of use.
- (H) Call Detail is available as an optional feature to this plan. For an additional MRC, Customers who select this option will receive itemized call detail of their zero-rated calls. Activation and deactivation of the feature will begin within the next billing cycle following a Customer's request. Customers may add or remove the Call Detail optional feature from this plan at no charge.
- (I) This plan will remain in effect until either: (a) canceled or changed by the Customer; (b) the Customer terminates Service with the Company; (c) the Company no longer offers this plan to new Customers or to existing Customers moving to new locations or changing the number of access lines under the BTN(s) to which this plan applies; or (d) the Company notifies the Customer this plan is no longer available, whichever occurs first.
- (J) Term Agreement

At the end of a Customer's one (1) year term agreement, if the Customer does not resubscribe for a new one (1) year term, or cancel services, they may continue with this plan on an out of term month-to-month basis at the out of term month-to-month rates defined in Section 4.3.8 (M) of this pricing and service guide. Customers on out-of-term month-to-month rates may contact the Company if they wish to move to the non-term rate option.

### SECTION 4 – BUSINESS DOMESTIC SERVICE DESCRIPTIONS, RATES AND CHARGES

- 4.3 Business Optional Calling Plans (continued)
  - 4.3.8 All for Less® Unlimited (continued)
    - (K) Early Termination Fee
      - .1 Customers that subscribe to this plan combined with the one (1) year term agreement, and who wish to: (a) terminate long distance service with the Company: or (b) change its calling plan prior to the expiration of the Term Agreement, may be charged an early termination fee ("ETF").

The ETF shall be \$5.00 multiplied by the number of months remaining in the term.

Original Page 19

- .2 The Company will not charge an ETF:
  - .a Customer may change its calling plan to the All for Less<sup>®</sup> 500 as specified in Section 4.3 of this pricing and service guide, or the AT&T<sup>®</sup> Business Block of Time<sup>SM</sup> II Plan (rate option 700 minutes or greater) as specified in Section 4.3 of this pricing and service guide; in such case, ETF shall not apply.
  - .b if the Company no longer offers this plan, and if the Customer moves to a new location or changes the number of access lines under the BTN(s) to which this Plan applies then the Customer's service under the plan will terminate, and the Customer will not be subject to termination charges.
  - .c if Customer cancels this plan while under a term agreement and at the same time the Customer agrees to replace some or all their existing service with Internet Protocol (IP) service, Wireless, or any functionally equivalent service from an Affiliate of the Company for the purpose of placing outbound and/or inbound live voice communications outside of the Customer's local calling area.
  - .d when subscribing to this plan under the non-term rate option.

SECTION 4 – BUSINESS DOMESTIC SERVICE DESCRIPTIONS, RATES AND CHARGES

- 4.3 Business Optional Calling Plans (continued)
  - 4.3.8 All for Less® Unlimited (continued)
    - (L) Restrictions

This plan is provided for standard voice calling involving live dialog between individuals. This plan is not intended for use as a substitute for dedicated or open circuits, or similar applications. This plan may not be used for data connections, including but not limited to, modem to modem calls, remote access applications, Internet access, or Intranet access (including access to corporate LANs). Additionally, this plan may not be used for dedicated point-to-point connections between equipment that leaves a circuit connected without a contemporary, continuous voice communication.

Original Page 20

Effective: September 1, 2019

The plan may not be used for auto dialers; PBX trunks; ground start line or trunks; ISDN service, including PRI; foreign exchange services; Remote Call Forwarding/Telebranch Service; public telephone service; public access smart-pay phones; analog to digital conversion digital PBX service, including local access provisioned via T-1 facilities; WATS service; PBX/PABX/EABX services; non-square electronic key telephone systems; hybrid key telephone systems; predictive calling/dialing systems; automatic outbound dialing systems; any type of automatic call distribution system; or the functional equivalent of any such systems listed above. This plan may be used for fax transmissions, excluding broadcast fax applications (which are prohibited).

If the Company determines that the Customer is in violation of any of the restrictions or terms and conditions of this plan as specified above, the Customer will forfeit eligibility for rates under this plan and will be moved to AT&T® Business Calling<sup>SM</sup> Monthly unless an alternative plan is selected by the Customer. If moved to AT&T® Business Calling<sup>SM</sup> Monthly, the rates associated with AT&T® Business Calling<sup>SM</sup> Monthly specified in Section 4.3.3 of this pricing and service guide will apply in lieu of the rates specified in Section 4.3.8 (M) of this pricing and service guide.

If, during a billing cycle, the Customer has more than 10 calls to any single number of a duration of more than 3 hours each or any single call of greater than 10 hours of duration, the Customer will be presumed to be in violation of these restrictions. In such case, Customer may be asked to provide reasonable proof to the Company that it is not using the Service for a prohibited purpose. Failure to provide such proof to the Company shall be treated as a violation of the terms and conditions of this Plan.

SECTION 4 – BUSINESS DOMESTIC SERVICE DESCRIPTIONS, RATES AND CHARGES

- 4.3 Business Optional Calling Plans (continued)
  - 4.3.8 All for Less® Unlimited (continued)
    - (M) Rates and Charges

The MRC and TFS usage rates are as follows:

Customer Commitment	MRC (Per Access Line)	Inbound (TFS) Per Minute Usage Rate
1 Year Term <sup>1</sup>	\$12.50	\$0.055
Out-of-Term Month-to-Month <sup>1</sup>	\$12.50	\$0.055
Non-Term Option <sup>2</sup>	\$5.00	\$0.055

.1 Existing Customers with a 1-Year Term Agreement prior to June 1, 2015 will receive a monthly bill credit reward amount equal to \$2.50 per line for the duration of the 1-Year Term Agreement. Effective June 1, 2015, new Customers or existing Customers who renew their All for Less® Unlimited plan for a subsequent 1-year term on or after June 1, 2015 will receive a monthly bill credit reward amount equal to \$7.50 per line for the duration of the 1-Year Term Agreement.

During the 1-Year Term, Federal Universal Service Fee, Federal Regulatory Fee, taxes and other similar fees and surcharges will be calculated on the MRC prior to the calculation of the monthly bill credit reward amount.

Original Page 21

- .2 See Section 4.4 of this pricing and service guide for the AT&T Long Distance Toll Free<sup>SM</sup> Service MRC.
- .3 Call Detail: The MRC is \$5.00 per line up to ten (10) lines.

<sup>&</sup>lt;sup>1</sup> Effective September 1, 2017, the 1-Year term option is no longer available to new Customers. Existing Customers at the end of their current 1-Year term may request to renew this plan for a subsequent 1-Year term, or they may continue to subscribe to this plan on the out-of-term month-to-month basis, until they move locations and/or make changes to their service, whichever occurs first.

<sup>&</sup>lt;sup>2</sup> Rate applicable to new Customers on or after September 1, 2017.

SECTION 4 – BUSINESS DOMESTIC SERVICE DESCRIPTIONS, RATES AND CHARGES

### 4.3 Business Optional Calling Plans (continued)

### 4.3.9 All for Less<sup>®</sup> 500

- (A) This plan is an outbound calling and optional inbound Switched Toll Free Service (TFS) long distance calling plan. This plan is established at the BTN level. Aggregation of multiple BTNs is not available with this plan. If the Customer or Applicant selects a different long distance calling plan for specific WTNs the Customer is required to establish a separate BTN for each variation. This plan is available to Business Customers in Alabama, Florida, Georgia, Kentucky, Louisiana, Mississippi, North Carolina, South Carolina and Tennessee.
- (B) This plan is available to Business Customers who:
  - .1 utilize Switched Access to reach the long distance network for outbound calling and optionally, to receive calls from the long distance network for inbound TFS;

Original Page 22

Effective: September 1, 2019

- .2 subscribe to the Company for interLATA (interstate and intrastate) and intraLATA (local toll) Service for outbound long distance calling on each AT&T business access line per service location, and may also subscribe to the Company's TFS for inbound long distance calling;
- .3 currently or newly subscribe to access line(s) from an Affiliated ILEC of the Company (excluding where that Affiliate is operating as a CLEC);
- .4 subscribe to this plan only on business access lines subscribing to AT&T Business Local Calling<sup>SM</sup> from an Affiliated ILEC of the Company;
- .5 commit to subscribe to this service for an initial one (1) year term agreement;
- .6 request to be provisioned under this plan; and
- .7 meet one or more of the Qualification Options defined in Section 4.3.9 (C), herein.

## (C) Qualification Options

### **OPTION 1**

Newly subscribe to a qualifying Internet service as defined and offered by an Affiliate of the Company.

## **OPTION 2**

- newly subscribe to a qualifying wireless service as defined and offered by an Affiliated Wireless provider of the Company; and
- agree the Company may combine its wireless Affiliate billing with the Affiliated ILEC billing except when Customers are participating in the wireless Affiliate's SBS Wireless Deposit Waiver Program.

## **OPTION 3**

Currently subscribe to Internet Service from an Affiliate of the Company and agree to upgrade current internet service to any 6Mbps or higher, Internet speed.

#### **OPTION 4**

- 1. Currently subscribe to business access line(s) from an Affiliated ILEC of the Company (excluding where that Affiliate is operating as a CLEC) and state an intention to change local carriers for economic reasons and/or in response to a competitive offer; and
- 2. currently subscribe to Internet service as defined and offered by an Affiliate of the Company and/or wireless service provided by an Affiliate Wireless provider of the Company.

SECTION 4 – BUSINESS DOMESTIC SERVICE DESCRIPTIONS, RATES AND CHARGES

### 4.3 Business Optional Calling Plans (continued)

### 4.3.9 All for Less® 500 (continued)

(D) Customers may subscribe to this plan for outbound Service only or for both outbound service and inbound TFS. Customer should specify at the time of ordering if the plan is to be used for outbound calling, or both outbound and inbound TFS calling. Where a Customer subscribes to TFS, inbound toll-free calls may originate on any type of access and are terminated via Switched Access to the Customer's location. (See Section 3, Section 4.4 and Section 4.5 and of this pricing and service guide for optional features, rules, regulations, TFS MRC and general information regarding TFS)

Original Page 23

Effective: September 1, 2019

- (E) This plan provides the Customer with a block of 500 minutes for placing (1+) Direct-Dialed domestic outbound long distance calling or for both (1+) Direct-Dialed domestic outbound calling and inbound switched TFS calls. All usage in excess of the 500 minute block will be billed at a fixed rate per minute. Any minutes not used in a billing cycle will not be carried over to the next billing cycle. No credits will be given for any unused minutes. Calls are billed in increments of one (1) second subject to a minimum connect time (initial period) of thirty (30) seconds.
- (F) This plan will remain in effect until either: (a) canceled or changed by the Customer; (b) the Customer terminates Service with the Company; (c) the Company no longer offers this plan to new Customers or to existing Customers moving to new locations or changing the number of access lines under the BTN(s) to which this plan applies; or (d) the Company notifies the Customer this plan is no longer available, whichever occurs first.

### (G) Term Agreement

At the end of a Customer's one (1) year term agreement, if the Customer does not resubscribe for a new one (1) year term, or cancel services, they may continue with this plan on an out of term month-to-month basis at the out of term month-to-month rates defined in Section 4.3.9 (I) of this pricing and service guide.

SECTION 4 – BUSINESS DOMESTIC SERVICE DESCRIPTIONS, RATES AND CHARGES

- 4.3 Business Optional Calling Plans (continued)
  - 4.3.9 All for Less® 500 (continued)
    - (H) Early Termination Fee
      - .1 Customers that subscribe to this plan combined with the one (1) year term agreement, and who wish to: (a) terminate long distance service with the Company: or (b) change its calling plan prior to the expiration of the Term Agreement, may be charged an early termination fee ("ETF").

The ETF shall be \$5.00 multiplied by the number of months remaining in the term.

Original Page 24

Effective: September 1, 2019

- .2 The Company will not charge an ETF:
  - .a Customer may change its calling plan to the All for Less<sup>®</sup> 500 Unlimited as specified in Section 4.3.8 of this pricing and service guide, or the AT&T<sup>®</sup> Business Block of Time<sup>SM</sup> II Plan (rate option 700 minutes or greater) as specified in Section 4.3.5 of this pricing and service guide; in such case, ETF shall not apply.
  - .b If the Company no longer offers this plan, and if the Customer moves to a new location or changes the number of access lines under the BTN(s) to which this Plan applies then the Customer's service under the plan will terminate, and the Customer will not be subject to termination charges.
  - .c The Company will not charge an ETF if Customer cancels this plan while under a term agreement if at the same time the Customer agrees to replace some or all of their existing service with Internet Protocol (IP) service, Wireless, or any functionally equivalent service from an Affiliate of the Company for the purpose of placing outbound and/or inbound live voice communications outside of the Customer's local calling area.
- (I) Rates and Charges

The MRC and usage rates are as follows:

Customer Commitment	MRC	Minutes Allotted in MRC	Additional Per-Minute Rate Over Allotment
1-Year Term	\$28.00	500	\$0.057
Out-of-Term Month-to-Month	\$28.00	500	\$0.057

- .1 Customers will receive a monthly bill credit reward amount equal to \$3.00 per month for the duration of the 1-Year Term Agreement. During the 1-Year Term, Federal Universal Service Fee, Federal Regulatory Fee, taxes and other similar fees and surcharges will be calculated on the MRC prior to the calculation of the monthly bill credit reward amount.
- .2 See Section 4.4 of this Business Service Guide for the AT&T Long Distance Toll Free<sup>SM</sup> Service MRC.

Original Page 25

Effective: September 1, 2019

SECTION 4 – BUSINESS DOMESTIC SERVICE DESCRIPTIONS, RATES AND CHARGES

- 4.4 AT&T Long Distance Toll Free<sup>SM</sup> Service (Formerly known as Business Easy Toll Free Plan)
  - 4.4.1 AT&T Long Distance Toll Free<sup>SM</sup> is an inward toll-free (8xx) calling service offered to business Customers from switched access lines. Service is offered on a flat rate basis, twenty-four hours per day, seven days per week. This Service permits termination of calls from diverse geographic locations to the Customer's local exchange lines. With this Service, the Customer is billed for the call rather than the call originator. The Customer will be billed for all calls terminating at the toll-free number, including misdialed calls. AT&T Long Distance Toll Free<sup>SM</sup> Service is not available for terminations to wireless telecommunications devices. Toll-free number orders completed on or after January 3, 2006 require subscription to a domestic optional calling (non-MTS) plan.

To be eligible for this Service, Customers must subscribe to an access line provided by the Company's Affiliated ILEC.

Toll-free usage under this Service is subject to the rates, billing increments, term and/or volume discounts, monthly usage commitments, and monthly recurring charge waiver provisions of the associated optional calling plan subscribed to by the Customer. For package plan customers, toll-free usage will be included in package minutes. The Company will automatically adjust toll-free rates and plan provisions to reflect changes in business service subscriptions. The Business MTS per-minute rate specified in Section 4.2 of this pricing and service guide will apply to Customers who discontinue their domestic optional calling plan but wish to retain AT&T Long Distance Toll Free<sup>SM</sup> Service. AT&T Long Distance Toll Free<sup>SM</sup> Customers who receive the Business MTS service and rate must continue to maintain an access line provided by the Company's Affiliated ILEC.

4.4.2 Optional Features and Services

Optional features and services are available for this plan. The terms, conditions and charges for these features and services are identified in Sections 3.1.5 and 4.5 of this pricing and service guide.

4.4.3 The MRC is \$18.00.

SECTION 4 – BUSINESS DOMESTIC SERVICE DESCRIPTIONS, RATES AND CHARGES

- 4.5 Business Optional Features and Services
  - 4.5.1 Optional features and services are available to Business Customers subscribing to domestic Business Services. Call charges will be billed based on the calling plan selected by the Customer.

The following non-recurring and monthly charges apply per billing account for domestic outbound (1+) and domestic toll-free optional features and services.

Original Page 26

Effective: September 1, 2019

(A) Outbound Services

.1	Feature Non-Recurring Installation Charge:	\$25.00
.2	Feature Monthly Recurring Charge:	\$10.00

(B) Inbound Service (Toll-Free)

.1	Feature Non-Recurring Installation Charge:	\$25.00
.2	Feature Monthly Recurring Charge:	\$5.00

SECTION 4 – BUSINESS DOMESTIC SERVICE DESCRIPTIONS, RATES AND CHARGES

#### 4.5 Business Optional Features and Services (continued)

### 4.5.2 AT&T NetPROTECT

AT&T NetPROTECT capabilities enables Customer to reduce the risk of Remote Toll Fraud.

### (A) Definitions

Covered Calls – International Outbound Long Distance Direct Dial calls from a CTS, international Operator Assistance calls from a CTS billed to the CTS location, and domestic toll-free calls to a CTS.

Original Page 27

Effective: September 1, 2019

Customer Telephone System (CTS) – a Private Branch Exchange (PBX) or electronic key system or a Customer Site served by Centrex service that is (a) owned or leased (or in the case of Centrex, subscribed to) by Customer, (b) subject to Customer's direct control, (c) located in the U.S. Mainland, Hawaii, Puerto Rico or the U.S. Virgin Islands, and (d) not located in a Licensed Space Arrangement.

Basic CTS – a CTS covered by NetPROTECT Basic.

Plus CTS – a CTS certified by AT&T to be covered by NetPROTECT Plus.

Premium CTS – a CTS certified by AT&T to be covered by NetPROTECT Premium (a Centrex location may not be a Premium CTS).

Remote Toll Fraud – unauthorized Covered Calls made by remotely accessing a CTS, but not including calls made by, in collusion with, or as the result of the intentional acts (including the provision of information or assistance to unauthorized parties) of, Customer or its past or present employees, agents, vendors or independent contractors. A series of such unauthorized Covered Calls that arise from the same instance of a breach of CTS security comprise a single Remote Toll Fraud incident.

AT&T Notification – a telephone call by AT&T to Customer to notify Customer of suspected Remote Toll Fraud.

Customer Notification – a telephone call by Customer to the AT&T NetPROTECT Fraud Monitoring Center at 1-800-821-8235 (or such other telephone number designated by AT&T) to notify AT&T of suspected Remote Toll Fraud.

#### (B) General NetPROTECT Requirements

Customer must immediately make a Customer Notification when it detects suspected Remote Toll Fraud. Customer will cooperate reasonably and promptly with AT&T's investigation of suspected Remote Toll Fraud, including providing a written description of the means by which the Remote Toll Fraud occurred (if known), the changes made to the CTS to stop the Remote Toll Fraud, and any password(s) used to remotely access the CTS.

To make a claim under an AT&T NetPROTECT liability cap or liability waiver for Remote Toll Fraud not subject to AT&T Notification, Customer must (a) identify in writing any calls claimed to be Remote Toll Fraud within six months after the date of the AT&T bill that contains charges for such calls, and (b) establish (e.g., by way of call detail, station message detail recording system configuration documentation, etc.) that each call is Remote Toll Fraud. To the extent the AT&T NetPROTECT liability cap or liability waiver reduces or eliminates any charges for which Customer otherwise would be liable, AT&T will have the right to pursue, on its own behalf, any claims Customer may have against third parties arising in connection with the associated Remote Toll Fraud.

SECTION 4 – BUSINESS DOMESTIC SERVICE DESCRIPTIONS, RATES AND CHARGES

### 4.5 Optional Features and Services (continued)

### 4.5.2 AT&T NetPROTECT (continued)

### (C) AT&T NetPROTECT Basic

AT&T NetPROTECT Basic Service is available to all Customers except those who are acting in the capacity of a reseller or wholesaler. AVOICS customers, IP Toll Free, Remote Worker without Secure Voice Tunnel Option, customers who provide Audio and/or Video Teleconference service, and Prepaid Card Providers are not covered by AT&T NetPROTECT Basic.

Original Page 28

Effective: September 1, 2019

AT&T NetPROTECT Basic applies (at no charge) with respect to Covered calls to or from a Basic CTS, unless Customer subscribes to AT&T NetPROTECT Premium or AT&T NetPROTECT Plus. Under AT&T NetPROTECT Basic, Customer's liability is capped at \$25,000 for charges for Remote Toll Fraud calls that commence before an AT&T Notification or a Customer Notification. The liability cap applies for a one time only Remote Toll Fraud incident, based on the combined charges at all Basic CTSs affected by such incident.

If Customer makes a Customer Notification with respect to a Remote Toll Fraud incident before AT&T makes an AT&T Notification, the liability cap will be reduced by 50% with respect to that incident. To be eligible for such liability cap reduction, Customer must (within six months after the Customer Notification) provide a written explanation of the means used to detect, and the changes made to the CTS to stop, the Remote Toll Fraud.

Customer is liable for all charges for Remote Toll Fraud calls that commence after an AT&T Notification or a Customer Notification.

### (D) AT&T NetPROTECT Plus

AT&T NetPROTECT Plus Service is available to all Customers except those who are acting in the capacity of a reseller or wholesaler. AVOICS customers, IP Toll Free, Remote Worker without Secure Voice Tunnel Option, customers who provide Audio and/or Video Teleconference service, and Prepaid Card Providers are not covered by AT&T NetPROTECT Plus.

AT&T NetPROTECT Plus is a value-added Remote Toll Fraud protection and notification plan that applies with respect to Covered Calls to or from a Plus CTS. When AT&T certifies Customer's CTS as a Plus CTS, Customer's liability is initially capped at \$2,000 for charges for Remote Toll Fraud calls that commence before an AT&T Notification or a Customer Notification. The liability cap applies once across all Plus CTSs affected by each Remote Toll Fraud incident. After each Remote Toll Fraud incident, the liability cap will be increased by \$2,000.

If Customer makes a Customer Notification with respect to a Remote Toll Fraud incident before AT&T makes an AT&T Notification, the liability cap will be reduced by 50% with respect to that incident. To be eligible for such liability cap reduction, Customer must (within six months after the Customer Notification) provide a written explanation of the means used to detect, and the changes made to the CTS to stop, the Remote Toll Fraud.

SECTION 4 – BUSINESS DOMESTIC SERVICE DESCRIPTIONS, RATES AND CHARGES

- 4.5 Optional Features and Services (continued)
  - 4.5.2 AT&T NetPROTECT (continued)
    - (D) AT&T NetPROTECT Plus (continued)

When suspected Remote Toll Fraud is detected, AT&T will attempt to make an AT&T Notification and assist Customer's efforts to stop the suspected Remote Toll Fraud. AT&T will block all Covered Calls out of the Plus CTS until the problem has been isolated and corrected if Customer gave AT&T written authorization for such blocking at the time of service implementation. At any time during a suspected Remote Toll Fraud incident, AT&T will block all or a portion of Covered Calls out of the Plus CTS, at Customer's request (if Customer did not provide written authorization for such blocking prior to the incident, Customer must send AT&T a written confirmation of such authorization within three hours after a verbal request). AT&T will restore blocked service at any time during the incident at Customer's request.

Original Page 29

Effective: September 1, 2019

Customer is liable for all charges for Remote Toll Fraud calls that commence after an AT&T Notification or a Customer Notification. If a Remote Toll Fraud incident triggers application of the liability cap, the affected CTS will not be covered again by the liability cap until thirty (30) consecutive days have passed without subsequent Remote Toll Fraud incident affecting the same CTS.

CUSTOMER must provide the following information to the AT&T NetPROTECT Service Establishment Group to be eligible for the liability cap:

- a written list of all telephone numbers, both outbound and toll-free (if applicable), assigned to each Plus CTS location (the liability cap will not apply for usage associated with numbers not included on such list)
- a completed Customer profile form for each Plus CTS, describing the type of CTS, service configuration, calling volumes and patterns to foreign destinations
- current contact information (*e.g.*, telephone number, electronic mail address, pager and PIN number, wireless telephone number and/or facsimile number) at which an authorized Customer representative can be contacted 24 hours per day, 7 days per week (such representative shall have the authority to take necessary action to stop suspected Remote Toll Fraud affecting the Plus CTS)
- immediate written updates to the information provided under this Section, signed by an authorized Customer representative
- description of any change to Plus CTS that could affect the security of the Plus CTS (to be provided before the change is implemented)

After the NetPROTECT Service Establishment Group has determined that Customer has satisfied all NetPROTECT Plus requirements, it will provide a CTS Certification package to Customer, which package will identify the service effective date that the specified CTS will be considered a Plus CTS. If the NetPROTECT Service Establishment Group determines that Customer has failed to satisfy the NetPROTECT Plus requirements, the NetPROTECT Service Establishment Group will advise Customer of the requirement(s) that have not been satisfied.

SECTION 4 - BUSINESS DOMESTIC SERVICE DESCRIPTIONS, RATES AND CHARGES

### 4.5 Business Optional Features and Services (continued)

### 4.5.2 AT&T NetPROTECT (continued)

### (D) AT&T NetPROTECT Plus (continued)

Customer must notify the AT&T NetPROTECT Service Establishment Group in writing thirty (30) days prior to its date of discontinuance. If Customer replaces AT&T NetPROTECT Plus with AT&T NetPROTECT Premium, the non-recurring charge for installation of AT&T NetPROTECT Premium will be reduced by the non-recurring charges paid by Customer for installation of AT&T NetPROTECT Plus.

Original Page 30

Effective: September 1, 2019

## (E) AT&T NetPROTECT Premium

AT&T NetPROTECT Premium Service is available to all Customers except those who are acting in the capacity of a reseller or wholesaler. AVOICS customers, IP Toll Free, Remote Worker without Secure Voice Tunnel Option, customer who provide Audio and/or Video Teleconference service, and Prepaid Card Providers are not covered by NetPROTECT Premium.

AT&T NetPROTECT Premium is a value-added Remote Toll Fraud protection and notification plan that applies with respect to Covered Calls to or from a Premium CTS. When AT&T certifies Customer's CTS as a Premium CTS, Customer is not liable for charges for Remote Toll Fraud calls that commence during the Protected Period for that Premium CTS.

The Protected Period for a Premium CTS is the period from the commencement of AT&T NetPROTECT Premium protection for the Premium CTS through two hours after any AT&T Notification or Customer Notification involving the Premium CTS. When the Protected Period had terminated as a result of an AT&T Notification or a Customer Notification, it will not recommence for the Covered Premium CTS in question until seven days after such termination (or until seven days after a subsequent AT&T Notification or Customer Notification that would have terminated the Protected Period if the Protected Period had been in effect for that Premium CTS).

The Protected Period does not include any period during which Customer has (a) failed to satisfy any of its responsibilities related to the Premium CTS or (b) failed to implement a restriction recommended by AT&T during an active Remote Toll Fraud incident. Any Remote Toll Fraud that involves a Premium CTS and that occurs when the Protected Period is not in effect for that Premium CTS will also be deemed not to have occurred during the Protected Period for any other Premium CTS involved in the same Remote Toll Fraud incident (regardless of whether the Protected Period otherwise would have been in effect for such Premium CTS).

### SECTION 4 – BUSINESS DOMESTIC SERVICE DESCRIPTIONS, RATES AND CHARGES

- 4.5 Business Optional Features and Services (continued)
  - 4.5.2 AT&T NetPROTECT (continued)
    - (E) AT&T NetPROTECT Premium (continued)

Customer must comply with the following requirements to be eligible for the AT&T NetPROTECT Premium liability waiver:

• Customer must change all manufacturer's or vendor's default passwords on all Premium CTS equipment to a random password with a minimum of seven alphanumeric characters.

Original Page 31

- Customer must protect all network test units and all remote maintenance ports, on all Premium CTS equipment by (a) requiring human intervention for external access to all such network test units and remote maintenance ports, (b) installing a security device that uses password encryption or a callback device, or (c) requiring use of a password that complies with the following password requirements:
  - o Passwords must be changed at least every three months.
  - o Passwords cannot be used more than once.
  - O Customer must be able to produce accurate, dated records of all passwords used and to whom these passwords were distributed.
  - o Passwords must be a minimum of seven characters in length.
  - o Passwords must be comprised of a minimum of three letters and a minimum of three numbers and/or special characters (such as #, !, or \*).
  - o Passwords may not include sequential or repetitive numbers or letters that form common English words.
- Customer must block or prevent all call transfer capability external to each
  Covered Premium CTS by voice mail and auto attendant systems on each
  Covered Premium CTS except call transfer capability to another Cover Premium
  CTS.
- Customer must protect Direct Toll-Free System Access on each Covered Premium CTS by: (a) requiring human intervention for all external access, (b) installing a security device that uses password encryption or a callback device, or (c) utilizing single string security codes containing at least seven digits plus a secondary code of at least 4 digits (or another combination of codes that provides less than a 1 in 12,000 chance of dialing an authorized code number on a single, randomly dialed attempt).
- Customer must deactivate all trunk verification codes on each Premium CTS.
- Customer must provide a written list of all telephone numbers, both outbound and toll-free (if applicable), assigned to each Premium CTS location (the liability waiver will not apply for usage associated with numbers not included on such list).
- Customer must provide a written list of the PBX/Key System Model and Version, PBX/Key System Vendor and Maintenance Contact Number(s), Voice Mail Model, and Auto Attendant Model, as applicable.

SECTION 4 – BUSINESS DOMESTIC SERVICE DESCRIPTIONS, RATES AND CHARGES

- 4.5 Business Optional Features and Services (continued)
  - 4.5.2 AT&T NetPROTECT (continued)
    - (E) AT&T NetPROTECT Premium (continued)

Customer must comply with the following requirements to be eligible for the AT&T NetPROTECT Premium liability waiver: (continued)

Original Page 32

- Customer must provide a completed Customer profile form for each Premium CTS, describing the type of CTS, service configuration, calling volumes and patterns to foreign destinations.
- Customer must provide current contact information (*e.g.*, current telephone number, electronic mail address, pager and PIN number, wireless telephone number and/or facsimile number) at which an authorized Customer representative can be contacted 24 hours per day, 7 days per week (such representative shall have the authority to take necessary action to stop suspected Remote Toll Fraud affecting the Premium CTS).
- Customer must provide a completed network security certification form as provided by the AT&T NetPROTECT Service Establishment Group.
- Customer must provide immediate written updates to the information provided under this Section, signed by an authorized Customer representative.
- Customer must provide description of any change to Premium CTS that could affect the security of the Premium CTS (to be provided before the change is implemented).
- Customer must obtain all Toll-Free calling capability associated with a Premium CTS from AT&T.
- If the Premium CTS is connected by private line to any other CTS, such other CTS must also be a Premium CTS (except that a Premium CTS may be connected by private line to a CTS not located within the U.S. Mainland, Hawaii, Puerto Rico or the U.S. Virgin Islands, in which case the liability waiver does not apply to any usage charges for Remote Toll Fraud that accessed a Premium CTS on such private line).
- Customer must allow AT&T to connect Customer with Customer's CTS vendor or a qualified external organization that can work remotely with Customer to eliminate Remote Toll Fraud.
- Customer must either purchase or perform an annual security review of its
  equipment and including all adjuncts and incoming and outgoing trunks to
  determine what security measures are required to better secure Customer's
  network. Customer must provide a copy of the results of this review to the
  AT&T NetPROTECT Fraud Management Center for review. Customer must
  implement security recommendations of the AT&T NetPROTECT Fraud
  Management Center resulting from this review.

SECTION 4 - BUSINESS DOMESTIC SERVICE DESCRIPTIONS, RATES AND CHARGES

- 4.5 Business Optional Features and Services (continued)
  - 4.5.2 AT&T NetPROTECT (continued)
    - (E) AT&T NetPROTECT Premium (continued)

Customer must comply with the following requirements to be eligible for the AT&T NetPROTECT Premium liability waiver: (continued)

Original Page 33

- Customer must permit AT&T to review Customer's equipment configuration
  upon demand, on a random or spot check basis (no more than twice a year per
  CTS location). If Remote Toll Fraud occurs, another CTS review will be
  performed immediately to determine the cause of the Remote Toll Fraud.
- Customer must permit an AT&T representative to enter and conduct a contract compliance review of any Premium CTS location if requested to investigate suspected Remote Toll Fraud.
- After the NetPROTECT Service Establishment Group has determined that Customer has satisfied all NetPROTECT Premium requirements, it will provide a CTS Certification package to Customer, which package will identify the service effective date that the specified CTS will be considered a Premium CTS. If the NetPROTECT Service Establishment Group determines that Customer has failed to satisfy the NetPROTECT Premium requirements, the NetPROTECT Service Establishment Group will advise Customer of the requirement(s) that have not been satisfied.
- Customer must notify the AT&T Service Establishment Group in writing at least thirty (30) days prior to the date of discontinuance. If Customer replaces AT&T NetPROTECT Premium with AT&T NetPROTECT Plus, all non-recurring installation charges associated with AT&T NetPROTECT Plus will be waived.
- As new methods of Remote Toll Fraud become apparent, AT&T periodically will inform Customer of such new methods and the recommended remedial action required of Customer to protect all Premium CTS locations. If Customer does not comply with these recommendations, the NetPROTECT Premium Liability Waiver will not apply with respect to any charges associated with such new methods of Remote Toll Fraud. Additionally, if Customer does not secure all Covered Premium CTS equipment against the new methods of Remote Toll Fraud, AT&T will not be obligated to continue AT&T NetPROTECT Premium Service.

SECTION 4 – BUSINESS DOMESTIC SERVICE DESCRIPTIONS, RATES AND CHARGES

- 4.5 Business Optional Features and Services (continued)
  - 4.5.2 AT&T NetPROTECT (continued)
    - (F) AT&T NetPROTECT Rate Tables
      - .1 NetPROTECT Plus

Plus Customer	Non-Recurring	Monthly Recurring
Telephone Systems (CTS)	Charge	Charge
First 100 CTS (1-100), per CTS	\$100.00	-
Next 100 CTS (101-200), per CTS	\$75.00	=
All Additional CTS (Over 200), per CTS	\$50.00	-
First 500 CTS (1-500), per CTS	-	\$10.00
All Additional CTS (Over 500), per CTS	-	

Original Page 34

Effective: September 1, 2019

### .2 NetPROTECT Premium

Premium Customer Telephone Systems (CTS)	Non-Recurring Charge	Monthly Recurring Charge
First 100 CTS (1-100), per CTS	\$300.00	\$200.00
Next 100 CTS (101-200), per CTS	\$250.00	\$150.00
All Additional CTS (Over 200), per CTS	\$200.00	\$100.00
Location Move Charge, per CTS	\$25.00	_

Original Page 1

Effective: September 1, 2019

SECTION 5 – RESIDENTIAL DOMESTIC SERVICE DESCRIPTIONS, RATES AND CHARGES

### 5.1 General

Customers are billed based on their usage of the Company's services. Rates may vary based upon service type, call duration, time of day rate period, day of week, mileage, call type and/or calling plan. Fixed recurring charges, not dependent upon usage, are billed in advance. Usage-based charges are billed after each usage cycle.

Customers will be charged for each individual call placed through the Company. Call duration is determined as described in Section 3 of this pricing and service guide. For distance or time of day sensitive offerings, charges are determined based on mileage calculations and applicable rate periods found in Section 3 unless otherwise specified in this pricing and service guide.

Outbound direct dial services are available from originating locations within the United States. Calls may be placed to locations within the United States.

SECTION 5 - RESIDENTIAL DOMESTIC SERVICE DESCRIPTIONS, RATES AND CHARGES

### 5.2 Residential Message Telecommunications Service (MTS)

5.2.1 Residential MTS is a direct dialed outbound long distance service offered to residential Customers from switched access lines. Service is offered on a flat rate basis, twenty-four hours per day, seven days per week. Customers must dial 1 plus the area code, if applicable, and the terminating number to complete a call utilizing this service.

Customers who do not designate and qualify for a specific optional calling plan are automatically placed on Residential MTS. To be eligible for this service Customers must subscribe to local exchange service provided by the Company's Affiliated ILEC. Customers may subscribe to MTS service for the provision of intraLATA only calling (local toll), interLATA only calling (interstate and intrastate), or intraLATA and interLATA calling combined.

Original Page 2

Effective: September 1, 2019

### 5.2.2 Rates and Charges

All calls are billed in one (1) minute increments after an initial period, for billing purpose, of one (1) minute. Holiday discounts do not apply.

If monthly outbound domestic 1+ Direct-Dialed usage charges equal or exceed the Monthly Minimum Usage Charge (MUC) in a billing period, the MUC will not apply. If monthly outbound domestic 1+ Direct-Dialed usage charges in a billing period are less than the MUC, the MUC that will apply will be the difference between that month's usage charges and the MUC.

(A) Monthly Minimum Usage Charge (MUC)

Customer Subscribes To:	MUC
intraLATA Only MTS	\$0.00
interLATA Only MTS	\$5.99
intraLATA and interLATA MTS	\$5.99

### (B) Per-Minute Usage Rates

Jurisdiction	Per-Minute Usage Rate
Interstate	\$0.42
Intrastate	
Alabama, Florida, Kentucky, Louisiana, Mississippi, North Carolina, South Carolina and Tennessee	\$0.42
Georgia InterLATA and IntraLATA	
Mileage 0-22	\$0.00
Mileage 23 +	\$0.42

SECTION 5 – RESIDENTIAL DOMESTIC SERVICE DESCRIPTIONS, RATES AND CHARGES

### 5.3 Residential Optional Calling Plans

## 5.3.1 AT&T ONE RATE® Nationwide Calling 1

(A) The AT&T ONE RATE® Nationwide Calling 1 plan is an outbound only direct dialed interstate and intrastate long distance usage calling plan offered to Residential Customers on switched access lines. Service is offered on a per-minute basis, twenty-four hours per day, seven days per week. A single monthly recurring charge applies. Customers must dial 1 plus the area code, if applicable, and the called telephone number from their presubscribed telephone line.

Original Page 3

- (B) To be eligible for this plan, Customers must subscribe to and maintain an access line from the Company's Affiliated ILEC and must subscribe to the Company for the provision of interstate and intrastate InterLATA and/or intrastate IntraLATA service.
- (C) All calls are billed in one (1) minute increments after an initial period, for billing purposes, of one (1) minute.
- (D) The Monthly Recurring Charge (MRC) is \$3.99, and the per-minute usage rate is \$0.1200.

SECTION 5 – RESIDENTIAL DOMESTIC SERVICE DESCRIPTIONS, RATES AND CHARGES

### 5.3 Residential Optional Calling Plans (continued)

## 5.3.2 AT&T ONE RATE® Nationwide Advantage

(A) The AT&T ONE RATE® Nationwide Advantage is a bundled outbound only direct dialed interstate and intrastate long distance usage calling plan offered to Residential Customers on switched access lines. Service is offered on a per-minute basis, twenty-four hours per day, seven days per week. A single monthly recurring charge applies. Customers must dial 1 plus the area code, if applicable, and the called telephone number from their presubscribed telephone line.

Original Page 4

- (B) To be eligible for this plan, Customers must subscribe to and maintain an access line from the Company's Affiliated ILEC and must subscribe to the Company for the provision of interstate and intrastate InterLATA and/or intrastate IntraLATA service.
- (C) All calls are billed in one (1) minute increments after an initial period, for billing purposes, of one (1) minute.
- (D) The MRC is \$6.99, and the per-minute usage rate is \$0.0700.

SECTION 5 - RESIDENTIAL DOMESTIC SERVICE DESCRIPTIONS, RATES AND CHARGES

- 5.3 Residential Optional Calling Plans (continued)
  - 5.3.3 AT&T Unlimited Nationwide Calling One
    - (A) AT&T Unlimited Nationwide Calling One is a bundled outbound only interstate and intrastate long distance usage calling plan that is offered to Residential Customers on switched access lines. Customers must dial 1 plus the area code, if applicable, and the called telephone number from their presubscribed telephone line. This plan provides the Customer with unlimited minutes of interexchange long distance usage for a single monthly recurring charge (certain terms and conditions, as described below, apply). This plan does not provide call detail information on the Customer's bill.

Original Page 5

- (B) In order to be eligible for this plan, the Company must be able to verify that the Customer meets the following terms and conditions. Customers who no longer meet these terms and conditions will not be eligible for this plan and may be offered an alternative plan or have their service suspended, restricted or canceled without prior notice:
  - .1 Customers must subscribe to and maintain an access line from the Company's Affiliated ILEC.
  - .2 Customers must subscribe to the Company for the provision of interstate, intrastate IntraLATA and intrastate InterLATA long distance service.
  - .3 This plan may be combined with a verticals feature package as defined and offered by the Company's Affiliated ILEC.
  - .4 This plan is only available for a maximum of three (3) lines at the same location.
  - .5 This plan is not available on an account that is the recipient of charges billed from another location.
  - .6 Unlimited plan usage does not include multi-party conference calls (except those calls placed by using Three-Way Calling feature are included), calls to 900, 976, 700 numbers or other calls to access information services, directory assistance, operator services, international calling and toll-free calling services.
  - .7 This plan cannot be used for any use inconsistent with typical residential voice service. If usage under this plan is not consistent with typical residential voice Customer usage, at the Company's sole discretion, the Company may offer the Customer an alternative plan or suspend, restrict or cancel the Customer's service without prior notice, and the Customer may not be eligible to resubscribe to this plan. Calls that are not consistent with typical residential voice usage include but are not limited to: use for general business purposes, commercial facsimile, auto-dialing, resale, call centers, long distance access to internet service providers and telemarketing.
- (C) The MRC is \$26.99.

SECTION 5 - RESIDENTIAL DOMESTIC SERVICE DESCRIPTIONS, RATES AND CHARGES

- 5.3 Residential Optional Calling Plans (continued)
  - 5.3.4 AT&T Unlimited Nationwide Calling Advantage 1
    - (A) AT&T Unlimited Nationwide Calling Advantage 1 is a bundled outbound only interstate and intrastate long distance usage calling plan that is offered to Residential Customers on switched access lines. Customers must dial 1 plus the area code, if applicable, and the called telephone number from their presubscribed telephone line. This plan provides the Customer with unlimited minutes of interexchange long distance usage for a single monthly recurring charge (certain terms and conditions, as described below, apply). This plan does not provide call detail information on the Customer's bill.

Original Page 6

- (B) In order to be eligible for this plan, the Company must be able to verify that the Customer meets the following terms and conditions. Customers who no longer meet these terms and conditions will not be eligible for this plan and may be offered an alternative plan or have their service suspended, restricted or canceled without prior notice.
  - .1 Customers must subscribe to and maintain the following services from the Company's Affiliated ILEC:
    - .a access line, and;
    - verticals feature package as defined and offered by the Company's Affiliated ILEC.
  - .2 Customers must subscribe to the Company for the provision of interstate, intrastate IntraLATA and intrastate InterLATA long distance service.
  - .3 This plan is only available for a maximum of three (3) lines at the same location.
  - .4 This plan is not available on an account that is the recipient of charges billed from another location.
  - .5 Unlimited plan usage does not include multi-party conference calls (except those calls placed by using Three-Way Calling feature are included), calls to 900, 976, 700 numbers or other calls to access information services, directory assistance, operator services, international calling and toll-free calling services.
  - .6 This plan cannot be used for any use inconsistent with typical residential voice service. If usage under this plan is not consistent with typical residential voice Customer usage, at the Company's sole discretion, the Company may offer the Customer an alternative plan or suspend, restrict or cancel the Customer's service without prior notice, and the Customer may not be eligible to resubscribe to this plan. Calls that are not consistent with typical residential voice usage include but are not limited to: use for general business purposes, commercial facsimile, auto-dialing, resale, call centers, long distance access to internet service providers and telemarketing.
- (C) The monthly recurring charge is \$22.00.

SECTION 5 – RESIDENTIAL DOMESTIC SERVICE DESCRIPTIONS, RATES AND CHARGES

- 5.3 Residential Optional Calling Plans (continued)
  - 5.3.5 AT&T Unlimited Nationwide Calling Advantage 2
    - (A) AT&T Unlimited Nationwide Calling Advantage 2 is a bundled outbound only interstate and intrastate long distance usage calling plan that is offered to Residential Customers on switched access lines. Customers must dial 1 plus the area code, if applicable, and the called telephone number from their presubscribed telephone line. This plan provides the Customer with unlimited minutes of interexchange long distance usage for a single monthly recurring charge (certain terms and conditions, as described below, apply). This plan does not provide call detail information on the Customer's bill.

Original Page 7

Effective: September 1, 2019

(B) In order to be eligible for this plan, the Company must be able to verify that the Customer meets the following terms and conditions associated with one of the following options. Customers who no longer meet these terms and conditions will not be eligible for this plan and may be offered an alternative plan or have their service suspended, restricted or canceled without prior notice.

SECTION 5 - RESIDENTIAL DOMESTIC SERVICE DESCRIPTIONS, RATES AND CHARGES

- 5.3 Residential Optional Calling Plans (continued)
  - 5.3.5 AT&T Unlimited Nationwide Calling Advantage 2 (continued)
    - (B) (continued)

### Option 1:

.1 Customers must order this plan on-line via the Company's designated website or by contacting the Company's Affiliated ILEC's Customer Care Center.

Original Page 8

- .2 Customers must newly subscribe to an access line from the Company's Affiliated ILEC.
- .3 Customers must meet a minimum of one (1) of the following conditions:
  - a previously subscribed to local dial tone service of the Company's
     Affiliated ILEC and have cancelled that service, or;
  - b previously subscribed to long distance service from the Company and have cancelled that service, or;
  - .c be a current local telephone Customer in the Company's Affiliated ILEC's local territory who is now moving dial tone service from a competitor of the Company to the Company's Affiliated IEC.
- .4 Customers must subscribe to the Company for the provision of interstate, intrastate IntraLATA and intrastate InterLATA service.
- .5 This plan is only available for a maximum of three (3) lines at the same location.
- .6 This plan may be combined with verticals feature package as defined and offered by the Company's Affiliated ILEC.
- .7 This plan is not available on an account that is the recipient of charges billed from another location.
- .8 Hearing impaired and disabled customers who access the Company's website and are unable to order this plan on-line may order via text telephone.
- .9 In the event the Company's on-line ordering system is not functioning or processing orders due to a system outage or malfunction, Customers may call the Company's customer care center or sales representative to request assistance for on-line ordering of AT&T Unlimited Nationwide Calling Advantage 2.
- .10 Unlimited plan usage does not include multi-party conference calls (except those calls placed by using Three-Way Calling feature are included), calls to 900, 976, 700 numbers or other calls to access information services, directory assistance, operator services, international calling and toll-free calling services.
- .11 This plan cannot be used for any use inconsistent with typical residential voice service. If usage under this plan is not consistent with typical residential voice Customer usage, at the Company's sole discretion, the Company may offer the Customer an alternative plan or suspend, restrict or cancel the Customer's service without prior notice, and the Customer may not be eligible to resubscribe to this plan. Calls that are not consistent with typical residential voice usage include but are not limited to: use for general business purposes, commercial facsimile, auto-dialing, resale, call centers, long distance access to internet service providers and telemarketing.

SECTION 5 - RESIDENTIAL DOMESTIC SERVICE DESCRIPTIONS, RATES AND CHARGES

- 5.3 Residential Optional Calling Plans (continued)
  - 5.3.5 AT&T Unlimited Nationwide Calling Advantage 2 (continued)
    - (B) (continued)

### Option 2:

- .1 Customers must currently subscribe to local dial tone service from the Company's Affiliated ILEC and state an intention to change local carriers in response to a competitive offer.
- .2 This plan is only available for a maximum of three (3) lines at the same location.

Original Page 9

- .3 Customers must currently subscribe to the Company for the provision of interstate, intrastate IntraLATA and intrastate InterLATA service.
- .4 This plan may be combined with verticals feature package as defined and offered by the Company's Affiliated ILEC.
- .5 This plan is not available on an account that is the recipient of charges billed from another location.
- .6 Unlimited plan usage does not include multi-party conference calls (except those calls placed by using Three-Way Calling feature are included), calls to 900, 976, 700 numbers or other calls to access information services, directory assistance, operator services, international calling and toll-free calling services.
- .7 This plan cannot be used for any use inconsistent with typical residential voice service. If usage under this plan is not consistent with typical residential voice Customer usage, at the Company's sole discretion, the Company may offer the Customer an alternative plan or suspend, restrict or cancel the Customer's service without prior notice, and the Customer may not be eligible to resubscribe to this plan. Calls that are not consistent with typical residential voice usage include but are not limited to: use for general business purposes, commercial facsimile, auto-dialing, resale, call centers, long distance access to internet service providers and telemarketing.

#### SECTION 5 - RESIDENTIAL DOMESTIC SERVICE DESCRIPTIONS, RATES AND CHARGES

- 5.3 Residential Optional Calling Plans (continued)
  - 5.3.5 AT&T Unlimited Nationwide Calling Advantage 2 (continued)
    - (B) (continued)

#### Option 3

.1 Customers must order this plan on-line via the Company's designated website.

Original Page 10

- .2 Customers must subscribe to and maintain the following services from the Company's Affiliated ILEC:
  - access line, and;
  - verticals feature package as defined and offered by the Company's Affiliated ILEC.
- .3 This plan is only available for a maximum of three (3) lines at the same location.
- .4 Customers must subscribe to the Company for the provision of interstate, intrastate IntraLATA and intrastate InterLATA service.
- .5 This plan is not available on an account that is the recipient of charges billed from another location.
- .6 Hearing impaired and disabled customers who access the Company's website and are unable to order this plan on-line may order via text telephone.
- .7 In the event the Company's on-line ordering system is not functioning or processing orders due to a system outage or malfunction, Customers may call the Company's customer care center or sales representative to request assistance for on-line ordering of AT&T Unlimited Nationwide Calling Advantage 2.
- .8 Unlimited plan usage does not include multi-party conference calls (except those calls placed by using Three-Way Calling feature are included), calls to 900, 976, 700 numbers or other calls to access information services, directory assistance, operator services, international calling and toll-free calling services.
- .9 This plan cannot be used for any use inconsistent with typical residential voice service. If usage under this plan is not consistent with typical residential voice Customer usage, at the Company's sole discretion, the Company may offer the Customer an alternative plan or suspend, restrict or cancel the Customer's service without prior notice, and the Customer may not be eligible to resubscribe to this plan. Calls that are not consistent with typical residential voice usage include but are not limited to: use for general business purposes, commercial facsimile, auto-dialing, resale, call centers, long distance access to internet service providers and telemarketing.
- (C) The Monthly Recurring Charge for Options 1, 2 and 3 is \$16.00.

SECTION 5 - RESIDENTIAL DOMESTIC SERVICE DESCRIPTIONS, RATES AND CHARGES

#### 5.3 Residential Optional Calling Plans (continued)

#### 5.3.6 BellSouth® Unlimited Plan II<sup>1</sup>

(A) The BellSouth® Unlimited Plan II is a direct dialed outbound long distance service offered to single line residential Customers on switched access lines. Service is offered on a flat rate basis, twenty-four hours per day, seven days per week. Customers must dial 1 plus the area code, if applicable, and the terminating number to complete a call utilizing this service. The BellSouth® Unlimited Plan II provides the Customer with unlimited minutes of interstate long distance usage for a flat rate monthly charge (certain restrictions, as outlined below, do apply). This plan does not provide call detail information on the Customer's monthly bill. This plan is not offered on an intraLATA or intrastate only basis.

Original Page 11

- (B) In order to be eligible for this plan, the Company must be able to verify that the Customer meets the following eligibility requirements. Customers who no longer meet these eligibility requirements will not be eligible for this plan and may be offered an alternative plan or have their service suspended, restricted or canceled without prior notice. Customers are eligible for this service if they meet the following requirements:
  - .1 Customers must reside in areas where the Company's Affiliated ILEC's standalone BellSouth® Complete Choice® plan is not available and must subscribe to the Company's Affiliated ILEC's BellSouth® Area Plus® service with the Complete Choice® option.
  - .2 Customers must presubscribe to the Company for interLATA long distance service.
  - .3 Customer lines associated with educational institutions, (colleges, universities, etc.) are not eligible for this plan.
  - .4 This plan cannot be used for any use inconsistent with residential service.
  - .5 This plan is not available for resale.
  - .6 This plan is not available to Customers with a multi-line account or an account that bills to another number or is the recipient of charges billed from another number unless the Customer establishes separate billing accounts for each line.
  - .7 Unlimited plan usage does not include multi-party conference calls (except those calls placed by using the Three-Way Calling feature are included), calls to 900, 976, 700 numbers and other calls to access information services, directory assistance, operator services, international calling and toll-free calling services.
  - .8 If usage under this plan is not consistent with typical residential Customer usage, at the Company's sole discretion, the Company may offer the Customer an alternative plan or suspend, restrict or cancel the Customer's service without prior notice. Calls that are not consistent with typical residential voice usage include but are not limited to: use for general business purposes, commercial facsimile, auto-dialing, resale, call centers, long distance access to internet service providers and telemarketing.
  - .9 The Company reserves the right to exclude certain terminating telephone numbers from this plan. Calls to these numbers will be billed at applicable MTS rates.
  - .10 The Monthly Service Charge is \$16.00.

<sup>&</sup>lt;sup>1</sup>This plan is marketed by the Company as Hernando Unlimited.

SECTION 5 - RESIDENTIAL DOMESTIC SERVICE DESCRIPTIONS, RATES AND CHARGES

- 5.3 Residential Optional Calling Plans (continued)
  - 5.3.7 AT&T Unlimited Nationwide Calling<sup>SM</sup> Plus 1
    - (A) AT&T Unlimited Nationwide Calling<sup>SM</sup> Plus 1 is a bundled outbound only interstate intrastate long distance usage calling plan designed for Residential Customers with a single BTN. Multiple BTN Aggregation is not available with this Service. Customers or End Users can access the Company's long distance Service by dialing 1 + the area code + the called telephone number from their presubscribed telephone line. This plan does not provide call detail information on the Customer's bill. For a single MRC this Service is available to new and existing Residential Customers who meet the following terms and conditions:

Original Page 12

- .1 use Switched Access to reach the long distance network;
- .2 subscribes to the Company for the provision of interLATA (interstate and intrastate) service and intraLATA (local toll) service for outbound long distance calling;
- .3 subscribes to and maintains the following services from an Affiliated ILEC of the Company:
  - .a Access line with a BTN and;
  - .b Verticals Feature Package as defined and offered by the Affiliated ILEC of the Company;
- .4 subscribe to and maintain one (1) or more Affiliate of the Company Products as defined and offered by the Affiliate of the Company;
- .5 demonstrate to the satisfaction of the Company at the time of subscribing to the Service that the Residential Customer also subscribes to the required products and/or services described above;
- .6 request to be provisioned under this Service and;
- .7 limits the use of Service to that which is of a standard, domestic, residential nature.
- (B) If the Customer uses this Service for non-standard residential or non-residential purposes, including but not limited to commercial or broadcast facsimile, resale, telemarketing, internet connections and autodialing, the Company may immediately suspend, restrict or cancel the Customer's Service. As a result of non-standard or non-residential use of Service, the Company may move the Customer to AT&T ONE RATE® Nationwide Advantage. If the Customer is moved off this Service because of the previously described reasons, the Customer may be ineligible to re-subscribe to this Service.
- (C) The MRC is \$10.00.

SECTION 5 – RESIDENTIAL DOMESTIC SERVICE DESCRIPTIONS, RATES AND CHARGES

#### 5.4 Residential Special Offers

#### 5.4.1 Residential Retention Offers

As described below, Retention Offers are available to qualifying Customers of the Company who currently or newly subscribe to one of the Company's qualifying unlimited plans.

Original Page 13

Effective: September 1, 2019

#### (A) AT&T Unlimited Calling Qualifying Plans

The following plans qualify for this offer:

- AT&T Unlimited Nationwide Calling Advantage 1
- AT&T Unlimited Nationwide Calling Advantage 2
- AT&T Unlimited Nationwide Calling Advantage 3
- AT&T Unlimited Nationwide Calling Plus 1
- Unlimited II Plan

#### (B) Eligibility

- .1 currently subscribe to the Company for interLATA (interstate and intrastate) service, intraLATA (local toll) service, or interLATA/intraLATA service combined and state an intention to change long distance service to a competitor of the Company, are offered a qualifying plan as defined in (A) above and continue to state an intention to change long distance service to a Competitor of the Company; or
- .2 currently subscribe to a qualifying plan as defined in (A) above and state an intention to change long distance Service to a competitor of the Company.
- .3 in either case, the Customer must subscribe to one of the qualifying plans defined in (A) above to qualify for this retention offer.
- .4 Customers must maintain the requirements of the qualifying plan as defined in Section 5.3 and Section 12 of this pricing and service guide and continue to subscribe to the qualifying plan to receive the \$5.00 monthly credit during the twelve (12) month benefit period.

#### (C) Rates and Charges

Customers subscribing to a qualifying plan as defined in (A) above will receive a \$5.00 monthly credit for a twelve (12) month benefit period. At the end of the benefit period the customer will be charged the then current rates for the qualifying plan to which the Customer subscribes to as defined in Section 5.3 and Section 12 of this pricing and service guide.

(D) This offer cannot be combined with any other promotion offered by the Company or with the Company Reward offers.

Original Page 1

Effective: September 1, 2019

SECTION 6 – BUSINESS INTERNATIONAL SERVICE DESCRIPTIONS, RATES AND CHARGES

#### 6.1 General

- 6.1.1 Customers are billed based on their usage of the Company's services. Rates may vary by service type, call duration, time of day, rate period, day of week, mileage, call type, calling plan and foreign location. Fixed recurring charges, not dependent upon usage, are billed in advance. Usage-based charges are billed after each usage cycle.
- 6.1.2 Customers will be charged for each individual call placed through the Company. Call duration is determined as described in Section 3 of this pricing and service guide. For distance or time of day sensitive offerings, charges are determined based on the applicable rate periods found in Section 3 unless otherwise specified in this pricing and service guide.
- 6.1.3 Outbound direct dial services are available from originating locations within the United States. Calls may be placed between the United States and those countries listed herein. Inbound international toll-free service is available to Customers. Toll-free calls may be originated from locations within Canada.
- 6.1.4 Calls to or from a foreign country or area, or calls routed through a foreign country or area, are subject to such restrictions as may be enforced from time to time by the authorities in that country or area.
- 6.1.5 An additional per-minute International Mobile Termination Charge (IMTC) is applied to all international calls originating in the United States and terminating in certain countries to (1) wireless devices; and (2) ranges of fixed wireless telephone numbers which the foreign telecommunications company designates to be for special services. Because many foreign telecommunications companies impose extra fees on the Company to complete these calls, the Company charges the IMTC, in addition to all other applicable charges. The IMTC amount varies by country and changes from time to time. Please check codes and rates periodically for current information. Codes can be found at <a href="www.att.com/mobileterm">www.att.com/mobileterm</a>. The per-minute charges in Section 6.4.1 will apply.

Original Page 2

Effective: September 1, 2019

SECTION 6 – BUSINESS INTERNATIONAL SERVICE DESCRIPTIONS, RATES AND CHARGES

#### 6.2 Business International Message Telecommunications Service (IMTS)

The Business IMTS is a direct dialed outbound international long distance service offered to business Customers from switched access lines. Service is offered twenty-four hours per day, seven days per week. Charges depend on the country called, the time of day in which the call was made and the duration of the call. Customers must dial 1 plus the country code or area code, if applicable, and the terminating number to complete a call utilizing this service.

To be eligible for this service Customers must subscribe to local exchange service provided by the Company's Affiliated ILEC.

All calls are billed in one (1) minute increments after an initial period, for billing purpose, of one (1) minute. This service is only offered in conjunction with a Company's Business Domestic Service. A monthly recurring charge does not apply for this service.

Inbound international toll-free service originated from locations within Canada is available to Customers of this service who also subscribe to AT&T Long Distance Toll Free<sup>SM</sup> Service in Section 4.4 of this pricing and service guide. The rate table in Section 6.2.2 also contains the per-minute rates for AT&T Long Distance Toll Free<sup>SM</sup> Service originating in Canada and terminating throughout the United States.

#### 6.2.1 Time of Day Rate Periods

Time of day rates, as defined in Section 3 of this pricing and service guide apply to this plan.

Original Page 3 Effective: September 1, 2019

## BUSINESS AND RESIDENTIAL SERVICES PRICING AND SERVICE GUIDE

SECTION 6 – BUSINESS INTERNATIONAL SERVICE DESCRIPTIONS, RATES AND CHARGES

### 6.2 Business International Message Telecommunications Service (continued)

#### 6.2.2 Per Minute Rate Tables

	Sta	ndard	Eco	Economy		
		Each		Each		
	Initial		Initial	Additional		
Country Name	Minute	Minute	Minute	Minute		
Afghanistan	\$21.08	\$21.08	\$21.08	\$21.08		
Albania	\$13.02	\$13.02	\$13.02	\$13.02		
Algeria	\$8.73	\$8.73	\$8.73	\$8.73		
Andorra	\$5.37	\$5.37	\$5.37	\$5.37		
Angola	\$13.95	\$13.95	\$13.95	\$13.95		
Anguilla	\$6.04	\$6.04	\$6.04	\$6.04		
Antarctica (Casey Base)	\$11.88	\$11.88	\$11.88	\$11.88		
Antarctica (Scott Base)	\$11.88	\$11.88	\$11.88	\$11.88		
Antigua	\$6.04	\$6.04	\$6.04	\$6.04		
Argentina	\$6.72	\$6.72	\$6.72	\$6.72		
Armenia	\$10.77	\$10.77	\$10.77	\$10.77		
Aruba	\$6.04	\$6.04	\$6.04	\$6.04		
Ascension Island	\$8.85	\$8.85	\$8.85	\$8.85		
Australia	\$5.55	\$5.55	\$5.55	\$5.55		
Austria	\$5.54	\$5.54	\$5.54	\$5.54		
Azerbaijan	\$10.30	\$10.30	\$10.30	\$10.30		
Bahamas	\$4.92	\$4.92	\$4.92	\$4.92		
Bahrain	\$8.67	\$8.67	\$8.67	\$8.67		
Bangladesh	\$10.77	\$10.77	\$10.77	\$10.77		
Barbados	\$6.04	\$6.04	\$6.04	\$6.04		
Belarus	\$10.77	\$10.77	\$10.77	\$10.77		
Belgium	\$5.29	\$5.29	\$5.29	\$5.29		
Belize	\$6.72	\$6.72	\$6.72	\$6.72		
Benin	\$8.07	\$8.07	\$8.07	\$8.07		
Bermuda	\$5.10	\$5.10	\$5.10	\$5.10		
Bhutan	\$16.16	\$16.16	\$16.16	\$16.16		
Bolivia	\$6.95	\$6.95	\$6.95	\$6.95		
Bosnia	\$6.72	\$6.72	\$6.72	\$6.72		
Botswana	\$7.79	\$7.79	\$7.79	\$7.79		
Brazil	\$6.72	\$6.72	\$6.72	\$6.72		
British Virgin Islands	\$6.04	\$6.04	\$6.04	\$6.04		
Brunei	\$8.45	\$8.45	\$8.45	\$8.45		
Bulgaria	\$8.07	\$8.07	\$8.07	\$8.43		
	\$13.12	\$13.12	\$13.12	\$13.12		
Burkina Faso Burundi	\$15.12	\$15.12	\$15.12	\$15.12		
Cambodia	\$13.48	\$13.48	\$13.48	\$13.48		
Cameroon	\$8.07	\$8.07	\$8.07	\$8.07		
Canada - Outbound	\$1.41	\$1.41	\$1.41	\$1.41		
Canada - Inbound	\$1.41	\$1.41	\$1.41	\$1.41		
Cape Verde Island	\$10.77	\$10.77	\$10.77	\$10.77		
Cayman Islands	\$6.04	\$6.04	\$6.04	\$6.04		
Central African Republic	\$13.95	\$13.95	\$13.95	\$13.95		
Chad	\$15.69	\$15.69	\$15.69	\$15.69		
Chile	\$6.72	\$6.72	\$6.72	\$6.72		
China	\$10.77	\$10.77	\$10.77	\$10.77		
Christmas & Cocos Islands	\$5.55	\$5.55	\$5.55	\$5.55		
Colombia	\$6.72	\$6.72	\$6.72	\$6.72		
Comoros Islands	\$15.20	\$15.20	\$15.20	\$15.20		

Original Page 4 Effective: September 1, 2019

## BUSINESS AND RESIDENTIAL SERVICES PRICING AND SERVICE GUIDE

- 6.2 Business International Message Telecommunications Service (continued)
  - 6.2.2 Per Minute Rate Tables (continued)

	Standard		Eco	onomy
		Each		Each
C ( PT		Additional		
Country Name	Minute		Minute	
Congo	\$12.88	\$12.88	\$12.88	\$12.88
Cook Islands	\$14.75	\$14.75	\$14.75	\$14.75
Costa Rica	\$6.04	\$6.04	\$6.04	\$6.04
Croatia	\$7.48	\$7.48	\$7.48	\$7.48
Cuba	\$8.73	\$8.73	\$8.73	\$8.73
Cyprus	\$7.70	\$7.70	\$7.70	\$7.70
Czech Republic	\$8.07	\$8.07	\$8.07	\$8.07
Denmark	\$5.07	\$5.07	\$5.07	\$5.07
Diego Garcia	\$13.80	\$13.80	\$13.80	\$13.80
Djibouti	\$11.69	\$11.69	\$11.69	\$11.69
Dominica	\$6.04	\$6.04	\$6.04	\$6.04
Dominican Republic	\$6.04	\$6.04	\$6.04	\$6.04
East Timor	\$13.00	\$13.00	\$13.00	\$13.00
Easter Island	\$6.72	\$6.72	\$6.72	\$6.72
Ecuador	\$6.49	\$6.49	\$6.49	\$6.49
Egypt	\$8.07	\$8.07	\$8.07	\$8.07
El Salvador	\$6.72	\$6.72	\$6.72	\$6.72
Equatorial Guinea	\$14.20	\$14.20	\$14.20	\$14.20
Eritrea	\$10.77	\$10.77	\$10.77	\$10.77
Estonia	\$10.77	\$10.77	\$10.77	\$10.77
Ethiopia	\$10.77	\$10.77	\$10.77	\$10.77
Faeroe Islands	\$4.98	\$4.98	\$4.98	\$4.98
Falkland Islands	\$12.88	\$12.88	\$12.88	\$12.88
Fiji Islands	\$10.77	\$10.77	\$10.77	\$10.77
Finland	\$4.70	\$4.70	\$4.70	\$4.70
France	\$5.00	\$5.00	\$5.00	\$5.00
French Antilles	\$5.14	\$5.14	\$5.14	\$5.14
French Guiana	\$7.08	\$7.08	\$7.08	\$7.08
French Polynesia	\$10.18	\$10.18	\$10.18	\$10.18
Gabon	\$8.07	\$8.07	\$8.07	\$8.07
Gambia	\$8.67	\$8.67	\$8.67	\$8.67
Georgia	\$10.77	\$10.77	\$10.77	\$10.77
Germany	\$4.83	\$4.83	\$4.83	\$4.83
Ghana	\$8.73	\$8.73	\$8.73	\$8.73
Gibraltar	\$6.48	\$6.48	\$6.48	\$6.48
Greece	\$6.72	\$6.72	\$6.72	\$6.72
Greenland	\$6.10	\$6.10	\$6.10	\$6.10
Grenada	\$6.04	\$6.04	\$6.04	\$6.04
Guadeloupe	\$6.04	\$6.04	\$6.04	\$6.04
Guantanamo	\$7.56	\$7.56	\$7.56	\$7.56
Guatemala	\$6.56	\$6.56	\$6.56	\$6.56
Guinea	\$10.77	\$10.77	\$10.77	\$10.77
Guinea-Bissau	\$14.62	\$14.62	\$14.62	\$10.77
Guinea-Bissau Guyana	\$8.07	\$8.07	\$8.07	\$8.07
Haiti	\$6.30	\$6.30	\$6.30	\$6.30
Honduras				
	\$7.48	\$7.48	\$7.48	\$7.48
Hong Kong	\$6.43 \$7.24	\$6.43	\$6.43	\$6.43 \$7.24
Hungary		\$7.24	\$7.24	
Iceland	\$5.94	\$5.94	\$5.94	\$5.94

Original Page 5 Effective: September 1, 2019

## BUSINESS AND RESIDENTIAL SERVICES PRICING AND SERVICE GUIDE

- 6.2 Business International Message Telecommunications Service (continued)
  - 6.2.2 Per Minute Rate Tables (continued)

	Sta	ndard	Economy		
	T242.a1	Each	T242-a1	Each Additional	
Country Name				Additional Minute	
India	\$10.77		\$10.77		
Indonesia	\$8.44	\$8.44	\$8.44	\$8.44	
Inmarsat (SNAC)	\$23.91		\$23.91		
Iran	\$8.73	\$8.73	\$8.73	\$8.73	
Iraq	\$10.41	\$10.41	\$10.41		
Ireland	\$5.18	\$5.18	\$5.18	\$5.18	
Israel	\$7.79	\$7.79	\$7.79	\$7.79	
Italy	\$6.23	\$6.23	\$6.23	\$6.23	
Ivory Coast	\$8.75	\$8.75	\$8.75	\$8.75	
Jamaica	\$6.13	\$6.13	\$6.13	\$6.13	
Japan	\$6.14	\$6.14	\$6.14	\$6.14	
Jordan	\$8.07	\$8.07	\$8.07	\$8.07	
Kazakhstan	\$10.77	\$10.77	\$10.77	\$10.77	
Kenya	\$8.07	\$8.07	\$8.07	\$8.07	
Kiribati	\$10.77	\$10.77	\$10.77	\$10.77	
Korea (North)	\$16.16	\$16.16	\$16.16		
Korea (South)	\$7.48	\$7.48	\$7.48	\$7.48	
Kosovo	\$4.55	\$4.55	\$4.55	\$4.55	
Kuwait	\$6.72	\$6.72	\$6.72	\$6.72	
Kyrgyzstan	\$10.77	\$10.77	\$10.77	\$10.77	
Laos	\$16.16	\$16.16	\$16.16	\$16.16	
Latvia	\$10.77	\$10.77	\$10.77	\$10.77	
Lebanon	\$10.77	\$10.77	\$10.77	\$10.77	
Lesotho	\$8.43	\$8.43	\$8.43	\$8.43	
Liberia	\$8.07	\$8.07	\$8.07	\$8.07	
Libya	\$8.89	\$8.89	\$8.89	\$8.89	
Liechtenstein	\$4.00	\$4.00	\$4.00	\$4.00	
Lithuania	\$10.18	\$10.18	\$10.18		
Luxembourg	\$3.99	\$3.99	\$3.99	\$3.99	
Macao	\$10.69	\$10.69	\$10.69	\$10.69	
Macedonia	\$6.72	\$6.72	\$6.72	\$6.72	
Madagascar	\$14.50	\$14.50	\$14.50	\$14.50	
Malawi	\$7.82	\$7.82	\$7.82	\$7.82	
Malaysia	\$6.57	\$6.57	\$6.57	\$6.57	
Maldives	\$13.13	\$13.13	\$13.13	\$13.13	
Mali Republic	\$10.77	\$10.77	\$10.77	\$10.77	
Malta Republic	\$8.07	\$8.07	\$8.07	\$8.07	
Marshall Islands	\$9.43	\$9.43	\$9.43	\$9.43	
Mauritania	\$12.88		\$12.88		
Mauritius	\$12.89	\$12.89	\$12.89	\$12.89	
Mexico 1	\$2.16	\$2.16	\$2.16	\$2.16	
Mexico 2	\$4.51	\$4.51	\$4.51	\$4.51	
Micronesia	\$9.77	\$9.77	\$9.77	\$9.77	
Moldova	\$10.77	\$10.77	\$10.77	\$10.77	
Monaco	\$4.55	\$4.55	\$4.55	\$4.55	
Mongolia	\$15.38	\$15.38	\$15.38	\$15.38	
Montenegro	\$6.72	\$6.72	\$6.72	\$6.72	
Montserrat	\$6.04	\$6.04	\$6.04	\$6.04	
Morocco	\$8.75	\$8.75	\$8.75	\$8.75	

Original Page 6

Effective: September 1, 2019

- 6.2 Business International Message Telecommunications Service (continued)
  - 6.2.2 Per Minute Rate Tables (continued)

	Standard		Economy		
	514	Each	Ec	Each	
	Initial		Initial	Additional	
Country Name				Minute	
Mozambique	\$13.05		\$13.05		
Myanmar	\$16.16		\$16.16	\$16.16	
Namibia	\$6.72	\$6.72	\$6.72	\$6.72	
Nauru	\$13.35		\$13.35	\$13.35	
Nepal	\$10.77		\$10.77	\$10.77	
Netherlands	\$4.55	\$4.55	\$4.55	\$4.55	
Netherlands Antilles	\$6.12	\$6.12	\$6.12	\$6.12	
Networks	\$25.31	\$25.31	\$25.31	\$25.31	
New Caledonia	\$12.11	\$12.11	\$12.11	\$12.11	
New Zealand	\$7.04	\$7.04	\$7.04	\$7.04	
Nicaragua	\$6.30	\$6.30	\$6.30	\$6.30	
Niger Republic	\$10.77	\$10.77	\$10.77	\$10.77	
Nigeria	\$6.72	\$6.72	\$6.72	\$6.72	
Niue	\$15.87	\$15.87	\$15.87	\$15.87	
Norfolk Island	\$11.88		\$11.88	\$11.88	
Norway	\$4.70	\$4.70	\$4.70	\$4.70	
Oman	\$8.69	\$8.69	\$8.69	\$8.69	
Pakistan	\$10.77		\$10.77	\$10.77	
Palau	\$13.33		\$13.33	\$13.33	
Palestinian Authority	\$8.07	\$8.07	\$8.07	\$8.07	
Panama	\$6.04	\$6.04	\$6.04	\$6.04	
Papua New Guinea	\$7.50	\$7.50	\$7.50	\$7.50	
Paraguay	\$8.07	\$8.07	\$8.07	\$8.07	
Peru	\$6.72	\$6.72	\$6.72	\$6.72	
Philippines	\$6.85	\$6.85	\$6.85	\$6.85	
Poland	\$6.72	\$6.72	\$6.72	\$6.72	
Portugal	\$6.04	\$6.04	\$6.04	\$6.04	
	\$8.07	\$8.07	\$8.07	\$8.07	
Qatar Reunion Island (includes Mayotte Island)			\$12.88	\$12.88	
Romania	\$8.07	\$8.07	\$8.07	\$8.07	
Russia	\$10.77		\$10.77	\$10.77	
Rwanda, Republic of	\$13.48		\$13.48	\$13.48	
San Marino	\$5.37	\$5.37	\$5.37	\$5.37	
Sao Tome	\$15.25	\$15.25	\$15.25	\$15.25	
Saudi Arabia	\$8.07	\$8.07	\$8.07	\$8.07	
	\$9.60	\$9.60		\$9.60	
Senegal Sorbio	\$6.72		\$9.60		
Serbia Service Depublic of		\$6.72	\$6.72	\$6.72	
Seychelles, Republic of	\$13.80	\$13.80	\$13.80	\$13.80	
Sierra Leone	\$10.77	\$10.77	\$10.77	\$10.77	
Singapore	\$5.83	\$5.83	\$5.83	\$5.83	
Slovakia	\$7.99	\$7.99	\$7.99	\$7.99	
Slovenia	\$7.82	\$7.82	\$7.82	\$7.82	
Solomon Island	\$14.14	\$14.14	\$14.14	\$14.14	
Somalia, Democratic Rep	\$12.27	\$12.27	\$12.27	\$12.27	
South Africa	\$6.72	\$6.72	\$6.72	\$6.72	
Spain	\$6.45	\$6.45	\$6.45	\$6.45	
Sri Lanka	\$10.77	\$10.77	\$10.77	\$10.77	
St. Helena	\$12.35	\$12.35	\$12.35	\$12.35	

Original Page 7 Effective: September 1, 2019

## BUSINESS AND RESIDENTIAL SERVICES PRICING AND SERVICE GUIDE

- 6.2 Business International Message Telecommunications Service (continued)
  - 6.2.2 Per Minute Rate Tables (continued)

	Standard Economy					
	56	Each	Det	Each		
	Initial		Initial	Additional		
Country Name	Minute		Minute			
St. Kitts/Nevis	\$6.04	\$6.04	\$6.04	\$6.04		
St. Lucia	\$6.04	\$6.04	\$6.04	\$6.04		
St. Maarten	\$6.12	\$6.12	\$6.12	\$6.12		
St. Pierre Miquelon	\$5.37	\$5.37	\$5.37	\$5.37		
St. Vincent/Grenadines	\$6.04	\$6.04	\$6.04	\$6.04		
South Sudan	\$13.48	\$13.48	\$13.48	\$13.48		
Sudan	\$13.48	\$13.48	\$13.48	\$13.48		
Suriname	\$9.20	\$9.20	\$9.20	\$9.20		
Swaziland	\$8.07	\$8.07	\$8.07	\$8.07		
Sweden	\$4.55	\$4.55	\$4.55	\$4.55		
Switzerland	\$4.68	\$4.68	\$4.68	\$4.68		
Syrian Arab Republic	\$10.77	\$10.77	\$10.77	\$10.77		
Taiwan	\$7.12	\$7.12	\$7.12	\$7.12		
Tajikistan	\$10.94	\$10.94	\$10.94	\$10.94		
Tanzania	\$8.07	\$8.07	\$8.07	\$8.07		
Thailand	\$6.77	\$6.77	\$6.77	\$6.77		
Togo	\$8.07	\$8.07	\$8.07	\$8.07		
Tokelau	\$23.91	\$23.91	\$23.91	\$23.91		
Tonga Island	\$10.77	\$10.77	\$10.77	\$10.77		
Trinidad & Tobago	\$6.04	\$6.04	\$6.04	\$6.04		
Tunisia	\$8.07	\$8.07	\$8.07	\$8.07		
Turkey	\$7.04	\$7.04	\$7.04	\$7.04		
Turkmenistan	\$9.43	\$9.43	\$9.43	\$9.43		
Turks and Caicos Islands	\$6.04	\$6.04	\$6.04	\$6.04		
Tuvalu	\$16.16	\$16.16	\$16.16	\$16.16		
Uganda	\$8.07	\$8.07	\$8.07	\$8.07		
Ukraine	\$10.77	\$10.77	\$10.77	\$10.77		
United Arab Emirates	\$7.42	\$7.42	\$7.42	\$7.42		
United Kingdom	\$4.60	\$4.60	\$4.60	\$4.60		
Uruguay	\$6.72	\$6.72	\$6.72	\$6.72		
Uzbekistan	\$10.77	\$10.77	\$10.77	\$10.77		
Vanuatu	\$8.07	\$8.07	\$8.07	\$8.07		
Vatican City	\$6.23	\$6.23	\$6.23	\$6.23		
Venezuela	\$5.83	\$5.83	\$5.83	\$5.83		
Vietnam	\$10.77	\$10.77	\$10.77	\$10.77		
Wallis Futuna	\$14.14	\$14.14	\$14.14	\$14.14		
Western Samoa	\$10.77	\$10.77	\$10.77	\$10.77		
Yemen Arab Republic	\$8.07	\$8.07	\$8.07	\$8.07		
Zaire	\$8.07	\$8.07	\$8.07	\$8.07		
Zambia	\$7.25	\$7.25	\$7.25	\$7.25		
Zimbabwe	\$7.82	\$7.82	\$7.82	\$7.82		

SECTION 6 – BUSINESS INTERNATIONAL SERVICE DESCRIPTIONS, RATES AND CHARGES

#### 6.3 Business Optional Calling Plans

- 6.3.1 AT&T Business International Calling<sup>SM</sup> Basic
  - (A) AT&T Business International Calling<sup>SM</sup> Basic is a direct dialed outbound international long distance service offered to Customers on switched access lines. Service is offered to each country on a flat rate basis, twenty-four hours per day, seven days per week. Customers must dial 1 plus the country code or area code, if applicable, and the terminating number to complete a call utilizing this Service.

Original Page 8

Effective: September 1, 2019

- (B) All calls are billed in sixty (60) second increments after an initial period, for billing purposes, of sixty (60) seconds. This Service is only offered in conjunction with one of the Company's Business Domestic Service.
- (C) For Customers who also subscribe AT&T Long Distance Toll Free<sup>SM</sup> Service in Section 4.4 of this pricing and service guide, the inbound international toll-free service perminute usage rate for calls originating in Canada is found in Section 6.3.1 (E) below.
- (D) International Mobile Termination Charge

See Section 6.4.1 of this pricing and service guide for the applicable International Mobile Termination Charge.

(E) The MRC is \$5.00, and the per-minute usage rates are as follows:

Country	Initial Minute	Each Additional Minute	Country	Initial Minute	Each Additional Minute
Afghanistan	\$2.56	\$2.56	Bolivia	\$0.84	\$0.84
Albania	\$1.54	\$1.54	Bosnia	\$1.10	\$1.10
Algeria	\$0.97	\$0.97	Botswana	\$1.14	\$1.14
Andorra	\$0.64	\$0.64	Brazil	\$0.57	\$0.57
Angola	\$2.06	\$2.06	British Virgin Islands	\$0.64	\$0.64
Anguilla	\$0.89	\$0.89	Brunei	\$1.18	\$1.18
Antarctica (Casey Base)	\$3.04	\$3.04	Bulgaria	\$0.90	\$0.90
Antarctica (Scott Base)	\$3.04	\$3.04	Burkina Faso	\$1.26	\$1.26
Antigua	\$0.73	\$0.73	Burundi	\$2.45	\$2.45
Argentina	\$0.74	\$0.74	Cambodia	\$3.13	\$3.13
Armenia	\$1.32	\$1.32	Cameroon	\$1.39	\$1.39
Aruba	\$0.64	\$0.64	Canada – Outbound	\$0.12	\$0.12
Ascension Island	\$2.36	\$2.36	Canada - Inbound	\$0.12	\$0.12
Australia	\$0.25	\$0.25	Cape Verde Island	\$1.14	\$1.14
Austria	\$0.26	\$0.26	Cayman Islands	\$0.65	\$0.65
Azerbaijan	\$1.70	\$1.70	Central African Republic	\$2.10	\$2.10
Bahamas	\$0.36	\$0.36	Chad	\$3.48	\$3.48
Bahrain	\$1.12	\$1.12	Chile	\$0.53	\$0.53
Bangladesh	\$1.60	\$1.60	China	\$0.43	\$0.43
Barbados	\$0.82	\$0.82	Christmas Island	\$0.25	\$0.25
Belarus	\$1.01	\$1.01	Cocos Island	\$0.25	\$0.25
Belgium	\$0.26	\$0.26	Colombia	\$0.68	\$0.68
Belize	\$0.98	\$0.98	Comoros Islands	\$4.93	\$4.93
Benin	\$0.92	\$0.92	Congo	\$1.56	\$1.56
Bermuda	\$0.44	\$0.44	Cook Islands	\$2.32	\$2.32
Bhutan	\$2.57	\$2.57	Costa Rica	\$0.54	\$0.54

Original Page 9

Effective: September 1, 2019

- 6.3 Business Optional Calling Plans (continued)
  - 6.3.1 AT&T Business International Calling<sup>SM</sup> Basic (continued)
    - (E) The per-minute usage rates are as follows: (continued)

Country	Initial Minute	Each Additional Minute	Country	Initial Minute	Each Additional Minute
Croatia	\$0.77	\$0.77	Guyana	\$1.37	\$1.37
Cuba	\$2.26	\$2.26	Haiti	\$1.04	\$1.04
Cyprus	\$0.82	\$0.82	Honduras	\$0.96	\$0.96
Czech Republic	\$0.78	\$0.78	Hong Kong	\$0.24	\$0.24
Denmark	\$0.26	\$0.26	Hungary	\$0.71	\$0.71
Diego Garcia	\$6.20	\$6.20	Iceland	\$0.72	\$0.72
Djibouti	\$1.46	\$1.46	India	\$1.04	\$1.04
Dominica	\$0.82	\$0.82	Indonesia	\$0.77	\$0.77
Dominican Republic	\$0.48	\$0.48	Inmarsat (SNAC)	\$15.43	\$15.43
East Timor	\$3.38	\$3.38	Iran	\$1.76	\$1.76
Easter Island	\$0.53	\$0.53	Iraq	\$1.92	\$1.92
Ecuador	\$0.78	\$0.78	Ireland	\$0.26	\$0.26
Egypt	\$1.11	\$1.11	Israel	\$0.42	\$0.42
El Salvador	\$0.74	\$0.74	Italy	\$0.31	\$0.31
Equatorial Guinea	\$3.61	\$3.61	Ivory Coast	\$1.73	\$1.73
Eritrea	\$1.82	\$1.82	Jamaica	\$1.00	\$1.00
Estonia	\$1.27	\$1.27	Japan	\$0.18	\$0.18
Ethiopia	\$1.57	\$1.57	Jordan	\$1.29	\$1.29
Faeroe Island	\$0.84	\$0.84	Kazakhstan	\$0.59	\$0.59
Falkland Island	\$2.48	\$2.48	Kenya	\$1.19	\$1.19
Fiji Island	\$1.63	\$1.63	Kiribati	\$1.85	\$1.85
Finland	\$0.26	\$0.26	Korea (North)	\$4.26	\$4.26
France	\$0.28	\$0.28	Korea (South)	\$0.24	\$0.24
French Antilles	\$0.90	\$0.90	Kosovo	\$0.26	\$0.26
French Guiana	\$0.92	\$0.92	Kuwait	\$1.32	\$1.32
French Polynesia	\$1.85	\$1.85	Kyrgyzstan	\$1.54	\$1.54
Gabon	\$1.52	\$1.52	Laos	\$3.38	\$3.38
Gambia	\$0.95	\$0.95	Latvia	\$1.31	\$1.31
Georgia	\$1.44	\$1.44	Lebanon	\$1.68	\$1.68
Germany	\$0.31	\$0.31	Lesotho	\$1.19	\$1.19
Ghana	\$0.95	\$0.95	Liberia	\$0.90	\$0.90
Gibraltar	\$1.43	\$1.43	Libya	\$1.34	\$1.34
Networks (Global Satellite)	\$7.20	\$7.20	Liechtenstein	\$0.24	\$0.24
Greece	\$0.36	\$0.36	Lithuania	\$1.68	\$1.68
Greenland	\$1.16	\$1.16	Luxembourg	\$0.26	\$0.26
Grenada	\$0.88	\$0.88	Macao	\$1.07	\$1.07
Guadeloupe	\$0.89	\$0.89	Macedonia	\$0.82	\$0.82
Guantanamo Bay	\$2.15	\$2.15	Madagascar	\$3.10	\$3.10
Guatemala	\$0.88	\$0.88	Malawi	\$1.01	\$1.01
Guinea	\$1.14	\$1.14	Malaysia	\$0.53	\$0.53
Guinea-Bissau	\$2.88	\$2.88	Maldives	\$2.02	\$2.02

ce Effective: September 1, 2019 **BUSINESS AND RESIDENTIAL SERVICES** 

Original Page 10

## PRICING AND SERVICE GUIDE SECTION 6 – BUSINESS INTERNATIONAL SERVICE DESCRIPTIONS, RATES AND CHARGES

- 6.3 Business Optional Calling Plans (continued)
  - 6.3.1 AT&T Business International Calling<sup>SM</sup> Basic (continued)
    - (E) The per-minute usage rates are as follows: (continued)

		Each			Each
Country	Initial Minute	Additional	Country	Initial	Additional
Country Mali Republic	Minute \$1.45	<b>Minute</b> \$1.45	Country Romania	<b>Minute</b> \$1.19	<b>Minute</b> \$1.19
Malta Republic	\$1.43	\$1.43	Russia	\$0.59	\$0.59
Marshall Island	\$1.04	\$1.04	Rwanda, Republic of	\$1.64	\$1.64
Mauritania	\$1.37	\$1.37	San Marino	\$1.14	\$1.14
Mauritius	\$1.73	\$1.73	Sao Tome	\$3.80	\$3.80
Mayotte Island	\$1.79	\$1.79	Saudi Arabia	\$1.37	\$1.37
Mexico Band 1	\$0.31	\$0.31	Senegal	\$1.99	\$1.99
Mexico Band 2	\$0.31	\$0.31	Serbia	\$0.93	\$0.93
Micronesia	\$1.24	\$1.24	Seychelles Island	\$2.08	\$2.08
Moldova	\$1.70	\$1.70	Sierra Leone	\$1.39	\$1.39
Monaco	\$0.26	\$0.26	Singapore	\$0.43	\$0.43
Mongolia	\$2.36	\$2.36	Slovakia	\$0.86	\$0.86
Montenegro	\$0.93	\$0.93	Slovenia	\$0.82	\$0.82
Montserrat	\$0.88	\$0.88	Solomon Island	\$2.44	\$2.44
Morocco	\$1.14	\$1.14	Somalia Democratic Rep	\$3.38	\$3.38
Mozambique	\$1.76	\$1.76	South Africa	\$0.66	\$0.66
Myanmar	\$3.14	\$3.14	Spain	\$0.28	\$0.28
Namibia	\$1.10	\$1.10	Sri Lanka	\$1.49	\$1.49
Nauru	\$2.98	\$2.98	St. Helena	\$2.08	\$2.08
Nepal	\$1.79	\$1.79	St. Kitts/Nevis	\$0.73	\$0.73
Netherland	\$0.28	\$0.28	St. Lucia	\$0.76	\$0.76
Netherlands Antilles	\$0.71	\$0.71	St. Maarten	\$0.71	\$0.71
Networks	\$9.90	\$9.90	St. Pierre Miquelon	\$0.92	\$0.92
New Caledonia	\$2.02	\$2.02	St. Vincent/Grenadines	\$0.86	\$0.86
New Zealand	\$0.36	\$0.36	South Sudan	\$1.91	\$1.91
Nicaragua	\$0.92	\$0.92	Sudan	\$1.91	\$1.91
Niger Republic	\$1.28	\$1.28	Suriname	\$1.73	\$1.73
Nigeria	\$1.40	\$1.40	Swaziland	\$1.06	\$1.06
Niue	\$6.20	\$6.20	Sweden	\$0.26	\$0.26
Norfolk Island	\$3.04	\$3.04	Switzerland	\$0.24	\$0.24
Norway	\$0.24	\$0.24	Syrian Arab Republic	\$1.74	\$1.74
Oman Pakistan	\$1.45	\$1.45	Taiwan	\$0.24	\$0.24
Palau	\$1.36 \$1.86	\$1.36 \$1.86	Tajikistan Tanzania	\$0.72 \$1.40	\$0.72 \$1.40
Palestinian Authority	\$0.47	\$0.47	Thailand	\$0.65	\$0.65
Panama	\$0.47	\$0.47	Togo	\$1.52	\$1.52
Papua New Guinea	\$1.72	\$1.72	Tokelau	\$17.98	\$17.98
Paraguay	\$0.86	\$0.86	Tonga Island	\$2.16	\$2.16
Peru	\$0.83	\$0.83	Trinidad & Tobago	\$0.88	\$0.88
Philippines	\$0.59	\$0.83	Tunisia Tobago	\$1.09	\$1.09
Poland	\$0.55	\$0.55	Turkey	\$0.82	\$0.82
Portugal	\$0.26	\$0.26	Turkmenistan	\$1.49	\$1.49
Qatar	\$1.45	\$1.45	Turks and Caicos Islands	0.82	\$0.82
Reunion Island	\$1.79	\$1.79	Tuvalu	\$3.02	\$3.02

Original Page 11

Effective: September 1, 2019

SECTION 6 – BUSINESS INTERNATIONAL SERVICE DESCRIPTIONS, RATES AND CHARGES

- 6.3 Optional Calling Plans (continued)
  - 6.3.1 AT&T Business International Calling<sup>SM</sup> Basic (continued)
    - (E) The per-minute usage rates are as follows: (continued)

Country	Initial Minute	Each Additional Minute	Country	Initial Minute	Each Additional Minute
Uganda	\$1.12	\$1.12	Ukraine	\$0.82	\$0.82
Ukraine	\$0.82	\$0.82	Venezuela	\$0.54	\$0.54
United Arab Emirates	\$0.84	\$0.84	Vietnam	\$1.34	\$1.34
United Kingdom	\$0.14	\$0.14	Wallis Futuna	\$4.03	\$4.03
Uruguay	\$0.86	\$0.86	Western Samoa	\$1.43	\$1.43
Uzbekistan	\$0.77	\$0.77	Yemen Arab Republic	\$1.45	\$1.45
Vanuatu	\$8.70	\$8.70	Zaire	\$1.09	\$1.09
Vatican City	\$0.31	\$0.31	Zambia	\$1.18	\$1.18
Uganda	\$1.12	\$1.12	Zimbabwe	\$1.09	\$1.09

(F) Term Plans

Term plan discounts are not available with this plan.

(G) This Service will remain in effect until either: (a) cancelled or changed by the Customer or (b) until the Company no longer offers this Service to new Customers or existing subscribers moving to a new location, whichever occurs first. Changes to this Service will be effective on the day the Customer's order is processed.

SECTION 6 – BUSINESS INTERNATIONAL SERVICE DESCRIPTIONS, RATES AND CHARGES

#### 6.3 Business Optional Calling Plans (continued)

#### 6.3.2 AT&T Business International Calling<sup>SM</sup> Standard

(A) AT&T Business International Calling<sup>SM</sup> Standard is a direct dialed outbound international long distance service offered to Customers on switched access lines. Service is offered to each country on a flat rate basis, twenty-four hours per day, seven days per week. Customers must dial 1 plus the country code or area code, if applicable, and the terminating number to complete a call utilizing this Service.

Original Page 12

Effective: September 1, 2019

- (B) All calls are billed in six (6) second increments after an initial period, for billing purposes, of sixty (60) seconds. This Service is only offered in conjunction with one of the Company's Business Domestic Services.
- (C) Inbound international toll-free service originated from locations within Canada is available to Customers of this service who also subscribe to AT&T Long Distance Toll Free<sup>SM</sup> Service in Section 4.4 of this pricing and service guide. The rate table in Section 6.3.2 (E) also contains the per-minute rate for inbound toll-free service originating in Canada and terminating throughout the United States.
- (D) International Mobile Termination Charge

See Section 6.4.1 of this pricing and serivce guide for the applicable International Mobile Termination Charge.

(E) The MRC is \$10.00, and the per-minute usage rates are as follows:

Country	Initial Minute	Each Additional Minute	Country	Initial Minute	Each Additional Minute
Afghanistan	\$1.28	\$1.28	Bolivia	\$0.42	\$0.42
Albania	\$0.87	\$0.87	Bosnia	\$0.62	\$0.62
Algeria	\$0.55	\$0.55	Botswana	\$0.64	\$0.64
Andorra	\$0.36	\$0.36	Brazil	\$0.29	\$0.29
Angola	\$1.16	\$1.16	British Virgin Islands	\$0.36	\$0.36
Anguilla	\$0.50	\$0.50	Brunei	\$0.66	\$0.66
Antarctica (Casey Base)	\$1.52	\$1.52	Bulgaria	\$0.51	\$0.51
Antarctica (Scott Base)	\$1.52	\$1.52	Burkina Faso	\$0.71	\$0.71
Antigua	\$0.41	\$0.41	Burundi	\$1.38	\$1.38
Argentina	\$0.37	\$0.37	Cambodia	\$1.77	\$1.77
Armenia	\$0.75	\$0.75	Cameroon	\$0.78	\$0.78
Aruba	\$0.36	\$0.36	Canada – Outbound	\$0.05	\$0.05
Ascension Island	\$1.18	\$1.18	Canada - Inbound	\$0.05	\$0.05
Australia	\$0.14	\$0.14	Cape Verde Island	\$0.64	\$0.64
Austria	\$0.13	\$0.13	Cayman Islands	\$0.30	\$0.30
Azerbaijan	\$0.96	\$0.96	Central African Republic	\$1.18	\$1.18
Bahamas	\$0.18	\$0.18	Chad	\$1.93	\$1.93
Bahrain	\$0.63	\$0.63	Chile	\$0.27	\$0.27
Bangladesh	\$0.84	\$0.84	China	\$0.22	\$0.22
Barbados	\$0.41	\$0.41	Christmas Island	\$0.14	\$0.14
Belarus	\$0.57	\$0.57	Cocos Island	\$0.14	\$0.14
Belgium	\$0.13	\$0.13	Colombia	\$0.34	\$0.34
Belize	\$0.49	\$0.49	Comoros Islands	\$2.79	\$2.79
Benin	\$0.52	\$0.52	Congo	\$0.88	\$0.88
Bermuda	\$0.22	\$0.22	Cook Islands	\$1.31	\$1.31
Bhutan	\$1.45	\$1.45	Costa Rica	\$0.27	\$0.27

Effective: September 1, 2019

Original Page 13

## BUSINESS AND RESIDENTIAL SERVICES PRICING AND SERVICE GUIDE

- 6.3 Business Optional Calling Plans (continued)
  - 6.3.2 AT&T Business International Calling<sup>SM</sup> Standard (continued)
    - (E) The per-minute usage rates are as follows: (continued)

	Initial	Each Additional		Initial	Each Additional
Country	Minute	Minute	Country	Minute	Minute
Croatia	\$0.43	\$0.43	Guinea-Bissau	\$1.44	\$1.44
Cuba	\$1.13	\$1.13	Guyana	\$0.77	\$0.77
Cyprus	\$0.46	\$0.46	Haiti	\$0.52	\$0.52
Czech Republic	\$0.44	\$0.44	Honduras	\$0.43	\$0.43
Denmark	\$0.13	\$0.13	Hong Kong	\$0.12	\$0.12
Diego Garcia	\$3.10	\$3.10	Hungary	\$0.33	\$0.33
Djibouti	\$0.76	\$0.76	Iceland	\$0.40	\$0.40
Dominica	\$0.42	\$0.42	India	\$0.52	\$0.52
Dominican Republic	\$0.24	\$0.24	Indonesia	\$0.39	\$0.39
East Timor	\$2.41	\$2.41	Inmarsat (SNAC)	\$8.99	\$8.99
Easter Island	\$0.27	\$0.27	Iran	\$0.99	\$0.99
Ecuador	\$0.39	\$0.39	Iraq	\$0.96	\$0.96
Egypt	\$0.59	\$0.59	Ireland	\$0.13	\$0.13
El Salvador	\$0.37	\$0.37	Israel	\$0.21	\$0.21
Equatorial Guinea	\$2.04	\$2.04	Italy	\$0.13	\$0.13
Eritrea	\$0.98	\$0.98	Ivory Coast	\$0.97	\$0.97
Estonia	\$0.72	\$0.72	Jamaica	\$0.50	\$0.50
Ethiopia	\$0.86	\$0.86	Japan	\$0.09	\$0.09
Faeroe Island	\$0.47	\$0.47	Jordan	\$0.71	\$0.71
Falkland Island	\$1.24	\$1.24	Kazakhstan	\$0.30	\$0.30
Fiji Island	\$0.82	\$0.82	Kenya	\$0.67	\$0.67
Finland	\$0.13	\$0.13	Kiribati	\$1.04	\$1.04
France	\$0.13	\$0.13	Korea (North)	\$2.41	\$2.41
French Antilles	\$0.51	\$0.51	Korea (South)	\$0.12	\$0.12
French Guiana	\$0.51	\$0.51	Kosovo	\$0.13	\$0.13
French Polynesia	\$1.04	\$1.04	Kuwait	\$0.74	\$0.74
Gabon	\$0.86	\$0.86	Kyrgyztan	\$0.87	\$0.87
Gambia	\$0.54	\$0.54	Laos	\$1.91	\$1.91
Georgia	\$0.81	\$0.81	Latvia	\$0.74	\$0.74
Germany	\$0.13	\$0.13	Lebanon	\$0.95	\$0.95
Ghana	\$0.54	\$0.54	Lesotho	\$0.67	\$0.67
Gibraltar	\$0.80	\$0.80	Liberia	\$0.51	\$0.51
Global Satellite	\$6.00	\$6.00	Libya	\$0.76	\$0.76
Greece	\$0.18	\$0.18	Liechtenstein	\$0.12	\$0.12
Greenland	\$0.58	\$0.58	Lithuania	\$0.95	\$0.95
Grenada	\$0.44	\$0.44	Luxembourg	\$0.13	\$0.13
Guadeloupe	\$0.50	\$0.50	Macao	\$0.60	\$0.60
Guantanamo Bay	\$1.21	\$1.21	Macedonia	\$0.46	\$0.46
Guatemala	\$0.41	\$0.41	Madagascar	\$1.67	\$1.67
Guinea	\$0.64	\$0.64	Malawi	\$0.57	\$0.57

BUSINESS AND RESIDENTIAL SERVICES

Original Page 14

Effective: September 1, 2019

PRICING AND SERVICE GUIDE

SECTION 6 – BUSINESS INTERNATIONAL SERVICE DESCRIPTIONS, RATES AND CHARGES

- 6.3 Optional Calling Plans (continued)
  - 6.3.2 AT&T Business International Calling<sup>SM</sup> Standard (continued)
    - (E) The per-minute usage rates are as follows: (continued)

Country	Initial Minute	Each Additional Minute	Country	Initial Minute	Each Additional Minute
Malaysia	\$0.27	\$0.27	Portugal	\$0.13	\$0.13
Maldives	\$1.14	\$1.14	Qatar	\$0.82	\$0.82
Mali Republic	\$0.82	\$0.82	Reunion Island	\$0.90	\$0.90
Malta Republic	\$0.57	\$0.57	Romania	\$0.67	\$0.67
Marshall Island	\$0.57	\$0.57	Russia	\$0.30	\$0.30
Mauritania	\$0.77	\$0.77	Rwanda, Republic of	\$0.93	\$0.93
Mauritius	\$0.97	\$0.97	San Marino	\$0.60	\$0.60
Mayotte Island	\$0.90	\$0.90	Sao Tome	\$1.90	\$1.90
Mexico Band 1	\$0.15	\$0.15	Saudi Arabia	\$0.73	\$0.73
Mexico Band 2	\$0.15	\$0.15	Senegal	\$1.13	\$1.13
Micronesia	\$0.62	\$0.62	Serbia	\$0.52	\$0.52
Moldova	\$0.96	\$0.96	Seychelles Island	\$1.17	\$1.17
Monaco	\$0.13	\$0.13	Sierra Leone	\$0.78	\$0.78
Mongolia	\$1.33	\$1.33	Singapore	\$0.23	\$0.23
Montenegro	\$0.52	\$0.52	Slovakia	\$0.49	\$0.49
Montserrat	\$0.44	\$0.44	Slovenia	\$0.46	\$0.46
Morocco	\$0.64	\$0.64	Solomon Island	\$1.22	\$1.22
Mozambique	\$0.99	\$0.99	Somalia Democratic Republic	\$1.69	\$1.69
Myanmar	\$1.70	\$1.70	South Africa	\$0.32	\$0.32
Namibia	\$0.62	\$0.62	Spain	\$0.13	\$0.13
Nauru	\$1.49	\$1.49	Sri Lanka	\$0.84	\$0.84
Nepal	\$1.01	\$1.01	St. Helena	\$1.04	\$1.04
Netherlands	\$0.13	\$0.13	St. Kitts/Nevis	\$0.41	\$0.41
Netherlands Antilles	\$0.33	\$0.33	St. Lucia	\$0.41	\$0.41
Networks	\$6.00	\$6.00	St. Maarten	\$0.33	\$0.33
New Caledonia	\$1.14	\$1.14	St. Pierre Miquelon	\$0.46	\$0.46
New Zealand	\$0.20	\$0.20	St. Vincent/Grenadines	\$0.49	\$0.49
Nicaragua	\$0.43	\$0.43	South Sudan	\$1.08	\$1.08
Niger Republic	\$0.72	\$0.72	Sudan	\$1.08	\$1.08
Nigeria	\$0.65	\$0.65	Suriname	\$0.97	\$0.97
Niue	\$3.10	\$3.10	Swaziland	\$0.59	\$0.59
Norfolk Island	\$1.52	\$1.52	Sweden	\$0.13	\$0.13
Norway	\$0.12	\$0.12	Switzerland	\$0.12	\$0.12
Oman	\$0.82	\$0.82	Syrian Arab Republic	\$0.98	\$0.98
Pakistan	\$0.68	\$0.68	Taiwan	\$0.12	\$0.12
Palau	\$1.05	\$1.05	Tajikistan	\$0.36	\$0.36
Palestinian Authority	\$0.39	\$0.39	Tanzania	\$0.79	\$0.79
Panama	\$0.39	\$0.39	Thailand	\$0.33	\$0.33
Papua New Guinea	\$0.86	\$0.86	Togo	\$0.86	\$0.86
Paraguay	\$0.49	\$0.49	Tokelau	\$8.99	\$8.99
Peru	\$0.42	\$0.42	Tonga Island	\$1.22	\$1.22
Philippines	\$0.30	\$0.30	Trinidad & Tobago	\$0.44	\$0.44
Poland	\$0.31	\$0.31	Tunisia	\$0.61	\$0.61

Original Page 15

Effective: September 1, 2019

SECTION 6 – BUSINESS INTERNATIONAL SERVICE DESCRIPTIONS, RATES AND CHARGES

- 6.3 Business Optional Calling Plans (continued)
  - 6.3.2 AT&T Business International Calling<sup>SM</sup> Standard (continued)
    - (E) The per-minute usage rates are as follows: (continued)

Country	Initial Minute	Each Additional Minute	Country	Initial Minute	Each Additional Minute
Turkey	\$0.38	\$0.38	Vanuatu	\$4.35	\$4.35
Turkmenistan	\$0.84	\$0.84	Vatican	\$0.18	\$0.18
Turks and Caicos Islands	\$0.46	\$0.46	Venezuela	\$0.27	\$0.27
Tuvalu	\$1.51	\$1.51	Vietnam	\$0.67	\$0.67
Uganda	\$0.63	\$0.63	Wallis Futuna	\$2.28	\$2.28
Ukraine	\$0.41	\$0.41	Western Samoa	\$0.80	\$0.80
United Arab Emirates	\$0.47	\$0.47	Yemen Arab Republic	\$0.82	\$0.82
United Kingdom	\$0.06	\$0.06	Zaire	\$0.61	\$0.61
Uruguay	\$0.49	\$0.49	Zambia	\$0.66	\$0.66
Uzbekistan	\$0.39	\$.039	Zimbabwe	\$0.61	\$0.61

(F) Term Plans

Term plan discounts are not available with this plan.

(G) This Service will remain in effect until either: (a) cancelled or changed by the Customer or (b) until the Company no longer offers this Service to new Customers or existing subscribers moving to a new location, whichever occurs first. Changes to this Service will be effective on the day the Customer's order is processed.

Original Page 16

Effective: September 1, 2019

SECTION 6 – BUSINESS INTERNATIONAL SERVICE DESCRIPTIONS, RATES AND CHARGES

#### 6.4 International Mobile Termination Charge

The International Mobile Termination Charge (IMTC) is an additional per-minute charge that is applied to all international calls originating in the United States and terminating in certain countries to (1) wireless devices; and (2) ranges of fixed wireless telephone numbers which the foreign telecommunications company designates to be for special services. Because many foreign telecommunications companies impose extra fees on the Company to complete these calls, the Company charges the IMTC, in addition to all other applicable charges. The IMTC amount varies by country and changes from time to time. Please check codes and rates periodically for current information. Codes can be found at <a href="https://www.att.com/mobileterm">www.att.com/mobileterm</a>. The following per-minute charge listed in Section 6.4.1 below will apply.

SECTION 6 – BUSINESS INTERNATIONAL SERVICE DESCRIPTIONS, RATES AND CHARGES

### 6.4 International Mobile Termination Charge (continued)

### 6.4.1 Per Country Mobile Termination Charges

		Each
G .	Initial	Additional
Country	Minute	Minute
Afghanistan	\$0.0300	\$0.0300
Albania	\$0.1800	\$0.1800
Algeria	\$0.0800	\$0.0800
Andorra	\$0.2300	\$0.2300
Angola	\$0.0800	\$0.0800
Anguilla	\$0.1100	\$0.1100
Argentina	\$0.1800	\$0.1800
Armenia	\$0.1300	\$0.1300
Aruba	\$0.1000	\$0.1000
Australia (including Christmas & Cocos Isls)	\$0.1300	\$0.1300
Austria	\$0.1200	\$0.1200
Azerbaijan	\$0.1200	\$0.1200
Bahrain	\$0.0300	\$0.0300
Bangladesh	\$0.0200	\$0.0200
Barbados	\$0.0900	\$0.0900
Belarus	\$0.0400	\$0.0400
Belgium	\$0.1600	\$0.1600
Belize	\$0.0200	\$0.0200
Benin	\$0.0500	\$0.0500
Bolivia	\$0.0400	\$0.0400
Bosnia	\$0.1600	\$0.1600
Botswana	\$0.1300	\$0.1300
Brazil	\$0.1900	\$0.1900
British Virgin Island	\$0.1100	\$0.1100
Brunei	\$0.0200	\$0.0200
Bulgaria	\$0.3300	\$0.3300
Burkina Faso	\$0.0600	\$0.0600
Cambodia	\$0.0200	\$0.0200
Cameroon	\$0.0900	\$0.0900
Cape Verde Island	\$0.1100	\$0.1100
Cayman Islands	\$0.0900	\$0.0900
Chile	\$0.1800	\$0.1800
China	\$0.0000	\$0.0000
Christmas Island	(See Australia)	(See Australia)
Cocos Island	(See Australia)	(See Australia)
Colombia	\$0.0300	\$0.0300
Comoros	\$0.0900	\$0.0900
Costa Rica	\$0.0200	\$0.0200
Croatia	\$0.1800	\$0.1800
Cyprus	\$0.0500	\$0.0500
Czech Republic	\$0.1800	\$0.1800
Denmark	\$0.1600	\$0.1600
Djibouti	\$0.0000	\$0.0000
Dominica	\$0.1000	\$0.1000
Dominican Republic	\$0.0700	\$0.0700
	Ψ0.0700	40.0700

G .	Initial	Each Additional
Country	Minute	Minute
Easter Islands	\$0.1800	\$0.1800
Ecuador	\$0.1100	\$0.1100
Egypt	\$0.0200	\$0.0200
El Salvador	\$0.1000	\$0.1000
Equatorial Guinea	\$0.0200	\$0.0200
Estonia	\$0.2900	\$0.2900
Ethiopia	\$0.0400	\$0.0400
Fiji	\$0.0900	\$0.0900
Finland	\$0.1100	\$0.1100
France	\$0.1200	\$0.1200
French Antilles	(See Martinique)	
French Guiana	\$0.0900	\$0.0900
French Polynesia	\$0.0800	\$0.0800
Gabon	\$0.0500	\$0.0500
Gambia	\$0.0000	\$0.0000
Georgia	\$0.0900	\$0.0900
Germany	\$0.1600	\$0.1600
Ghana	\$0.0700	\$0.0700
Gibraltar	\$0.2100	\$0.2100
Greece	\$0.1700	\$0.1700
Greenland	\$0.1700	\$0.1700
Grenada	\$0.1000	\$0.1000
Guadeloupe	\$0.2800	\$0.2800
Guatemala	\$0.0500	\$0.0500
Guinea Per	\$0.0300	\$0.0300
Haiti	\$0.0800	\$0.0800
Honduras	\$0.0200	\$0.0200
Hungary	\$0.1500	\$0.1500
Iceland	\$0.1900	\$0.1900
India	\$0.0000	\$0.0000
Indonesia	\$0.0200	\$0.0200
Iran	\$0.0000	\$0.0000
Iraq	\$0.0700	\$0.0700
Ireland	\$0.1700	\$0.1700
Israel	\$0.0800	\$0.0800
Italy	\$0.1600	\$0.1600
Ivory Coast	\$0.0600	\$0.0600
Jamaica Jamaica	\$0.1100	\$0.1100
Japan	\$0.1100	\$0.1100
Jordan	\$0.0600	\$0.0600
Kazakhstan	\$0.0300	\$0.0300
Kenya	\$0.0500	\$0.0500
Korea (South)	\$0.0300	\$0.0300
Kosovo	\$0.0300	\$0.2500
Kuwait	\$0.2500	\$0.2500
uxuwan	φυ.υσυυ	φυ.υ.ου

Original Page 17

Effective: September 1, 2019

Original Page 18

## BUSINESS AND RESIDENTIAL SERVICES PRICING AND SERVICE GUIDE

SECTION 6 – BUSINESS INTERNATIONAL SERVICE DESCRIPTIONS, RATES AND CHARGES

### 6.4 International Mobile Termination Charge (continued)

#### 6.4.1 Per Country Mobile Termination Charges (continued)

Country	Initial Minute	Each Additional Minute
Kyrgyzstan	\$0.0000	\$0.0000
Laos	\$0.0000	\$0.0000
Latvia	\$0.1300	\$0.1300
Lebanon	\$0.0900	\$0.0900
Lesotho	\$0.0600	\$0.0600
Libyan	\$0.0500	\$0.0500
Liechtenstein	\$0.5000	\$0.5000
Lithuania	\$0.1400	\$0.1400
Luxembourg	\$0.2200	\$0.2200
Macedonia	\$0.2200	\$0.2200
Madagascar	\$0.0000	\$0.0000
Malawi	\$0.0200	\$0.0200
Malaysia	\$0.0400	\$0.0400
Mali	\$0.1100	\$0.1100
Malta	\$0.1700	\$0.1700
Martinique (includes	\$0.2100	\$0.2100
Martinique (includes St. Barthelemy, St. Martin and French Antilles)		
Mauritania	\$0.0900	\$0.0900
Mauritius	\$0.0400	\$0.0400
Mayotte	(See Reunion Isl)	(See Reunion Isl)
Mexico (All Zones)	\$0.1500	\$0.1500
Moldova	\$0.0900	\$0.0900
Monaco	\$0.2500	\$0.2500
Mongolia	\$0.0700	\$0.0700
Montenegro	\$0.2600	\$0.2600
Morocco	\$0.1500	\$0.1500
Mozambique	\$0.1400	\$0.1400
Myanmar (Burma)	\$0.1000	\$0.1000
Namibia	\$0.1400	\$0.1400
Nepal	\$0.1000	\$0.1000
Netherlands	\$0.2200	\$0.2200
Netherland Antilles	\$0.0300	\$0.0300
New Zealand	\$0.2100	\$0.2100
Nicaragua	\$0.0800	\$0.0800
Nigeria	\$0.0600	\$0.0600
Norway	\$0.1500	\$0.1500
Oman	\$0.0700	\$0.0700
Pakistan	\$0.0000	\$0.0000
Panama	\$0.0900	\$0.0900
Paraguay	\$0.1400	\$0.1400
Peru	\$0.1600	\$0.1600
Philippines	\$0.0400	\$0.0400
Poland	\$0.1700	\$0.1700
Portugal	\$0.1600	\$0.1600
Qatar	\$0.0400	\$0.0400
Reunion Island (& Mayotte Island)	\$0.2100	\$0.2100
Romania	\$0.1400	\$0.1400

	1	
	Initial	Each Additional
Country	Minute	Minute
Russia	\$0.0300	\$0.0300
Rwanda	\$0.0200	\$0.0200
Saudi Arabia	\$0.0500	\$0.0500
Senegal	\$0.0900	\$0.0900
Serbia	\$0.1800	\$0.1800
Sierra Leone	\$0.0000	\$0.0000
Singapore	\$0.0000	\$0.0000
Slovakia	\$0.1800	\$0.1800
Slovenia	\$0.2800	\$0.2800
South Africa	\$0.1300	\$0.1300
South Sudan	\$0.2300	\$02300
Spain	\$0.1400	\$0.1400
Sri Lanka	\$0.0300	\$0.0300
St. Kitts/Nevis	\$0.1100	\$0.1100
St. Lucia	\$0.1000	\$0.1000
St. Eucla	φ0.1000	ψ0.1000
St. Maarten		
	\$0.0300	\$0.0300
St. Pierre/	\$0.1900	\$0.1900
Miquelon		
St. Vincent	\$0.1000	\$0.1000
(& the Grenadines)	,	
Sudan	\$0.0300	\$0.0300
Suriname	\$0.0000	\$0.0000
Swaziland	\$0.1900	\$0.1900
Sweden	\$0.1400	\$0.1400
Switzerland	\$0.2100	\$0.2100
Syria	\$0.0500	\$0.0500
Taiwan	\$0.0700	\$0.0700
Tanzania	\$0.0900	\$0.0900
Thailand	\$0.0000	\$0.0000
Trinidad & Tobago	\$0.0900	\$0.0900
Tunisia	\$0.1800	\$0.1800
Turkey	\$0.1400	\$0.1400
Turks & Caicos	\$0.1100	\$0.1100
Islands Uganda	\$0.0200	\$0.0200
Ukraine	\$0.0200	\$0.0200
United Arab	\$0.0700	\$0.0700
Emirates	\$0.0200	\$0.0200
United Kingdom	\$0.2000	\$0.2000
Uruguay	\$0.1300	\$0.2000
Uzbekistan	\$0.0200	\$0.0200
Vatican City	\$0.0200	\$0.0200
Vanezuela	\$0.1500	\$0.1500
Vietnam	\$0.0200	\$0.0200
Western Samoa	\$0.0200	\$0.0200
Yemen	\$0.0200	\$0.0200
Zaire	\$0.0200	\$0.0200
Zambia	\$0.1200	\$0.1200
Zimbabwe	\$0.2800	\$0.2800

Original Page 1

Effective: September 1, 2019

SECTION 7 - RESIDENTIAL INTERNATIONAL SERVICE DESCRIPTIONS, RATES AND CHARGES

#### 7.1 General

- 7.1.1 Customers are billed based on their usage of the Company's services. Rates may vary by service type, call duration, time of day, rate period, day of week, mileage, call type, calling plan and foreign location. Fixed recurring charges, not dependent upon usage, are billed in advance. Usage-based charges are billed after each usage cycle.
- 7.1.2 Customers will be charged for each individual call placed through the Company. Call duration is determined as described in Section 3 of this pricing and service guide. For distance or time of day sensitive offerings, charges are determined based on mileage calculations and applicable rate periods found in Section 3 unless otherwise specified in this pricing and service guide.
- 7.1.3 Outbound direct dial services are available from originating locations within the United States. Calls may be placed between the United States and those countries listed herein.
- 7.1.4 Operator Services are available from originating locations within the United States where Customers have the ability to dial directly to the Company's network. Calls may be placed between the United States and those countries listed herein.
- 7.1.5 Calls to or from a foreign country or area, or calls routed through a foreign country or area, are subject to such restrictions as may be enforced from time to time by the authorities in that country or area.
- 7.1.6 An additional per-minute International Mobile Termination Charge (IMTC) is applied to all international calls originating in the United States and terminating in certain countries to (1) wireless devices; and (2) ranges of fixed wireless telephone numbers which the foreign telecommunications company designates to be for special services. Because many foreign telecommunications companies impose extra fees on the Company to complete these calls, the Company charges the IMTC, in addition to all other applicable charges. The IMTC amount varies by country and changes from time to time. Please check codes and rates periodically for current information. Codes and rates can be found at <a href="www.att.com/mobileterm">www.att.com/mobileterm</a>. The applicable International Mobile Termination Charge can also be found in Section 7.4.1 of this pricing and service guide.

Original Page 2

Effective: September 1, 2019

SECTION 7 – RESIDENTIAL INTERNATIONAL SERVICE DESCRIPTIONS, RATES AND CHARGES

#### 7.2 Residential International Message Telecommunications Service (IMTS)

The Residential IMTS is a direct dialed outbound international long distance service offered to residential Customers from switched access lines. Service is offered twenty-four hours per day, seven days per week. Charges depend on the country called, the time of day in which the call was made and the duration of the call. Customers must dial 1 plus the country code or area code, if applicable, and the terminating number to complete a call utilizing this service. To be eligible for this service Customers must subscribe to local exchange service provided by the Company's Affiliated ILEC.

All calls are billed in one (1) minute increments after an initial period, for billing purpose, of one (1) minute. This service is only offered in conjunction with a Company Domestic Service. Amonthly recurring charge does not apply for this service.

#### 7.2.1 Time of Day Rate Periods

Time of day rates, as defined in Section 3 of this pricing and service guide, apply to this plan as well. Holiday discounts do not apply to this service.

Effective: September 1, 2019

Original Page 3

## BUSINESS AND RESIDENTIAL SERVICES PRICING AND SERVICE GUIDE

SECTION 7 – RESIDENTIAL INTERNATIONAL SERVICE DESCRIPTIONS, RATES AND CHARGES

### 7.2 Residential International Message Telecommunications Service (IMTS) (continued)

#### 7.2.2 Per Minute Rate Tables

	Standard		Eco	Economy		
		Each		Each		
Commenter Norma		Additional		Additional		
Country Name	Minute		Minute			
Afghanistan	\$4.75	\$4.75	\$4.75	\$4.75		
Albania	\$5.00	\$5.00	\$5.00	\$5.00		
Algeria	\$5.00	\$5.00	\$5.00	\$5.00		
Andorra	\$4.00	\$4.00	\$4.00	\$4.00		
Angola	\$5.00	\$5.00	\$5.00	\$5.00		
Anguilla	\$4.50	\$4.50	\$4.50	\$4.50		
Antarctica (Scott Base)	\$5.00	\$5.00	\$5.00	\$5.00		
Antigua	\$4.00	\$4.00	\$4.00	\$4.00		
Argentina	\$5.00	\$5.00	\$5.00	\$5.00		
Armenia	\$5.00	\$5.00	\$5.00	\$5.00		
Aruba	\$4.00	\$4.00	\$4.00	\$4.00		
Ascension Island	\$5.00	\$5.00	\$5.00	\$5.00		
Australia	\$4.25	\$4.25	\$4.25	\$4.25		
Australia External Territories*	\$4.25	\$4.25	\$4.25	\$4.25		
Austria	\$4.00	\$4.00	\$4.00	\$4.00		
Azerbaijan	\$5.00	\$5.00	\$5.00	\$5.00		
Bahamas	\$3.75	\$3.75	\$3.75	\$3.75		
Bahrain	\$5.00	\$5.00	\$5.00	\$5.00		
Bangladesh	\$5.00	\$5.00	\$5.00	\$5.00		
Barbados	\$4.00	\$4.00	\$4.00	\$4.00		
Belarus	\$5.00	\$5.00	\$5.00	\$5.00		
Belgium	\$4.00	\$4.00	\$4.00	\$4.00		
Belize	\$5.00	\$5.00	\$5.00	\$5.00		
Benin	\$5.00	\$5.00	\$5.00	\$5.00		
Bermuda	\$3.50	\$3.50	\$3.50	\$3.50		
Bhutan	\$5.00	\$5.00	\$5.00	\$5.00		
Bolivia	\$5.00	\$5.00	\$5.00	\$5.00		
Bosnia and Herzegovina	\$5.00	\$5.00	\$5.00	\$5.00		
Botswana	\$5.00	\$5.00	\$5.00	\$5.00		
Brazil	\$5.00	\$5.00	\$5.00	\$5.00		
British Virgin Islands	\$4.00	\$4.00	\$4.00	\$4.00		
Brunei	\$5.00	\$5.00	\$5.00	\$5.00		
Bulgaria	\$5.00	\$5.00	\$5.00	\$5.00		
Burkina Faso	\$5.00	\$5.00	\$5.00	\$5.00		
Burundi	\$5.00	\$5.00	\$5.00	\$5.00		
Cambodia	\$5.00	\$5.00	\$5.00	\$5.00		
Cameroon	\$5.00	\$5.00	\$5.00	\$5.00		
Canada	\$1.55	\$1.55	\$1.55	\$1.55		
Cape Verde Island	\$5.00	\$5.00	\$5.00	\$5.00		
Cayman Islands	\$4.50	\$4.50	\$4.50	\$4.50		
Central African Republic	\$5.00	\$5.00	\$5.00	\$5.00		
Chad	\$5.00	\$5.00	\$5.00	\$5.00		
Chile	\$4.50	\$4.50	\$4.50	\$4.50		
China	\$5.00	\$5.00	\$5.00	\$5.00		
Colombia	\$5.00	\$5.00	\$5.00	\$5.00		
Comoros Islands	\$5.00	\$5.00	\$5.00	\$5.00		
Congo	\$5.00	\$5.00	\$5.00	\$5.00		
Cook Islands	\$5.00	\$5.00	\$5.00	\$5.00		
Costa Rica	\$4.50	\$4.50	\$4.50	\$4.50		
Croatia	\$5.00	\$5.00	\$5.00	\$5.00		

<sup>\*</sup> Australian External Territories include – Antarctica-Casey Base and Christmas, Cocos and Norfolk Islands

Effective: September 1, 2019

Original Page 4

## PRICING AND SERVICE GUIDE SECTION 7 – RESIDENTIAL INTERNATIONAL SERVICE DESCRIPTIONS, RATES AND CHARGES

**BUSINESS AND RESIDENTIAL SERVICES** 

### 7.2 Residential International Message Telecommunications Service (IMTS) (continued)

	Standard		Economy		
		Each		Each	
		Additional		Additional	
Country Name	Minute		Minute		
Cuba	\$5.00	\$5.00	\$5.00	\$5.00	
Cyprus	\$5.00	\$5.00	\$5.00	\$5.00	
Czech Republic	\$5.00	\$5.00	\$5.00	\$5.00	
Denmark	\$4.00	\$4.00	\$4.00	\$4.00	
Diego Garcia	\$5.00	\$5.00	\$5.00	\$5.00	
Djibouti	\$5.00	\$5.00	\$5.00	\$5.00	
Dominica	\$4.50	\$4.50	\$4.50	\$4.50	
Dominican Republic	\$4.75	\$4.75	\$4.75	\$4.75	
East Timor	\$5.00	\$5.00	\$5.00	\$5.00	
Easter Island	\$4.50	\$4.50	\$4.50	\$4.50	
Ecuador	\$5.00	\$5.00	\$5.00	\$5.00	
Egypt	\$5.00	\$5.00	\$5.00	\$5.00	
El Salvador	\$5.00	\$5.00	\$5.00	\$5.00	
Equatorial Guinea	\$5.00	\$5.00	\$5.00	\$5.00	
Eritrea	\$5.00	\$5.00	\$5.00	\$5.00	
Estonia	\$5.00	\$5.00	\$5.00	\$5.00	
Ethiopia	\$5.00	\$5.00	\$5.00	\$5.00	
Faeroe Island	\$3.50	\$3.50	\$3.50	\$3.50	
Falkland Island	\$5.00	\$5.00	\$5.00	\$5.00	
Fiji Island	\$5.00	\$5.00	\$5.00	\$5.00	
Finland	\$3.50	\$3.50	\$3.50	\$3.50	
France	\$4.25	\$4.25	\$4.25	\$4.25	
French Antilles	\$4.00	\$4.00	\$4.00	\$4.00	
French Guiana	\$4.50	\$4.50	\$4.50	\$4.50	
French Polynesia	\$5.00	\$5.00	\$5.00	\$5.00	
Gabon	\$5.00	\$5.00	\$5.00	\$5.00	
Gambia	\$5.00	\$5.00	\$5.00	\$5.00	
Georgia	\$5.00	\$5.00	\$5.00	\$5.00	
Germany	\$3.75	\$3.75	\$3.75	\$3.75	
Ghana	\$5.00	\$5.00	\$5.00	\$5.00	
Gibraltar	\$4.50	\$4.50	\$4.50	\$4.50	
Greece	\$5.00	\$5.00	\$5.00	\$5.00	
Greenland	\$4.50	\$4.50	\$4.50	\$4.50	
Grenada	\$4.50	\$4.50	\$4.50	\$4.50	
Guadeloupe	\$4.50	\$4.50	\$4.50	\$4.50	
Guatemala	\$5.00	\$5.00	\$5.00	\$5.00	
Guinea	\$5.00	\$5.00	\$5.00	\$5.00	
Guinea-Bissau	\$5.00	\$5.00	\$5.00	\$5.00	
Guyana	\$5.00	\$5.00	\$5.00	\$5.00	
Haiti	\$5.00	\$5.00	\$5.00	\$5.00	
Honduras	\$5.00	\$5.00	\$5.00	\$5.00	
Hong Kong	\$5.00	\$5.00	\$5.00	\$5.00	
Hungary	\$4.50	\$4.50	\$4.50	\$4.50	
Iceland	\$4.50	\$4.50	\$4.50	\$4.50	
India	\$5.00	\$5.00	\$5.00	\$5.00	
Indonesia	\$5.00	\$5.00	\$5.00	\$5.00	

Original Page 5 Effective: September 1, 2019

## BUSINESS AND RESIDENTIAL SERVICES PRICING AND SERVICE GUIDE

SECTION 7 – RESIDENTIAL INTERNATIONAL SERVICE DESCRIPTIONS, RATES AND CHARGES

### 7.2 Residential International Message Telecommunications Service (MTS) (continued)

	St	andard	Economy	
		Each		Each
G . N	Initial	Additional	Initial	Additional
Country Name	Minute	Minute	Minute	Minute
Inmarsat (SNAC)	\$5.00	\$5.00	\$5.00	\$5.00
Iran	\$5.00	\$5.00	\$5.00	\$5.00
Iraq	\$5.00	\$5.00	\$5.00	\$5.00
Ireland	\$4.25	\$4.25	\$4.25	\$4.25
Israel	\$5.00	\$5.00	\$5.00	\$5.00
Italy	\$4.75	\$4.75	\$4.75	\$4.75
Ivory Coast	\$5.00	\$5.00	\$5.00	\$5.00
Jamaica	\$5.00	\$5.00	\$5.00	\$5.00
Japan	\$4.50	\$4.50	\$4.50	\$4.50
Jordan	\$5.00	\$5.00	\$5.00	\$5.00
Kazakhstan	\$5.00	\$5.00	\$5.00	\$5.00
Kenya	\$5.00	\$5.00	\$5.00	\$5.00
Kiribati	\$5.00	\$5.00	\$5.00	\$5.00
Korea (North)	\$5.00	\$5.00	\$5.00	\$5.00
Korea (South)	\$5.00	\$5.00	\$5.00	\$5.00
Kosovo	\$3.50	\$3.50	\$3.50	\$3.50
Kuwait	\$5.00	\$5.00	\$5.00	\$5.00
Kyrgyzstan	\$5.00	\$5.00	\$5.00	\$5.00
Laos	\$5.00	\$5.00	\$5.00	\$5.00
Latvia	\$5.00	\$5.00	\$5.00	\$5.00
Lebanon	\$5.00	\$5.00	\$5.00	\$5.00
Lesotho	\$5.00	\$5.00	\$5.00	\$5.00
Liberia	\$5.00	\$5.00	\$5.00	\$5.00
Libya	\$5.00	\$5.00	\$5.00	\$5.00
Liechtenstein	\$3.25	\$3.25	\$3.25	\$3.25
Lithuania	\$5.00	\$5.00	\$5.00	\$5.00
Luxembourg	\$3.25	\$3.25	\$3.25	\$3.25
Macao	\$5.00	\$5.00	\$5.00	\$5.00
Macedonia	\$5.00	\$5.00	\$5.00	\$5.00
Madagascar	\$5.00	\$5.00	\$5.00	\$5.00
Malawi	\$5.00	\$5.00	\$5.00	\$5.00
Malaysia	\$5.00	\$5.00	\$5.00	\$5.00
Maldives	\$5.00	\$5.00	\$5.00	\$5.00
Mali Republic	\$5.00	\$5.00	\$5.00	\$5.00
Malta Republic	\$5.00	\$5.00	\$5.00	\$5.00
Marshall Island	\$5.00	\$5.00	\$5.00	\$5.00
Mauritania	\$5.00	\$5.00	\$5.00	\$5.00
Mauritius	\$5.00	\$5.00	\$5.00	\$5.00
Mayotte Island (se		(sland)		
Mexico 1	\$3.50	\$3.50	\$3.50	\$3.50
Mexico 2	\$3.50	\$3.50	\$3.50	\$3.50
Micronesia	\$5.00	\$5.00	\$5.00	\$5.00
Moldova	\$5.00	\$5.00	\$5.00	\$5.00
Monaco	\$3.50	\$3.50	\$3.50	\$3.50
Mongolia	\$5.00	\$5.00	\$5.00	\$5.00
Montenegro	\$5.00	\$5.00	\$5.00	\$5.00

Original Page 6

Effective: September 1, 2019

SECTION 7 – RESIDENTIAL INTERNATIONAL SERVICE DESCRIPTIONS, RATES AND CHARGES

### 7.2 Residential International Message Telecommunications Service (IMTS) (continued)

	Standard		Eco	Economy	
	Each			Each	
		Additional		Additional	
Country Name				Minute	
Montserrat	\$4.50	\$4.50	\$4.50	\$4.50	
Morocco	\$5.00	\$5.00	\$5.00	\$5.00	
Mozambique	\$5.00	\$5.00	\$5.00	\$5.00	
Myanmar	\$5.00	\$5.00	\$5.00	\$5.00	
Namibia	\$5.00	\$5.00	\$5.00	\$5.00	
Nauru	\$5.00	\$5.00	\$5.00	\$5.00	
Nepal	\$5.00	\$5.00	\$5.00	\$5.00	
Netherlands	\$3.50	\$3.50	\$3.50	\$3.50	
Netherland Antilles	\$4.00	\$4.00	\$4.00	\$4.00	
Networks/Global Sat.	\$5.00	\$5.00	\$5.00	\$5.00	
New Caledonia	\$5.00	\$5.00	\$5.00	\$5.00	
New Zealand	\$5.00	\$5.00	\$5.00	\$5.00	
Nicaragua	\$5.00	\$5.00	\$5.00	\$5.00	
Niger Republic	\$5.00	\$5.00	\$5.00	\$5.00	
Nigeria	\$5.00	\$5.00	\$5.00	\$5.00	
Niue	\$5.00	\$5.00	\$5.00	\$5.00	
Norway	\$3.50	\$3.50	\$3.50	\$3.50	
Oman	\$5.00	\$5.00	\$5.00	\$5.00	
Pakistan	\$5.00	\$5.00	\$5.00	\$5.00	
Palau	\$5.00	\$5.00	\$5.00	\$5.00	
Palestinian Authority	\$5.00	\$5.00	\$5.00	\$5.00	
Panama	\$5.00	\$5.00	\$5.00	\$5.00	
Papua New Guinea	\$5.00	\$5.00	\$5.00	\$5.00	
Paraguay	\$5.00	\$5.00	\$5.00	\$5.00	
Peru	\$5.00	\$5.00	\$5.00	\$5.00	
Philippines	\$5.00	\$5.00	\$5.00	\$5.00	
Poland	\$5.00	\$5.00	\$5.00	\$5.00	
Portugal	\$4.50	\$4.50	\$4.50	\$4.50	
Qatar	\$5.00	\$5.00	\$5.00	\$5.00	
Reunion Island (includes Mayotte Island)	\$5.00	\$5.00	\$5.00	\$5.00	
Romania	\$5.00	\$5.00	\$5.00	\$5.00	
Russia	\$5.00	\$5.00	\$5.00	\$5.00	
Rwanda, Republic of	\$5.00	\$5.00	\$5.00	\$5.00	
San Marino	\$4.00	\$4.00	\$4.00	\$4.00	
Sao Tome	\$5.00	\$5.00	\$5.00	\$5.00	
Saudi Arabia	\$5.00	\$5.00	\$5.00	\$5.00	
Senegal	\$5.00	\$5.00	\$5.00	\$5.00	
Serbia	\$5.00	\$5.00	\$5.00	\$5.00	
Seychelles Island	\$5.00	\$5.00	\$5.00	\$5.00	
Sierra Leone	\$5.00	\$5.00	\$5.00	\$5.00	
Singapore	\$4.50	\$4.50	\$4.50	\$4.50	

Original Page 7 Effective: September 1, 2019

## BUSINESS AND RESIDENTIAL SERVICES PRICING AND SERVICE GUIDE

SECTION 7 – RESIDENTIAL INTERNATIONAL SERVICE DESCRIPTIONS, RATES AND CHARGES

### 7.2 Residential International Message Telecommunications Service (IMTS) (continued)

	Standard		Eco	onomy	
	Each		Each Initial Additional		
Communication Normal		Additional			
Country Name Slovakia	Minute		Minute \$5.00		
	\$5.00	\$5.00	\$5.00	\$5.00 \$5.00	
Slovenia	\$5.00	\$5.00	\$5.00		
Solomon Islands	\$5.00	\$5.00	\$5.00	\$5.00	
Somalia Democratic Rep	\$5.00	\$5.00	\$5.00	\$5.00	
South Africa	\$5.00	\$5.00	\$5.00	\$5.00	
South Sudan	\$5.00	\$5.00	\$5.00	\$5.00	
Spain	\$5.00	\$5.00	\$5.00	\$5.00	
Sri Lanka	\$5.00	\$5.00	\$5.00	\$5.00	
St. Helena	\$5.00	\$5.00	\$5.00	\$5.00	
St. Kitts/Nevis	\$4.50	\$4.50	\$4.50	\$4.50	
St. Lucia	\$4.50	\$4.50	\$4.50	\$4.50	
St. Maarten	\$4.00	\$4.00	\$4.00	\$4.00	
St. Pierre and Miquelon	\$3.50	\$3.50	\$3.50	\$3.50	
St. Vincent/Grenadines	\$4.50	\$4.50	\$4.50	\$4.50	
Sudan	\$5.00	\$5.00	\$5.00	\$5.00	
Suriname	\$5.00	\$5.00	\$5.00	\$5.00	
Swaziland	\$5.00	\$5.00	\$5.00	\$5.00	
Sweden	\$3.50	\$3.50	\$3.50	\$3.50	
Switzerland	\$3.50	\$3.50	\$3.50	\$3.50	
Syria	\$5.00	\$5.00	\$5.00	\$5.00	
Taiwan	\$5.00	\$5.00	\$5.00	\$5.00	
Tajikistan	\$5.00	\$5.00	\$5.00	\$5.00	
Tanzania	\$5.00	\$5.00	\$5.00	\$5.00	
Thailand	\$5.00	\$5.00	\$5.00	\$5.00	
Togo	\$5.00	\$5.00	\$5.00	\$5.00	
Tokelau	\$5.00	\$5.00	\$5.00	\$5.00	
Tonga Island	\$5.00	\$5.00	\$5.00	\$5.00	
Trinidad and Tobago	\$4.50	\$4.50	\$4.50	\$4.50	
Tunisia	\$5.00	\$5.00	\$5.00	\$5.00	
Turkey	\$5.00	\$5.00	\$5.00	\$5.00	
Turkmenistan	\$5.00	\$5.00	\$5.00	\$5.00	
Turks and Caicos Islands	\$4.50	\$4.50	\$4.50	\$4.50	
Tuvalu	\$5.00	\$5.00	\$5.00	\$5.00	
Uganda	\$5.00	\$5.00	\$5.00	\$5.00	
Ukraine	\$5.00	\$5.00	\$5.00	\$5.00	
United Arab Emirates	\$5.00	\$5.00	\$5.00	\$5.00	
United Kingdom	\$3.50	\$3.50	\$3.50	\$3.50	
Uruguay	\$5.00	\$5.00	\$5.00	\$5.00	
Uzbekistan	\$5.00	\$5.00	\$5.00	\$5.00	
Vanuatu	\$5.00	\$5.00	\$5.00	\$5.00	
Vatican City	\$4.50	\$4.50	\$4.50	\$4.50	
Venezuela	\$4.00	\$4.00	\$4.00	\$4.00	
Vietnam	\$5.00	\$5.00	\$5.00	\$5.00	
Wallis and Futuna Islands	\$5.00	\$5.00	\$5.00	\$5.00	
Western Samoa	\$5.00	\$5.00	\$5.00	\$5.00	
Yemen	\$5.00	\$5.00	\$5.00	\$5.00	
Zaire (Dem Congo)	\$5.00	\$5.00	\$5.00	\$5.00	
Zambia	\$4.50	\$4.50	\$4.50	\$4.50	
Zimbabwe	\$5.00	\$5.00	\$5.00	\$5.00	

SECTION 7 - RESIDENTIAL INTERNATIONAL SERVICE DESCRIPTIONS, RATES AND CHARGES

#### 7.3 Residential Optional Calling Plans

#### 7.3.1 AT&T Unlimited Canada Calling

(A) AT&T Unlimited Canada Calling is a direct dialed outbound international long distance service offered to residential Customers from switched access lines. Service is offered on a flat rate basis, twenty-four hours per day, seven days per week. AT&T Unlimited Canada Calling provides unlimited minutes of international outbound Canada usage for a flat rate monthly charge. Per minute charges apply to calls to other international countries based on the country called and the duration of the call in addition to the Monthly Recurring Charge. An international wireless termination charge applies to all applicable calls. Customers must dial 1 plus the country code or area code, if applicable, and the terminating number to complete a call utilizing this service.

Original Page 8

Effective: September 1, 2019

(B) Calls not included in unlimited calling are billed in one (1) minute increments after an initial period, for billing purposes, of one (1) minute. This plan does not provide call detail information on the Customer's monthly bill for unlimited calls to Canada, except for applicable mobile termination charges, but does provide call detail information for international calls to other countries. Customers may not subscribe to more than one international calling plan on the same account.

#### (C) Eligibility Requirements

In order to be eligible for this plan, the Company must be able to verify that the Customer meets the following eligibility requirements. Customers who no longer meet these eligibility requirements will not be eligible for this plan and may be offered an alternative plan or have their service suspended, restricted or canceled without notice.

- .1 Customers must subscribe to a local plan as defined and offered by the Company's Affiliated ILEC and must meet all the plan eligibility requirements. If the Company moves the Customer from one of the domestic unlimited plans above to another domestic plan due to prohibited usage, the Customer will no longer be eligible for this plan and will be offered an alternative international plan at the Company's sole discretion.
- .2 This plan cannot be used for any use inconsistent with residential service.
- .3 This plan is not available on an account that is the recipient of charges billed from another location.
- .4 Customer lines associated with educational institutions (colleges, universities, etc.) are not eligible for this plan.
- .5 Unlimited calling to Canada does not include multi-party conference calls (except those calls placed by using the Three-Way Calling features are included), calls to 900 numbers or other calls to access information services, directory assistance, operator services, international calling to countries other than Canada, domestic calls, and toll-free calling services.
- .6 This plan is not available for resale.
- .7 If usage under this plan is not consistent with typical residential Customer usage, at the Company's sole discretion, the Company may offer the Customer an alternative plan or suspend, restrict or cancel the Customer's service without prior notice. Calls that are not consistent with typical residential voice usage include but are not limited to: use for general business purposes, commercial facsimile, auto-dialing, resale, international call forwarding, call centers and telemarketing.

SECTION 7 – RESIDENTIAL INTERNATIONAL SERVICE DESCRIPTIONS, RATES AND CHARGES

### 7.3 Residential Optional Calling Plans (continued)

#### 7.3.1 AT&T Unlimited Canada Calling (continued)

(D) International Mobile Termination Charge
 See Section 7.4.1 of this pricing and service guide for the applicable International Mobile Termination Charge.

### (E) Rates and Charges

The monthly recurring charge (MRC) is \$6.99, and the per-minute rates are as follows:

		Each
Country	Initial Minute	Additional Minute
Afghanistan	\$1.76	\$1.76
Albania	\$1.04	\$1.04
Algeria	\$0.49	\$0.49
Andorra	\$0.46	\$0.46
Angola	\$1.48	\$1.48
Anguilla	\$0.57	\$0.57
Antarctica (Scott Base)	\$1.77	\$1.77
Antigua	\$0.47	\$0.47
Argentina	\$0.19	\$0.19
Armenia	\$0.59	\$0.59
Aruba	\$0.44	\$0.44
Ascension Island	\$1.14	\$1.14
Australia	\$0.10	\$0.10
Australia External	\$0.10	\$0.10
Territories*		
Austria	\$0.12	\$0.12
Azerbaijan	\$0.87	\$0.87
Bahamas	\$0.25	\$0.25
Bahrain	\$0.65	\$0.65
Bangladesh	\$0.55	\$0.55
Barbados	\$0.52	\$0.52
Belarus	\$0.35	\$0.35
Belgium	\$0.10	\$0.10
Belize	\$0.49	\$0.49

	Initial	Each Additional
Country	Minute	Minute
Benin	\$0.65	\$0.65
Bermuda	\$0.19	\$0.19
Bhutan	\$1.84	\$1.84
Bolivia	\$0.33	\$0.33
Bosnia	\$0.35	\$0.35
Botswana	\$0.84	\$0.84
Brazil	\$0.17	\$0.17
British Virgin Islands	\$0.45	\$0.45
Brunei	\$0.86	\$0.86
Bulgaria	\$0.35	\$0.35
Burkina Faso	\$1.04	\$1.04
Burundi	\$1.78	\$1.78
Cambodia	\$2.32	\$2.32
Cameroon	\$0.49	\$0.49
Canada	\$0.00	\$0.00
Cape Verde Island	\$0.82	\$0.82
Cayman Islands	\$0.25	\$0.25
Central African Republic	\$1.51	\$1.51
Chad	\$2.46	\$2.46
Chile	\$0.19	\$0.19
China	\$0.15	\$0.15
Colombia	\$0.17	\$0.17
Comoros Islands	\$2.29	\$2.29

Original Page 9

<sup>\*</sup> External Territories Include - Antarctica-Casey Base and Christmas, Cocos and Norfolk Islands

Original Page 10 Effective: September 1, 2019

## BUSINESS AND RESIDENTIAL SERVICES PRICING AND SERVICE GUIDE

SECTION 7 – RESIDENTIAL INTERNATIONAL SERVICE DESCRIPTIONS, RATES AND CHARGES

### 7.3 Residential Optional Calling Plans (continued)

### 7.3.1 AT&T Unlimited Canada Calling (continued)

### (E) Rates and Charges (continued)

	Each	
<b>G</b> 4	Initial	Additional
Country	Minute	Minute
Congo	\$0.75	\$0.75
Cook Islands	\$1.65	\$1.65
Costa Rica	\$0.27	\$0.27
Croatia	\$0.35	\$0.35
Cuba	\$0.92	\$0.92
Cyprus	\$0.56	\$0.56
Czech Republic	\$0.29	\$0.29
Denmark	\$0.12	\$0.12
Diego Garcia	\$2.15	\$2.15
Djibouti	\$0.97	\$0.97
Dominica	\$0.41	\$0.41
Dominican Republic	\$0.17	\$0.17
East Timor	\$2.39	\$2.39
Easter Island	\$0.19	\$0.19
Ecuador	\$0.30	\$0.30
Egypt	\$0.39	\$0.39
El Salvador	\$0.23	\$0.23
Equatorial Guinea	\$2.19	\$2.19
Eritrea	\$0.89	\$0.89
Estonia	\$0.69	\$0.69
Ethiopia	\$1.01	\$1.01
Faeroe Island	\$0.52	\$0.52
Falkland Island	\$1.21	\$1.21
Fiji Island	\$1.04	\$1.04
Finland	\$0.10	\$0.10
France	\$0.10	\$0.10
French Antilles	\$0.65	\$0.65
French Guiana	\$0.66	\$0.66
French Polynesia	\$1.24	\$1.24
Gabon	\$1.09	\$1.09
Gambia	\$0.59	\$0.59
Georgia	\$0.80	\$0.80
Germany	\$0.10	\$0.10
Ghana	\$0.47	\$0.47
Gibraltar	\$1.04	\$1.04
Greece	\$0.10	\$0.10
Greenland	\$0.58	\$0.58
Grenada	\$0.45	\$0.45
Guadeloupe	\$0.61	\$0.61
Guatemala	\$0.23	\$0.23
Guinea	\$0.82	\$0.82
Guinea-Bissau	\$1.65	\$1.65
Guyana	\$0.84	\$0.84
Haiti	\$0.49	\$0.49
114141	ψυιτλ	ψυ.τν

Country	Initial Minute	Each Additional Minute
Honduras	\$0.40	\$0.40
Hong Kong	\$0.10	\$0.10
Hungary	\$0.27	\$0.27
Iceland	\$0.49	\$0.49
India	\$0.31	\$0.31
Indonesia	\$0.33	\$0.33
Inmarsat (SNAC)	\$9.95	\$9.95
Iran	\$0.59	\$0.59
Iraq	\$0.79	\$0.79
Ireland	\$0.10	\$0.10
Israel	\$0.10	\$0.10
Italy	\$0.10	\$0.10
Ivory Coast	\$0.69	\$0.69
Jamaica	\$0.38	\$0.38
Japan	\$0.08	\$0.08
Jordan	\$0.59	\$0.59
Kazakhstan	\$0.19	\$0.19
Kenya	\$0.67	\$0.67
Kiribati	\$1.16	\$1.16
Korea (North)	\$1.14	\$1.14
Korea (South)	\$0.10	\$0.10
Kosovo	\$0.12	\$0.12
Kuwait	\$0.47	\$0.47
Kyrgyzstan	\$0.31	\$0.31
Laos	\$2.43	\$2.43
Latvia	\$0.35	\$0.35
Lebanon	\$0.55	\$0.55
Lesotho	\$0.84	\$0.84
Liberia	\$0.59	\$0.59
Libya	\$0.97	\$0.97
Liechtenstein	\$0.12	\$0.12
Lithuania	\$0.35	\$0.35
Luxembourg	\$0.12	\$0.12
Macao Macedonia	\$0.77 \$0.55	\$0.77 \$0.55
Madagascar	\$2.18	\$2.18
Malawi	\$2.18	\$2.18
Malaysia Maldives	\$0.18 \$1.45	\$0.18 \$1.45
Mali Republic	\$1.43	\$1.43
Malta Republic	\$0.72	\$0.72
Marshall Island	\$0.72	\$0.72
Mauritania	\$1.00	\$1.00
Mauritius	\$1.25	\$1.25
riamilian	Ψ1.20	Ψ1.23

ce Effective: September 1, 2019

BUSINESS AND RESIDENTIAL SERVICES

Original Page 11

## PRICING AND SERVICE GUIDE SECTION 7 – RESIDENTIAL INTERNATIONAL SERVICE DESCRIPTIONS, RATES AND CHARGES

### 7.3 Residential Optional Calling Plans (continued)

### 7.3.1 AT&T Unlimited Canada Calling (continued)

(E) Rates and Charges (continued)

	Initial	Each Additional		Initial	Each Additional
Country	Minute	Minute	Country	Minute	Minute
Mayotte Island (see Reunion Island)	-	=	Slovenia	\$0.56	\$0.56
Mexico 1	\$0.09	\$0.09	Solomon Island	\$1.20	\$1.20
Mexico 2	\$0.15	\$0.15	Somalia Democratic Rep	\$1.64	\$1.64
Micronesia	\$0.77	\$0.77	South Africa	\$0.36	\$0.36
Moldova	\$0.49	\$0.49	Spain	\$0.10	\$0.10
Monaco	\$0.12	\$0.12	Sri Lanka	\$1.02	\$1.02
Mongolia	\$1.70	\$1.70	St. Helena	\$1.15	\$1.15
Montenegro	\$0.42	\$0.42	St. Kitts/Nevis	\$0.49	\$0.49
Montserrat	\$0.52	\$0.52	St. Lucia	\$0.41	\$0.41
Morocco	\$0.59	\$0.59	St. Maarten	\$0.42	\$0.42
Mozambique	\$1.28	\$1.28	St. Pierre Miquelon	\$0.48	\$0.48
Myanmar (Burma)	\$2.17	\$2.17	St. Vincent/Grenadines	\$0.55	\$0.55
Namibia	\$0.80	\$0.80	South Sudan	\$0.89	\$0.89
Nauru	\$1.38	\$1.38	Sudan	\$0.89	\$0.89
Nepal	\$1.30	\$1.30	Suriname	\$1.25	\$1.25
Netherlands	\$0.10	\$0.10	Swaziland	\$0.76	\$0.76
Netherlands Antilles	\$0.42	\$0.42	Sweden	\$0.10	\$0.10
Networks	\$3.95	\$3.95	Switzerland	\$0.10	\$0.10
New Caledonia	\$1.45	\$1.45	Syrian Arab Republic	\$0.89	\$0.89
New Zealand	\$0.12	\$0.12	Taiwan	\$0.08	\$0.08
Nicaragua	\$0.38	\$0.38	Tajikistan	\$0.23	\$0.23
Niger Republic	\$0.93	\$0.93	Tanzania	\$0.49	\$0.49
Nigeria	\$0.56	\$0.56	Thailand	\$0.24	\$0.24
Niue	\$3.31	\$3.31	Togo	\$1.11	\$1.11
Norway	\$0.12	\$0.12	Tokelau	\$0.84	\$0.84
Oman	\$1.07	\$1.07	Tonga Island	\$1.57	\$1.57
Pakistan	\$0.31	\$0.31	Trinidad	\$0.34	\$0.34
Palau	\$1.30	\$1.30	Tunisia	\$0.49	\$0.49
Palestinian Authority	\$0.30	\$0.30	Turkey	\$0.39	\$0.39
Panama	\$0.35	\$0.35	Turkmenistan	\$0.31	\$0.31
Papua New Guinea	\$1.06	\$1.06	Turks and Caicos Islands	\$0.52	\$0.52
Paraguay	\$0.40	\$0.40	Tuvalu	\$1.85	\$1.85
Peru	\$0.25	\$0.25	Uganda	\$0.49	\$0.49
Philippines	\$0.16	\$0.16	Ukraine	\$0.29	\$0.29
Poland	\$0.18	\$0.18	United Arab Emirates	\$0.39	\$0.39
Portugal	\$0.10	\$0.10	United Kingdom	\$0.07	\$0.07
Qatar	\$0.95	\$0.95	Uruguay	\$0.40	\$0.40
Reunion Island (includes Mayotte Island)	\$1.02	\$1.02	Uzbekistan	\$0.31	\$0.31
Romania	\$0.35	\$0.35	Vanuatu	\$2.37	\$2.37
Russia	\$0.19	\$0.19	Vatican City	\$0.10	\$0.10
Rwanda, Republic of	\$1.19	\$1.19	Venezuela	\$0.20	\$0.20
San Marino	\$0.89	\$0.89	Vietnam	\$0.50	\$0.50
Sao Tome	\$1.59	\$1.59	Wallis Futuna	\$2.38	\$2.38
Saudi Arabia	\$0.49	\$0.49	Western Samoa	\$1.02	\$1.02
Senegal	\$0.69	\$0.69	Yemen Arab Republic	\$0.98	\$0.98
Serbia	\$0.42	\$0.42	Zaire (Dem Congo)	\$0.59	\$0.59
Seychelles Island	\$1.51	\$1.51	Zambia	\$0.59	\$0.59
Sierra Leone	\$0.59	\$0.59	Zimbabwe	\$0.49	\$0.49
Singapore	\$0.14	\$0.14			

SECTION 7 - RESIDENTIAL INTERNATIONAL SERVICE DESCRIPTIONS, RATES AND CHARGES

#### 7.3 Residential Optional Calling Plans (continued)

#### 7.3.2 AT&T® Worldwide Occasional Calling Plan

- (A) The AT&T® Worldwide Occasional Calling Plan is a direct dialed outbound international long distance service offered to residential Customers from switched access lines. Service is offered on a flat rate basis, twenty-four hours per day, seven days per week. Customers must dial 1 plus the country code or area code, if applicable, and the terminating number to complete a call utilizing this service. All calls are billed in one (1) minute increments after an initial period, for billing purposes, of one (1) minute. This plan is only offered in conjunction with a Company Domestic Service.
- (B) International Mobile Termination Charge
   See Section 7.4.1 of this pricing and serivce guide for the applicable International Mobile Termination Charge.
- (C) Rates and Charges

The MRC is \$2.99, and the per-minute rates are as follows:

		Each
	Initial	Additional
Country	Minute	Minute
Afghanistan	\$1.79	\$1.79
Albania	\$1.88	\$1.88
Algeria	\$1.88	\$1.88
Andorra	\$1.29	\$1.29
Angola	\$1.88	\$1.88
Anguilla	\$1.41	\$1.41
Antarctica (Scott Base)	\$1.68	\$1.68
Antigua	\$1.37	\$1.37
Argentina	\$1.68	\$1.68
Armenia	\$1.88	\$1.88
Aruba	\$1.31	\$1.31
Ascension Island	\$1.88	\$1.88
Australia	\$1.32	\$1.32
Australia External	\$1.32	\$1.32
Territories*		
Austria	\$1.28	\$1.28
Azerbaijan	\$1.88	\$1.88
Bahamas	\$1.06	\$1.06
Bahrain	\$1.84	\$1.84
Bangladesh	\$1.88	\$1.88
Barbados	\$1.37	\$1.37
Belarus	\$1.88	\$1.88
Belgium	\$1.29	\$1.29
Belize	\$1.65	\$1.65
Benin	\$1.86	\$1.86

	Ţ.	
		Each
~ .	Initial	Additional
Country	Minute	Minute
Bermuda	\$1.15	\$1.15
Bhutan	\$1.88	\$1.88
Bolivia	\$1.80	\$1.80
Bosnia and Herzegovina	\$1.60	\$1.60
Botswana	\$1.67	\$1.67
Brazil	\$1.79	\$1.79
British Virgin Islands	\$1.32	\$1.32
Brunei	\$1.77	\$1.77
Bulgaria	\$1.84	\$1.84
Burkina Faso	\$1.88	\$1.88
Burundi	\$1.88	\$1.88
Cambodia	\$1.88	\$1.88
Cameroon	\$1.88	\$1.88
Canada	\$0.42	\$0.42
Cape Verde Island	\$1.88	\$1.88
Cayman Islands	\$1.38	\$1.38
Central African Republic	\$1.88	\$1.88
Chad	\$1.88	\$1.88
Chile (includes Easter Island)	\$1.53	\$1.53
China	\$1.88	\$1.88
Colombia	\$1.82	\$1.82
Comoros Islands	\$1.88	\$1.88
Congo	\$1.88	\$1.88
Cook Islands	\$1.88	\$1.88

Original Page 12

<sup>\*</sup> Australian Territories include: Antarctica-Casey Base and Christmas, Cocos and Norfolk Islands

ce Effective: September 1, 2019

BUSINESS AND RESIDENTIAL SERVICES

Original Page 13

## PRICING AND SERVICE GUIDE SECTION 7 – RESIDENTIAL INTERNATIONAL SERVICE DESCRIPTIONS, RATES AND CHARGES

### 7.3 Residential Optional Calling Plans (continued)

### 7.3.2 AT&T® Worldwide Occasional Calling Plan

### (C) Rates and Charges (continued)

Country	Initial Minute	Each Additional Minute	Country	Initial Minute	Each Additional Minute
Costa Rica	\$1.47	\$1.47	Iceland	\$1.43	\$1.43
Croatia	\$1.59	\$1.59	India	\$1.88	\$1.88
Cuba	\$1.88	\$1.88	Indonesia	\$1.88	\$1.88
Cyprus	\$1.65	\$1.65	Inmarsat (SNAC)	\$3.75	\$3.75
Czech Republic	\$1.72	\$1.72	Iran	\$1.88	\$1.88
Denmark	\$1.72	\$1.72	Iraq	\$1.88	\$1.88
Diego Garcia	\$1.88	\$1.88	Ireland	\$1.26	\$1.26
Diibouti	\$1.88	\$1.88	Israel	\$1.87	\$1.87
Dominica	\$1.47	\$1.47	Italy	\$1.45	\$1.45
Dominican Republic	\$1.47	\$1.47	Ivory Coast	\$1.88	\$1.88
East Timor	\$1.88	\$1.88	Jamaica	\$1.62	\$1.62
Easter Island (See Chile)	ψ1.00	ψ1.00	Japan	\$1.41	\$1.41
Ecuador	\$1.71	\$1.71	Jordan	\$1.74	\$1.74
Egypt	\$1.88	\$1.88	Kazakhstan	\$1.88	\$1.88
El Salvador	\$1.76	\$1.76	Kenya	\$1.88	\$1.88
Equatorial Guinea	\$1.88	\$1.88	Kiribati	\$1.88	\$1.88
Eritrea	\$1.88	\$1.88	Korea (North)	\$1.88	\$1.88
Estonia	\$1.88	\$1.88	Korea (South)	\$1.74	\$1.74
Ethiopia	\$1.88	\$1.88	Kosovo	\$1.11	\$1.11
Faeroe Island	\$1.18	\$1.18	Kuwait	\$1.65	\$1.65
Falkland Island	\$1.88	\$1.88	Kyrgyzstan	\$1.88	\$1.88
Fiji Island	\$1.88	\$1.88	Laos	\$1.88	\$1.88
Finland	\$1.15	\$1.15	Latvia	\$1.88	\$1.88
France	\$1.23	\$1.23	Lebanon	\$1.88	\$1.88
French Antilles (including Martinique, St. Barthelemy & St. Martin)	\$1.35	\$1.35	Lesotho	\$1.72	\$1.72
French Guiana	\$1.53	\$1.53	Liberia	\$1.77	\$1.77
French Polynesia	\$1.88	\$1.88	Libya	\$1.88	\$1.88
Gabon	\$1.88	\$1.88	Liechtenstein	\$0.98	\$0.98
Gambia	\$1.86	\$1.86	Lithuania	\$1.88	\$1.88
Georgia	\$1.88	\$1.88	Luxembourg	\$0.97	\$0.97
Germany	\$1.17	\$1.17	Macao	\$1.88	\$1.88
Ghana	\$1.88	\$1.88	Macedonia	\$1.59	\$1.59
Gibraltar	\$1.50	\$1.50	Madagascar	\$1.88	\$1.88
Greece	\$1.84	\$1.84	Malawi	\$1.66	\$1.66
Greenland	\$1.41	\$1.41	Malaysia	\$1.60	\$1.60
Grenada	\$1.50	\$1.50	Maldives	\$1.88	\$1.88
Guadeloupe	\$1.38	\$1.38	Mali Republic	\$1.88	\$1.88
Guatemala	\$1.77	\$1.77	Malta Republic	\$1.88	\$1.88
Guinea	\$1.88	\$1.88	Marshall Island	\$1.88	\$1.88
Guinea-Bissau	\$1.88	\$1.88	Mauritania	\$1.88	\$1.88
Guyana	\$1.88	\$1.88	Mauritius	\$1.88	\$1.88
Haiti	\$1.63	\$1.63	Mayotte Island (See Reun		
Honduras	\$1.70	\$1.70	Mexico 1	\$0.86	\$0.86
Hong Kong	\$1.56	\$1.56	Mexico 2	\$0.86	\$0.86
Hungary	\$1.54	\$1.54	Micronesia	\$1.88	\$1.88

Original Page 14

Effective: September 1, 2019

SECTION 7 – RESIDENTIAL INTERNATIONAL SERVICE DESCRIPTIONS, RATES AND CHARGES

### 7.3 Residential Optional Calling Plans (continued)

### 7.3.2 AT&T® Worldwide Occasional Calling Plan (continued)

### (C) Rates and Charges (continued)

Committee	Initial	Each Additional	Committee	Initial	Each Additional
Country	Minute	Minute	Country	Minute	Minute
Moldova	\$1.88	\$1.88	Solomon Island	\$1.88	\$1.88
Monaco	\$1.11	\$1.11	Somalia Democratic Rep	\$1.88	\$1.88
Mongolia	\$1.88	\$1.88	South Africa	\$1.57	\$1.57
Montenegro	\$1.73	\$1.73	South Sudan	\$1.88	\$1.88
Montserrat	\$1.45	\$1.45	Spain	\$1.55	\$1.55
Morocco	\$1.88	\$1.88	Sri Lanka	\$1.88	\$1.88
Mozambique	\$1.88	\$1.88	St. Helena	\$1.88	\$1.88
Myanmar	\$1.88	\$1.88	St. Kitts/Nevis	\$1.44	\$1.44
Namibia	\$1.56	\$1.56	St. Lucia	\$1.45	\$1.45
Nauru	\$1.88	\$1.88	St. Maarten	\$1.34	\$1.34
Nepal	\$1.88	\$1.88	St. Pierre Miquelon	\$1.17	\$1.17
Netherlands	\$1.11	\$1.11	St. Vincent/Grenadines	\$1.50	\$1.50
Netherlands Antilles	\$1.34	\$1.34	Sudan	\$1.88	\$1.88
Networks	\$3.75	\$3.75	Suriname	\$1.88	\$1.88
New Caledonia	\$1.88	\$1.88	Swaziland	\$1.77	\$1.77
New Zealand	\$1.68	\$1.68	Sweden	\$1.11	\$1.11
Nicaragua	\$1.68	\$1.68	Switzerland	\$1.14	\$1.14
Niger Republic	\$1.88	\$1.88	Syria	\$1.88	\$1.88
Nigeria	\$1.77	\$1.77	Taiwan	\$1.73	\$1.73
Niue	\$1.88	\$1.88	Tajikistan	\$1.88	\$1.88
Norway	\$1.15	\$1.15	Tanzania	\$1.88	\$1.88
Oman	\$1.85	\$1.85	Thailand	\$1.65	\$1.65
Pakistan	\$1.88	\$1.88	Togo	\$1.88	\$1.88
Palau	\$1.88	\$1.88	Tokelau	\$1.88	\$1.88
Palestinian Authority	\$1.87	\$1.87	Tonga Island	\$1.88	\$1.88
Panama	\$1.58	\$1.58	Trinidad and Tobago	\$1.47	\$1.47
Papua New Guinea	\$1.68	\$1.68	Tunisia	\$1.84	\$1.84
Paraguay	\$1.88	\$1.88	Turkey	\$1.62	\$1.62
Peru	\$1.74	\$1.74	Turkmenistan	\$1.88	\$1.88
Philippines	\$1.88	\$1.88	Turks and Caicos Islands	\$1.39	\$1.39
Poland	\$1.59	\$1.59	Tuvalu	\$1.88	\$1.88
Portugal	\$1.52	\$1.52	Uganda	\$1.88	\$1.88
Qatar	\$1.83	\$1.83	Ukraine	\$1.88	\$1.88
Reunion Island (including Mayotte Island)	\$1.88	\$1.88	United Arab Emirates	\$1.56	\$1.56
Romania	\$1.88	\$1.88	United Kingdom	\$1.06	\$1.06
Russia	\$1.88	\$1.88	Uruguay	\$1.69	\$1.69
Rwanda, Republic of	\$1.88	\$1.88	Uzbekistan	\$1.88	\$1.88
San Marino	\$1.23	\$1.23	Vanuatu	\$1.64	\$1.64
Sao Tome	\$1.88	\$1.88	Vatican City	\$1.45	\$1.45
Saudi Arabia	\$1.88	\$1.88	Venezuela	\$1.31	\$1.31
Senegal	\$1.88	\$1.88	Vietnam	\$1.88	\$1.88
Serbia	\$1.73	\$1.73	Wallis Futuna	\$1.88	\$1.88
Seychelles Island	\$1.88	\$1.88	Western Samoa	\$1.88	\$1.88
Sierra Leone	\$1.88	\$1.88	Yemen Arab Republic	\$1.81	\$1.81
Singapore	\$1.42	\$1.42	Zaire (Dem Congo)	\$1.78	\$1.78
Slovakia	\$1.66	\$1.66	Zambia	\$1.53	\$1.53
Slovenia	\$1.68	\$1.68	Zimbabwe	\$1.70	\$1.70

SECTION 7 - RESIDENTIAL INTERNATIONAL SERVICE DESCRIPTIONS, RATES AND CHARGES

### 7.3 Residential Optional Calling Plans (continued)

### 7.3.3 AT&T® Worldwide Value Calling<sup>SM</sup> Plan

- (A) The AT&T® Worldwide Value Calling<sup>SM</sup> Plan is a direct dialed outbound international long distance service offered to residential Customers from switched access lines. Service is offered on a flat rate basis, twenty-four hours per day, seven days per week. Customers must dial 1 plus the country code or area code, if applicable, and the terminating number to complete a call utilizing this service. All calls are billed in one (1) minute increments after an initial period, for billing purposes, of one (1) minute. This plan is only offered in conjunction with a Company Domestic Service.
- (B) International Mobile Termination Charge
   See Section 7.4.1 of this pricing and serivce guide for the applicable International Mobile Termination Charge.
- (C) Rates and Charges

The MRC is \$5.99, and the per-minute rates are as follows:

		Each
~	Initial	Additional
Country	Minute	Minute
Afghanistan	\$1.76	\$1.76
Albania	\$1.04	\$1.04
Algeria	\$0.49	\$0.49
Andorra	\$0.46	\$0.46
Angola	\$1.48	\$1.48
Anguilla	\$0.57	\$0.57
Antarctica (Casey Base)	\$0.09	\$0.09
Antarctica (Scott Base)	\$0.08	\$0.08
Antigua	\$0.47	\$0.47
Argentina	\$0.17	\$0.17
Armenia	\$0.59	\$0.59
Aruba	\$0.44	\$0.44
Ascension Island	\$1.14	\$1.14
Australia	\$0.09	\$0.09
Austria	\$0.09	\$0.09
Azerbaijan	\$0.87	\$0.87
Bahamas	\$0.25	\$0.25
Bahrain	\$0.65	\$0.65
Bangladesh	\$0.50	\$0.50
Barbados	\$0.51	\$0.51
Belarus	\$0.35	\$0.35
Belgium	\$0.09	\$0.09
Belize	\$0.49	\$0.49
Benin	\$0.65	\$0.65
Bermuda	\$0.19	\$0.19
Bhutan	\$1.84	\$1.84
Bolivia	\$0.33	\$0.33

Country	Initial Minute	Each Additional Minute
Bosnia	\$0.35	\$0.35
Botswana	\$0.84	\$0.84
Brazil	\$0.15	\$0.15
British Virgin Islands	\$0.45	\$0.45
Brunei	\$0.86	\$0.86
Bulgaria	\$0.30	\$0.30
Burkina Faso	\$1.04	\$1.04
Burundi	\$1.78	\$1.78
Cambodia	\$2.32	\$2.32
Cameroon	\$0.49	\$0.49
Canada	\$0.05	\$0.05
Cape Verde Island	\$0.82	\$0.82
Cayman Islands	\$0.25	\$0.25
Central African Republic	\$1.51	\$1.51
Chad	\$2.46	\$2.46
Chile	\$0.19	\$0.19
China	\$0.11	\$0.11
Christmas Island	\$0.09	\$0.09
Cocos Island	\$0.09	\$0.09
Colombia	\$0.17	\$0.17
Comoros Islands	\$2.29	\$2.29
Congo	\$0.75	\$0.75
Cook Islands	\$1.65	\$1.65
Costa Rica	\$0.27	\$0.27
Croatia	\$0.33	\$0.33
Cuba	\$0.92	\$0.92
Cyprus	\$0.56	\$0.56

Original Page 15

Original Page 16

Effective: September 1, 2019

SECTION 7 – RESIDENTIAL INTERNATIONAL SERVICE DESCRIPTIONS, RATES AND CHARGES

- 7.3 Optional Calling Plans (continued)
  - 7.3.3 AT&T® Worldwide Value Calling<sup>SM</sup> Plan (continued)
    - (C) Rates and Charges (continued)

Country	Initial Minute	Each Additional Minute	Country	Initial Minute	Each Additional Minute
Czech Republic	\$0.22	\$0.22	Inmarsat (SNAC)	\$9.95	\$9.95
Denmark	\$0.09	\$0.09	Iran	\$0.59	\$0.59
Diego Garcia	\$2.15	\$2.15	Iraq	\$0.79	\$0.79
Diibouti	\$0.97	\$0.97	Ireland	\$0.09	\$0.09
Dominica	\$0.41	\$0.41	Israel	\$0.09	\$0.09
Dominican Republic	\$0.17	\$0.17	Italy	\$0.09	\$0.09
East Timor	\$2.39	\$2.39	Ivory Coast	\$0.69	\$0.69
Easter Island (see Chile)		•	Jamaica	\$0.36	\$0.36
Ecuador	\$0.32	\$0.32	Japan	\$0.09	\$0.09
Egypt	\$0.35	\$0.35	Jordan	\$0.59	\$0.59
El Salvador	\$0.23	\$0.23	Kazakhstan	\$0.16	\$0.16
Equatorial Guinea	\$2.19	\$2.19	Kenya	\$0.67	\$0.67
Eritrea	\$0.89	\$0.89	Kiribati	\$1.16	\$1.16
Estonia	\$0.69	\$0.69	Korea (North)	\$3.30	\$3.30
Ethiopia	\$1.01	\$1.01	Korea (South)	\$0.09	\$0.09
Faeroe Island	\$0.52	\$0.52	Kosovo	\$0.09	\$0.09
Falkland Island	\$1.21	\$1.21	Kuwait	\$0.47	\$0.47
Fiji Island	\$1.03	\$1.03	Kyrgyzstan	\$0.31	\$0.31
Finland	\$0.09	\$0.09	Laos	\$2.43	\$2.43
France	\$0.09	\$0.09	Latvia	\$0.35	\$0.35
French Antilles	\$0.65	\$0.65	Lebanon	\$0.55	\$0.55
French Guiana	\$0.66	\$0.66	Lesotho	\$0.84	\$0.84
French Polynesia	\$1.24	\$1.24	Liberia	\$0.59	\$0.59
Gabon	\$1.09	\$1.09	Libya	\$0.97	\$0.97
Gambia	\$0.59	\$0.59	Liechtenstein	\$0.09	\$0.09
Georgia	\$0.79	\$0.79	Lithuania	\$0.35	\$0.35
Germany	\$0.09	\$0.09	Luxembourg	\$0.09	\$0.09
Ghana	\$0.45	\$0.45	Macao	\$0.77	\$0.77
Gibraltar	\$1.04	\$1.04	Macedonia	\$0.55	\$0.55
Greece	\$0.09	\$0.09	Madagascar	\$2.18	\$2.18
Greenland	\$0.58	\$0.58	Malawi	\$0.49	\$0.49
Grenada	\$0.45	\$0.45	Malaysia	\$0.18	\$0.18
Guadeloupe	\$0.61	\$0.61	Maldives	\$1.45	\$1.45
Guatemala	\$0.24	\$0.24	Mali Republic	\$1.06	\$1.06
Guinea	\$0.82	\$0.82	Malta Republic	\$0.72	\$0.72
Guinea-Bissau	\$1.65	\$1.65	Marshall Island	\$0.74	\$0.74
Guyana	\$0.84	\$0.84	Mauritania	\$1.00	\$1.00
Haiti	\$0.49	\$0.49	Mauritius	\$1.25	\$1.25
Honduras	\$0.41	\$0.41	Mayotte Island (see Reu		
Hong Kong	\$0.11	\$0.11	Mexico 1	\$0.09	\$0.09
Hungary	\$0.27	\$0.27	Mexico 2	\$0.10	\$0.10
Iceland	\$0.49	\$0.49	Micronesia	\$0.77	\$0.77
India	\$0.28	\$0.28	Moldova	\$0.49	\$0.49
Indonesia	\$0.29	\$0.29	Monaco	\$0.09	\$0.09

SECTION 7 – RESIDENTIAL INTERNATIONAL SERVICE DESCRIPTIONS, RATES AND CHARGES

- 7.3 Residential Optional Calling Plans (continued)
  - 7.3.3 AT&T® Worldwide Value Calling<sup>SM</sup> Plan (continued)
    - (C) Rates and Charges (continued)

Country	Initial Minute	Each Additional Minute
Mongolia	\$1.70	\$1.70
Montenegro	\$0.42	\$0.42
Montserrat	\$0.52	\$0.52
Morocco	\$0.57	\$0.57
Mozambique	\$1.28	\$1.28
Myanmar (Burma)	\$2.17	\$2.17
Namibia	\$0.80	\$0.80
Nauru	\$1.38	\$1.38
Nepal	\$1.30	\$1.30
Netherlands	\$0.09	\$0.09
Netherlands Antilles	\$0.42	\$0.42
Networks	\$3.95	\$3.95
Networks	\$3.95	\$3.95
New Caledonia	\$1.45	\$1.45
New Zealand	\$0.08	\$0.08
Nicaragua	\$0.40	\$0.40
Niger Republic	\$0.93	\$0.93
Nigeria	\$0.52	\$0.52
Niue	\$3.31	\$3.31
Norfolk Island	\$0.09	\$0.09
Norway	\$0.09	\$0.09
Oman	\$1.07	\$1.07
Pakistan	\$0.28	\$0.28
Palau	\$1.30	\$1.30
Palestinian Authority	\$0.30	\$0.30
Panama	\$0.35	\$0.35
Papua New Guinea	\$1.06	\$1.06
Paraguay	\$0.40	\$0.40
Peru	\$0.22	\$0.22
Philippines	\$0.17	\$0.17
Poland	\$0.16	\$0.16
Portugal	\$0.09	\$0.09
Qatar	\$0.95	\$0.95
Reunion Island (includes Mayotte Island)	\$1.02	\$1.02
Romania	\$0.33	\$0.33
Russia	\$0.16	\$0.16
Rwanda, Republic of	\$1.19	\$1.19
San Marino	\$0.89	\$0.89
Sao Tome	\$1.59	\$1.59
Saudi Arabia	\$0.45	\$0.45
Senegal	\$0.69	\$0.69
Serbia	\$0.42	\$0.42
Seychelles Island	\$1.51	\$1.51
Sierra Leone	\$0.59	\$0.59
Singapore	\$0.14	\$0.14
Slovakia	\$0.25	\$0.25
Slovenia	\$0.56	\$0.56

		Each Additional
Country	Minute	Minute
Solomon Island	\$1.20	\$1.20
Somalia Democratic Rep	\$0.95	\$0.95
South Africa	\$0.36	\$0.36
Spain	\$0.09	\$0.09
Sri Lanka	\$1.02	\$1.02
St. Helena	\$1.15	\$1.15
St. Kitts/Nevis	\$0.49	\$0.49
St. Lucia	\$0.41	\$0.41
St. Maarten	\$0.42	\$0.42
St. Pierre Miquelon	\$0.48	\$0.48
St. Vincent/Grenadines	\$0.55	\$0.55
South Sudan	\$0.89	\$0.89
Sudan	\$0.89	\$0.89
Suriname	\$1.25	\$1.25
Swaziland	\$0.76	\$0.76
Sweden	\$0.09	\$0.09
Switzerland	\$0.09	\$0.09
Syrian Arab Republic	\$0.89	\$0.89
Taiwan	\$0.08	\$0.08
Tajikistan	\$0.23	\$0.23
Tanzania	\$0.49	\$0.49
Thailand	\$0.23	\$0.23
Togo	\$1.11	\$1.11
Tokelau	\$0.84	\$0.84
Tonga Island	\$1.57	\$1.57
Trinidad	\$0.34	\$0.34
Tunisia	\$0.49	\$0.49
Turkey	\$0.39	\$0.39
Turkmenistan	\$0.31	\$0.31
Turks and Caicos Islands	\$0.52	\$0.52
Tuvalu	\$1.85	\$1.85
Uganda	\$0.49	\$0.49
Ukraine	\$0.22	\$0.22
United Arab Emirates	\$0.39	\$0.39
Cinted 7 trao Emirates	Ψ0.57	Ψ0.57
United Kingdom	\$0.08	\$0.08
Uruguay	\$0.40	\$0.40
Uzbekistan	\$0.31	\$0.31
Vanuatu	\$2.37	\$2.37
Vatican City	\$0.09	\$0.09
Venezuela	\$0.20	\$0.20
Vietnam	\$0.49	\$0.49
Wallis Futuna	\$2.38	\$2.38
Western Samoa	\$1.02	\$1.02
Yemen Arab Republic	\$0.98	\$0.98
Zaire (Dem Congo)	\$0.59	\$0.59
Zambia	\$0.59	\$0.59
Zimbabwe	\$0.39	\$0.39
ZIIIIUaUWC	φυ <b>.</b> 42	φ <b>υ.</b> 47

Original Page 17

Original Page 18

Effective: September 1, 2019

SECTION 7 – RESIDENTIAL INTERNATIONAL SERVICE DESCRIPTIONS, RATES AND CHARGES

### 7.4 International Mobile Termination Charge

The International Mobile Termination Charge (IMTC) is an additional per-minute charge that is applied to all international calls originating in the United States and terminating in certain countries to (1) wireless devices; and (2) ranges of fixed wireless telephone numbers which the foreign telecommunications company designates to be for special services. Because many foreign telecommunications companies impose extra fees on the Company to complete these calls, the Company charges the IMTC, in addition to all other applicable charges. The IMTC amount varies by country and changes from time to time. Please check codes and rates periodically for current information. Codes can be found at <a href="https://www.att.com/mobileterm">www.att.com/mobileterm</a>. The following per-minute charge listed in Section 7.4.1 below will apply.

SECTION 7 – RESIDENTIAL INTERNATIONAL SERVICE DESCRIPTIONS, RATES AND CHARGES

# 7.4 International Mobile Termination Charge (continued)

# 7.4.1 Per Country Mobile Termination Charges

Country	Initial Minute	Each Additional Minute
Afghanistan	\$0.0300	\$0.0300
Albania	\$0.1800	\$0.1800
Algeria	\$0.0800	\$0.0800
Andorra	\$0.2300	\$0.2300
Angola	\$0.0800	\$0.0800
Anguilla	\$0.1100	\$0.1100
Argentina	\$0.1800	\$0.1800
Armenia	\$0.1300	\$0.1300
Aruba	\$0.1000	\$0.1000
Australia (including	\$0.1300	\$0.1300
Christmas & Cocos Isls)	ψ0.1300	ψ0.1300
Austria	\$0.1200	\$0.1200
Azerbaijan	\$0.1200	\$0.1200
Bahrain	\$0.0300	\$0.0300
Bangladesh	\$0.0200	\$0.0200
Barbados	\$0.0900	\$0.0900
Belarus	\$0.0400	\$0.0400
Belgium	\$0.1600	\$0.1600
Belize	\$0.0200	\$0.0200
Benin	\$0.0500	\$0.0500
Bolivia	\$0.0400	\$0.0300
Bosnia	\$0.1600	\$0.1600
Botswana	\$0.1300	\$0.1300
Brazil	\$0.1900	\$0.1900
British Virgin Island	\$0.1100	\$0.1100
Brunei	\$0.0200	\$0.0200
Bulgaria	\$0.3300	\$0.3300
Burkina Faso	\$0.0600	\$0.0600
Cambodia	\$0.0000	\$0.0000
Cameroon	\$0.0200	\$0.0200
Cape Verde Island	\$0.0900	\$0.1100
Cayman Islands	\$0.0900	\$0.0900
Chile	\$0.1800	\$0.1800
China	\$0.1800	\$0.1800
Christmas Island		(See Australia)
Cocos Island Colombia	\$0.0300	(See Australia) \$0.0300
		\$0.0300
Comoros Costo Pico	\$0.0900	
Costa Rica	\$0.0200	\$0.0200
Croatia	\$0.1800	\$0.1800
Cyprus	\$0.0500	\$0.0500
Czech Republic	\$0.1800	\$0.1800
Denmark	\$0.1600	\$0.1600
Djibouti	\$0.0000	\$0.0000
Dominica Dominica	\$0.1000	\$0.1000
Dominican Republic	\$0.0700	\$0.0700

G .	Initial	Each Additional
Country	Minute	Minute
Easter Islands	\$0.1800	\$0.1800
Ecuador	\$0.1100	\$0.1100
Egypt	\$0.0200	\$0.0200
El Salvador	\$0.1000	\$0.1000
Equatorial Guinea	\$0.0200	\$0.0200
Estonia	\$0.2900	\$0.2900
Ethiopia	\$0.0400	\$0.0400
Fiji	\$0.0900	\$0.0900
Finland	\$0.1100	\$0.1100
France	\$0.1200	\$0.1200
French Antilles	(See Martinique)	
French Guiana	\$0.0900	\$0.0900
French Polynesia	\$0.0800	\$0.0800
Gabon	\$0.0500	\$0.0500
Gambia	\$0.0000	\$0.0000
Georgia	\$0.0900	\$0.0900
Germany	\$0.1600	\$0.1600
Ghana	\$0.0700	\$0.0700
Gibraltar	\$0.2100	\$0.2100
Greece	\$0.1700	\$0.1700
Greenland	\$0.1700	\$0.1700
Grenada	\$0.1000	\$0.1000
Guadeloupe	\$0.2800	\$0.2800
Guatemala	\$0.0500	\$0.0500
Guinea Per	\$0.0300	\$0.0300
Haiti	\$0.0800	\$0.0800
Honduras	\$0.0200	\$0.0200
Hungary	\$0.1500	\$0.1500
Iceland	\$0.1900	\$0.1900
India	\$0.0000	\$0.0000
Indonesia	\$0.0200	\$0.0200
Iran	\$0.0000	\$0.0000
Iraq	\$0.0700	\$0.0700
Ireland	\$0.1700	\$0.1700
Israel	\$0.0800	\$0.0800
Italy	\$0.1600	\$0.1600
Ivory Coast	\$0.0600	\$0.0600
Jamaica Jamaica	\$0.1100	\$0.1100
Japan	\$0.1100	\$0.1100
Jordan	\$0.0600	\$0.0600
Kazakhstan	\$0.0300	\$0.0300
Kenya	\$0.0500	\$0.0500
Korea (South)	\$0.0300	\$0.0300
Kosovo	\$0.0300	\$0.2500
Kuwait	\$0.2500	\$0.2500
uxuwan	φυ.υσυυ	φυ.υ.ου

Original Page 19

BUSINESS AND RESIDENTIAL SERVICES

PRICING AND SERVICE GUIDE

SECTION 7 – RESIDENTIAL INTERNATIONAL SERVICE DESCRIPTIONS, RATES AND CHARGES

# 7.4 International Mobile Termination Charge (continued)

# 7.4.1 Per Country Mobile Termination Charges (continued)

Country	Initial Minute	Each Additional Minute
Kyrgyzstan	\$0.0000	\$0.0000
Laos	\$0.0000	\$0.0000
Latvia	\$0.1300	\$0.1300
Lebanon	\$0.0900	\$0.0900
Lesotho	\$0.0600	\$0.0600
Libyan	\$0.0500	\$0.0500
Liechtenstein	\$0.5000	\$0.5000
Lithuania	\$0.1400	\$0.1400
Luxembourg	\$0.2200	\$0.2200
Macedonia	\$0.2200	\$0.2200
Madagascar	\$0.0000	\$0.0000
Malawi	\$0.0200	\$0.0200
Malaysia	\$0.0400	\$0.0400
Mali	\$0.1100	\$0.1100
Malta	\$0.1700	\$0.1700
Martinique (includes	\$0.2100	\$0.2100
Martinique (includes St. Barthelemy, St. Martin and French Antilles)		
Mauritania	\$0.0900	\$0.0900
Mauritius	\$0.0400	\$0.0400
Mayotte	(See Reunion Isl)	(See Reunion Isl)
Mexico (All Zones)	\$0.1500	\$0.1500
Moldova	\$0.0900	\$0.0900
Monaco	\$0.2500	\$0.2500
Mongolia	\$0.0700	\$0.0700
Montenegro	\$0.2600	\$0.2600
Morocco	\$0.1500	\$0.1500
Mozambique	\$0.1400	\$0.1400
Myanmar (Burma)	\$0.1000	\$0.1000
Namibia	\$0.1400	\$0.1400
Nepal	\$0.1000	\$0.1000
Netherlands	\$0.2200	\$0.2200
Netherland Antilles	\$0.0300	\$0.0300
New Zealand	\$0.2100	\$0.2100
Nicaragua	\$0.0800	\$0.0800
Nigeria	\$0.0600	\$0.0600
Norway	\$0.1500	\$0.1500
Oman	\$0.0700	\$0.0700
Pakistan	\$0.0000	\$0.0000
Panama	\$0.0900	\$0.0900
Paraguay	\$0.1400	\$0.1400
Peru	\$0.1600	\$0.1600
Philippines	\$0.0400	\$0.0400
Poland	\$0.1700	\$0.1700
Portugal	\$0.1600	\$0.1600
Qatar	\$0.0400	\$0.0400
Reunion Island (& Mayotte Island)	\$0.2100	\$0.2100
Romania	\$0.1400	\$0.1400

Initial Minute	Each Additional Minute
	\$0.0300
\$0.0200	\$0.0200
\$0.0500	\$0.0500
	\$0.0900
\$0.1800	\$0.1800
\$0.0000	\$0.0000
\$0.0000	\$0.0000
\$0.1800	\$0.1800
\$0.2800	\$0.2800
\$0.1300	\$0.1300
\$0.2300	\$02300
\$0.1400	\$0.1400
\$0.0300	\$0.0300
\$0.1100	\$0.1100
\$0.1000	\$0.1000
\$0.0300	\$0.0300
\$0.1900	\$0.1900
\$0.1000	\$0.1000
	\$0.0300
\$0.0000	\$0.0000
\$0.1900	\$0.1900
	\$0.1400
	\$0.2100
	\$0.0500
	\$0.0700
1	\$0.0900
	\$0.0000
	\$0.0900
	\$0.1800
	\$0.1400
	\$0.1100
\$0.0200	\$0.0200
	\$0.0700
	\$0.0200
\$0.2000	\$0.2000
	\$0.1300
	\$0.0200
	\$0.1600
\$0.1500	\$0.1500
	\$0.0200
	\$0.1500
	\$0.0200
	\$0.0000 \$0.1200
\$0.1200	\$0.1200
	\$0.0300 \$0.0000 \$0.1800 \$0.2800 \$0.1800 \$0.2800 \$0.1800 \$0.1800 \$0.1800 \$0.1800 \$0.1800 \$0.1800 \$0.1800 \$0.1900 \$0.1900 \$0.1900 \$0.1900 \$0.1400 \$0.0300 \$0.1900 \$0.1400 \$0.1900

Original Page 20

#### SECTION 8 - OPERATOR SERVICES

### 8.1 Directory Assistance Services

### 8.1.1 Directory Assistance

### (A) Directory Assistance

Directory Assistance is available to Business and Residential Customers of the Company for locations within the United States where such information is available to the Company. A Directory Assistance charge applies to each call to the Directory Assistance Bureau. Up to two (2) requests may be made on each call to Directory Assistance. In order to obtain two (2) requests, the Customer must make the request upon reaching a live operator. The Directory Assistance charge applies to each call regardless of whether the Directory Assistance Bureau is able to furnish the requested telephone number. When more than one Directory Assistance Bureau handles requests for listings within the same area code, two listing will be provided only when the listing information is accessible to the Directory Assistance Operator that handles the request.

Original Page 1

Effective: September 1, 2019

Jurisdiction	Per Call Rate
Interstate	\$2.49
Intrastate	
Alabama, Florida, Kentucky, Mississippi,	\$1.20
North Carolina, South Carolina and Tennessee	
Georgia/ <sup>1/</sup>	
IntraLATA	\$0.94
InterLATA	\$1.20
Louisiana	\$0.85

### (B) Florida Exemption for Customers with Disabilities

Customers who have a visual or physical disability that prevents use of a telephone directory are exempt from the charges of Directory Assistance calls for up to and including 50 calls per month. This exemption applies to calls billed to one residential telephone line per Customer certified by the Local Exchange Carrier as disabled and applies to Directory Assistance calls for personal use only. Calls in excess of 50, where billing is available, will be billed the tariffed Directory Assistance charges set forth in (A) and (C) herein. Proof of certification is required.

#### (C) Directory Assistance Call Completion

The Company Directory Assistance operator will complete the call to the number requested by the Customer without requiring the Customer to redial the number. A Directory Assistance Call Completion charge applies for this service. This charge is in addition to the charge for determining the telephone number requested by the Customer and in addition to any usage and per call charges associated with placing the call.

Jurisdiction	Per Call Rate
Interstate	\$0.50
Intrastate	\$0.45
Alabama, Florida, Georgia, Kentucky, Louisiana,	
Mississippi, North Carolina, South Carolina and Tennessee	

<sup>/1/</sup> Where the Company's system cannot distinguish between an intraLATA and an interLATA Directory Assistance Call, the Company will charge the intraLATA rate for those calls.

SECTION 8 – OPERATOR SERVICES

### 8.2 Domestic Operator Services

### 8.2.1 General

Operator Services are provided to Residential and Business Customers with presubscribed Access Lines. Operator Services allows the caller to place a call and arrange for billing to the presubscribed Access Line or to an alternate billing arrangement. All calls are billed in one (1) minute increments after an initial period, for billing purposes, of one (1) minute. Holiday discounts, as defined in the pricing and service guide, apply. Usage charges and an appropriate service charge will be assessed on a per call basis. For calls made using a commercial credit card, acceptance of the card will be dependent upon the Company's ability to verify the card as valid. Only those cards accepted by the Company may be used for Operator Services.

Original Page 2

Effective: September 1, 2019

# 8.2.2 Per-Minute Per Call Usage Charges

Interstate
Per Call Rate
\$1.15

Intrastate	Per Call Rate
Alabama, Florida, Kentucky, Mississippi, North Carolina, and Tennessee	\$0.35
South Carolina	\$0.33
Georgia InterLATA and IntraLATA	
Mileage 0-22	\$0.00
Mileage 23 +	\$0.35
Louisiana	
Mileage 0-10	\$0.14
Mileage 11-16	\$0.19
Mileage 17-22	\$0.24
Mileage 23 +	\$0.25

SECTION 8 - OPERATOR SERVICES

### 8.2 Domestic Operator Services (continued)

### 8.2.3 Service Charges

One of the following per call service charges applies to each live or automated operator assisted station to station calls placed using the Company's services. Per call charges vary by type of call and type of billing method selected by the Customer. The Fully Automated Per Call Service Charge applies if a call; (a) is placed by a party identified as disabled and as a result of that disability cannot complete the call and the caller requests operator assistance for call completion; (b) it cannot be completed by the caller due to equipment failure or trouble on the long distance network and the caller requests operator assistance for call completion.

Original Page 3

Effective: September 1, 2019

Interstate	Comm. Credit Card <sup>/1/</sup>	Billed to Line <sup>/1/</sup>
Fully Automated	\$4.50	\$4.50
Operator Assisted	\$4.99	\$12.50
Operator Dialed	\$4.99	\$12.50

Intrastate	Comm. Credit Card <sup>/1/</sup>	Billed to Line <sup>/1/</sup>
Alabama		
Fully Automated	\$1.25	\$2.10
Operator Assisted	\$2.10	\$3.10
Operator Dialed	\$2.10	\$3.10
Florida		
Fully Automated	\$1.55	\$1.55
Operator Assisted	\$1.55	\$1.55
Operator Dialed	\$1.55	\$1.55
Georgia		
Fully Automated	\$4.95	\$3.95
Operator Assisted	\$5.50	\$6.50
Operator Dialed	\$5.50	\$6.50
Kentucky		
Fully Automated	\$4.45	\$3.55
Operator Assisted	\$4.95	\$5.85
Operator Dialed	\$4.95	\$5.85

/1/ Where this billing option is available. Billed to Line may also be billed as Sent Paid service.

Original Page 4

Effective: September 1, 2019

SECTION 8 – OPERATOR SERVICES

# 8.2 Domestic Operator Services (continued)

# 8.2.3 Service Charges (continued)

_	Comm.	Billed
Intrastate	Credit	to
	Card <sup>/1/</sup>	Line <sup>/1/</sup>
Louisiana		
Fully Automated	\$1.35	\$1.95
Operator Assisted	\$1.95	\$1.95
Operator Dialed	\$4.05	\$4.05
Mississippi		
Fully Automated	\$0.90	\$1.80
Operator Assisted	\$1.80	\$2.25
Operator Dialed	\$1.80	\$2.25
North Carolina		
Fully Automated	\$1.50	\$2.25
Operator Assisted	\$2.25	\$2.25
Operator Dialed	\$2.25	\$2.25
South Carolina		
Fully Automated	\$4.45	\$4.45
Operator Assisted	\$4.95	\$9.45
Operator Dialed	\$4.95	\$9.45
Tennessee		
Fully Automated	\$4.95	\$3.15
Operator Assisted	\$4.95	\$3.15
Operator Dialed	\$4.95	\$3.15

/1/ Where this billing option is available. Billed to Line may also be billed as Sent Paid service.

SECTION 8 – OPERATOR SERVICES

### 8.3 International Operator Services

### 8.3.1 General

International Operator Services are available on a presubscribed and casual calling basis to Business and Residential Customers and are available from originating locations within the United States where Customers have the ability to dial directly to the Company network. Calls may be placed between the United States and those countries listed herein. Operator Services allow the Customer to place a call and arrange for billing to the presubscribed Residential Access Line or to an alternate billing arrangement. Calls are rounded up to the next whole minute for billing purposes and are billed to the Customer through the monthly bill of the Customer's local exchange carrier.

Original Page 5

Effective: September 1, 2019

### 8.3.2 Per Minute Rates - International Locations

Country	Initial Minute	Each Additional Minute	Country		Each Additional Minute
Afghanistan	\$7.00	\$7.00	Bulgaria	\$2.55	\$2.55
Albania	\$4.41	\$4.41	Burkina Faso	\$3.89	\$3.89
Algeria	\$2.59	\$2.59	Burundi	\$4.85	\$4.85
Andorra	\$1.96	\$1.96	Cameroon	\$2.65	\$2.65
Angola	\$4.95	\$4.95	Canada	\$0.75	\$0.75
Anguilla	\$1.95	\$1.95	Cape Verde Islands	\$3.34	\$3.34
Antarctica (Casey Base)	\$5.57	\$5.57	Cayman Islands	\$1.75	\$1.75
Antarctica (Scott Base)	\$5.57	\$5.57	Central African Republic	\$4.79	\$4.79
Antigua	\$1.81	\$1.81	Chad	\$5.33	\$5.33
Argentina	\$2.51	\$2.51	Chile	\$1.83	\$1.83
Armenia	\$3.93	\$3.93	Cameroon	\$2.65	\$2.65
Aruba	\$1.71	\$1.71	China	\$3.47	\$3.47
Ascension Island	\$3.28	\$3.28	Christmas Island	\$2.08	\$2.08
Australia	\$2.08	\$2.08	Cocos Island	\$2.08	\$2.08
Austria	\$1.85	\$1.85	Colombia	\$2.54	\$2.54
Azerbaijan	\$3.92	\$3.92	Comoros	\$5.44	\$5.44
Bahamas	\$1.52	\$1.52	Congo, Republic of	\$2.46	\$2.46
Bahrain	\$2.65	\$2.65	Cook Islands	\$5.81	\$5.81
Bangladesh	\$4.85	\$4.85	Costa Rica	\$2.10	\$2.10
Barbados	\$1.82	\$1.82	Croatia	\$2.18	\$2.18
Belarus	\$3.98	\$3.98	Cuba	\$1.89	\$1.89
Belgium	\$1.92	\$1.92	Cyprus	\$2.23	\$2.23
Belize	\$2.18	\$2.18	Czech Republic	\$2.27	\$2.27
Benin	\$2.46	\$2.46	Denmark	\$1.84	\$1.84
Bermuda	\$1.68	\$1.68	Diego Garcia	\$4.49	\$4.49
Bhutan	\$5.54	\$5.54	Djibouti	\$3.69	\$3.69
Bolivia	\$2.33	\$2.33	Dominica	\$2.21	\$2.21
Bosnia & Herzegovina	\$2.19	\$2.19	Dominican Republic	\$1.82	\$1.82
Botswana	\$2.30	\$2.30	East Timor	\$9.83	\$9.83
Brazil	\$2.49	\$2.49	Easter Island	\$1.83	\$1.83
British Virgin Islands	\$1.73	\$1.73	Ecuador	\$2.46	\$2.46
Brunei	\$2.87	\$2.87	Egypt	\$2.76	\$2.76

Original Page 6 Effective: September 1, 2019

SECTION 8 – OPERATOR SERVICES

# 8.3 International Operator Services (continued)

# 8.3.2 Per Minute Rates - International Locations (continued)

		Each			Each
		Additional			Additional
Country		Minute	Country		Minute
El Salvador	\$2.07	\$2.07	Ivory Coast	\$3.14	\$3.14
Equatorial Guinea	\$4.94	\$4.94	Jamaica	\$1.94	\$1.94
Eritrea	\$3.13	\$3.13	Japan	\$2.15	\$2.15
Estonia	\$4.37	\$4.37	Jordan	\$2.53	\$2.53
Ethiopia	\$2.72	\$2.72	Kazakhstan	\$3.98	\$3.98
Faeroe Islands	\$1.81	\$1.81	Kenya	\$2.68	\$2.68
Falkland Islands	\$4.22	\$4.22	Kiribati	\$4.49	\$4.49
Fiji Islands	\$3.45	\$3.45	Korea, North	\$6.42	\$6.42
Finland	\$1.92	\$1.92	Korea, South	\$2.23	\$2.23
France	\$1.64	\$1.64	Kosovo	\$1.62	\$1.62
French Antilles	\$1.74	\$1.74	Kuwait	\$2.35	\$2.35
French Guiana	\$2.10	\$2.10	Kyrgyzstan	\$3.82	\$3.82
French Polynesia	\$3.15	\$3.15	Laos	\$7.22	\$7.22
Gabon	\$2.65	\$2.65	Latvia	\$3.94	\$3.94
Gambia	\$2.47	\$2.47	Lebanon	\$4.27	\$4.27
Georgia	\$3.96	\$3.96	Lesotho	\$2.45	\$2.45
Germany	\$1.63	\$1.63	Liberia	\$2.53	\$2.53
Ghana	\$2.73	\$2.73	Libya	\$2.65	\$2.65
Gibraltar	\$2.43	\$2.43	Liechtenstein	\$1.88	\$1.88
Greece	\$2.59	\$2.59	Lithuania	\$3.90	\$3.90
Greenland	\$2.22	\$2.22	Luxembourg	\$1.85	\$1.85
Grenada	\$1.87	\$1.87	Macao	\$3.36	\$3.36
Guadeloupe	\$1.75	\$1.75	Macedonia	\$2.23	\$2.23
Guatemala	\$2.28	\$2.28	Madagascar	\$5.28	\$5.28
Guinea	\$3.13	\$3.13	Malawi	\$2.24	\$2.24
Guinea-Bissau	\$5.21	\$5.21	Malaysia	\$2.90	\$2.90
Guyana	\$2.69	\$2.69	Maldives	\$4.37	\$4.37
Haiti	\$2.11	\$2.11	Mali Republic	\$3.45	\$3.45
Honduras	\$2.34	\$2.34	Malta	\$2.57	\$2.57
Hong Kong	\$2.46	\$2.46	Marshall Islands	\$3.11	\$3.11
Hungary	\$2.09	\$2.09	Mauritania	\$3.68	\$3.68
Iceland	\$2.14	\$2.14	Mauritius	\$4.35	\$4.35
India	\$3.48	\$3.48	Mayotte Island (See R	eunion I	sland)
Indonesia	\$3.01	\$3.01	Mexico – Rate Step 1	\$0.75	\$0.75
INMARSAT - SNAC	\$11.97	\$11.97	Mexico – Rate Step 2	\$1.50	\$1.50
Iran	\$3.02	\$3.02	Micronesia	\$2.88	\$2.88
Iraq	\$3.87	\$3.87	Moldova	\$3.45	\$3.45
Ireland	\$1.68	\$1.68	Monaco	\$1.62	\$1.62
Israel	\$2.43	\$2.43	Mongolia	\$5.64	\$5.64
Italy	\$1.88	\$1.88	Montenegro	\$2.37	\$2.37

Original Page 7 Effective: September 1, 2019

# SECTION 8 – OPERATOR SERVICES

# 8.3 International Operator Services (continued)

# 8.3.2 Per Minute Rates - International Locations (continued)

	Initial	Each Additional		Initial	Each Additional
Country	Minute		Country	Minute	
Montserrat	\$2.08	\$2.08	Serbia	\$2.37	\$2.37
Morocco	\$2.49	\$2.49	Seychelles Islands	\$4.74	\$4.74
Mozambique	\$4.41	\$4.41	Sierra Leone	\$3.35	\$3.35
Myanmar	\$7.62	\$7.62	Singapore	\$2.35	\$2.35
Namibia	\$2.44	\$2.44	Slovakia	\$2.12	\$2.12
Nauru	\$4.97	\$4.97	Slovenia	\$2.31	\$2.31
Nepal	\$4.18	\$4.18	Solomon Islands	\$4.47	\$4.47
Netherlands	\$1.26	\$1.26	Somalia	\$4.64	\$4.64
Netherlands Antilles	\$1.83	\$1.83	South Africa	\$2.10	\$2.10
Networks	\$10.64	\$10.64	Spain	\$2.11	\$2.11
New Caledonia	\$3.38	\$3.38	Spanish Sahara	\$5.40	\$5.40
New Zealand	\$2.35	\$2.35	Sri Lanka	\$4.34	\$4.34
Nicaragua	\$2.06	\$2.06	St. Helena	\$4.52	\$4.52
Niger Republic	\$3.49	\$3.49	St. Kitts/Nevis	\$1.94	\$1.94
Nigeria	\$2.08	\$2.08	St. Lucia	\$1.88	\$1.88
Niue	\$5.21	\$5.21	St. Maarten	\$1.83	\$1.83
Norfolk Island	\$5.57	\$5.57	St. Pierre/Miquelon	\$1.66	\$1.66
Norway	\$1.64	\$1.64	St. Vincent/Grenadines	\$1.92	\$1.92
Oman	\$2.63	\$2.63	South Sudan	\$4.72	\$4.72
Pakistan	\$5.04	\$5.04	Sudan	\$4.72	\$4.72
Palau, Republic of	\$4.51	\$4.51	Suriname	\$3.32	\$3.32
Palestinian Authority	\$4.31	\$4.31	Swaziland	\$2.49	\$2.49
Panama, Republic of	\$2.11	\$2.11	Sweden	\$1.65	\$1.65
Papua New Guinea	\$2.88	\$2.88	Switzerland	\$1.77	\$1.77
Paraguay	\$2.90	\$2.90	Syria	\$4.01	\$4.01
Peru	\$2.39	\$2.39	Taiwan	\$2.65	\$2.65
Philippines	\$2.77	\$2.77	Tajikistan	\$3.84	\$3.84
Pitcairn	\$5.04	\$5.04	Tanzania	\$2.72	\$2.72
Poland	\$1.93	\$1.93	Thailand	\$2.84	\$2.84
Portugal	\$2.16	\$2.16	Togo	\$2.87	\$2.87
Qatar	\$2.73	\$2.73	Tokelau	\$9.00	\$9.00
Reunion Island (includes	\$3.52	\$3.52	Tonga Islands	\$3.85	\$3.85
Mayotte Island)					
Romania	\$2.86	\$2.86	Trinidad & Tobago	\$2.00	\$2.00
Russia	\$3.98	\$3.98	Tunisia	\$2.46	\$2.46
Rwanda	\$3.91	\$3.91	Turkey	\$2.44	\$2.44
San Marino	\$1.97	\$1.97	Turkmenistan	\$1.88	\$1.88
Sao Tome	\$4.87	\$4.87	Turks & Caicos	\$3.81	\$3.81
Saudi Arabia	\$2.68	\$2.68	Tuvalu	\$6.10	\$6.10
Senegal Republic	\$2.98	\$2.98	Uganda	\$2.74	\$2.74

SECTION 8 – OPERATOR SERVICES

# 8.3 International Operator Services (continued)

# 8.3.2 Per Minute Rates - International Locations (continued)

Country	Initial Minute	Each Additional Minute
Ukraine	\$3.96	\$3.96
United Arab Emirates	\$2.33	\$2.33
United Kingdom	\$1.54	\$1.54
Uruguay	\$2.26	\$2.26
Uzbekistan	\$3.90	\$3.90
Vanuatu	\$6.02	\$6.02
Vatican City	\$1.88	\$1.88
Venezuela	\$1.71	\$1.71

	Initial	Each Additional
Country	Minute	Minute
Vietnam	\$3.48	\$3.48
Wallis & Futuna Islands	\$5.43	\$5.43
Western Sahara	\$5.40	\$5.40
Western Samoa	\$4.07	\$4.07
Yemen Arab Republic	\$2.46	\$2.46
Zaire	\$2.34	\$2.34
Zambia	\$2.34	\$2.34
Zimbabwe	\$2.38	\$2.38

Original Page 8

Effective: September 1, 2019

# 8.3.3 International Per Call Surcharges

	Other International			
	Locations	Canada	Mexico	Cuba
<b>Billing Option - Commercial Cred</b>	it Card <sup>/1/</sup>			
Fully Automated Station-to-Station	N/A	N/A	N/A	N/A
Operator Assisted Station-to-Station	N/A	N/A	N/A	N/A
Operator Dialed Station-to-Station	N/A	N/A	N/A	N/A
Billing Option - Sent Paid				
Fully Automated Station-to-Station	\$9.00	\$9.00	\$9.00	\$9.00
Operator Assisted (where available)	\$9.00	\$9.00	\$9.00	\$9.00
Operator Dialed (where available)	\$9.00	\$9.00	\$9.00	\$9.00

# 8.3.4 International Mobile Termination Charge

See Sections 6.4.1 and 7.4.8 of this pricing and service guide for the applicable International Mobile Termination Charge.

/1/ Where this billing option is available.

Original Page 1

Effective: September 1, 2019

#### SECTION 9 - MISCELLANEOUS CHARGES AND FEES

#### 9.1 Returned Check/Bank Draft/Electronic Funds Transfer

When another telecommunications carrier provides the billing function on behalf of the Company, the other carrier's bad check charge applies. Otherwise, the Company will assess the Customer a return check charge of \$20.00 for any check that is returned for any reason by the financial institution on which it is drawn.

# 9.2 Pay Telephone Surcharge

An undiscountable surcharge shall apply to each call utilizing the Company's services which originate from an instrument which the Company identifies as a domestic Pay Telephone. Unless otherwise specified in this pricing and service guide, services for which a Pay Telephone Surcharge applies include, but are not limited to, calls placed to a toll-free number provided by the Company with its inbound long distance services. Surcharges will be billed to the inbound long distance Customer rather than the party originating the call.

The following per call rates apply:

Jurisdiction	Per Call Rate
Interstate	\$0.65
Intrastate	
Alabama	\$0.62
Florida, Georgia, Kentucky, Louisiana and North Carolina	\$0.65
Mississippi, South Carolina and Tennessee	\$0.50

#### 9.3 Federal Universal Service Fee

The Company is required by the Federal Communications Commission (FCC) to contribute to the Federal Universal Service Fund (FUSF) which supports universal service in high cost areas, assist low income users of telecommunications, and assist schools, libraries and rural health care facilities to obtain telecommunications and information services. The FCC permits the Company to assess a monthly surcharge for recovery of contributions paid by the Company to FUSF. The FCC sets the rate that the Company is required to contribute, and the FCC can increase and decrease the rate on a quarterly basis. The Company will assess all Customers a Federal Universal Service Fee. The Company reserves the right to change the amount of the assessment based on changes made to the FUSF rate by the FCC. The FCC Contribution factor subject to change quarterly can be found at <a href="http://www.fcc.gov/omd/contribution-factor.html">http://www.fcc.gov/omd/contribution-factor.html</a>.

The Federal Universal Service Fee is not subject to discounts nor contributes to any monthly minimums or revenue commitments which may apply to Company services subscribed to by the Customer.

Original Page 2

Effective: September 1, 2019

### SECTION 9 - MISCELLANEOUS CHARGES AND FEES

### 9.4 Presubscribed Interexchange Carrier Charge (PICC)

For Customers who select the Company as their Primary Interexchange Carrier, the Company will assess the Customer a monthly recurring charge, called a Presubscribed Interexchange Carrier Charge (PICC). This charge will be assessed on a per line basis as shown below. PICCs will not be prorated for a partial month of service and are not subject to discounts nor contribute to any monthly minimums or revenue commitments which may apply to Company services subscribed to by the Customer. Credits for interruption of service as described in Section 2.14 of this pricing and service guide do not apply to PICCs.

Per Multi-Line Business Line or Trunk\$0.00Per Centrex Station Line\$0.00Per ISDN Basic Rate Digital Subscriber Line:\$0.00Per ISDN Primary Rate Interface:\$0.00

### 9.5 Primary Interexchange Carrier (PIC) Change Charge

The Company may, at its option, elect to incur the PIC change charge on the Customer's behalf for new Customers who presubscribe their interstate service to the Company through Company-designated sales channels for Company-designated marketing campaigns and where the appropriate arrangements are in place between the Company and the eligible Customer's local exchange carrier. In the event that no such arrangements are in place, the Company may, at its option, reimburse the Customer, or issue the appropriate credits on the Customer's invoice upon Customer's proof of payment of such PIC change charges.

### 9.6 Carrier Cost Recovery Fee

Residential Customers will be charged \$2.24 per month Carrier Cost Recovery Fee to recover certain costs associated with providing state-to-state and international long distance service including expenses for national regulatory fees and programs, and connection and account servicing charges. This fee applies for each month in which the customer has any AT&T Long Distance Service state-to-state and/or international charges on their bill. The Carrier Cost Recovery Fee is applied in full whether or not the Customer's billing period covers a full month. This fee is not a tax or charge required by the government.

### 9.7 Federal Regulatory Fee

The Federal Regulatory Fee is a non-discountable monthly charge to recover amounts paid to the federal government for regulatory costs and telecommunications services for the hearing impaired and costs associated with local number portability administration. This fee is applied to Business Customers only. The Federal Regulatory Fee is equal to the Customer's total net charges for interstate and international telecommunications services, after application of all applicable discounts and credits, multiplied by 6.38%.

Original Page 3

Effective: September 1, 2019

### SECTION 9 - MISCELLANEOUS CHARGES AND FEES

#### 9.8 State Specific

### **Florida**

### **Emergency Call Exemptions**

The following calls are exempted from all charges: Emergency calls to recognizable authorized civil agencies including police, fire, ambulance, bomb squad and poison control. AT&T Long Distance Service will only handle these calls if the caller dials all of the digits to route and bill the call. Credit will be given for any billed charges pursuant to this exemption on a subsequent bill after verified notification by the billed Customer within thirty (30) days of billing.

### Telecommunications Relay Service Discount

Subject to the availability of relay service to the end user, intrastate toll calls received from the relay service shall be discounted by 50 percent of the applicable rate for a voice non-relay call except that where either the calling or called party indicates that either party is both hearing and visually impaired, the call will be discounted 60 percent off the applicable rate for voice non-relay calls. The above discounts apply only to time-sensitive elements of a charge for the call and shall not apply to per call charges or surcharges. Credit for the discount may be given on a subsequent bill.

### Louisiana

### Universal Service Fee

A monthly recurring Louisiana Universal Service Fee is applicable to each Customer's total monthly net intrastate charges for telecommunications services that are provided within the state of Louisiana. This fee is a non-discountable monthly recurring charge to recover the Company's required contributions to the Louisiana Universal Service Fund. The Company will revise this fee if the method and/or amount of its required contribution to the Louisiana Universal Fund changes.

#### **South Carolina**

### Universal Service Fund

A monthly recurring surcharge equal to the revenue percentage established by the Administrator of the South Carolina Universal Service Fund will be applied to Customers' total monthly interstate and intrastate telecommunications services in order to support South Carolina's Universal Service Fund.

Original Page 1

Effective: September 1, 2019

### **SECTION 10 - PROMOTIONS**

### 10.1 General

From time to time, the Company shall, at its option, promote subscription or stimulate network usage by offering to waive some or all the nonrecurring or recurring charges for the Customer (if eligible) of target services for a limited duration. Such promotions shall be made available to all similarly situated Customers in the target market area and will comply with all applicable Commission regulations.

#### **SECTION 10 - PROMOTIONS**

### 10.2 Promotional Offerings

- 10.2.1 AT&T Business Unlimited Calling<sup>SM</sup> Retention Promotion VIII
  - (A) The sign-up period for this promotion is September 1, 2018 through August 31, 2019. Service must be activated by September 30, 2019. This promotion is available to Business Customers who:
    - .1 currently subscribe to the Company for interLATA (interstate and intrastate) and intraLATA (local toll) Service; and state an intention to disconnect long distance service with the Company;

Original Page 2

- .2 newly subscribe to AT&T Business Unlimited Calling as defined in Section 4.3.1 of this pricing and service guide;
- .3 newly subscribe to or have a one (1), two (2) or three (3) year local service term with an Affiliated ILEC of the Company; and,
- .4 request to participate in this promotional offering.
- (B) Customers subscribing to this promotion will be charged the following promotional MRC for 12 consecutive months in lieu of the MRC charge specified in Section 4.3.1 (E) of this pricing and service guide:

Direct Dial Service	Promotional MRC
Primary Line	\$10.00
Each Additional Line after the Primary Line	\$10.00

- (C) Customers must continue to subscribe to and maintain the requirements of AT&T Business Unlimited Calling as defined in 4.3.1 of this pricing and service guide to continue to qualify for this promotion. Should Customer disconnect service with AT&T Long Distance Service or change its calling plan from AT&T Business Unlimited Calling prior to the expiration of the 12-month promotional period, Customer will forfeit future promotion benefit.
- (D) Customers who continue to subscribe and meet the requirements of AT&T Business Unlimited Calling at the end of the 12-month promotional period will be charged the MRC charges specified in Section 4.3.1 (E) of this pricing and service guide.
- (E) This promotion cannot be combined with any other promotional offers associated with AT&T Business Unlimited Calling.

#### **SECTION 10 - PROMOTIONS**

### 10.2 Promotional Offerings (continued)

- 10.2.2 AT&T Business Calling<sup>SM</sup> Monthly Retention Promotion #503-D
  - (A) The sign-up period for this promotion is September 1, 2018 through August 31, 2019. Service must be activated by September 30, 2019. This promotion is available to Business Customers who:
    - .1 currently subscribe to the Company for interLATA (interstate and intrastate) service, intraLATA (local toll) service only, or interLATA/intraLATA service combined; and state an intention to disconnect long distance service with the Company;

Original Page 3

- .2 newly or currently subscribe to the AT&T Business Calling<sup>SM</sup> Monthly plan as defined in Section 4.3.3 of this pricing and service guide;
- .3 newly subscribe to or have a one (1), two (2) or three (3) year local service term with an Affiliated ILEC of the Company; and,
- .4 request to participate in this promotional offering.
- (B) Qualifying Customers subscribing to this promotion will be charged the following Promotional MRC and Promotional Per-Minute Usage Rate for a twelve (12) month benefit period in lieu of the MRC and Usage Rate for the AT&T Business Calling Monthly plan Direct Dial Service as specified in Section 4.3.3(E) of this pricing and service guide for the initial twelve (12) month period under the plan.

Promotional	Promotional
MRC	Per-Minute Rate
\$5.00	\$0.06

- (C) Customers must continue to subscribe to and maintain the requirements of the AT&T Business Calling<sup>SM</sup> Monthly plan as defined in Section 4.3.3 of this pricing and service guide to continue to qualify for this promotion. Should Customer disconnect service with AT&T Long Distance Service or change its calling plan from the AT&T Business Calling<sup>SM</sup> Monthly plan prior to the 12-month promotional period, Customer will forfeit future promotion benefit.
- (D) Customers who continue to subscribe and meet the requirements of the AT&T Business Calling<sup>SM</sup> Monthly plan at the end of the 12-month promotional period will be charged the MRC and per-minute rate specified in Section 4.3.3 (E) of this pricing and service guide.
- (E) This promotion cannot be combined with any other promotional offers associated with the AT&T Business Calling<sup>SM</sup> Monthly plan.

#### **SECTION 10 - PROMOTIONS**

- 10.2 Promotional Offerings (continued)
  - 10.2.3 AT&T Business Unlimited Calling<sup>SM</sup> Retention Promotion E
    - (A) The sign-up period for this promotion is September 1, 2019 through August 31, 2020. Service must be activated by September 30, 2020. This promotion is available to Business Customers who:
      - .1 currently subscribe to the Company for interLATA (interstate and intrastate) and intraLATA (local toll) Service; and state an intention to disconnect long distance service with the Company;

Original Page 4

- newly subscribe to AT&T Business Unlimited Calling as defined in Section 4.3.1 of this pricing and service guide; and,
- .3 request to participate in this promotional offering.
- (B) Customers subscribing to this promotion will be charged the following promotional MRC for 12 consecutive months in lieu of the MRC charge specified in Section 4.3.1 (E) of this Business Service Guide:

Direct Dial Service	Promotional MRC
Primary Line	\$10.00
Each Additional Line after the Primary Line	\$10.00

- (C) Customers must continue to subscribe to and maintain the requirements of AT&T Business Unlimited Calling as defined in 4.3.1 of this pricing and service guide to continue to qualify for this promotion. Should Customer disconnect service with AT&T Long Distance Service or change its calling plan from AT&T Business Unlimited Calling prior to the expiration of the 12-month promotional period, Customer will forfeit future promotion benefit.
- (D) Customers who continue to subscribe and meet the requirements of AT&T Business Unlimited Calling at the end of the 12-month promotional period will be charged the MRC charges specified in Section 4.3.1 (E) of this pricing and service guide.
- (E) This promotion cannot be combined with any other promotional offers associated with AT&T Business Unlimited Calling.

#### **SECTION 10 - PROMOTIONS**

- 10.2 Promotional Offerings (continued)
  - 10.2.4 AT&T Business Calling<sup>SM</sup> Monthly Retention Promotion #503-E
    - (A) The sign-up period for this promotion is September 1, 2019 through August 31, 2020. Service must be activated by September 30, 2020. This promotion is available to Business Customers who:
      - .1 currently subscribe to the Company for interLATA (interstate and intrastate) service, intraLATA (local toll) service only, or interLATA/intraLATA service combined; and state an intention to disconnect long distance service with the Company;

Original Page 5

- .2 newly or currently subscribe to the AT&T Business Calling<sup>SM</sup> Monthly plan as defined in Section 4.3.3 of this pricing and service guide; and,
- .3 request to participate in this promotional offering.
- (B) Qualifying Customers subscribing to this promotion will be charged the following Promotional MRC and Promotional Per-Minute Usage Rate for a twelve (12) month benefit period in lieu of the MRC and Usage Rate for the AT&T Business Calling Monthly plan Direct Dial Service as specified in Section 4.3.3(E) of this pricing and service guide for the initial twelve (12) month period under the plan.

Promotional	Promotional
MRC	Per-Minute Rate
\$5.00	\$0.06

- (C) Customers must continue to subscribe to and maintain the requirements of the AT&T Business Calling<sup>SM</sup> Monthly plan as defined in Section 4.3.3 of this pricing and service guide to continue to qualify for this promotion. Should Customer disconnect service with AT&T Long Distance Service or change its calling plan from the AT&T Business Calling<sup>SM</sup> Monthly plan prior to the 12-month promotional period, Customer will forfeit future promotion benefit.
- (D) Customers who continue to subscribe and meet the requirements of the AT&T Business Calling<sup>SM</sup> Monthly plan at the end of the 12-month promotional period will be charged the MRC and per-minute rate specified in Section 4.3.3 (E) of this pricing and service guide.
- (E) This promotion cannot be combined with any other promotional offers associated with the AT&T Business Calling<sup>SM</sup> Monthly plan.

Original Page 1

Effective: September 1, 2019

SECTION 11 - MARKET TRIALS

### 11.1 General Information

From time to time, the Company shall, at its option, promote subscription or stimulate network usage by offering market trials to certain Customer (if eligible) of target areas for a limited duration. Market trials will be made available to a limited number of Customers within a target area to test new products and services prior to implementation on a larger scale.

Original Page 1

Effective: September 1, 2019

### SECTION 12 – GRANDFATHERED SERVICES

#### 12.1 General

Grandfathered Services are those services no longer offered to new Customers following the date specified for each service in this section of this pricing and service guide. For existing Customers, the effect of grandfathering a service will depend on the form or type of grandfathering stated for each grandfathered service offering. Grandfathered services as listed in this section of the pricing and service guide are classified as grandfathered according to the following types.

Type 1 Customers may continue to use the grandfathered service arrangement and may add to, change, move or transfer the service as required. Type 2 Customers may continue to use the grandfathered service arrangement and may make changes to the arrangement as long as service continues to be provided at the same Customer Premises. However, no additions to the arrangement or transfers/moves of service to new locations are permitted. Type 3 Customers may continue to use the grandfathered service arrangement and may move or transfer the service as required. However, no additions or changes to the service configuration is permitted. Type 4 Customers may continue to use the grandfathered service arrangement. However, no additions, changes, moves or transfer of service are permitted. Type 5 Used to indicate that the terms and conditions associated with grandfathering the service are service-specific and stated in the pricing and service guide pages grandfathering the offering.

Grandfathered services are furnished subject to all rules and regulations of this pricing and service guide the same as would be applicable if the service offering were not grandfathered. In addition, the Company reserves the right to revise rates, terms, conditions, rules and regulations associated with grandfathered services from time to time.

#### SECTION 12 - GRANDFATHERED SERVICES

### 12.2 Type 1 – Grandfathered Services

# 12.2.1 AT&T® Worldwide Calling (Grandfathered 7/2/2007)

- (A) This plan is a direct dialed outbound international long distance service offered to residential Customers from switched access lines. Service is offered on a flat rate basis, twenty-four hours per day, seven days per week. Per minute charges apply and will be based on the country called and the duration of the call. Customers must dial 1 plus the country code or area code, if applicable, and the terminating number to complete a call utilizing this service.
- (B) Customers must subscribe to an access line from the Company's Affiliated ILEC.
- (C) All calls are billed in one (1) minute increments after an initial period, for billing purposes, of one (1) minute. This plan is only offered in conjunction with a Company Domestic Service. Customers who no longer meet the eligibility requirements identified above will be transferred to another calling plan.
- (D) International Mobile Termination Charge

See Section 6.4.1 of this pricing and service guide for the applicable International Mobile Termination Charge.

(E) Rates and Charges

The MRC is \$6.99, and the per-minute usage rates are as follows:

		Each
	Initial	Additional
Country	Minute	Minute
Afghanistan	\$1.68	\$1.68
Albania	\$1.28	\$1.28
Algeria	\$0.67	\$0.67
Andorra	\$0.57	\$0.57
Angola	\$1.82	\$1.82
Anguilla	\$0.70	\$0.70
Antarctica (Casey Base)	\$0.15	\$0.15
Antarctica (Scott Base)	\$0.15	\$0.15
Antigua	\$0.58	\$0.58
Argentina	\$0.23	\$0.23
Armenia	\$0.76	\$0.76
Aruba	\$0.55	\$0.55
Ascension Island	\$1.40	\$1.40
Australia	\$0.15	\$0.15
Austria	\$0.15	\$0.15
Azerbaijan	\$1.08	\$1.08
Bahamas	\$0.34	\$0.34
Bahrain	\$0.85	\$0.85
Bangladesh	\$0.68	\$0.68
Barbados	\$0.65	\$0.65
Belarus	\$0.47	\$0.47
Belgium	\$0.15	\$0.15
Belize	\$0.67	\$0.67
Benin	\$0.80	\$0.80
Bermuda	\$0.26	\$0.26
Bhutan	\$2.28	\$2.28
Bolivia	\$0.45	\$0.45

Country	Initial Minute	Each Additional Minute
Bosnia and Herzegovina	\$0.48	\$0.48
Botswana	\$1.03	\$1.03
Brazil	\$0.22	\$0.22
British Virgin Islands	\$0.56	\$0.56
Brunei	\$1.06	\$1.06
Bulgaria	\$0.47	\$0.47
Burkina Faso	\$1.28	\$1.28
Burundi	\$2.17	\$2.17
Cambodia	\$2.88	\$2.88
Cameroon	\$0.67	\$0.67
Canada	\$0.09	\$0.09
Cape Verde Island	\$1.00	\$1.00
Cayman Islands	\$0.35	\$0.35
Central African Republic	\$1.88	\$1.88
Chad	\$3.06	\$3.06
Chile	\$0.26	\$0.26
China	\$0.17	\$0.17
Christmas Island	\$0.15	\$0.15
Cocos Island	\$0.15	\$0.15
Colombia	\$0.23	\$0.23
Comoros Islands	\$3.06	\$3.06
Congo	\$0.97	\$0.97
Cook Islands	\$2.04	\$2.04
Costa Rica	\$0.35	\$0.35
Croatia	\$0.44	\$0.44
Cuba	\$1.11	\$1.11
Cyprus	\$0.70	\$0.70

Original Page 2

# BUSINESS AND RESIDENTIAL SERVICES Effective: September 1, 2019

Original Page 3

# PRICING AND SERVICE GUIDE SECTION 12 – GRANDFATHERED SERVICES

# 12.2 Type 1 - Grandfathered Services (continued)

# 12.2.1 AT&T® Worldwide Calling (continued)

(E) Rates and Charges (continued)

The per-minute usage rates are as follows: (continued)

			_	
Country	Initial Minute	Each Additional Minute		Country
Czech Republic	\$0.36	\$0.36		Ireland
Denmark	\$0.15	\$0.15	-	Israel
Diego Garcia	\$1.85	\$1.85	1	Italy
Diibouti	\$1.19	\$1.19	-	Ivory Coast
Dominica	\$0.51	\$0.51	1	Jamaica
Dominican Republic	\$0.24	\$0.24	1	Japan
East Timor	\$2.37	\$2.37	1	Jordan
Easter Island	See Chile	See Chile	1	Kazakhstan
Ecuador	\$0.41	\$0.41	1	Kenya
Egypt	\$0.51	\$0.51	1	Kiribati
El Salvador	\$0.32	\$0.32	1	Korea (North)
Equatorial Guinea	\$2.93	\$2.93		Korea (South)
Eritrea	\$1.13	\$1.13		Kosovo
Estonia	\$0.86	\$0.86		ıwait
Ethiopia	\$1.23	\$1.23		yzstan
Faeroe Island	\$0.64	\$0.64	Laos	, cuii
alkland Island	\$1.47	\$1.47	Latvia	
iji Island	\$1.28	\$1.28	Lebanon	
Finland	\$0.15	\$0.15	Lesotho	
France	\$0.15	\$0.15	Liberia	
French Antilles	\$0.80	\$0.80	Libya	
French Guiana	\$0.81	\$0.81	Liechtenstein	
French Polynesia	\$1.65	\$1.65	Lithuania	
Gabon	\$1.43	\$1.43	Luxembourg	
Gambia	\$0.76	\$0.76	Macao	
Georgia	\$0.78	\$0.78	Macedonia	
Germany	\$0.15	\$0.15	Madagascar	
Ghana	\$0.58	\$0.58	Malawi	
Gibraltar	\$1.28	\$1.28	Malaysia	
Greece	\$0.15	\$0.15	Maldives	
Greenland	\$0.72	\$0.72	Mali Republic	
Grenada	\$0.58	\$0.58	Malta Republic	
Guadeloupe	\$0.77	\$0.77	Marshall Island	
Guatemala	\$0.32	\$0.32	Mauritania	
Guinea	\$1.00	\$1.00	Mauritius	
Guinea-Bissau	\$2.04	\$2.04	Mayotte Island	(se
Guyana	\$1.07	\$1.07	Mexico 1	(50
Haiti	\$0.61	\$0.61	Mexico 2	
Honduras	\$0.53	\$0.53	Micronesia	
Hong Kong	\$0.15	\$0.15	Moldova	
Hungary	\$0.36	\$0.36	Monaco	
Iceland	\$0.61	\$0.61	Mongolia	
India	\$0.41	\$0.41	Montenegro	
Indonesia	\$0.45	\$0.45	Montserrat	
Inmarsat (SNAC)	\$5.00	\$5.00	Morocco	
Iran	\$0.75	\$0.75	Mozambique	
11 411	\$0.73	\$0.73	Myanmar	

Original Page 4

Effective: September 1, 2019

# SECTION 12 – GRANDFATHERED SERVICES

# 12.2 Type 1 - Grandfathered Services (continued)

- 12.2.1 AT&T® Worldwide Calling (continued)
  - (E) Rates and Charges (continued)

The per-minute usage rates are as follows: (continued)

	Each			Each
Initial	Additional		Initial	Additional
				Minute
				\$1.28
				\$1.41
				\$0.61
				\$0.56
				\$0.53
				\$0.59
\$1.78	\$1.78	St. Vincent/Grenadines	\$0.70	\$0.70
\$0.15	\$0.15	South Sudan	\$1.20	\$1.20
		Sudan	\$1.20	\$1.20
\$1.13	\$1.13	Suriname	\$1.53	\$1.53
\$0.70	\$0.70	Swaziland	\$0.94	\$0.94
\$2.27	\$2.27	Sweden	\$0.15	\$0.15
\$0.15	\$0.15	Switzerland	\$0.15	\$0.15
\$0.15	\$0.15	Syrian Arab Republic	\$1.20	\$1.20
\$1.31	\$1.31	Taiwan	\$0.13	\$0.13
\$0.41	\$0.41	Tajikistan	\$0.32	\$0.32
\$1.58	\$1.58	Tanzania	\$0.67	\$0.67
\$0.38		Thailand	\$0.34	\$0.34
\$0.46		Togo	\$1.35	\$1.35
		Tokelau		\$1.03
\$0.52		Tonga Island	\$1.93	\$1.93
\$0.30	\$0.30	Trinidad and Tobago	\$0.44	\$0.44
\$0.23	\$0.23	Tunisia	\$0.67	\$0.67
\$0.23	\$0.23	Turkey	\$0.48	\$0.48
			\$0.43	\$0.43
				\$0.64
\$1.24	\$1.24	Tuvalu	\$2.26	\$2.26
\$0.44	\$0.44	Uganda	\$0.67	\$0.67
\$0.23	\$0.23	Ukraine	\$0.33	\$0.33
\$1.45	\$1.45	United Arab Emirates	\$0.51	\$0.51
\$1.09	\$1.09	United Kingdom	\$0.12	\$0.12
\$1.96	\$1.96	Uruguay	\$0.52	\$0.52
\$0.63	\$0.63	Uzbekistan	\$0.43	\$0.43
\$0.94	\$0.94	Vanuatu	\$2.94	\$2.94
\$0.56	\$0.56	Vatican City	\$0.15	\$0.15
\$1.85	\$1.85	Venezuela	\$0.28	\$0.28
\$0.76		Vietnam	\$0.68	\$0.68
\$0.21	\$0.21	Wallis Futuna	\$2.89	\$2.89
\$0.35	\$0.35	Western Samoa	\$1.28	\$1.28
\$0.69	\$0.69	Yemen Arab Republic		\$1.20
				\$0.80
				\$0.80
		Zimbabwe		\$0.67
			,	,
	Minute \$0.98 \$1.47 \$1.58 \$0.15 \$0.53 \$4.80 \$1.78 \$0.15 \$0.52 \$1.13 \$0.70 \$2.27 \$0.15 \$0.15 \$1.31 \$0.41 \$1.58 \$0.38 \$0.46 \$1.30 \$0.52 \$0.30 \$0.23 \$0.23 \$0.15 \$1.17 \$1.24 \$0.44 \$0.23 \$1.17 \$1.24	Minute         Minute           \$0.98         \$0.98           \$1.47         \$1.47           \$1.58         \$1.58           \$0.15         \$0.15           \$0.53         \$0.53           \$4.80         \$4.80           \$1.78         \$1.78           \$0.15         \$0.15           \$0.52         \$0.52           \$1.13         \$1.13           \$0.70         \$0.70           \$2.27         \$2.27           \$0.15         \$0.15           \$0.15         \$0.15           \$0.15         \$0.15           \$0.15         \$0.15           \$0.15         \$0.15           \$0.15         \$0.15           \$0.15         \$0.15           \$0.15         \$0.15           \$1.31         \$1.31           \$0.41         \$0.41           \$1.58         \$0.38           \$0.38         \$0.38           \$0.38         \$0.38           \$0.46         \$1.30           \$0.52         \$0.52           \$0.30         \$0.23           \$0.23         \$0.23           \$0.23         \$0.23           \$0.15<	Initial Minute	Initial   Minute   Support   Suppo

#### SECTION 12 - GRANDFATHERED SERVICES

### 12.2 Type 1 - Grandfathered Services (continued)

#### 12.2.2 International Savings Value Plan (Grandfathered 7/2/2007)

(A) This plan is a direct dialed outbound international long distance service offered to residential Customers from switched access lines. Service is offered on a flat rate basis, twenty-four hours per day, seven days per week. Per minute charges apply and will be based on the country called and the duration of the call. Customers must dial 1 plus the country code or area code, if applicable, and the terminating number to complete a call utilizing this service.

Original Page 5

- (B) Customers of this plan must also subscribe to a local calling plan/service as defined and offered by the Company's Affiliated ILEC must also subscribe to at least two verticals features as defined and offered by the Company's Affiliated ILEC.
- (C) All calls are billed in one (1) minute increments after an initial period, for billing purposes, of one (1) minute. This plan is only offered in conjunction with BellSouth® interstate long distance service. Customers who no longer meet the eligibility requirements identified above will be transferred to Residential MTS. The Customer will be billed a minimum monthly commitment charge if actual usage is below the commitment level in any month. Usage over the minimum monthly commitment charge may not be carried over to future months to satisfy the commitment.
- (D) International Mobile Termination Charge
  - See Section 6.4.1 of this pricing and service guide for the applicable International Mobile Termination Charge.
- (E) The Minimum Monthly Commitment charge is \$14.95, and the per-minute usage rates are as follows:

Country	Initial Minute	Each Additional Minute	Country	Initial Minute	Each Additional Minute
Afghanistan	\$1.73	\$1.73	Belize	\$0.49	\$0.49
Albania	\$1.04	\$1.04	Benin	\$0.65	\$0.65
Algeria	\$0.49	\$0.49	Bermuda	\$0.19	\$0.19
Andorra	\$0.46	\$0.46	Bhutan	\$1.84	\$1.84
Angola	\$1.48	\$1.48	Bolivia	\$0.33	\$0.33
Anguilla	\$0.57	\$0.57	Bosnia	\$0.35	\$0.35
Antarctica (Casey Base)	\$0.09	\$0.09	Botswana	\$0.84	\$0.84
Antarctica (Scott Base)	\$0.08	\$0.08	Brazil	\$0.15	\$0.15
Antigua	\$0.47	\$0.47	British Virgin Islands	\$0.45	\$0.45
Argentina	\$0.17	\$0.17	Brunei	\$0.86	\$0.86
Armenia	\$0.59	\$0.59	Bulgaria	\$0.30	\$0.30
Aruba	\$0.44	\$0.44	Burkina Faso	\$1.04	\$1.04
Ascension Island	\$1.14	\$1.14	Burundi	\$1.78	\$1.78
Australia	\$0.09	\$0.09	Cambodia	\$2.32	\$2.32
Austria	\$0.09	\$0.09	Cameroon	\$0.49	\$0.49
Azerbaijan	\$0.87	\$0.87	Canada	\$0.05	\$0.05
Bahamas	\$0.25	\$0.25	Cape Verde Island	\$0.82	\$0.82
Bahrain	\$0.65	\$0.65	Cayman Islands	\$0.25	\$0.25
Bangladesh	\$0.50	\$0.50	Central African Republic	\$1.51	\$1.51
Barbados	\$0.51	\$0.51	Chad	\$2.46	\$2.46
Belarus	\$0.35	\$0.35	Chile	\$0.17	\$0.17
Belgium	\$0.09	\$0.09	China	\$0.11	\$0.11

# BUSINESS AND RESIDENTIAL SERVICES Effective: September 1, 2019

Original Page 6

# **PRICING AND SERVICE GUIDE**SECTION 12 – GRANDFATHERED SERVICES

# 12.2 Type 1 - Grandfathered Services (continued)

# 12.2.2 International Savings Value Plan (continued)

# (E) (continued)

Country         Minute         Country         Minute         Minute         Minute           Christmas Island         \$0.09         \$0.09         \$0.09         \$0.49         \$0.49           Cocos Island         \$0.09         \$0.09         \$0.09         \$0.41         \$0.41           Combros Islands         \$2.29         \$2.29         \$0.75         \$0.75         \$0.27           Congo         \$0.75         \$0.27         \$0.27         \$0.27           Cost Rica         \$0.27         \$0.27         \$0.27           Croatia         \$0.33         \$0.33         \$0.32           Cotha         \$0.89         \$0.89         \$0.89           Cyprus         \$0.56         \$0.56         \$0.56           Demmark         \$0.09         \$0.09         \$0.09           Dijbouti         \$0.97         \$0.97         \$0.97		Initial	Each Additional		Initial	Each Additional
Haiti   S0.49   S0.49   Cocos Island   S0.09   S0.09   Honduras   S0.41   S0.42   S0.27   S0.27   S0.27   S0.27   S0.27   S0.27   S0.27   S0.27   S0.27   S0.28   S0.29   S0.59   S0	Country		Minute	Country		
Cocos Island         \$0.09         \$0.09         Honduras         \$0.41         \$0.41           Colombia         \$0.16         \$0.16         \$0.09         \$0.09         \$0.09         \$0.09           Comoros Islands         \$1.65         \$1.65         \$1.65         \$1.65         \$0.27         \$0.21         \$0.39         \$0.49         \$0.29         \$0.29         \$0.29         \$0.29         \$0.29         \$0.29         \$0.29         \$0.29         \$0.29         \$0.29 </td <td></td> <td></td> <td></td> <td></td> <td></td> <td></td>						
Colombia   \$0.16   \$0.16   \$0.16   Hong Kong   \$0.09						
Comoros Islands         \$2.29         \$2.29         Hungary         \$0.27         \$0.27           Congo         \$0.75         \$0.75         leeland         \$0.49         \$0.49           Cook Islands         \$1.65         \$1.65         lindia         \$0.28         \$0.28           Costa Rica         \$0.27         \$0.27         lindia         \$0.29         \$0.29           Cotatia         \$0.33         \$0.33         lindonesia         \$0.29         \$0.29           Cyprus         \$0.56         \$0.56         liran         \$0.59         \$0.29           Cyprus         \$0.56         \$0.56         liran         \$0.59         \$0.59           Cyprus         \$0.50         \$0.50         liran         \$0.59         \$0.59           Denmark         \$0.09         \$0.09         lital         \$0.78         \$0.78           Dijbouti         \$0.97         \$0.97         ltaly         \$0.09         \$0.09           Dominican Republic         \$0.17         \$0.17         ltaly         \$0.69         \$0.69           Easter Island         \$0.17         \$0.17         kazakhstan         \$0.16         \$0.59           El Salvador         \$0.33         \$0.31						
Congo         \$0.75         \$0.75         \$0.75         India         \$0.49         \$0.49         \$0.49           Cook Islands         \$1.65         \$1.65         India         \$0.28         \$0.28         \$0.29         \$0.59         \$0.59         \$0.59         \$0.59         \$0.59         \$0.59         \$0.59         \$0.59         \$0.59         \$0.09         \$0.09         \$0.09         \$0.09         \$0.09         \$0.09         \$0.09         \$0.09         \$0.09         \$0.09         \$0.09         \$0.09         \$0.09         \$0.09         \$0.09						
Cook Islands         \$1.65         \$1.65         India         \$0.28         \$0.28         \$0.29           Costa Rica         \$0.27         \$0.27         Indonesia         \$0.29         \$0.29           Crotatia         \$0.33         \$0.33         \$0.33         \$0.33         \$0.32         \$9.32						
Costa Rica         \$0.27         \$0.23         \$0.23         \$0.29         \$0.29         \$0.29           Croatia         \$0.33         \$0.33         \$0.33         \$0.35         \$0.35         \$0.89         \$9.32         \$9.32         \$9.32         \$9.32         \$9.32         \$9.32         \$9.32         \$9.32         \$9.32         \$9.32         \$9.32         \$9.32         \$9.32         \$9.32         \$9.32         \$9.32         \$9.32         \$9.39         \$9.59         \$9.59         \$9.59         \$9.59         \$9.59         \$9.59         \$9.59         \$9.59         \$9.59         \$9.59         \$9.59         \$9.59         \$9.59         \$9.59         \$9.59         \$9.59         \$9.09						
Croatia						\$0.29
Cuba         \$0.89         \$0.89           Cyprus         \$0.56         \$0.56           Cyprus         \$0.56         \$0.56           Czech Republic         \$0.22         \$0.22           Denmark         \$0.09         \$0.09           Diego Garcia         \$1.50         \$1.50           Diego Garcia         \$1.50         \$1.50           Diminican         \$0.39         \$0.39           Dominican Republic         \$0.17         \$0.17           Bominican Republic         \$0.17         \$0.17           East Timor         \$1.94         \$1.94           East Timor         \$1.94         \$1.94           Easter Island         \$0.17         \$0.17           Eaudor         \$0.31         \$0.31           Eaypt         \$0.35         \$0.35           \$0.33         \$0.31         \$0.31           Elyaloria Guinea         \$2.19         \$2.19           Ejypt         \$0.67         \$0.67           Eaver Island         \$0.23         \$0.23           Korea (North)         \$1.15         \$1.15           Ejypt Sudaria         \$0.68         \$0.68           Korea (North)         \$0.9						
Cyprus         \$0.56         \$0.56           Czech Republic         \$0.22         \$0.22           Denmark         \$0.09         \$0.09           Bonamark         \$0.09         \$0.09           Bound         \$1.50         \$1.50           Diego Garcia         \$1.50         \$1.50           Dominica         \$0.39         \$0.39           Dominican Republic         \$0.17         \$0.17           Bominican Republic         \$0.17         \$0.17           Bast Timor         \$1.94         \$1.94           Baster Island         \$0.17         \$0.17           So.13         \$0.31         \$0.31           Egypt         \$0.35         \$0.35           El Salvador         \$0.23         \$0.23           El Salvador         \$0.23         \$0.23           Equatorial Guinea         \$2.19         \$2.19           Estonia         \$0.68         \$0.68           Estonia         \$0.68         \$0.68           Estonia         \$0.68         \$0.68           Bastoria         \$1.01         \$1.01           Ericea Island         \$0.52         \$0.52           Estonia         \$0.68         \$0.68						
Czech Republic         \$0.22         \$0.22         Ireland         \$0.09         \$0.09           Demmark         \$0.09         \$0.09         \$0.09         \$0.09         \$0.09           Diego Garcia         \$1.50         \$1.50         \$1.50         \$0.97         \$0.97           Dijbouti         \$0.97         \$0.97         \$0.97         \$0.99         \$0.09           Dominican Republic         \$0.17         \$0.17         \$0.17         \$0.17         \$0.17           Bominican Republic         \$0.17         \$0.17         \$0.17         \$0.17         \$0.17         \$0.17         \$0.17         \$0.17         \$0.17         \$0.17         \$0.17         \$0.16         \$0.59         \$0.99         \$0.09         \$0.09						
Denmark         \$0.09         <		\$0.22				
Diego Garcia         \$1.50         \$1.50         Italy         \$0.09         \$0.09           Dijbouti         \$0.97         \$0.97         \$0.97         \$0.69         \$0.69         \$0.69           Dominica         \$0.39         \$0.39         \$0.39         Jordan         \$0.36         \$0.36         \$0.36         \$0.36         \$0.36         \$0.39         \$0.09	Denmark					
Djibouti						
Dominica         \$0.39         \$0.39         \$0.39         Jamaica         \$0.36         \$0.36           Dominican Republic         \$0.17         \$0.17         \$0.17         \$0.09         \$0.09         \$0.09           East Timor         \$1.94         \$1.94         \$1.94         \$0.59         \$0.05           Easter Island         \$0.17         \$0.17         \$0.17         \$0.16         \$0.50           Ecuador         \$0.31         \$0.31         \$0.31         \$0.67         \$0.67           Eyypt         \$0.35         \$0.35         \$0.35         \$0.35         \$0.35         \$0.67         \$0.67           Eyypt         \$0.35         \$0.35         \$0.35         \$0.35         \$0.65         \$1.16         \$1.16         \$1.15         \$1.15         \$1.15         \$1.15         \$1.15         \$1.16         \$1.15         \$1.15         \$1.15         \$0.47         \$0						
Dominican Republic         \$0.17         \$0.17         \$0.194         \$1.94         \$1.94         \$0.19         \$0.59         \$0.67 <td></td> <td></td> <td></td> <td></td> <td></td> <td></td>						
East Timor         \$1.94         \$1.94         Jordan         \$0.59         \$0.59           Easter Island         \$0.17         \$0.17         \$0.17         \$0.16         \$0.16           Ecuador         \$0.31         \$0.31         \$0.31         \$0.67         \$0.67           Egypt         \$0.35         \$0.35         Kiribati         \$1.16         \$1.16           El Salvador         \$0.23         \$0.23         Korea (North)         \$1.15         \$1.15           Equatorial Guinea         \$2.19         \$2.19         Korea (North)         \$1.15         \$1.15           Equatorial Guinea         \$2.19         \$2.19         Korea (South)         \$0.09         \$0.09           Estirea         \$0.89         \$0.89         Kosovo         \$0.09         \$0.09           Estonia         \$0.68         \$0.68         Kuwait         \$0.47         \$0.47           Ethiopia         \$1.01         \$1.01         Kyrgyzstan         \$0.31         \$0.31           Falkland Island         \$1.21         \$1.21         Laos         \$2.43         \$2.43           Fiji Island         \$1.03         \$1.03         Laos         \$0.25         \$0.35           Fiji Island         \$0.09 <td></td> <td></td> <td></td> <td></td> <td></td> <td></td>						
Easter Island         \$0.17         \$0.17           Ecuador         \$0.31         \$0.31           \$0.35         \$0.35         \$0.35           El Salvador         \$0.23         \$0.23           El Salvador         \$0.23         \$0.23           Equatorial Guinea         \$2.19         \$2.19           Eritrea         \$0.89         \$0.89           Estonia         \$0.68         \$0.68           Estonia         \$0.68         \$0.68           Ethiopia         \$1.01         \$1.01           Faeroe Island         \$0.52         \$0.52           Falkland Island         \$1.21         \$1.21           Fiji Island         \$1.03         \$1.03           Finland         \$0.09         \$0.09           French Antilles         \$0.65         \$0.65           French Guiana         \$0.66         \$0.66           French Polynesia         \$1.24         \$1.24           Gabon         \$1.09         \$1.09           Georgia         \$0.79         \$0.79           Macao         \$0.45         \$0.45           Mahaa         \$0.45         \$0.45           Malawi         \$0.49         \$0.49 <tr< td=""><td>East Timor</td><td></td><td></td><td></td><td></td><td></td></tr<>	East Timor					
Ecuador         \$0.31         \$0.35         \$0.35         Kenya         \$0.67         \$0.67           Egypt         \$0.35         \$0.35         \$0.23         \$0.23         Kiribati         \$1.16         \$1.15 <t< td=""><td>Easter Island</td><td></td><td></td><td>Kazakhstan</td><td></td><td></td></t<>	Easter Island			Kazakhstan		
Egypt         \$0.35         \$0.35         Kiribati         \$1.16         \$1.16           El Salvador         \$0.23         \$0.23         \$0.23         Korea (North)         \$1.15         \$1.15           Equatorial Guinea         \$2.19         \$2.19         Korea (South)         \$0.09         \$0.09           Estonia         \$0.68         \$0.68         Kosovo         \$0.09         \$0.09           Estonia         \$0.68         \$0.68         Kuwait         \$0.47         \$0.47           Ethiopia         \$1.01         \$1.01         Kyrgyzstan         \$0.31         \$0.31           Falkland Island         \$1.21         \$1.21         Latvia         \$0.35         \$0.35           Falkland Island         \$1.03         \$1.03         Lebanon         \$0.35         \$0.35           Fiji Island         \$0.09         \$0.09         Lebanon         \$0.55         \$0.55           Finland         \$0.09         \$0.09         Lebanon         \$0.55         \$0.55           French Antilles         \$0.65         \$0.65         Lebanon         \$0.59         \$0.99           French Guiana         \$0.66         \$0.66         Libya         \$0.59         \$0.97           French Pol	Ecuador			Kenya		
El Salvador			\$0.35			
Equatorial Guinea         \$2.19         \$2.19         Korea (South)         \$0.09         \$0.09           Eritrea         \$0.89         \$0.89         Kosovo         \$0.09         \$0.09           Estonia         \$0.68         \$0.68         \$0.68         Kusovo         \$0.09         \$0.09           Ethiopia         \$1.01         \$1.01         Kusait         \$0.47         \$0.47           Ethiopia         \$1.01         \$1.01         Kusait         \$0.47         \$0.47           Ethiopia         \$1.01         \$1.01         Kusait         \$0.47         \$0.47           Falkland Island         \$1.21         \$1.21         Laos         \$2.43         \$2.43           Fiji Island         \$1.03         \$1.03         Laos         \$2.43         \$2.43           France         \$0.09         \$0.09         Lebanon         \$0.55         \$0.55           France         \$0.09         \$0.09         Lebanon         \$0.55         \$0.55           French Antilles         \$0.65         \$0.65         Lesotho         \$0.84         \$0.84           French Guiana         \$0.66         \$0.65         Liberia         \$0.59         \$0.97         \$0.97           Gambia	El Salvador			Korea (North)		
Eritrea         \$0.89         \$0.89         Kosovo         \$0.09         \$0.09           Estonia         \$0.68         \$0.68         \$0.68         Kuwait         \$0.47         \$0.47           Ethiopia         \$1.01         \$1.01         Kyrgyzstan         \$0.31         \$0.31           Faeroe Island         \$0.52         \$0.52         Laos         \$2.43         \$2.43           Falkland Island         \$1.21         \$1.21         Latvia         \$0.35         \$0.35           Fijji Island         \$1.03         \$1.03         Lebanon         \$0.55         \$0.55           Fiji Island         \$0.09         \$0.09         Lebanon         \$0.55         \$0.55           Finland         \$0.09         \$0.09         Lesotho         \$0.84         \$0.84           France         \$0.09         \$0.09         Liberia         \$0.59         \$0.59           French Antilles         \$0.65         \$0.65         Libya         \$0.97         \$0.97           French Polynesia         \$1.24         \$1.24         Litechtenstein         \$0.09         \$0.09           French Polynesia         \$1.09         \$0.09         Macao         \$0.77         \$0.77           Georgia						\$0.09
Estonia         \$0.68         \$0.68         \$0.68         Kuwait         \$0.47         \$0.47           Ethiopia         \$1.01         \$1.01         \$1.01         Kyrgyzstan         \$0.31         \$0.31           Faeroe Island         \$0.52         \$0.52         \$0.52         \$0.52         \$0.52           Falkland Island         \$1.03         \$1.21         \$1.21         \$0.35         \$0.35           Fiji Island         \$1.03         \$1.03         \$1.03         \$0.55         \$0.35           Fiji Island         \$0.09         \$0.09         \$0.09         \$0.55         \$0.55         \$0.55           Finland         \$0.09         \$0.09         \$0.09         \$0.84         \$0.84         \$0.84           France         \$0.09         \$0.09         \$0.09         \$0.59         \$0.59         \$0.59         \$0.59           French Antilles         \$0.65	Eritrea			Kosovo		
Ethiopia         \$1.01         \$1.01         \$1.01         Kyrgyzstan         \$0.31         \$0.31           Faeroe Island         \$0.52         \$0.52         \$0.52         \$0.52         \$0.52         \$0.31         \$0.32         \$0.35         \$0.35         \$0.35         \$0.35         \$0.35         \$0.35         \$0.55         \$0.55         \$0.55         \$0.55         \$0.55         \$0.59         \$0.59         \$0.09         \$0.09         \$0.09         \$0.09	Estonia			Kuwait	\$0.47	\$0.47
Falkland Island         \$1.21         \$1.21         Latvia         \$0.35         \$0.35           Fiji Island         \$1.03         \$1.03         Lebanon         \$0.55         \$0.55           Finland         \$0.09         \$0.09         Leotho         \$0.84         \$0.84           France         \$0.09         \$0.09         Liberia         \$0.59         \$0.59           French Antilles         \$0.65         \$0.65         Liberia         \$0.59         \$0.59           French Guiana         \$0.66         \$0.66         Libya         \$0.97         \$0.97           French Polynesia         \$1.24         \$1.24         Lithuania         \$0.09         \$0.09           Gabon         \$1.09         \$1.09         Lithuania         \$0.35         \$0.35           Gambia         \$0.59         \$0.59         Macao         \$0.77         \$0.77           Georgia         \$0.79         \$0.79         Macao         \$0.77         \$0.77           Germany         \$0.09         \$0.09         Madagascar         \$2.18         \$2.18           Ghana         \$0.45         \$0.45         Malawi         \$0.49         \$0.49           Greece         \$0.09         \$0.09			\$1.01	Kyrgyzstan	\$0.31	\$0.31
Fiji Island         \$1.03         \$1.03         Lebanon         \$0.55         \$0.55           Finland         \$0.09         \$0.09         Lesotho         \$0.84         \$0.84           France         \$0.09         \$0.09         Liberia         \$0.59         \$0.59           French Antilles         \$0.65         \$0.65         Libya         \$0.97         \$0.97           French Guiana         \$0.66         \$0.66         Libya         \$0.97         \$0.97           French Polynesia         \$1.24         \$1.24         Lithuania         \$0.35         \$0.35           Gabon         \$1.09         \$1.09         Luxembourg         \$0.09         \$0.09           Gambia         \$0.59         \$0.59         Macao         \$0.77         \$0.77           Georgia         \$0.79         \$0.79         Macao         \$0.55         \$0.55           Germany         \$0.09         \$0.09         Madagascar         \$2.18         \$2.18           Ghana         \$0.45         \$0.45         Malawi         \$0.49         \$0.49           Greece         \$0.09         \$0.09         Maldives         \$1.45         \$1.45           Greenland         \$0.58         \$0.58 <t< td=""><td>Faeroe Island</td><td>\$0.52</td><td>\$0.52</td><td>Laos</td><td></td><td>\$2.43</td></t<>	Faeroe Island	\$0.52	\$0.52	Laos		\$2.43
Finland         \$0.09         \$0.09         \$0.09         Lesotho         \$0.84         \$0.84           France         \$0.09         \$0.09         Liberia         \$0.59         \$0.59           French Antilles         \$0.65         \$0.65         Libya         \$0.97         \$0.97           French Guiana         \$0.66         \$0.66         Libya         \$0.97         \$0.97           French Polynesia         \$1.24         \$1.24         Lithuania         \$0.09         \$0.09           Gabon         \$1.09         \$1.09         Lithuania         \$0.35         \$0.35           Gambia         \$0.59         \$0.59         Macao         \$0.09         \$0.09           Georgia         \$0.79         \$0.79         Macao         \$0.77         \$0.77           Germany         \$0.09         \$0.09         Madagascar         \$2.18         \$2.18           Ghana         \$0.45         \$0.45         Malawi         \$0.49         \$0.49           Gibraltar         \$1.04         \$1.04         Malawi         \$0.49         \$0.49           Greece         \$0.09         \$0.09         Malikee         \$1.45         \$1.45           Greenland         \$0.45         \$0.	Falkland Island	\$1.21	\$1.21	Latvia	\$0.35	\$0.35
France         \$0.09         \$0.09           French Antilles         \$0.65         \$0.65           French Guiana         \$0.66         \$0.66           French Polynesia         \$1.24         \$1.24           Gabon         \$1.09         \$1.09           Gambia         \$0.59         \$0.59           Georgia         \$0.79         \$0.79           Georgia         \$0.45         \$0.45           Ghana         \$0.45         \$0.45           Gibraltar         \$1.04         \$1.04           Greece         \$0.09         \$0.09           Greenland         \$0.58         \$0.58           Grenada         \$0.45         \$0.45           Guadeloupe         \$0.61         \$0.61           Guinea         \$0.82         \$0.82           Guinea-Bissau         \$1.65         \$1.65	Fiji Island	\$1.03	\$1.03	Lebanon	\$0.55	\$0.55
French Antilles         \$0.65         \$0.65           French Guiana         \$0.66         \$0.66           French Polynesia         \$1.24         \$1.24           Gabon         \$1.09         \$1.09           Gambia         \$0.59         \$0.59           Georgia         \$0.79         \$0.79           Germany         \$0.09         \$0.09           Ghana         \$0.45         \$0.45           Gibraltar         \$1.04         \$1.04           Greece         \$0.09         \$0.09           Greenland         \$0.58         \$0.58           Grenada         \$0.45         \$0.45           Greenland         \$0.58         \$0.58           Greenland         \$0.82         \$0.45           Guinea         \$0.82         \$0.82           Guinea-Bissau         \$1.65         \$1.65	Finland			Lesotho	\$0.84	\$0.84
French Guiana         \$0.66         \$0.66         \$0.66         \$0.66         \$0.09         \$0.09           French Polynesia         \$1.24         \$1.24         \$1.24         \$1.24         \$0.35         \$0.35           Gabon         \$1.09         \$1.09         \$1.09         \$0.09         \$0.09         \$0.09         \$0.09         \$0.09         \$0.09         \$0.09         \$0.09         \$0.09         \$0.07         \$0.77         \$0.77         \$0.77         \$0.77         \$0.77         \$0.77         \$0.77         \$0.77         \$0.77         \$0.77         \$0.77         \$0.55         \$0.49         \$0.49         \$0.49         \$0.49         \$0.17         \$0.17         \$0.17         \$0.17         \$0.17         \$0.17         \$0.17	France	\$0.09	\$0.09		\$0.59	\$0.59
French Polynesia         \$1.24         \$1.24         Lithuania         \$0.35         \$0.35           Gabon         \$1.09         \$1.09         Luxembourg         \$0.09         \$0.09           Gambia         \$0.59         \$0.59         Macao         \$0.77         \$0.77           Georgia         \$0.79         \$0.79         Macedonia         \$0.55         \$0.55           Germany         \$0.09         \$0.09         Madagascar         \$2.18         \$2.18           Ghana         \$0.45         \$0.45         Malawi         \$0.49         \$0.49           Gibraltar         \$1.04         \$1.04         Malawi         \$0.49         \$0.49           Greece         \$0.09         \$0.09         Maldives         \$1.45         \$1.45           Greenland         \$0.58         \$0.58         Mali Republic         \$1.06         \$1.06           Guadeloupe         \$0.61         \$0.61         Marshall Island         \$0.72         \$0.72           Guinea         \$0.82         \$0.82         Mauritius         \$1.25         \$1.25           Guinea-Bissau         \$1.65         \$1.65         Mayotte Island (See Reunion Island)         -         -	French Antilles	\$0.65	\$0.65	Libya	\$0.97	
Gabon         \$1.09         \$1.09         Luxembourg         \$0.09         \$0.09           Gambia         \$0.59         \$0.59         Macao         \$0.77         \$0.77           Georgia         \$0.79         \$0.79         Macedonia         \$0.55         \$0.55           Germany         \$0.09         \$0.09         Madagascar         \$2.18         \$2.18           Ghana         \$0.45         \$0.45         Malawi         \$0.49         \$0.49           Gibraltar         \$1.04         \$1.04         Malawi         \$0.49         \$0.49           Greece         \$0.09         \$0.09         Maldives         \$1.45         \$1.45           Greenland         \$0.58         \$0.58         Mali Republic         \$1.06         \$1.06           Grenada         \$0.45         \$0.45         Marshall Island         \$0.72         \$0.72           Guadeloupe         \$0.61         \$0.61         Mauritania         \$1.00         \$1.00           Guinea         \$0.82         \$0.82         Mauritius         \$1.25         \$1.25           Guinea-Bissau         \$1.65         \$1.65         Mayotte Island (See Reunion Island)         -         -	French Guiana	\$0.66	\$0.66	Liechtenstein	\$0.09	\$0.09
Gambia         \$0.59         \$0.59           Georgia         \$0.79         \$0.79           Germany         \$0.09         \$0.09           Ghana         \$0.45         \$0.45           Gibraltar         \$1.04         \$1.04           Greece         \$0.09         \$0.09           Greenland         \$0.58         \$0.58           Grenada         \$0.45         \$0.45           Greenland         \$0.58         \$0.58           Grenada         \$0.45         \$0.45           Guadeloupe         \$0.61         \$0.61           Guatemala         \$0.23         \$0.23           Guinea         \$0.82         \$0.82           Guinea-Bissau         \$1.65         \$1.65	French Polynesia	\$1.24	\$1.24	Lithuania	\$0.35	\$0.35
Georgia         \$0.79         \$0.79         Macedonia         \$0.55         \$0.55           Germany         \$0.09         \$0.09         Madagascar         \$2.18         \$2.18           Ghana         \$0.45         \$0.45         Malawi         \$0.49         \$0.49           Gibraltar         \$1.04         \$1.04         Malawi         \$0.49         \$0.49           Greece         \$0.09         \$0.09         Maldives         \$1.45         \$1.45           Greenland         \$0.58         \$0.58         Mali Republic         \$1.06         \$1.06           Grenada         \$0.45         \$0.45         Malta Republic         \$0.72         \$0.72           Guadeloupe         \$0.61         \$0.61         Marshall Island         \$0.74         \$0.74           Guinea         \$0.82         \$0.82         Mauritius         \$1.25         \$1.25           Guinea-Bissau         \$1.65         \$1.65         Mayotte Island (See Reunion Island)         -         -	Gabon	\$1.09		Luxembourg		\$0.09
Germany         \$0.09         \$0.09           Ghana         \$0.45         \$0.45           Gibraltar         \$1.04         \$1.04           Greece         \$0.09         \$0.09           Greenland         \$0.58         \$0.58           Grenada         \$0.45         \$0.45           Guadeloupe         \$0.61         \$0.61           Guatemala         \$0.23         \$0.23           Guinea         \$0.82         \$0.82           Guinea-Bissau         \$1.65         \$1.65	Gambia	\$0.59		Macao	\$0.77	\$0.77
Ghana         \$0.45         \$0.45           Gibraltar         \$1.04         \$1.04           Greece         \$0.09         \$0.09           Greenland         \$0.58         \$0.58           Grenada         \$0.45         \$0.45           Guadeloupe         \$0.61         \$0.61           Guatemala         \$0.23         \$0.23           Guinea         \$0.82         \$0.82           Guinea-Bissau         \$1.65         \$1.65	Georgia					
Ghana         \$0.45         \$0.45           Gibraltar         \$1.04         \$1.04           Greece         \$0.09         \$0.09           Greenland         \$0.58         \$0.58           Grenada         \$0.45         \$0.45           Guadeloupe         \$0.61         \$0.61           Guatemala         \$0.23         \$0.23           Guinea         \$0.82         \$0.82           Guinea-Bissau         \$1.65         \$1.65	Germany	\$0.09	\$0.09	Madagascar		
Greece         \$0.09         \$0.09         Maldives         \$1.45         \$1.45           Greenland         \$0.58         \$0.58         Mali Republic         \$1.06         \$1.06           Grenada         \$0.45         \$0.45         Mali Republic         \$0.72         \$0.72           Guadeloupe         \$0.61         \$0.61         Marshall Island         \$0.74         \$0.74           Guatemala         \$0.23         \$0.23         Mauritania         \$1.00         \$1.00           Guinea         \$0.82         \$0.82         Mauritius         \$1.25         \$1.25           Guinea-Bissau         \$1.65         \$1.65         Mayotte Island (See Reunion Island)         -         -	Ghana	\$0.45		Malawi	\$0.49	\$0.49
Greece         \$0.09         \$0.09         Maldives         \$1.45         \$1.45           Greenland         \$0.58         \$0.58         Mali Republic         \$1.06         \$1.06           Grenada         \$0.45         \$0.45         Mali Republic         \$0.72         \$0.72           Guadeloupe         \$0.61         \$0.61         Marshall Island         \$0.74         \$0.74           Guatemala         \$0.23         \$0.23         Mauritania         \$1.00         \$1.00           Guinea         \$0.82         \$0.82         Mauritius         \$1.25         \$1.25           Guinea-Bissau         \$1.65         \$1.65         Mayotte Island (See Reunion Island)         -         -	Gibraltar		\$1.04	Malaysia		
Grenada         \$0.45         \$0.45         Malta Republic         \$0.72         \$0.72           Guadeloupe         \$0.61         \$0.61         Marshall Island         \$0.74         \$0.74           Guatemala         \$0.23         \$0.23         Mauritania         \$1.00         \$1.00           Guinea         \$0.82         \$0.82         Mauritius         \$1.25         \$1.25           Guinea-Bissau         \$1.65         \$1.65         Mayotte Island (See Reunion Island)         -         -	Greece	\$0.09	\$0.09		\$1.45	\$1.45
Guadeloupe         \$0.61         \$0.61         Marshall Island         \$0.74         \$0.74           Guatemala         \$0.23         \$0.23         Mauritania         \$1.00         \$1.00           Guinea         \$0.82         \$0.82         Mauritius         \$1.25         \$1.25           Guinea-Bissau         \$1.65         \$1.65         Mayotte Island (See Reunion Island)         -         -	Greenland					
Guatemala         \$0.23         \$0.23         Mauritania         \$1.00         \$1.00           Guinea         \$0.82         \$0.82         Mauritius         \$1.25         \$1.25           Guinea-Bissau         \$1.65         \$1.65         Mayotte Island (See Reunion Island)         -         -	Grenada					
Guinea         \$0.82         \$0.82         Mauritius         \$1.25         \$1.25           Guinea-Bissau         \$1.65         \$1.65         Mayotte Island (See Reunion Island)         -         -	Guadeloupe					
Guinea-Bissau \$1.65 \$1.65 Mayotte Island (See Reunion - Island)	Guatemala					
Island)	Guinea		\$0.82	Mauritius	\$1.25	\$1.25
	Guinea-Bissau	\$1.65	\$1.65		-	-
$\frac{3}{3}$	Guyana	\$0.84	\$0.84	Mexico 1	\$0.10	\$0.10

Original Page 7

Effective: September 1, 2019

# SECTION 12 – GRANDFATHERED SERVICES

# 12.2 Type 1 - Grandfathered Services (continued)

# 12.2.2 International Savings Value Plan (continued)

# (E) (continued)

	T 1	Each		T 1	Each
Country	Initial	Additional Minute	Country	Initial	Additional Minute
Mexico	<b>Minute</b> \$0.10	\$0.10	Slovenia	\$0.56	\$0.56
Micronesia	\$0.77	\$0.77	Solomon Island	\$1.20	\$1.20
Moldova	\$0.49	\$0.49	Somalia Democratic Rep	\$1.64	\$1.64
Monaco	\$0.09	\$0.09	South Africa	\$0.36	\$0.36
Mongolia	\$1.70	\$1.70	Spain	\$0.09	\$0.09
Montenegro	\$0.42	\$0.42	Sri Lanka	\$1.02	\$1.02
Montserrat	\$0.52	\$0.52	St. Helena	\$1.15	\$1.15
Morocco	\$0.57	\$0.57	St. Kitts/Nevis St. Lucia	\$0.49	\$0.49
Mozambique	\$1.28	\$1.28		\$0.41	\$0.41
Myanmar	\$2.17	\$2.17	St. Maarten	\$0.42	\$0.42
Namibia	\$0.80	\$0.80	St. Pierre Miquelon	\$0.48	\$0.48
Nauru	\$1.18	\$1.18	St. Vincent/Grenadines	\$0.55	\$0.55
Nepal	\$1.30	\$1.30	South Sudan	\$0.89	\$0.89
Netherlands	\$0.09	\$0.09	Sudan	\$0.89	\$0.89
Netherland Antilles	\$0.42	\$0.42	Suriname	\$1.25	\$1.25
Networks	\$3.95	\$3.95	Swaziland	\$0.76	\$0.76
New Caledonia	\$1.45	\$1.45	Sweden	\$0.09	\$0.09
New Zealand	\$0.08	\$0.08	Switzerland	\$0.09	\$0.09
Nicaragua	\$0.38	\$0.38	Syrian Arab Republic	\$0.89	\$0.89
Niger Republic	\$0.93	\$0.93	Taiwan	\$0.08	\$0.08
Nigeria	\$0.52	\$0.52	Tajikistan	\$0.23	\$0.23
Niue	\$1.84	\$1.84	Tanzania	\$0.49	\$0.49
Norfolk Island	\$0.09	\$0.09	Thailand	\$0.23	\$0.23
Norway	\$0.09	\$0.09	Togo	\$1.11	\$1.11
Oman	\$1.07	\$1.07	Tokelau	\$0.84	\$0.84
Pakistan	\$0.28	\$0.28	Tonga Island	\$1.57	\$1.57
Palau	\$1.30	\$1.30	Trinidad	\$0.34	\$0.34
Palestinian Authority	\$0.30	\$0.30	Tunisia	\$0.49	\$0.49
Panama	\$0.35	\$0.35	Turkey	\$0.39	\$0.39
Papua New Guinea	\$1.06	\$1.06	Turkmenistan	\$0.31	\$0.31
Paraguay	\$0.40	\$0.40	Turks and Caicos Islands	\$0.52	\$0.52
Peru	\$0.22	\$0.22	Tuvalu	\$1.85	\$1.85
Philippines	\$0.16	\$0.16	Uganda	\$0.49	\$0.49
Poland	\$0.16	\$0.16	Ukraine	\$0.22	\$0.22
Portugal	\$0.09	\$0.09	United Arab Emirates	\$0.39	\$0.39
Qatar	\$0.95	\$0.95	United Kingdom	\$0.08	\$0.08
Reunion Island (includes Mayotte Island)	\$1.02	\$1.02	Uruguay	\$0.40	\$0.40
Romania	\$0.33	\$0.33	Uzbekistan	\$0.31	\$0.31
Russia	\$0.16	\$0.16	Vanuatu	\$2.37	\$2.37
Rwanda, Republic of	\$1.19	\$1.19	Vatican City	\$0.09	\$0.09
San Marino	\$0.89	\$0.89	Venezuela	\$0.20	\$0.20
Sao Tome	\$1.59	\$1.59	Vietnam	\$0.49	\$0.49
Saudi Arabia	\$0.45	\$0.45	Wallis Futuna	\$2.38	\$2.38
Senegal	\$0.69	\$0.69	Western Samoa	\$1.02	\$1.02
Serbia	\$0.42	\$0.42	Yemen Arab Republic	\$0.98	\$0.98
Seychelles Island	\$1.51	\$1.51	Zaire	\$0.59	\$0.59
Sierra Leone	\$0.59	\$0.59	Zambia	\$0.59	\$0.59
Singapore	\$0.10	\$0.10	Zimbabwe	\$0.49	\$0.49
Slovakia	\$0.25	\$0.25			
	, , , , , , , , , , , , , , , , , , , ,				

#### SECTION 12 – GRANDFATHERED SERVICES

### 12.3 Type 2 – Grandfathered Services

### 12.3.1 Business Fixed Rate Plan (Grandfathered 9/4/2007)

(A) The Business Fixed Rate plan is a direct dialed outbound long distance service offered to business Customers from switched access lines. Service is offered on a flat rate basis, twenty-four hours per day, seven days per week. Customers must dial 1 plus the area code, if applicable, and the terminating number to complete a call utilizing this service.

Original Page 8

Effective: September 1, 2019

- (B) All calls are billed in six (6) second increments after an initial period, for billing purposes, of thirty (30) seconds. This service is not offered on an intraLATA only basis.
- (C) The MRC is \$46.00 per account, and the per-minute usage rate is \$0.609.

### 12.3.2 Business Fixed Rate Plus Plan (Grandfathered 1/3/2004)

- (A) The Business Fixed Rate Plus plan is a direct dialed outbound long distance service offered to business Customers from switched access lines. Service is offered on a flat rate basis, twenty-four hours per day, seven days per week. Customers must dial 1 plus the area code, if applicable, and the terminating number to complete a call utilizing this service.
- (B) All calls are billed in six (6) second increments after an initial period, for billing purposes, of thirty (30) seconds. This service is not offered on an intraLATA only basis.
- (C) The monthly recurring charge (MRC) is \$49.00, and the per-minute usage rate is \$0.454. The MRC and will be waived in instances where the Customer's monthly billing is \$50.00 or more of combined intrastate and interstate usage. Surcharges, taxes and other similar fees shall not be included in the calculation for the waiving of the MRC.

### 12.3.3 Business Fixed Rate Ultra® Plan (Grandfathered 1/3/2004)

- (A) The Business Fixed Rate Ultra® plan is a direct dialed outbound long distance service offered to business Customers from switched access lines. Service is offered on a flat rate basis, twenty-four hours per day, seven days per week. Customers must dial 1 plus the area code, if applicable, and the terminating number to complete a call utilizing this service.
- (B) All calls are billed in six (6) second increments after an initial period, for billing purposes, of thirty (30) seconds. This service is not offered on an intraLATA only basis.
- (C) The MRC will be waived in instances where the Customer's monthly billing is \$200.00 or more of combined intrastate and interstate usage. Surcharges, taxes and other similar fees shall not be included in the calculation for the waiving of the MRC.
- (D) The MRC is \$54.00, and the per-minute usage rate is \$0.447.

#### SECTION 12 – GRANDFATHERED SERVICES

- 12.3 Type 2 Grandfathered Services (continued)
  - 12.3.4 Business One Plan (Grandfathered 9/4/2007)
    - (A) The Business One Plan is a direct dialed domestic outbound long distance service offered to business Customers on switched access lines. Service is offered on a flat rate basis, twenty-four hours per day, seven days per week. Customers must dial 1 plus the area code, if applicable, and the terminating number to complete a call utilizing this service.

Original Page 9

- (B) In order to be eligible for this plan, Customers must presubscribe to the Company or the Company's Affiliated ILEC for intraLATA service and must also presubscribe to the Company for interLATA long distance service.
- (C) All calls are billed in six (6) second increments after an initial period, for billing purposes, of thirty (30) seconds. This service is only offered in conjunction with the corresponding international Business One Plan. This service is not offered on an intraLATA only basis.
- (D) The MRC is \$7.95 and the per minute usage rate is \$0.0690. The MRC will be waived in instances where the Customer's monthly billing is \$50.00 or more of combined domestic and international usage. Surcharges, taxes and other similar fees shall not be included in the calculation for the waiving of the MRC.
- (E) The following discount will be given off the total domestic direct dialed outbound and domestic toll free inbound monthly usage charges based upon the actual domestic direct dialed outbound and domestic toll free inbound monthly usage. Surcharges, taxes and other similar fees shall not be included in the calculation for determination of the qualifying monthly usage.

Monthly Usage	Discount %
\$0.00 to \$49.99	0.00 %
\$50.00 to \$199.99	2.89 %
\$200.00 and above	5.8%

#### SECTION 12 – GRANDFATHERED SERVICES

- 12.3 Type 2 Grandfathered Services (continued)
  - 12.3.5 Business Unlimited IntraLATA Plan (Grandfathered 9/4/2007)
    - (A) The Business Unlimited IntraLATA plan is a direct dial domestic outbound long distance voice service available to business Customers with one to twenty switched access lines, one BellSouth® Primary Rate ISDN (PRI) or one BellSouth® Integrated Solutions T-1 (T-1). Service is offered on a flat rate basis, twenty-four hours per day, seven days per week. Customers must dial 1 plus the area code, if applicable, and the terminating number to complete a call utilizing this service. The Business Unlimited IntraLATA plan provides unlimited interexchange (both interstate and intrastate) intraLATA toll calling which originates and terminates within the same intraLATA toll calling area for a monthly charge. Intrastate interLATA toll calls are provided at usage rates specified below. BellSouth® Primary Rate ISDN and BellSouth® Integrated Solutions T-1 are services provided by the Company's Affiliated ILEC.

Original Page 10

- (B) All calls are billed in six (6) second increments after an initial period, for billing purposes, of thirty (30) seconds. This service is not offered on an intraLATA only basis. This plan does not provide call detail information on the Customer's monthly bill for eligible unlimited intraLATA calling. Customers may choose to receive this call detail information for an additional monthly charge.
- (C) Toll free service is available with this plan for Customers who also subscribe to AT&T Long Distance Toll Free<sup>SM</sup> Service. The provisions and rates and charges of the AT&T Long Distance Toll Free<sup>SM</sup> Service will apply as described in Section 4.4 of this pricing and service guide except that usage rates are specified below.
- (D) In order to be eligible for this plan, the Company must be able to verify that the Customer meets the following eligibility requirements. If the Company determines that the Customer's use of service violates any of these restrictions the Company, at its sole option, may move the Customer to another plan or suspend, restrict or cancel the Customer's service without prior notice. Customers are eligible for this service if they meet the following requirements:
  - 1. Must have a business location with 20 business lines or fewer, or one PRI or one T-1 at a single location on a single billing account. Customers who exceed these line sizes are not eligible for this plan. All lines, PRI or T-1 at the location must subscribe to this plan;
  - .2 Must presubscribe to the Company for both intraLATA and interLATA long distance service on all lines covered by this plan; and,
  - .3 Must be a local service Customer of the Company's Affiliated ILEC on each line, PRI or T-1 at each location under this plan and subscribe to a local service contract offered by that entity for all lines covered by this plan and receive billing for this service from the Company's Affiliated ILEC.

#### SECTION 12 – GRANDFATHERED SERVICES

- 12.3 Type 2 Grandfathered Services (continued)
  - 12.3.5 Business Unlimited IntraLATA Plan (continued)
    - (E) Plan Limitations
      - .1 Customer lines, PRI or T-1 associated with educational institutions (colleges, universities, etc.) or other businesses that aggregate end user traffic are not eligible for this plan service.

Original Page 11

- .2 Unlimited intrastate intraLATA usage does not include multi-party conference calls (except those placed by using the Three-Way Calling features are included), calls to 900, 976, 700 numbers or other calls to access information services, directory assistance, operator services or international callings services and toll-free calling services.
- .3 This plan service is not available for resale.
- .4 Term discounts are not applicable to this service.
- .5 Service under this plan cannot be used with the following applications or services: call center environment or in connection with any similar such application, auto-dialers or any similar type or device, PBX or PBX-like equipment, data transmission equipment, Centrex service, foreign exchange service, public telephone services, broadcast facsimile services or the equivalents of any such services.
- .6 The Company reserves the right to exclude certain terminating telephone numbers from unlimited intraLATA calling under this plan. Calls to these numbers will be billed at the applicable MTS rates.
- .7 This plan is not available to Customers with an account that bills to another number or is the recipient of charges billed from another number.

### SECTION 12 - GRANDFATHERED SERVICES

# 12.3 Type 2 – Grandfathered Services (continued)

### 12.3.5 Business Unlimited IntraLATA Plan (continued)

# (F) Rates and Charges

### .1 MRC

Direct Dial Service

The following MRC applies based on the number of switched access lines or type of service subscribed to this plan.

Original Page 12

Effective: September 1, 2019

Number of Switched Access Lines	MRC
1 – 5 Lines	\$18.95
6 – 10 Lines	\$30.95
11 – 20 Lines, one PRI or one T-1	\$49.95

Toll-Free Service: See Section 4.4 for the Toll-free Inbound Service MRC.

### Call Detail

The following MRC applies to Customers choosing to receive call detail information based on the number of switched access lines or type of service subscribed to this plan.

Number of Switched Access Lines	MRC
1 – 5 Lines	\$5.00
6 – 10 Lines	\$15.00
11 – 20 Lines, one PRI or one T-1	\$35.00

# .2 Per Call Usage Rates

### Direct Dial

	Initial 30 Seconds	Each Add'l 6 Seconds
IntraLATA	\$0.000	\$0.000
InterLATA	\$0.035	\$0.007

#### Toll-free Inbound

	Initial	Each Add'l	
	30 Seconds	6 Seconds	
IntraLATA	\$0.035	\$0.007	
InterLATA	\$0.035	\$0.007	

#### SECTION 12 – GRANDFATHERED SERVICES

### 12.3 Type 2 – Grandfathered Services (continued)

### 12.3.6 International Business One Plan (Grandfathered 9/4/2007)

(A) The International Business One Plan is a direct dialed outbound international long distance service offered to business Customers from switched access lines. Service is offered on a flat rate basis, twenty-four hours per day, seven days per week. Per minute charges apply and will be based on the country called and the duration of the call. Customers must dial 1 plus the country code or area code, if applicable, and the terminating number to complete a call utilizing this service.

Original Page 13

Effective: September 1, 2019

- (B) To be eligible for this plan Customers must subscribe to the Company's Affiliated ILEC for local exchange service and must presubscribe to the Company for both intraLATA and interLATA usage.
- (C) All calls are billed in six (6) second increments after an initial period, for billing purposes, of thirty (30) seconds. This service is only offered in conjunction with the corresponding domestic Business One Plan. This service is not offered on an intraLATA only basis.
- (D) International Mobile Termination Charge

See Section 6.4.1 of this pricing and service guide for the applicable International Mobile Termination Charge.

(E) The MRC is \$7.95, and the per-minute usage rates are as follows:

		Each		Initial	Each
~ .	Initial	Additional	~	30	Additional
Country	30 Seconds	6 Seconds	Country	Seconds	6 Seconds
Afghanistan	\$0.6550	\$0.1310	Antigua	\$0.1850	\$0.0370
Albania	\$0.2850	\$0.0570	Argentina	\$0.1900	\$0.0380
Algeria	\$0.2450	\$0.0490	Armenia	\$0.3900	\$0.0620
Andorra	\$0.1050	\$0.0210	Aruba	\$0.1600	\$0.0320
Angola	\$0.3650	\$0.0730	Ascension Island	\$0.3500	\$0.0700
Anguilla	\$0.2900	\$0.0420	Australia	\$0.0750	\$0.0150
Antarctica (Casey Base)	\$0.8850	\$0.1770	Austria	\$0.0750	\$0.0150
Antarctica (Scott Base)	\$0.8850	\$0.1770	Azerbaijan	\$0.4000	\$0.0800

# ce Effective: September 1, 2019 BUSINESS AND RESIDENTIAL SERVICES

Original Page 14

## PRICING AND SERVICE GUIDE SECTION 12 – GRANDFATHERED SERVICES

- 12.3 Type 2 Grandfathered Services (continued)
  - 12.3.6 International Business One Plan (continued)
    - (E) The per-minute usage rates are as follows: (continued)

	T '/' I	Each		T '4' 1	Each
Country	Initial 30 Seconds	Additional 6 Seconds	Country	Initial 30 Seconds	Additional 6 Seconds
Bahamas	\$0.9000	\$0.0200	El Salvador	\$0.2050	\$0.0410
Bahrain	\$0.2650	\$0.0200	Equatorial Guinea	\$0.2030	\$0.0410
Bangladesh	\$0.2030	\$0.0330	Eritrea	\$0.4750	\$0.1340
Barbados	\$0.4200	\$0.0420	Estonia Estonia	\$0.4750	\$0.0430
Belarus	\$0.2400	\$0.0420	Ethiopia Ethiopia	\$0.2130	\$0.0430
Belgium	\$0.2400	\$0.0480	Faeroe Island	\$0.4230	\$0.0350
Belize	\$0.2650	\$0.0130	Falkland Island	\$0.1730	\$0.0330
Benin	\$0.2050	\$0.0330	Fiji Island	\$0.3650	\$0.1040
Bermuda	\$0.2330	\$0.0470	Finland	\$0.3030	\$0.0730
Bhutan	\$0.1300	\$0.0200	France	\$0.0750	\$0.0150
Bolivia	\$0.3900	\$0.1020	French Antilles	\$0.0730	\$0.0130
Bosnia	\$0.2250	\$0.0430	French Guiana	\$0.1930	\$0.0590
Botswana	\$0.2350	\$0.0310	French Polynesia	\$0.2300	\$0.0300
Brazil	\$0.2230	\$0.0430	Gabon	\$0.3930	\$0.0790
British Virgin Islands	\$0.1650	\$0.0320	Gambia	\$0.2700	\$0.0540
Brunei	\$0.1650	\$0.0530	Gambia Georgia	\$0.2700	\$0.0340
Bulgaria	\$0.2030	\$0.0350		\$0.4030	\$0.0310
Burkina Faso	\$0.1730	\$0.0530	Germany Ghana	\$0.0730	\$0.0130
Burundi	\$0.2900	\$0.0380	Gibraltar	\$0.2850	\$0.0460
Cambodia	\$0.5700	\$0.1140	Greece	\$0.2830	\$0.0370
	\$0.0400	\$0.1280		\$0.9000	\$0.0200
Cameroon Canada		\$0.0380	Greenland Grenada		
	\$0.0350			\$0.2250	\$0.0450
Cape Verde Island Cayman Islands	\$0.3000 \$0.1500	\$0.0600 \$0.0300	Guadeloupe Guantanamo	\$0.1950 \$0.6450	\$0.0390 \$0.1290
Central African Republic	\$0.1300	\$0.0300	Guatemala	\$0.0430	\$0.1290
Chad	\$1.1900	\$0.1080	Guinea-Bissau	\$0.2030	\$0.0410
Chile	\$0.1500	\$0.2220		\$0.4050	\$0.0810
China	\$0.1300	\$0.0300	Guyana Haiti	\$0.4230	\$0.0550
				\$0.2750	\$0.0330
Christmas Island Cocos Island	\$0.0750 \$0.0750	\$0.0150 \$0.0150	Honduras Hong Kong	\$0.2150	\$0.0430
Colombia	\$0.0730	\$0.0130		\$0.9000	\$0.0200
Comoros Islands	\$0.1900	\$0.0380	Hungary Iceland	\$0.1050	\$0.0350
	\$0.4630	\$0.0930	India		\$0.0530
Congo Cook Islands	\$0.5200	\$0.0640	India Indonesia	\$0.2650 \$0.2900	\$0.0330
Costa Rica	\$0.0000	\$0.1200	Inmarsat (Atlantic Ocean-East)	\$4.1250	\$0.8250
Croatia	\$0.1000	\$0.0320	Inmarsat (Atlantic Ocean-Bast)	\$4.1250	\$0.8250
Cuba	\$0.2030	\$0.0410	Inmarsat (Indian Ocean)	\$4.1250	\$0.8250
	\$0.3000	\$0.9000	Inmarsat (Pacific Ocean)	\$4.1250	\$0.8250
Cyprus Czech Republic	\$0.2900	\$0.0420	Inmarsat (SNAC)	\$4.1250	\$0.8250
Denmark	\$0.1030	\$0.0330	Iran	\$0.4900	\$0.8230
Diego Garcia Djibouti	\$0.7050 \$0.3700	\$0.1410 \$0.0740	Iraq Ireland	\$0.5000 \$0.0750	\$0.9000 \$0.0150
Dominican Republic	\$0.3700	\$0.0740	Israel	\$0.0730	\$0.0130
				\$0.1150	
Dominica East Timor	\$0.2050 \$1.2050	\$0.0410 \$0.2410	Italy Ivory Coast	\$0.0750	\$0.0150 \$0.0840
				\$0.4200	
Easter Island	\$0.1500 \$0.2300	\$0.0300	Jamaica Topon	\$0.2700	\$0.0540
Ecuador		\$0.0460	Japan Tordan	\$0.0750	\$0.0150
Egypt	\$0.2950	\$0.0590	Jordan	J \$0.3400	\$0.0680

BUSINESS AND RESIDENTIAL SERVICES

Effective: September 1, 2019

Original Page 15

## PRICING AND SERVICE GUIDE SECTION 12 – GRANDFATHERED SERVICES

- 12.3 Type 2 Grandfathered Services (continued)
  - 12.3.6 International Business One Plan (continued)
    - (E) The per-minute usage rates are as follows: (continued)

	Tuitial	Each		Initial	Each
Country	Initial 30 Seconds	Additional 6 Seconds	Country	30 Seconds	Additional 6 Seconds
Kazakhstan	\$0.1500	\$0.0300	Nicaragua	\$0.2150	\$0.0430
Kenya	\$0.3000	\$0.0600	Niger Republic	\$0.3400	\$0.0430
Kiribati	\$0.4500	\$0.0900	Nigeria	\$0.3250	\$0.0650
Korea (North)	\$0.6500	\$0.1300	Niue	\$0.6300	\$0.1260
Korea (South)	\$0.9000	\$0.0200	Norfolk Island	\$0.8850	\$0.1770
Kosovo	\$0.0750	\$0.0200	Norway	\$0.0750	\$0.0150
Kuwait	\$0.3450	\$0.0690	Oman	\$0.3400	\$0.0680
Kuwan	\$0.3900	\$0.0780	Pakistan	\$0.3750	\$0.0750
Laos	\$0.5600	\$0.0780	Palau	\$0.3750	\$0.0750
Latvia	\$0.2550	\$0.0510	Palestinian Authority	\$0.1950	\$0.0730
Lebanon	\$0.2330	\$0.0510	Panama	\$0.1930	\$0.0390
Lesotho	\$0.3050	\$0.0610	Papua New Guinea	\$0.2130	\$0.0500
Liberia	\$0.2300	\$0.0010	*	\$0.2300	
Libya	\$0.2300	\$0.0460	Paraguay Peru	\$0.2300	\$0.0460 \$0.0450
Liechtenstein	\$0.3900	\$0.0020	Philippines	\$0.2250	\$0.0430
Lithuania					
	\$0.3900	\$0.0620	Poland	\$0.1500	\$0.0300
Luxembourg	\$0.0750	\$0.0150	Portugal	\$0.0800	\$0.0160
Macao	\$0.2850	\$0.0570	Qatar	\$0.3450	\$0.0690
Macedonia	\$0.2150	\$0.0430	Reunion Island	\$0.3750	\$0.0750
Madagascar	\$1.0850	\$0.2170	Romania	\$0.2250	\$0.0450
Malawi	\$0.2400	\$0.0480	Russia	\$0.1500	\$0.0300
Malaysia	\$0.1550	\$0.0310	Rwanda, Republic of	\$0.4150	\$0.0830
Maldives	\$0.4600	\$0.0920	San Marino	\$0.2900	\$0.0580
Mali Republic	\$0.3900	\$0.0780	Sao Tome	\$0.7950	\$0.1590
Malta Republic	\$0.2050	\$0.0410	Saudi Arabia	\$0.3600	\$0.0720
Marshall Island	\$0.2750	\$0.0550	Senegal	\$0.4350	\$0.0870
Mauritania	\$0.3600	\$0.0720	Serbia	\$0.2500	\$0.0500
Mauritius	\$0.4200	\$0.0840	Seychelles Island	\$0.5000	\$0.9000
Mayotte Island	\$0.3750	\$0.0750	Sierra Leone	\$0.3600	\$0.0720
Mexico 1	\$0.0750	\$0.0150	Singapore	\$0.1250	\$0.0250
Mexico 2	\$0.1250	\$0.0250	Slovakia	\$0.2050	\$0.0410
Micronesia	\$0.3250	\$0.0650	Slovenia	\$0.1750	\$0.0350
Moldova	\$0.3600	\$0.0720	Solomon Island	\$0.5000	\$0.9000
Monaco	\$0.0750	\$0.0150	Somalia Democratic Rep	\$0.5750	\$0.1150
Mongolia	\$0.6000	\$0.1200	South Africa	\$0.1800	\$0.0360
Montenegro	\$0.2500	\$0.0500	Spain	\$0.0750	\$0.0150
Montserrat	\$0.2250	\$0.0450	Sri Lanka	\$0.3600	\$0.0720
Morocco	\$0.2900	\$0.0580	St. Helena	\$0.4050	\$0.0810
Mozambique	\$0.3750	\$0.0750	St. Kitts/Nevis	\$0.1900	\$0.0380
Myanmar	\$0.8000	\$0.1600	St. Lucia	\$0.1950	\$0.0390
Namibia	\$0.2550	\$0.0510	St. Maarten	\$0.1650	\$0.0330
Nauru	\$0.5350	\$0.1070	St. Pierre Miquelon	\$0.1700	\$0.0340
Nepal	\$0.3850	\$0.0770	St. Vincent/Grenadines	\$0.2150	\$0.0430
Netherland Antilles	\$0.1650	\$0.0330	South Sudan	\$0.4400	\$0.0880
Netherlands	\$0.0750	\$0.0150	Sudan	\$0.4400	\$0.0880
Networks	\$3.0000	\$0.6000	Suriname	\$0.4350	\$0.0870
New Caledonia	\$0.4200	\$0.0840	Swaziland	\$0.2050	\$0.0410
New Zealand	\$0.1900	\$0.0220	Sweden	\$0.0750	\$0.0150

Original Page 16

Effective: September 1, 2019

- 12.3 Type 2 Grandfathered Services (continued)
  - 12.3.6 International Business One Plan (continued)
    - (E) The per-minute usage rates are as follows: (continued)

	Initial	Each Additional		Initial	Each Additional
Country	30 Seconds	6 Seconds	Country	30 Seconds	
Switzerland	\$0.0750	\$0.0150	Ukraine	\$0.2250	\$0.0450
Syrian Arab Republic	\$0.3700	\$0.0740	United Arab Emirates	\$0.2250	\$0.0450
Taiwan	\$0.1250	\$0.0250	United Kingdom	\$0.0350	\$0.0070
Tajikistan	\$0.1850	\$0.0370	Uruguay	\$0.2250	\$0.0450
Tanzania	\$0.2900	\$0.0580	Uzbekistan	\$0.2250	\$0.0450
Thailand	\$0.1800	\$0.0360	Vanuatu	\$1.1500	\$0.2300
Togo	\$0.4250	\$0.0850	Vatican City	\$0.0750	\$0.0150
Tokelau	\$1.5650	\$0.3130	Venezuela	\$0.1650	\$0.0330
Tonga Island	\$0.5550	\$0.1110	Vietnam	\$0.3900	\$0.0780
Trinidad	\$0.2400	\$0.0480	Wallis Futuna	\$0.6500	\$0.1300
Tunisia	\$0.2150	\$0.0430	Western Samoa	\$0.3250	\$0.0650
Turkey	\$0.1900	\$0.0380	Yemen Arab Republic	\$0.3550	\$0.0710
Turkmenistan	\$0.3600	\$0.0720	Zaire	\$0.2800	\$0.0560
Turks and Caicos Islands	\$0.2050	\$0.0410	Zambia	\$0.2850	\$0.0570
Tuvalu	\$0.9550	\$0.1910	Zimbabwe	\$0.2250	\$0.0450
Uganda	\$0.2850	\$0.0570			

### SECTION 12 – GRANDFATHERED SERVICES

- 12.3 Type 2 Grandfathered Services (continued)
  - 12.3.7 Talk 500 Plan<sup>1</sup> (Grandfathered 12/17/2003)
    - (A) This plan is a direct dialed outbound long distance service offered to residential Customers from switched access lines. Service is offered on a flat rate basis, twenty-four hours per day, seven days per week. Customers must dial 1 plus the area code, if applicable, and the terminating number to complete a call utilizing this service. This plan provides the Customer with 500 minutes of interexchange (both interstate and intrastate) long distance usage for a flat rate monthly service charge. Additional usage over the initial 500 minutes will be billed on a flat rate per minute basis.

Original Page 17

- (B) Customers of this plan must subscribe to a local plan/serice and at least two verticals features as defined and offered by the Company's Affiliated ILEC.
- (C) All calls are billed in one (1) minute increments after an initial period, for billing purpose, of one (1) minute. This service is not offered on an intraLATA only basis.
- (D) The MRC is \$26.99, and the per-minute usage rate is \$0.0700 for calls completed after the 500 minute block of time has been used.

This service is no longer available to new Customers or existing Customers at new locations effective December 17, 2003.

### SECTION 12 – GRANDFATHERED SERVICES

- 12.3 Type 2 Grandfathered Services (continued)
  - 12.3.8 Unlimited Savings Value Plan (Grandfathered 11/1/2005)
    - (A) The Unlimited Savings Value plan is a direct dialed outbound long distance service offered to single line or multi-line residential Customers on switched access lines. Service is offered on a flat rate basis, twenty-four hours per day, seven days per week. Customers must dial 1 plus the area code, if applicable, and the terminating number to complete a call utilizing this service. The Unlimited Savings Value Plan provides the Customer with unlimited minutes of interexchange interstate long distance usage for a flat rate monthly charge (certain restrictions, as outlined below, do apply). This plan does not provide call detail information on the Customer's monthly bill. This service is not offered on an intraLATA or intrastate only basis.

Original Page 18

- (B) In order to be eligible for this plan, the Company must be able to verify that the Customer meets the following eligibility requirements. Customers who no longer meet these eligibility requirements will not be eligible for this plan and may be offered an alternative plan or have their service suspended, restricted or canceled without prior notice. Customers with one to three lines at a location are eligible for this service if they meet the following requirements:
  - .1 Customers must subscribe to and maintain one of the residential services and Vertical Feature Packages as defined and offered by the Affiliated ILEC of the Company.
  - .2 Customers must presubscribe to BellSouth Long Distance for both intraLATA and interLATA long distance service on all lines subscribed to under this plan.
  - .3 This plan cannot be used for any use inconsistent with residential service.
  - .4 This plan is not available on an account that is the recipient of charges billed from another location.
  - .5 Customer lines associated with educational institutions, (colleges, universities, etc.) are not eligible for this plan.
  - .6 Unlimited plan usage does not include multi-party conference calls (except those calls placed by using the Three-Way Calling Feature are included), calls to 900, 976, 700 numbers, directory assistance, operator services, international calling and toll-free calling services.
  - .7 This plan is not available for resale.

### SECTION 12 – GRANDFATHERED SERVICES

- 12.3 Type 2 Grandfathered Services (continued)
  - 12.3.8 Unlimited Savings Value Plan (continued)
    - (B) In order to be eligible for this plan, the Company must be able to verify that the Customer meets the following eligibility requirements: (continued)
      - .8 If usage under this plan is not consistent with typical residential Customer usage, at the Company's sole discretion, the Company may offer the Customer an alternative plan or suspend, restrict or cancel the Customer's service without prior notice. Calls that are not consistent with typical residential voice usage include but are not limited to: use for general business purposes, commercial facsimile, auto-dialing, resale, call centers, long distance access to internet service providers and telemarketing.

Original Page 19

- .9 The Company reserves the right to exclude certain terminating telephone numbers from this plan. Calls to these numbers will be billed at applicable MTS rates.
- (C) The MRC is \$31.99 for unlimited intrastate and interstate minutes of use.

### SECTION 12 – GRANDFATHERED SERVICES

### 12.3 Type 2 – Grandfathered Services (continued)

### 12.3.9 Talk 120 Value Lite Plan (Grandfathered 10/4/2006)

(A) This plan is a direct dialed outbound long distance service offered to residential Customers from switched access lines. Service is offered on a flat rate basis, twenty-four hours per day, seven days per week. Customers must dial 1 plus the area code, if applicable, and the terminating number to complete a call utilizing this service. This plan provides the Customer with 120 minutes of interexchange (both interstate and intrastate) long distance usage for a flat rate monthly service charge. Additional usage over the initial 120 minutes will be billed on a flat rate per minute basis.

Original Page 20

Effective: September 1, 2019

- (B) Customers of this plan must subscribe to a local plan/service and must must also subscribe to at least one Vertical Feature as defined and offered by the Company's Affiliated ILEC.
- (C) All calls are billed in one (1) minute increments after an initial period, for billing purposes, of one (1) minute. This service is not offered on an intraLATA only basis.
- (D) The MRC is \$14.99, and the per-minute usage rate is \$0.1200 for calls completed after the 120 minute block of time has been used.

### 12.3.10 Nickel Value Plan (Grandfathered 12/1/2005)

- (A) The BellSouth® Nickel Value plan is a direct dialed outbound long distance service offered to residential Customers from switched access lines. Service is offered on a flat rate basis, twenty-four hours per day, seven days per week. Customers must dial 1 plus the area code, if applicable, and the terminating number to complete a call utilizing this service.
- (B) Customers must subscribe to and maintain an access line and a Vertical Features Package as defined and offered by the Company's Affiliated ILEC. This plan is no longer available to Customers who subscribe to Two-Party Service offered by the Affiliated ILEC of the Company. This plan may also require subscription to at least one (1) or more products or services offered by a Company Affiliate as defined and offered by that Company Affiliate in conjunction with a subscription to this plan. Customers required to subscribe to one (1) or more products offered by a Company Affiliate must continue to subscribe to those products to be eligible for this plan.
- (C) All calls are billed in one (1) minute increments after an initial period, for billing purposes, of one (1) minute. This service is not offered on an intraLATA only basis.
- (D) The MRC is \$9.99, and the per-minute usage rate is \$0.0500.

### SECTION 12 - GRANDFATHERED SERVICES

- 12.3 Type 2 Grandfathered Services (continued)
  - 12.3.11 AT&T Core Value Plan (Grandfathered 8/16/2007)
    - (A) The AT&T Core Value plan is a direct dialed outbound long distance service offered to residential Customers from switched access lines. Service is offered on a flat rate basis, twenty-four hours per day, seven days per week. Customers must dial 1 plus the area code, if applicable, and the terminating number to complete a call utilizing this service.

Original Page 21

- (B) Customers must subscribe to and maintain an access line and a Vertical Feature or Vertical Features Package as defined and offered by the Affiliated ILEC of the Company. This plan is no longer available to Customers who subscribe to Two-Party Service offered by the Affiliated ILEC of the Company.
- (C) All calls are billed in one (1) minute increments after an initial period, for billing purposes, of one (1) minute. This service is only offered in conjunction with the corresponding intrastate AT&T Core Value plan. This service is not offered on an intraLATA only basis.
- (D) The MRC is \$5.99 per account, and the per-minute usage rate is \$0.1400.

Original Page 22 Effective: September 1, 2019

- 12.4 Type 3 Grandfathered Services
  - 12.4.1 Reserved for Future Use

### SECTION 12 – GRANDFATHERED SERVICES

### 12.5 Type 4 - Grandfathered Services

- 12.5.1 Business Country Select® Plan (Grandfathered 10/1/2005)
  - (A) The Business Country Select<sup>®</sup> plan is a direct dialed outbound international long distance service offered to Business Customers from switched access lines. Service is offered to each country on a flat rate basis, twenty-four hours per day, seven days per week. Customers must dial 1 plus the country code or area code, if applicable, and the terminating number to complete a call utilizing this service.

Original Page 23

Effective: September 1, 2019

- (B) All calls are billed in six (6) second increments after an initial period, for billing purpose, of thirty (30) seconds. This service is only offered in conjunction with any Business domestic service.
- (C) This plan allows Business Customers to designate one international country for which they will receive a discounted per minute rate. The Customer designates his/her selected country at the time the plan is selected and may change the selected country at any time. The Company reserves the right to limit the number of times the selected country is changed within a given period of time. Calls to non-selected countries will be billed at the regular rate.
- (D) Inbound international toll-free service originated from locations within Canada is available to Customers of this service who also subscribe to AT&T Long Distance Toll Free<sup>SM</sup> Service in Section 4.4 of this pricing and service guide. The rate table in Section 12.5.1 (F) also contains the per minute rates for Inbound toll-free service originating in Canada and terminating throughout the United States.
- (E) International Mobile Termination Charge

See Section 6.4.1 of this pricing and service guide for the applicable International Mobile Termination Charge.

### SECTION 12 – GRANDFATHERED SERVICES

- 12.5 Type 4 Grandfathered Services (continued)
  - 12.5.1 Business Country Select® Plan (continued)
    - (F) The MRC \$6.95 and is waved once a Customer's international usage exceeds \$50.00 for the current billing month. Usage may not be carried to the next billing cycle to meet the wavier requirement. The per minute usage rates are as follows:

Original Page 24

	<b>Selected Country Rate</b>		Regular Rate		
	Initial	Each Add'l	Initial	Each Add'l	
Country	30 Seconds		30 Seconds	6 Seconds	
Afghanistan	\$0.5900	\$0.1180	\$0.7200	\$0.1440	
Albania	\$0.2550	\$0.0510	\$0.3150	\$0.0630	
Algeria	\$0.2200	\$0.0440	\$0.2700	\$0.0540	
Andorra	\$0.0950	\$0.0190	\$0.1150	\$0.0230	
Angola	\$0.3300	\$0.0660	\$0.4000	\$0.0800	
Anguilla	\$0.1900	\$0.0380	\$0.2300	\$0.0460	
Antarctica (Casey Base)	\$0.7950	\$0.1590	\$0.9750	\$0.1950	
Antarctica (Scott Base)	\$0.7950	\$0.1590	\$0.9750	\$0.1950	
Antigua	\$0.1650	\$0.0330	\$0.2050	\$0.0410	
Argentina	\$0.1700	\$0.0340	\$0.2900	\$0.0420	
Armenia	\$0.2800	\$0.0560	\$0.3400	\$0.0680	
Aruba	\$0.1450	\$0.0290	\$0.1750	\$0.0350	
Ascension Island	\$0.3150	\$0.0630	\$0.3850	\$0.0770	
Australia	\$0.0700	\$0.0140	\$0.0850	\$0.0170	
Austria	\$0.0700	\$0.0140	\$0.0850	\$0.0170	
Azerbaijan	\$0.3600	\$0.0720	\$0.4400	\$0.0880	
Bahamas	\$0.0900	\$0.0180	\$0.1900	\$0.0220	
Bahrain	\$0.2400	\$0.0480	\$0.2900	\$0.0580	
Bangladesh	\$0.3800	\$0.0760	\$0.4600	\$0.0920	
Barbados	\$0.1900	\$0.0380	\$0.2300	\$0.0460	
Belarus	\$0.2150	\$0.0430	\$0.2650	\$0.0530	
Belgium	\$0.0700	\$0.0140	\$0.0850	\$0.0170	
Belize	\$0.2400	\$0.0480	\$0.2900	\$0.0580	
Benin	\$0.2900	\$0.0420	\$0.2600	\$0.0520	
Bermuda	\$0.1150	\$0.0230	\$0.1450	\$0.0290	
Bhutan	\$0.4600	\$0.0920	\$0.5600	\$0.1120	
Bolivia	\$0.2050	\$0.0410	\$0.2500	\$0.0500	
Bosnia	\$0.2300	\$0.0460	\$0.2800	\$0.0560	
Botswana	\$0.2050	\$0.0410	\$0.2500	\$0.0500	
Brazil	\$0.1450	\$0.0290	\$0.1750	\$0.0350	
British Virgin Islands	\$0.1500	\$0.0300	\$0.1800	\$0.0360	
Brunei	\$0.2400	\$0.0480	\$0.2900	\$0.0580	
Bulgaria	\$0.1600	\$0.0320	\$0.1950	\$0.0390	
Burkina Faso	\$0.2600	\$0.0520	\$0.3200	\$0.0640	
Burundi	\$0.5150	\$0.1030	\$0.6250	\$0.1250	
Cambodia	\$0.5750	\$0.1150	\$0.7050	\$0.1410	
Cameroon	\$0.2600	\$0.0520	\$0.3200	\$0.0640	
Canada Outbound	\$0.0300	\$0.0060	\$0.0400	\$0.0080	
Canada - Inbound	\$0.0300	\$0.0060	\$0.0400	\$0.0080	

Original Page 25

Effective: September 1, 2019

- 12.5 Type 4 Grandfathered Services (continued)
  - 12.5.1 Business Country Select® Plan (continued)
    - (F) The per minute usage rates are as follows: (continued)

	Selected C	ountry Rate	Regular Rate		
	Initial	Each Add'l		Each Add'l	
Country	30 Seconds		30 Seconds	6 Seconds	
Cape Verde Island	\$0.2700	\$0.0540	\$0.3300	\$0.0660	
Cayman Islands	\$0.1350	\$0.0270	\$0.1650	\$0.0330	
Central African Republic	\$0.4850	\$0.0970	\$0.5950	\$0.1190	
Chad	\$1.0000	\$0.2000	\$1.2200	\$0.2440	
Chile	\$0.1350	\$0.0270	\$0.1650	\$0.0330	
China	\$0.1750	\$0.0350	\$0.2150	\$0.0430	
Christmas Island	\$0.0700	\$0.0140	\$0.0850	\$0.0170	
Cocos Island	\$0.0700	\$0.0140	\$0.0850	\$0.0170	
Colombia	\$0.1700	\$0.0340	\$0.2900	\$0.0420	
Comoros Islands	\$0.4200	\$0.0840	\$0.5900	\$0.1020	
Congo	\$0.2900	\$0.0580	\$0.3500	\$0.0700	
Cook Islands	\$0.5400	\$0.1080	\$0.6600	\$0.1320	
Costa Rica	\$0.1450	\$0.0290	\$0.1750	\$0.0350	
Croatia	\$0.1850	\$0.0370	\$0.2250	\$0.0450	
Cuba	\$0.4900	\$0.0980	\$0.5500	\$0.1900	
Cyprus	\$0.1900	\$0.0380	\$0.2300	\$0.0460	
Czech Republic	\$0.1500	\$0.0300	\$0.1800	\$0.0360	
Denmark	\$0.0700	\$0.0140	\$0.0850	\$0.0170	
Diego Garcia	\$0.6350	\$0.1270	\$0.7750	\$0.1550	
Djibouti	\$0.3350	\$0.0670	\$0.4050	\$0.0810	
Dominican Republic	\$0.1150	\$0.0230	\$0.1400	\$0.0280	
Dominica	\$0.1850	\$0.0370	\$0.2250	\$0.0450	
East Timor	\$1.2050	\$0.2410	\$1.2050	\$0.2410	
Easter Island	\$0.1350	\$0.0270	\$0.1650	\$0.0330	
Ecuador	\$0.2050	\$0.0410	\$0.2550	\$0.0510	
Egypt	\$0.2650	\$0.0530	\$0.3250	\$0.0650	
El Salvador	\$0.1850	\$0.0370	\$0.2250	\$0.0450	
Equatorial Guinea	\$0.6050	\$0.1210	\$0.7350	\$0.1470	
Eritrea	\$0.4300	\$0.0860	\$0.5250	\$0.1050	
Estonia	\$0.1950	\$0.0390	\$0.2350	\$0.0470	
Ethiopia	\$0.3850	\$0.0770	\$0.4700	\$0.0940	
Faeroe Island	\$0.1600	\$0.0320	\$0.1950	\$0.0390	
Falkland Island	\$0.4700	\$0.0940	\$0.5700	\$0.1140	
Fiji Island	\$0.3300	\$0.0660	\$0.4000	\$0.0800	
Finland	\$0.0700	\$0.0140	\$0.0850	\$0.0170	
France	\$0.0700	\$0.0140	\$0.0850	\$0.0170	
French Antilles	\$0.1750	\$0.0350	\$0.2150	\$0.0430	
French Guiana	\$0.2250	\$0.0450	\$0.2750	\$0.0550	
French Polynesia	\$0.3550	\$0.0710	\$0.4350	\$0.0870	
Gabon	\$0.2650	\$0.0530	\$0.3250	\$0.0650	
Gambia	\$0.2450	\$0.0490	\$0.2950	\$0.0590	
Georgia	\$0.3650	\$0.0730	\$0.4450	\$0.0890	

BUSINESS AND RESIDENTIAL SERVICES

Original Page 26

Effective: September 1, 2019

## PRICING AND SERVICE GUIDE SECTION 12 – GRANDFATHERED SERVICES

- 12.5 Type 4 Grandfathered Services (continued)
  - 12.5.1 Business Country Select® Plan (continued)
    - (F) The per minute usage rates are as follows: (continued)

	Selected C	ountry Rate	Regular Rate			
	Initial	Each Add'l	Initial	Each Add'l		
Country	30 Seconds	6 Seconds	30 Seconds	6 Seconds		
Germany	\$0.0700	\$0.0140	\$0.0850	\$0.0170		
Ghana	\$0.2050	\$0.0410	\$0.2550	\$0.0510		
Gibraltar	\$0.2550	\$0.0510	\$0.3150	\$0.0630		
Greece	\$0.0900	\$0.0180	\$0.1900	\$0.0220		
Greenland	\$0.1750	\$0.0350	\$0.2150	\$0.0430		
Grenada	\$0.2050	\$0.0410	\$0.2500	\$0.0500		
Guadeloupe	\$0.1750	\$0.0350	\$0.2150	\$0.0430		
Guantanamo	\$0.5800	\$0.1160	\$0.7900	\$0.1420		
Guatemala	\$0.1850	\$0.0370	\$0.2250	\$0.0450		
Guinea	\$0.2700	\$0.0540	\$0.3300	\$0.0660		
Guinea-Bissau	\$0.3650	\$0.0730	\$0.4450	\$0.0890		
Guyana	\$0.3850	\$0.0770	\$0.4700	\$0.0940		
Haiti	\$0.2500	\$0.0500	\$0.3050	\$0.0610		
Honduras	\$0.1950	\$0.0390	\$0.2350	\$0.0470		
Hong Kong	\$0.0900	\$0.0180	\$0.1900	\$0.0220		
Hungary	\$0.1500	\$0.0300	\$0.1800	\$0.0360		
Iceland	\$0.1600	\$0.0320	\$0.1950	\$0.0390		
India	\$0.2400	\$0.0480	\$0.2900	\$0.0580		
Indonesia	\$0.1900	\$0.0380	\$0.2300	\$0.0460		
Inmarsat (SNAC)	\$3.7150	\$0.7430	\$4.5400	\$0.9080		
Iran	\$0.3700	\$0.0740	\$0.4500	\$0.0900		
Iraq	\$0.4500	\$0.0900	\$0.5500	\$0.1900		
Ireland	\$0.0700	\$0.0140	\$0.0850	\$0.0170		
Israel	\$0.1050	\$0.0210	\$0.1250	\$0.0250		
Italy	\$0.0700	\$0.0140	\$0.0850	\$0.0170		
Ivory Coast	\$0.3800	\$0.0760	\$0.4600	\$0.0920		
Jamaica	\$0.2450	\$0.0490	\$0.2950	\$0.0590		
Japan	\$0.0700	\$0.0140	\$0.0850	\$0.0170		
Jordan	\$0.3050	\$0.0610	\$0.3750	\$0.0750		
Kazakhstan	\$0.1350	\$0.0270	\$0.1650	\$0.0330		
Kenya	\$0.2700	\$0.0540	\$0.3300	\$0.0660		
Kiribati	\$0.4050	\$0.0810	\$0.4950	\$0.0990		
Korea (North)	\$0.5850	\$0.1170	\$0.7150	\$0.1430		
Korea (South)	\$0.0900	\$0.0180	\$0.1900	\$0.0220		
Kosovo	\$0.2250	\$0.0450	\$0.2750	\$0.0550		
Kuwait	\$0.3900	\$0.0620	\$0.3800	\$0.0760		
Kyrgyzstan	\$0.3500	\$0.0700	\$0.4300	\$0.0860		
Laos	\$0.5050	\$0.1010	\$0.6150	\$0.1230		
Latvia	\$0.2300	\$0.0460	\$0.2800	\$0.0560		

Original Page 27

Effective: September 1, 2019

- 12.5 Type 4 Grandfathered Services (continued)
  - 12.5.1 Business Country Select® Plan (continued)
    - (F) The per minute usage rates are as follows: (continued)

	Selected C	ountry Rate	Regular Rate			
	Initial	Each Add'l		Each Add'l		
Country	30 Seconds	6 Seconds	30 Seconds	6 Seconds		
Lebanon	\$0.2850	\$0.0570	\$0.3450	\$0.0690		
Lesotho	\$0.2750	\$0.0550	\$0.3350	\$0.0670		
Liberia	\$0.2050	\$0.0410	\$0.2550	\$0.0510		
Libya	\$0.2800	\$0.0560	\$0.3400	\$0.0680		
Liechtenstein	\$0.0700	\$0.0140	\$0.0850	\$0.0170		
Lithuania	\$0.2800	\$0.0560	\$0.3400	\$0.0680		
Luxembourg	\$0.0700	\$0.0140	\$0.0850	\$0.0170		
Macao	\$0.2550	\$0.0510	\$0.3150	\$0.0630		
Macedonia	\$0.1950	\$0.0390	\$0.2350	\$0.0470		
Madagascar	\$0.9750	\$0.1950	\$1.1950	\$0.2390		
Malawi	\$0.2150	\$0.0430	\$0.2650	\$0.0530		
Malaysia	\$0.1400	\$0.0280	\$0.1700	\$0.0340		
Maldives	\$0.4150	\$0.0830	\$0.5050	\$0.1010		
Mali Republic	\$0.3500	\$0.0700	\$0.4300	\$0.0860		
Malta Republic	\$0.1850	\$0.0370	\$0.2250	\$0.0450		
Marshall Island	\$0.2500	\$0.0500	\$0.3050	\$0.0610		
Mauritania	\$0.3250	\$0.0650	\$0.3950	\$0.0790		
Mauritius	\$0.3800	\$0.0760	\$0.4600	\$0.0920		
Mayotte Island	\$0.3400	\$0.0680	\$0.4150	\$0.0830		
Mexico 1	\$0.0700	\$0.0140	\$0.0850	\$0.0170		
Mexico 2	\$0.1150	\$0.0230	\$0.1400	\$0.0280		
Micronesia	\$0.2950	\$0.0590	\$0.3600	\$0.0720		
Moldova	\$0.3250	\$0.0650	\$0.3950	\$0.0790		
Monaco	\$0.2250	\$0.0450	\$0.2750	\$0.0550		
Mongolia	\$0.5400	\$0.1080	\$0.6600	\$0.1320		
Montenegro	\$0.2250	\$0.0450	\$0.2750	\$0.0550		
Montserrat	\$0.2050	\$0.0410	\$0.2500	\$0.0500		
Morocco	\$0.2600	\$0.0520	\$0.3200	\$0.0640		
Mozambique	\$0.3400	\$0.0680	\$0.4150	\$0.0830		
Myanmar	\$0.7200	\$0.1440	\$0.8800	\$0.1760		
Namibia	\$0.2300	\$0.0460	\$0.2800	\$0.0560		
Nauru	\$0.4800	\$0.0960	\$0.5900	\$0.1180		
Nepal	\$0.3450	\$0.0690	\$0.4250	\$0.0850		
Netherland Antilles	\$0.1500	\$0.0300	\$0.1800	\$0.0360		
Netherlands	\$0.0700	\$0.0140	\$0.0850	\$0.0170		
Networks	\$3.0000	\$0.6000	\$3.3000	\$0.6600		
New Caledonia	\$0.3800	\$0.0760	\$0.4600	\$0.0920		
New Zealand	\$0.9000	\$0.0200	\$0.1200	\$0.0240		
Nicaragua Nicaragua	\$0.1950	\$0.0390	\$0.2350	\$0.0470		
Niger Republic	\$0.3050	\$0.0610	\$0.3750	\$0.0750		
Nigeria Nigeria	\$0.2950	\$0.0590	\$0.3600	\$0.0720		
Niue	\$0.5650	\$0.1130	\$0.6950	\$0.1390		
Norfolk Island	\$0.7950	\$0.1590	\$0.9750	\$0.1950		

Original Page 28

Effective: September 1, 2019

- 12.5 Type 4 Grandfathered Services (continued)
  - 12.5.1 Business Country Select® Plan (continued)
    - (F) The per minute usage rates are as follows: (continued)

	Selected C	ountry Rate	Regular Rate		
	Initial	Each Add'l		Each Add'l	
Country	30 Seconds	6 Seconds	30 Seconds	6 Seconds	
Norway	\$0.0700	\$0.0140	\$0.0850	\$0.0170	
Oman	\$0.3050	\$0.0610	\$0.3750	\$0.0750	
Pakistan	\$0.3400	\$0.0680	\$0.4150	\$0.0830	
Palau	\$0.3400	\$0.0680	\$0.4150	\$0.0830	
Palestinian Authority	\$0.1950	\$0.0390	\$0.1950	\$0.0390	
Panama	\$0.1950	\$0.0390	\$0.2350	\$0.0470	
Papua New Guinea	\$0.2250	\$0.0450	\$0.2750	\$0.0550	
Paraguay	\$0.2050	\$0.0410	\$0.2550	\$0.0510	
Peru	\$0.2050	\$0.0410	\$0.2500	\$0.0500	
Philippines	\$0.1500	\$0.0300	\$0.1800	\$0.0360	
Poland	\$0.1350	\$0.0270	\$0.1650	\$0.0330	
Portugal	\$0.0700	\$0.0140	\$0.0900	\$0.0180	
Reunion Island	\$0.3400	\$0.0680	\$0.4150	\$0.0830	
Romania	\$0.2050	\$0.0410	\$0.2500	\$0.0500	
Russia	\$0.1350	\$0.0270	\$0.1650	\$0.0330	
Rwanda, Republic of	\$0.3750	\$0.0750	\$0.4550	\$0.0910	
San Marino	\$0.2600	\$0.0520	\$0.3200	\$0.0640	
Sao Tome	\$0.7150	\$0.1430	\$0.8750	\$0.1750	
Saudi Arabia	\$0.3250	\$0.0650	\$0.3950	\$0.0790	
Senegal	\$0.3900	\$0.0780	\$0.4800	\$0.0960	
Serbia	\$0.2250	\$0.0450	\$0.2750	\$0.0550	
Seychelles Island	\$0.4500	\$0.0900	\$0.5500	\$0.1900	
Sierra Leone	\$0.3250	\$0.0650	\$0.3950	\$0.0790	
Singapore	\$0.1150	\$0.0230	\$0.1400	\$0.0280	
Slovakia	\$0.1850	\$0.0370	\$0.2250	\$0.0450	
Slovenia	\$0.1600	\$0.0320	\$0.1950	\$0.0390	
Solomon Island	\$0.4500	\$0.0900	\$0.5500	\$0.1900	
Somalia Democratic Rep	\$0.5200	\$0.1040	\$0.6350	\$0.1270	
South Africa	\$0.1600	\$0.0320	\$0.2000	\$0.0400	
Spain	\$0.0700	\$0.0140	\$0.0850	\$0.0170	
Sri Lanka	\$0.3250	\$0.0650	\$0.3950	\$0.0790	
St. Helena	\$0.3650	\$0.0730	\$0.4450	\$0.0890	
St. Kitts/Nevis	\$0.1700	\$0.0340	\$0.2900	\$0.0420	
St. Lucia	\$0.1750	\$0.0350	\$0.2150	\$0.0430	
St. Maarten	\$0.1500	\$0.0300	\$0.1800	\$0.0360	
St. Pierre Miquelon	\$0.1550	\$0.0310	\$0.1850	\$0.0370	
St. Vincent/Grenadines	\$0.1950	\$0.0390	\$0.2350	\$0.0470	
South Sudan	\$0.3950	\$0.0790	\$0.4850	\$0.0970	
Sudan	\$0.3950	\$0.0790	\$0.4850	\$0.0970	
Suriname	\$0.3900	\$0.0780	\$0.4800	\$0.0960	
Swaziland	\$0.1850	\$0.0370	\$0.2250	\$0.0450	

Original Page 29

Effective: September 1, 2019

- 12.5 Type 4 Grandfathered Services (continued)
  - 12.5.1 Business Country Select® Plan (continued)
    - (F) The per minute usage rates are as follows: (continued)

	Selected C	ountry Rate	Regula	r Rate
	Initial	Each Add'l		Each Add'l
Country	30 Seconds	6 Seconds	30 Seconds	6 Seconds
Sweden	\$0.0700	\$0.0140	\$0.0850	\$0.0170
Switzerland	\$0.0700	\$0.0140	\$0.0850	\$0.0170
Syrian Arab Republic	\$0.3350	\$0.0670	\$0.4050	\$0.0810
Taiwan	\$0.1150	\$0.0230	\$0.1400	\$0.0280
Tajikistan	\$0.1650	\$0.0330	\$0.2050	\$0.0410
Tanzania	\$0.2600	\$0.0520	\$0.3200	\$0.0640
Thailand	\$0.1600	\$0.0320	\$0.2000	\$0.0400
Togo	\$0.3850	\$0.0770	\$0.4700	\$0.0940
Tokelau	\$1.4900	\$0.2820	\$1.7200	\$0.3440
Tonga Island	\$0.5000	\$0.9000	\$0.6900	\$0.1220
Trinidad	\$0.2150	\$0.0430	\$0.2650	\$0.0530
Tunisia	\$0.1950	\$0.0390	\$0.2350	\$0.0470
Turkey	\$0.1700	\$0.0340	\$0.2900	\$0.0420
Turkmenistan	\$0.3250	\$0.0650	\$0.3950	\$0.0790
Turks and Caicos Islands	\$0.1850	\$0.0370	\$0.2250	\$0.0450
Tuvalu	\$0.8600	\$0.1720	\$1.0500	\$0.2900
Uganda	\$0.2550	\$0.0510	\$0.3150	\$0.0630
Ukraine	\$0.2050	\$0.0410	\$0.2500	\$0.0500
United Arab Emirates	\$0.2050	\$0.0410	\$0.2500	\$0.0500
United Kingdom	\$0.0300	\$0.0060	\$0.0400	\$0.0080
Uruguay	\$0.2050	\$0.0410	\$0.2500	\$0.0500
Uzbekistan	\$0.2050	\$0.0410	\$0.2500	\$0.0500
Vanuatu	\$1.0350	\$0.2070	\$1.2650	\$0.2530
Vatican City	\$0.0700	\$0.0140	\$0.0850	\$0.0170
Venezuela	\$0.1500	\$0.0300	\$0.1800	\$0.0360
Vietnam	\$0.3500	\$0.0700	\$0.4300	\$0.0860
Wallis Futuna	\$0.5850	\$0.1170	\$0.7150	\$0.1430
Western Samoa	\$0.2950	\$0.0590	\$0.3600	\$0.0720
Yemen Arab Republic	\$0.3200	\$0.0640	\$0.3900	\$0.0780
Zaire	\$0.2500	\$0.0500	\$0.3900	\$0.0620
Zambia	\$0.2550	\$0.0510	\$0.3150	\$0.0630
Zimbabwe	\$0.2050	\$0.0410	\$0.2500	\$0.0500

### SECTION 12 – GRANDFATHERED SERVICES

- 12.5 Type 4 Grandfathered Services (continued)
  - 12.5.2 Business International Advantage Plan (Grandfathered 10/1/2005)
    - (A) The International Advantage plan is a direct dialed outbound international long distance service offered to Customers from switched access lines. Service is offered to each country on a flat rate basis, twenty-four hours per day, seven days per week. Customers must dial 1 plus the country code or area code, if applicable, and the terminating number to complete a call utilizing this service.

Original Page 30

Effective: September 1, 2019

- (B) All calls are billed in six (6) second increments after an initial period, for billing purpose, of thirty (30) seconds. This service is only offered in conjunction with any of the Company's Business Domestic Services.
- (C) Inbound international toll-free service originated from locations within Canada is available to Customers of this service who also subscribe to AT&T Long Distance Toll Free<sup>SM</sup> Service in Section 4.4 of this pricing and service guide. The rate table in Section 12.5.2 (E) also contains the per-minute rates for inbound toll-free service originating in Canada and terminating throughout the United States.
- (D) International Mobile Termination Charge

See Section 6.4.1 of this pricing and service guide for the applicable International Mobile Termination Charge.

Original Page 31

Effective: September 1, 2019

- 12.5 Type 4 Grandfathered Services (continued)
  - 12.5.2 Business International Advantage Plan (continued)
    - (E) The MRC is 9.95, and the per-minute usage rates are as follows:

Country	Initial 30 Seconds	Each Additional 6 Seconds	Country	Initial 30 Seconds	Each Additional 6 Seconds
Afghanistan	\$0.5900	\$0.1180	Christmas Island	\$0.0700	\$0.0140
Albania	\$0.2550	\$0.0510	Cocos Island	\$0.0700	\$0.0140
Algeria	\$0.2200	\$0.0440	Colombia	\$0.1700	\$0.0340
Andorra	\$0.0950	\$0.0190	Comoros Islands	\$0.4200	\$0.0840
Angola	\$0.3300	\$0.0660	Congo	\$0.2900	\$0.0580
Anguilla	\$0.1900	\$0.0380	Cook Islands	\$0.5400	\$0.1080
Antarctica (Casey Base)	\$0.7950	\$0.1590	Costa Rica	\$0.1450	\$0.0290
Antarctica (Scott Base)	\$0.7950	\$0.1590	Croatia	\$0.1850	\$0.0370
Antigua	\$0.1650	\$0.0330	Cuba	\$0.4900	\$0.0980
Argentina	\$0.1700	\$0.0340	Cyprus	\$0.1900	\$0.0380
Armenia	\$0.2800	\$0.0560	Czech Republic	\$0.1500	\$0.0300
Aruba	\$0.1450	\$0.0290	Denmark	\$0.0700	\$0.0140
Ascension Island	\$0.3150	\$0.0630	Diego Garcia	\$0.6350	\$0.1270
Australia	\$0.0700	\$0.0140	Djibouti	\$0.3350	\$0.0670
Austria	\$0.0700	\$0.0140	Dominican Republic	\$0.1150	\$0.0230
Azerbaijan	\$0.3600	\$0.0720	Dominica	\$0.1850	\$0.0370
Bahamas	\$0.0900	\$0.0180	East Timor	\$1.2050	\$0.2410
Bahrain	\$0.2400	\$0.0480	Easter Island	\$0.1350	\$0.0270
Bangladesh	\$0.3800	\$0.0760	Ecuador	\$0.2050	\$0.0410
Barbados	\$0.1900	\$0.0380	Egypt	\$0.2650	\$0.0530
Belarus	\$0.2150	\$0.0430	El Salvador	\$0.1850	\$0.0370
Belgium	\$0.0700	\$0.0140	Equatorial Guinea	\$0.6050	\$0.1210
Belize	\$0.2400	\$0.0480	Eritrea	\$0.4300	\$0.0860
Benin	\$0.2900	\$0.0420	Estonia	\$0.1950	\$0.0390
Bermuda	\$0.1150	\$0.0230	Ethiopia	\$0.3850	\$0.0770
Bhutan	\$0.4600	\$0.0920	Faeroe Island	\$0.1600	\$0.0320
Bolivia	\$0.2050	\$0.0410	Falkland Island	\$0.4700	\$0.0940
Bosnia	\$0.2300	\$0.0460	Fiji Island	\$0.3300	\$0.0660
Botswana	\$0.2050	\$0.0410	Finland	\$0.0700	\$0.0140
Brazil	\$0.1450	\$0.0290	France	\$0.0700	\$0.0140
British Virgin Islands	\$0.1500	\$0.0300	French Antilles	\$0.1750	\$0.0350
Brunei	\$0.2400	\$0.0480	French Guiana	\$0.2250	\$0.0450
Bulgaria	\$0.1600	\$0.0320	French Polynesia	\$0.3550	\$0.0710
Burkina Faso	\$0.2600	\$0.0520	Gabon	\$0.2650	\$0.0530
Burundi	\$0.5150	\$0.1030	Gambia	\$0.2450	\$0.0490
Cambodia	\$0.5750	\$0.1150	Georgia	\$0.3650	\$0.0730
Cameroon	\$0.2600	\$0.0520	Germany	\$0.0700	\$0.0140
Canada Outbound	\$0.0300	\$0.0060	Ghana	\$0.2050	\$0.0410
Canada Inbound	\$0.0300	\$0.0060	Gibraltar	\$0.2550	\$0.0510
Cape Verde Island	\$0.2700	\$0.0540	Greece	\$0.0900	\$0.0180
Cayman Islands	\$0.1350	\$0.0270	Greenland	\$0.1750	\$0.0350
Central African Republic	\$0.4850	\$0.0970	Grenada	\$0.2050	\$0.0410
Chad	\$1.0000	\$0.2000	Guadeloupe	\$0.1750	\$0.0350
Chile	\$0.1350	\$0.0270	Guantanamo	\$0.5800	\$0.1160
China	\$0.1750	\$0.0350	Guatemala	\$0.1850	\$0.0370

Original Page 32

Effective: September 1, 2019

- 12.5 Type 4 Grandfathered Services (continued)
  - 12.5.2 Business International Advantage Plan (continued)
    - (E) The per-minute usage rates are as follows: (continued)

		Each			Each
C	Initial	Additional	C 4	Initial	Additional
Country	30 Seconds		Country	30 Seconds	
Guinea	\$0.2700	\$0.0540	Mali Danielia	\$0.4150	\$0.0830
Guinea-Bissau	\$0.3650	\$0.0730	Mali Republic	\$0.3500	\$0.0700
Guyana	\$0.3850	\$0.0770	Malta Republic	\$0.1850	\$0.0370
Haiti	\$0.2500	\$0.0500	Marshall Island	\$0.2500	\$0.0500
Honduras	\$0.1950	\$0.0390	Mauritania	\$0.3250	\$0.0650
Hong Kong	\$0.0900	\$0.0180	Mauritius	\$0.3800	\$0.0760
Hungary	\$0.1500	\$0.0300	Mayotte Island	\$0.3400	\$0.0680
Iceland	\$0.1600	\$0.0320	Mexico 1	\$0.0700	\$0.0140
India	\$0.2400	\$0.0480	Mexico 2	\$0.1150	\$0.0230
Indonesia	\$0.1900	\$0.0380	Micronesia	\$0.2950	\$0.0590
Inmarsat (Atlantic Ocean-East)	\$3.7150	\$0.7430	Moldova	\$0.3250	\$0.0650
Inmarsat (Atlantic Ocean-West)		\$0.7430	Monaco	\$0.0700	\$0.0140
Inmarsat (Indian Ocean)	\$3.7150	\$0.7430	Mongolia	\$0.5400	\$0.1080
Inmarsat (SNAC)	\$3.7150	\$0.7430	Montenegro	\$0.2250	\$0.0450
Inmarsat (Pacific Ocean)	\$3.7150	\$0.7430	Montserrat	\$0.2050	\$0.0410
Iran	\$0.3700	\$0.0740	Morocco	\$0.2600	\$0.0520
Iraq	\$0.4500	\$0.0900	Mozambique	\$0.3400	\$0.0680
Ireland	\$0.0700	\$0.0140	Myanmar	\$0.7200	\$0.1440
Israel	\$0.1050	\$0.0210	Namibia	\$0.2300	\$0.0460
Italy	\$0.0700	\$0.0140	Nauru	\$0.4800	\$0.0960
Ivory Coast	\$0.3800	\$0.0760	Nepal	\$0.3450	\$0.0690
Jamaica	\$0.2450	\$0.0490	Netherland Antilles	\$0.1500	\$0.0300
Japan	\$0.0700	\$0.0140	Netherlands	\$0.0700	\$0.0140
Jordan	\$0.3050	\$0.0610	Networks	\$3.0000	\$0.6000
Kazakhstan	\$0.1350	\$0.0270	New Caledonia	\$0.3800	\$0.0760
Kenya	\$0.2700	\$0.0540	New Zealand	\$0.9000	\$0.0200
Kiribati	\$0.4050	\$0.0810	Nicaragua	\$0.1950	\$0.0390
Korea (North)	\$0.5850	\$0.1170	Niger Republic	\$0.3050	\$0.0610
Korea (South)	\$0.0900	\$0.0180	Nigeria	\$0.2950	\$0.0590
Kosovo	\$0.0700	\$0.0140	Niue	\$0.5650	\$0.1130
Kuwait	\$0.3900	\$0.0620	Norfolk Island	\$0.7950	\$0.1590
Kyrgyzstan	\$0.3500	\$0.0700	Norway	\$0.0700	\$0.0140
Laos	\$0.5050	\$0.1010	Oman	\$0.3050	\$0.0610
Latvia	\$0.2300	\$0.0460	Pakistan	\$0.3400	\$0.0680
Lebanon	\$0.2850	\$0.0570	Palau	\$0.3400	\$0.0680
Lesotho	\$0.2750	\$0.0550	Palestinian Authority	\$0.1950	\$0.0390
Liberia	\$0.2050	\$0.0410	Panama	\$0.1950	\$0.0390
Libya	\$0.2800	\$0.0560	Papua New Guinea	\$0.2250	\$0.0450
Liechtenstein	\$0.0700	\$0.0140	Paraguay	\$0.2050	\$0.0410
Lithuania	\$0.2800	\$0.0560	Peru	\$0.2050	\$0.0410
Luxembourg	\$0.0700	\$0.0140	Philippines	\$0.1500	\$0.0300
Macao	\$0.2550	\$0.0510	Poland	\$0.1350	\$0.0270
Macedonia	\$0.1950	\$0.0390	Portugal	\$0.0700	\$0.0140
Madagascar	\$0.9750	\$0.1950	Qatar	\$0.3900	\$0.0620
Malawi	\$0.2150	\$0.0430	Reunion Island	\$0.3400	\$0.0680
Malaysia	\$0.1400	\$0.0280	Romania	\$0.2050	\$0.0410

BUSINESS AND RESIDENTIAL SERVICES

Effective: September 1, 2019

Original Page 33

## PRICING AND SERVICE GUIDE SECTION 12 – GRANDFATHERED SERVICES

- 12.5 Type 4 Grandfathered Services (continued)
  - 12.5.2 Business International Advantage Plan (continued)
    - (E) The per-minute usage rates are as follows: (continued)

Country	Initial 30 Seconds	Each Additional 6 Seconds	Country	Initial 30 Seconds	Each Additional 6 Seconds
Russia	\$0.1350	\$0.0270	Taiwan	\$0.1150	\$0.0230
Rwanda, Republic of	\$0.3750	\$0.0750	Tajikistan	\$0.1650	\$0.0330
San Marino	\$0.2600	\$0.0520	Tanzania	\$0.2600	\$0.0520
Sao Tome	\$0.7150	\$0.1430	Thailand	\$0.1600	\$0.0320
Saudi Arabia	\$0.3250	\$0.0650	Togo	\$0.3850	\$0.0770
Senegal	\$0.3900	\$0.0780	Tokelau	\$1.4900	\$0.2820
Serbia	\$0.2250	\$0.0450	Tonga Island	\$0.5000	\$0.9000
Seychelles Island	\$0.4500	\$0.0900	Trinidad	\$0.2150	\$0.0430
Sierra Leone	\$0.3250	\$0.0650	Tunisia	\$0.1950	\$0.0390
Singapore	\$0.1150	\$0.0230	Turkey	\$0.1700	\$0.0340
Slovakia	\$0.1850	\$0.0370	Turkmenistan	\$0.3250	\$0.0650
Slovenia	\$0.1600	\$0.0320	Turks and Caicos Islands	\$0.1850	\$0.0370
Solomon Island	\$0.4500	\$0.0900	Tuvalu	\$0.8600	\$0.1720
Somalia Democratic Rep	\$0.5200	\$0.1040	Uganda	\$0.2550	\$0.0510
South Africa	\$0.1600	\$0.0320	Ukraine	\$0.2050	\$0.0410
Spain	\$0.0700	\$0.0140	United Arab Emirates	\$0.2050	\$0.0410
Sri Lanka	\$0.3250	\$0.0650	United Kingdom	\$0.0300	\$0.0060
St. Helena	\$0.3650	\$0.0730	Uruguay	\$0.2050	\$0.0410
St. Kitts/Nevis	\$0.1700	\$0.0340	Uzbekistan	\$0.2050	\$0.0410
St. Lucia	\$0.1750	\$0.0350	Vanuatu	\$1.0350	\$0.2070
St. Maarten	\$0.1500	\$0.0300	Vatican City	\$0.0700	\$0.0140
St. Pierre Miquelon	\$0.1550	\$0.0310	Venezuela	\$0.1500	\$0.0300
St. Vincent/Grenadines	\$0.1950	\$0.0390	Vietnam	\$0.3500	\$0.0700
South Sudan	\$0.3950	\$0.0790	Wallis Futuna	\$0.5850	\$0.1170
Sudan	\$0.3950	\$0.0790	Western Samoa	\$0.2950	\$0.0590
Suriname	\$0.3900	\$0.0780	Yemen Arab Republic	\$0.3200	\$0.0640
Swaziland	\$0.1850	\$0.0370	Zaire	\$0.2500	\$0.0500
Sweden	\$0.0700	\$0.0140	Zambia	\$0.2550	\$0.0510
Switzerland	\$0.0700	\$0.0140	Zimbabwe	\$0.2050	\$0.0410
Syrian Arab Republic	\$0.3350	\$0.0670			

### SECTION 12 – GRANDFATHERED SERVICES

- 12.5 Type 4 Grandfathered Services (continued)
  - 12.5.3 Business International Fixed Rate Plan (Grandfathered 10/1/2005)
    - (A) The Business International Fixed Rate plan is a direct dialed outbound international long distance service offered to Customers from switched access lines. Service is offered to each country on a flat rate basis, twenty-four hours per day, seven days per week. Customers must dial 1 plus the country code or area code, if applicable, and the terminating number to complete a call utilizing this service.

Original Page 34

Effective: September 1, 2019

- (B) All calls are billed in six (6) second increments after an initial period, for billing purpose, of thirty (30) seconds. This service is only offered in conjunction with any Company Business domestic service.
- (C) Inbound international toll-free service originated from locations within Canada is available to Customers of this service who also subscribe to AT&T Long Distance Toll Free<sup>SM</sup> Service in Section 4.4 of this pricing and service guide. The rate table in Section 12.5.3 (E) also contains the per-minute rates for inbound toll-free service originating in Canada and terminating throughout the United States (see Canada Inbound).
- (D) International Mobile Termination Charge

See Section 6.4.1 of this pricing and service guide for the applicable International Mobile Termination Charge.

Original Page 35

Effective: September 1, 2019

- 12.5 Type 4 Grandfathered Services (continued)
  - 12.5.3 Business International Fixed Rate Plan (continued)
    - (E) The MRC is \$31.95, and the per-minute usage rates are as follows:

		Each			Each
Country	Initial 30 Seconds	Additional	Country	Initial 30 Seconds	Additional
Afghanistan	\$3.040	\$0.608	Christmas Island	\$0.342	\$0.068
Albania	\$1.320	\$0.264	Cocos Island	\$0.342	\$0.068
Algeria	\$1.134	\$0.204	Colombia	\$0.880	\$0.176
Andorra	\$0.482	\$0.227	Comoros Islands	\$2.158	\$0.170
Angola	\$1.692	\$0.030	Congo	\$1.483	\$0.432
Anguilla	\$0.972	\$0.338	Cook Islands	\$2.787	\$0.297
Antarctica (Casey Base)	\$4.113	\$0.194	Costa Rica	\$0.741	\$0.337
Antarctica (Scott Base)	\$4.113	\$0.823	Croatia	\$0.741	\$0.148
Antigua	\$0.854	\$0.823	Cuba	\$2.112	\$0.190
Argentina	\$0.880	\$0.171	Cyprus	\$0.972	\$0.422
Armenia	\$1.436	\$0.170	Czech Republic	\$0.764	\$0.153
Aruba	\$0.741	\$0.287	Denmark	\$0.764	\$0.155
Ascension Island	\$1.623	\$0.148	Diego Garcia	\$3.273	\$0.655
Australia	\$0.342	\$0.068	Diibouti	\$1.716	\$0.033
Austria	\$0.342	\$0.068	Dominica	\$0.949	\$0.343
Azerbaijan	\$1.857	\$0.008	Dominican Republic	\$0.578	\$0.190
Bahamas	\$0.462	\$0.371	East Timor	\$5.599	\$1.120
Bahrain	\$1.228	\$0.092	East Timol Easter Island	\$0.693	\$0.139
Bangladesh			Ecuador	\$1.066	
Barbados	\$1.950 \$0.972	\$0.390 \$0.194		\$1.000	\$0.213 \$0.273
	\$1.112	\$0.194	Egypt El Salvador		\$0.273
Belarus Belgium	\$0.342	\$0.222	Equatorial Guinea	\$0.949 \$3.113	\$0.190
Belize	\$1.228	\$0.008	Eritrea	\$2.203	\$0.623
Benin	\$1.228	\$0.246	Estonia	\$0.995	\$0.441
Bermuda	\$0.600	\$0.218	Ethiopia Ethiopia	\$1.971	\$0.199
	\$2.368	\$0.120	Faeroe Island		
Bhutan Bolivia	\$1.041	\$0.474	Falkland Island	\$0.808 \$2.415	\$0.162 \$0.483
Bosnia	\$1.041	\$0.208	Fiji Island	\$1.692	\$0.463
Botswana	\$1.179	\$0.230	Finland	\$0.342	\$0.334
Brazil	\$0.741	\$0.208	France	\$0.342	\$0.068
British Virgin Islands	\$0.741	\$0.148	French Antilles	\$0.342	\$0.180
Brunei	\$1.228	\$0.133	French Guiana	\$1.161	\$0.180
Bulgaria	\$0.808	\$0.240	French Polynesia	\$1.833	\$0.232
Burkina Faso	\$1.347	\$0.162	Gabon	\$1.367	\$0.273
Burundi	\$2.647	\$0.209	Gambia	\$1.254	\$0.273
Cambodia	\$2.047	\$0.529	Gambia	\$1.234	\$0.231
Cameroon	\$1.347	\$0.269	Georgia	\$0.342	\$0.068
Canada - Inbound	\$0.158	\$0.209	Ghana	\$1.066	\$0.008
Canada - Outbound	\$0.158	\$0.032	Gibraltar	\$1.000	\$0.213
Cape Verde Island	\$1.390	\$0.032	Gibraitar Greece	\$1.320	
Cayman Islands	\$0.693	\$0.278	Greece Greenland	\$0.462	\$0.092
Central African Republic		\$0.139	Greenand Grenada	\$1.041	\$0.180 \$0.208
Chad	\$5.159	\$1.032	Grenada Guadeloupe	\$0.902	\$0.208
Chile	\$0.693	\$0.139	Guantanamo	\$2.996	\$0.180
China	\$0.902	\$0.180	Guatemala	\$0.949	\$0.190

Original Page 36

Effective: September 1, 2019

- 12.5 Type 4 Grandfathered Services (continued)
  - 12.5.3 Business International Fixed Rate Plan (continued)
    - (E) The per-minute usage rates are as follows: (continued)

		Each			Each
Country	Initial 30 Seconds	Additional	Country	Initial 30 Seconds	Additional
Guinea			Country	\$1.273	\$0.255
	\$1.390	\$0.278	Marshall Island	· ·	
Guinea-Bissau	\$1.879	\$0.376	Mauritania	\$1.670	\$0.334
Guyana	\$1.971	\$0.394	Mauritius	\$1.950	\$0.390
Haiti	\$1.273	\$0.255	Mayotte Island	\$1.739	\$0.348
Honduras	\$0.995	\$0.199	Mexico 1	\$0.342	\$0.068
Hong Kong	\$0.462	\$0.092	Mexico 2	\$0.578	\$0.116
Hungary	\$0.764	\$0.153	Micronesia	\$1.508	\$0.302
Iceland	\$0.808	\$0.162	Moldova	\$1.670	\$0.334
India	\$1.228	\$0.246	Monaco	\$0.342	\$0.068
Indonesia	\$0.972	\$0.194	Mongolia	\$2.787	\$0.557
Inmarsat (Atlantic Ocean-East)	\$19.186	\$3.837	Montenegro	\$1.161	\$0.232
Inmarsat (Atlantic Ocean-West)	\$19.186	\$3.837	Montserrat	\$1.041	\$0.208
Inmarsat (Indian Ocean)	\$19.186	\$3.837	Morocco	\$1.347	\$0.269
Inmarsat (SNAC)	\$19.186	\$3.837	Mozambique	\$1.739	\$0.343
Inmarsat (Pacific Ocean)	\$19.186	\$3.837	Myanmar	\$3.717	\$0.748
Iran	\$1.902	\$0.380	Namibia	\$1.179	\$0.236
Iraq	\$2.322	\$0.464	Nauru	\$2.484	\$0.497
Ireland	\$0.342	\$0.068	Nepal	\$1.785	\$0.357
Israel	\$0.531	\$0.106	Netherland Antilles	\$0.764	\$0.153
Italy	\$0.342	\$0.068	Netherlands	\$0.342	\$0.068
Ivory Coast	\$1.950	\$0.390	Networks	\$13.953	\$2.791
Jamaica	\$1.254	\$0.251	New Caledonia	\$1.950	\$0.390
Japan	\$0.342	\$0.068	New Zealand	\$0.505	\$0.101
Jordan	\$1.577	\$0.315	Nicaragua	\$0.995	\$0.199
Kazakhstan	\$0.693	\$0.139	Niger Republic	\$1.577	\$0.315
Kenya	\$1.390	\$0.278	Nigeria Nigeria	\$1.508	\$0.302
Kiribati	\$2.090	\$0.418	Niue	\$2.926	\$0.585
Korea (North)	\$3.018	\$0.604	Norfolk Island	\$4.113	\$0.823
Korea (South)	\$0.462	\$0.092	Norway	\$0.342	\$0.068
Kosovo	\$0.342	\$0.068	Oman	\$1.577	\$0.315
Kuwait	\$1.599	\$0.320	Pakistan	\$1.739	\$0.348
Kyrgyzstan	\$1.810	\$0.362	Palau	\$1.739	\$0.348
Laos	\$2.602	\$0.502	Palestinian Authority	\$0.902	\$0.180
Latvia	\$1.179	\$0.236	Panama	\$0.995	\$0.199
Lebanon	\$1.458	\$0.292	Papua New Guinea	\$1.161	\$0.232
Lesotho	\$1.412	\$0.282	Paraguay	\$1.066	\$0.232
Liberia	\$1.066	\$0.232	Peru	\$1.041	\$0.208
Libya	\$1.436	\$0.213	Philippines	\$0.764	\$0.208
Liechtenstein	\$0.342	\$0.267	Poland	\$0.704	\$0.133
	\$1.436	\$0.008		\$0.093	\$0.139
Lithuania			Portugal		
Luxembourg	\$0.342	\$0.068	Qatar	\$1.599	\$0.320
Macao	\$1.320	\$0.264	Reunion Island	\$1.739	\$0.348
Macedonia	\$0.995	\$0.199	Romania	\$1.041	\$0.208
Madagascar	\$5.042	\$1.008	Russia	\$0.693	\$0.139
Malawi	\$1.112	\$0.222	Rwanda, Republic of	\$1.927	\$0.385
Malaysia	\$0.717	\$0.143	San Marino	\$1.347	\$0.269
Maldives	\$2.136	\$0.427	Sao Tome	\$3.693	\$0.739
Mali Republic	\$1.810	\$0.362	Saudi Arabia	\$1.670	\$0.334
Malta Republic	\$0.949	\$0.190	Senegal	\$2.020	\$0.404

Original Page 37

Effective: September 1, 2019

- 12.5 Type 4 - Grandfathered Services (continued)
  - 12.5.3 Business International Fixed Rate Plan (continued)
    - (E) The per-minute usage rates are as follows: (continued)

		Each			Each
	Initial	Additional		Initial	Additional
Country	30 Seconds	6 Seconds	Country	30 Seconds	6 Seconds
Serbia	\$1.161	\$0.232	Thailand	\$0.834	\$0.167
Seychelles Island	\$2.322	\$0.464	Togo	\$1.971	\$0.394
Sierra Leone	\$1.670	\$0.334	Tokelau	\$7.274	\$1.455
Singapore	\$0.578	\$0.116	Tonga Island	\$2.578	\$0.516
Slovakia	\$0.949	\$0.190	Trinidad & Tobago	\$1.112	\$0.222
Slovenia	\$0.808	\$0.162	Tunisia	\$0.995	\$0.199
Solomon Island	\$2.322	\$0.464	Turkey	\$0.880	\$0.176
Somalia Democratic Rep	\$2.670	\$0.534	Turkmenistan	\$1.670	\$0.334
South Africa	\$0.834	\$0.167	Turks and Caicos Islands	\$0.949	\$0.190
South Sudan	\$2.042	\$0.408	Tuvalu	\$4.437	\$0.887
Spain	\$0.342	\$0.068	Uganda	\$1.320	\$0.264
Sri Lanka	\$1.670	\$0.334	Ukraine	\$1.041	\$0.208
St. Helena	\$1.879	\$0.376	United Arab Emirates	\$1.041	\$0.208
St. Kitts/Nevis	\$0.880	\$0.176	United Kingdom	\$0.158	\$0.032
St. Lucia	\$0.902	\$0.180	Uruguay	\$1.041	\$0.208
St. Maarten	\$0.764	\$0.153	Uzbekistan	\$1.041	\$0.208
St. Pierre Miquelon	\$0.787	\$0.157	Vanuatu	\$5.346	\$1.069
St. Vincent/Grenadines	\$0.995	\$0.199	Vatican City	\$0.342	\$0.068
Sudan	\$2.042	\$0.408	Venezuela	\$0.764	\$0.153
Suriname	\$2.020	\$0.404	Vietnam	\$1.810	\$0.362
Swaziland	\$0.949	\$0.190	Wallis Futuna	\$3.018	\$0.604
Sweden	\$0.342	\$0.068	Western Samoa	\$1.508	\$0.302
Switzerland	\$0.342	\$0.068	Yemen Arab Republic	\$1.645	\$0.329
Syrian Arab Republic	\$1.716	\$0.343	Zaire	\$1.298	\$0.260
Taiwan	\$0.578	\$0.116	Zambia	\$1.320	\$0.264
Tajikistan	\$0.854	\$0.171	Zimbabwe	\$1.041	\$0.208
Tanzania	\$1.347	\$0.269			

### SECTION 12 – GRANDFATHERED SERVICES

- 12.5 Type 4 Grandfathered Services (continued)
  - 12.5.4 Business 3500 Minute Package Plan (Grandfathered 9/30/2011)
    - (A) The Business 3500 Minute Package plan is a domestic direct dialed outbound and domestic inbound toll free (8XX) long distance service offered to business Customers from switched or dedicated access lines. Service is offered on a flat rate basis, twenty-four hours per day, seven days per week. For outbound calls Customers must dial 1 plus the area code, if applicable, and the terminating number to complete a call utilizing this service. The Business 3500 Minute Package plan provides the Customer with 3500 minutes of outbound and inbound interexchange (both interstate and intrastate) long distance usage for a flat rate monthly charge. Additional usage over the initial 3500 minutes will be billed on a flat rate per minute basis. No term discount is applicable to this service. The 3500 plan minutes cannot be used for calls to 900 services, international calling, directory assistance or operator services calls. To be eligible for this plan, Customers must presubscribe to the Company or the Company's Affiliated ILEC for intraLATA service and must also presubscribe to the Company for interLATA long distance service.

Original Page 38

- (B) To receive toll-free service Customers must subscribe to the AT&T Long Distance Toll Free<sup>SM</sup> Service. The provisions and rates and charges of the AT&T Long Distance Toll Free<sup>SM</sup> Service will apply as described in Section 4.4 of this pricing and service guide except that usage rates are specified in (D) below.
- (C) All calls are billed in one (1) minute increments after an initial period, for billing purposes, of one (1) minute. This service is not offered on an intraLATA only basis.
- (D) The MRC is \$920.00, and the per-minute usage rate is \$0.219 for calls completed after the 3,500 minute block of time has been used.

### SECTION 12 – GRANDFATHERED SERVICES

- 12.5 Type 4 Grandfathered Services (continued)
  - 12.5.5 Business Integrated Solutions 500 Minutes Plan (Grandfathered 9/30/2011)
    - (A) The Business Integrated Solutions 500 Minutes plan is a direct dialed domestic outbound and domestic inbound long distance service offered to business Customers on switched access lines. Service is offered on a flat rate basis, twenty-four hours per day, seven days per week. For outbound calls Customers must dial 1 plus the area code, if applicable, and the terminating number to complete a call utilizing this service. The Business Integrated Solutions 500 Minutes plan provides the Customer with 500 minutes of outbound and inbound interexchange (both interstate and intrastate) long distance usage per billing period for a flat rate monthly charge. Additional usage over the initial 500 minutes will be billed on a flat rate per minute basis. No term discount is applicable to this service. The 500 plan minutes cannot be used for calls to 900 services, international calling, directory assistance or operator services calls.

Original Page 39

- (B) Customers of this plan must subscribe to one of the following services offered by the Company's Affiliated ILEC: BellSouth® Integrated Solutions, ISDN and to a local service term agreement from the Company's Affiliated ILEC.
- (C) Customers who no longer meet the eligibility requirements identified above will be transferred to Business Message Telecommunications Service without further notice until the Customer selects another Company service.
- (D) All calls are billed in one (1) minute increments after an initial period, for billing purposes, of one (1) minute. This service is not offered on an intraLATA only basis.
- (E) To receive toll-free service (inbound) Customers must subscribe to the AT&T Long Distance Toll Free<sup>SM</sup> Service. The provisions and rates and charges of the AT&T Long Distance Toll Free<sup>SM</sup> Service will apply as described in Section 4.4 of this pricing and service guide except that usage rates are specified in (F) below.
- (F) The MRC is \$122.50, and the per-minute usage rate is \$0.235 for calls completed after the 500 minute block of time has been used.

### SECTION 12 – GRANDFATHERED SERVICES

- 12.5 Type 4 Grandfathered Services (continued)
  - 12.5.6 Business Integrated Solutions 1000 Minutes Plan (Grandfathered 9/30/2011)
    - (A) The Business Integrated Solutions 1000 Minutes plan is a direct dialed domestic outbound and domestic inbound long distance service offered to business Customers on switched access lines. Service is offered on a flat rate basis, twenty-four hours per day, seven days per week. For outbound calls Customers must dial 1 plus the area code, if applicable, and the terminating number to complete a call utilizing this service. The Business Integrated Solutions 1000 Minutes plan provides the Customer with 1000 minutes of outbound and inbound interexchange (both interstate and intrastate) long distance usage per billing period for a flat rate monthly charge. Additional usage over the initial 1000 minutes will be billed on a flat rate per minute basis. No term discount is applicable to this service. The 1000 plan minutes cannot be used for calls to 900 services, international calling, directory assistance or operator services calls.

Original Page 40

- (B) Customers of this plan must subscribe to one of the following services offered by the Company's Affiliated ILEC: BellSouth® Integrated Solutions, BellSouth® Centrex Service, BellSouth® Dedicated Internet Access or BellSouth® Primary Rate ISDN and to a local service term agreement from the Company's Affiliated ILEC.
- (C) Customers who no longer meet the eligibility requirements identified above will be transferred to Business Message Telecommunications Service without further notice until the Customer selects another Company service.
- (D) All calls are billed in one (1) minute increments after an initial period, for billing purposes, of one (1) minute. This service is not offered on an intraLATA only basis.
- (E) To receive toll-free service (inbound) Customers must subscribe to the AT&T Long Distance Toll Free<sup>SM</sup> Service. The provisions and rates and charges of the AT&T Long Distance Toll Free<sup>SM</sup> will apply as described in Section 4.4 of this pricing and service guide except that usage rates are specified (F) below.
- (F) The MRC is \$234.00, and the per-minute usage rate is \$0.223 for calls completed after the 1,000 minute block of time has been used.

### SECTION 12 – GRANDFATHERED SERVICES

- 12.5 Type 4 Grandfathered Services (continued)
  - 12.5.7 Business 7500 Minute Package Plan (Grandfathered 9/30/2011)
    - (A) The Business 7500 Minute Package plan is a direct dialed domestic outbound and domestic inbound long distance service offered to business Customers on switched access lines. Service is offered on a flat rate basis, twenty-four hours per day, seven days per week. For outbound calls Customers must dial 1 plus the area code, if applicable, and the terminating number to complete a call utilizing this service. The Business 7500 Minute Package plan provides the Customer with 7,500 minutes of outbound and inbound interexchange (both interstate and intrastate) long distance usage for a flat rate monthly charge. Additional usage over the initial 7,500 minutes will be billed on a flat rate per minute basis. No term discount is applicable to this service. The 7,500 plan minutes cannot be used for calls to 900 services, international calling, directory assistance or operator services calls.

Original Page 41

- (B) To be eligible for this plan Customers must subscribe to local exchange service provided by the Company's Affiliated ILEC.
- (C) All calls are billed in one (1) minute increments after an initial period, for billing purposes, of one (1) minute. This service is not offered on an intraLATA only basis.
- (D) To receive toll-free service (inbound) Customers must subscribe to the AT&T Long Distance Toll Free<sup>SM</sup> Service. The provisions and rates and charge of the AT&T Long Distance Toll Free<sup>SM</sup> Service will apply as described in Section 4.4 of this pricing and service guide except that usage rates are specified in (E) below.
- (E) The MRC is \$1,812.00, and the per-minute usage rate is \$0.203 for calls completed after the 7,500 block of time has been used.

### SECTION 12 – GRANDFATHERED SERVICES

- 12.5 Type 4 Grandfathered Services (continued)
  - 12.5.8 Business 10000 Minute Package Plan (Grandfathered 9/30/2011)
    - (A) The Business 10000 Minute Package plan is a direct dialed domestic outbound and domestic inbound long distance service offered to business Customers on switched access lines. Service is offered on a flat rate basis, twenty-four hours per day, seven days per week. For outbound calls Customers must dial 1 plus the area code, if applicable, and the terminating number to complete a call utilizing this service. The Business 10000 Minute Package plan provides the Customer with 10,000 minutes of outbound and inbound interexchange (both interstate and intrastate) long distance usage for a flat rate monthly charge. Additional usage over the initial 10,000 minutes will be billed on a flat rate per minute basis. No term discount is applicable to this service. The 10,000 plan minutes cannot be used for calls to 900 services, international calling, directory assistance or operator services calls.

Original Page 42

- (B) To be eligible for this plan Customers must subscribe to local exchange service provided by the Company's Affiliated ILEC.
- (C) All calls are billed in one (1) minute increments after an initial period, for billing purposes, of one (1) minute. This service is not offered on an intraLATA only basis.
- (D) To receive toll-free service (inbound) Customers must subscribe to the AT&T Long Distance Toll Free<sup>SM</sup> Service. The provisions and rates and charge of the AT&T Long Distance Toll Free<sup>SM</sup> Service will apply as described in Section 4.4 of this pricing and service guide except that usage rates are specified in (E) below.
- (E) The MRC is \$2,226.50, and the per-minute usage rate is \$0.185 for calls completed after the 10,000 block of time has been used.

### SECTION 12 – GRANDFATHERED SERVICES

- 12.5 Type 4 Grandfathered Services (continued)
  - 12.5.9 Business 500 Minute Package Plan (Grandfathered 9/30/2011)
    - (A) The Business 500 Minute Package plan is a direct dialed long distance service offered to business Customers from switched or dedicated access lines. Service is offered on a flat rate basis, twenty-four hours per day, seven days per week. Customers must dial 1 plus the area code, if applicable, and the terminating number to complete a call utilizing this service. The Business 500 Minute Package plan provides the Customer with 500 minutes of interexchange (both interstate and intrastate) long distance usage for a flat rate monthly charge. Additional usage over the initial 500 minutes will be billed on a flat rate per minute basis. No term discount is applicable to this service. The 500 plan minutes cannot be used for calls to 900 services, international calling, directory assistance or operator services calls. To be eligible for this plan, Customers must presubscribe to the Company or the Company's Affiliated ILEC for intraLATA service and must also presubscribe to the Company for interLATA long distance service.

Original Page 43

- (B) All calls are billed in one (1) minute increments after an initial period, for billing purposes, of one (1) minute. This service is not offered on an intraLATA only basis.
- (C) The MRC is \$151.00, and the per-minute usage rate is \$0.263 for calls completed after the 500 block of time has been used.

### SECTION 12 – GRANDFATHERED SERVICES

12.5 Type 4 - Grandfathered Services (continued)

12.5.10 Business 1000 Minute Package Plan (Grandfathered 9/30/2011)

(A) The Business 1000 Minute Package plan is a direct dialed long distance service offered to business Customers from switched or dedicated access lines. Service is offered on a flat rate basis, twenty-four hours per day, seven days per week. Customers must dial 1 plus the area code, if applicable, and the terminating number to complete a call utilizing this service. The Business 1000 Minute Package plan provides the Customer with 1000 minutes of interexchange (both interstate and intrastate) long distance usage for a flat rate monthly charge. Additional usage over the initial 1000 minutes will be billed on a flat rate per minute basis. No term discount is applicable to this service. The 1000 plan minutes cannot be used for calls to 900 services, international calling, directory assistance or operator services calls. In order to be eligible for this plan, Customers must presubscribe to the Company or the Company's Affiliated ILEC for intraLATA service and must also presubscribe to the Company for interLATA long distance service.

Original Page 44

- (B) All calls are billed in one (1) minute increments after an initial period, for billing purposes, of one (1) minute. This service is not offered on an intraLATA only basis.
- (C) The MRC is \$290.00, and the per-minute usage rate is \$0.263 for calls completed after the 1000 block of time has been used.

### SECTION 12 – GRANDFATHERED SERVICES

- 12.5 Type 4 Grandfathered Services (continued)
  - 12.5.11 Business 2000 Minute Package Plan (Grandfathered 9/30/2011)
    - (A) The Business 2000 Minute Package plan is a direct dialed long distance service offered to business Customers from switched or dedicated access lines. Service is offered on a flat rate basis, twenty-four hours per day, seven days per week. Customers must dial 1 plus the area code, if applicable, and the terminating number to complete a call utilizing this service. The Business 2000 Minute Package plan provides the Customer with 2000 minutes of interexchange (both interstate and intrastate) long distance usage for a flat rate monthly charge. Additional usage over the initial 2000 minutes will be billed on a flat rate per minute basis. No term discount is applicable to this service. The 2000 plan minutes cannot be used for calls to 900 services, international calling, directory assistance or operator services calls. To be eligible for this plan, Customers must presubscribe to the Company or the Company's Affiliated ILEC for intraLATA service and must also presubscribe to the Company for interLATA long distance service.

Original Page 45

- (B) All calls are billed in one (1) minute increments after an initial period, for billing purposes, of one (1) minute. This service is not offered on an intraLATA only basis.
- (C) The MRC is \$541.50, and the per-minute usage rate is \$0.245 for calls completed after the 2000 block of time has been used.

### SECTION 12 – GRANDFATHERED SERVICES

- 12.5 Type 4 Grandfathered Services (continued)
  - 12.5.12 Business 5000 Minute Package Plan (Grandfathered 9/30/2011)
    - (A) The Business 5000 Minute Package plan is a direct dialed long distance service offered to business Customers from switched or dedicated access lines. Service is offered on a flat rate basis, twenty-four hours per day, seven days per week. Customers must dial 1 plus the area code, if applicable, and the terminating number to complete a call utilizing this service. The Business 5000 Minute Package plan provides the Customer with 5000 minutes of interexchange (both interstate and intrastate) long distance usage for a flat rate monthly charge. Additional usage over the initial 5000 minutes will be billed on a flat rate per minute basis. No term discount is applicable to this service. The 5000 plan minutes cannot be used for calls to 900 services, international calling, directory assistance or operator services calls. To be eligible for this plan, Customers must presubscribe to the Company or the Company's Affiliated ILEC for intraLATA service and must also presubscribe to the Company for interLATA long distance service.

Original Page 46

- (B) All calls are billed in one (1) minute increments after an initial period, for billing purposes, of one (1) minute. This service is not offered on an intraLATA only basis.
- (C) The MRC is \$1,253.50, and the per-minute usage rate is \$0.219 for calls completed after the 5000 block of time has been used.

### SECTION 12 – GRANDFATHERED SERVICES

- 12.5 Type 4 Grandfathered Services (continued)
  - 12.5.13 Business Integrated Solutions 2000 Minutes Plan (Grandfathered 9/30/2011)
    - (A) The Business Integrated Solutions 2000 Minutes plan is a direct dialed outbound long distance service offered to business Customers from switched access lines. Service is offered on a flat rate basis, twenty-four hours per day, seven days per week. Customers must dial 1 plus the area code, if applicable, and the terminating number to complete a call utilizing this service. The Business Integrated Solutions 2000 Minutes plan provides the Customer with 2,000 minutes of Additional usage over the initial 2,000 minutes will be billed on a flat rate per minute basis. No term discount is applicable to this service. The 2000 plan minutes cannot be used for calls to 900 services, international calling, directory assistance or operator services calls.

Original Page 47

- (B) Customers of this plan must subscribe to one of the following services offered by the Company's Affiliated ILECy: BellSouth® Integrated Solutions, BellSouth® Centrex Service, BellSouth® Dedicated Internet Access or BellSouth® Primary ISDN and to a local service term agreement from the Company's Affiliated ILEC. Customers who no longer meet the eligibility requirements identified above will be transferred to Business Message Telecommunications Service without further notice until the Customer selects another Company service.
- (C) All calls are billed in one (1) minute increments after an initial period, for billing purposes, of one (1) minute. This service is not offered on an intraLATA only basis.
- (D) The MRC is \$449.00, and the per-minute usage rate is \$0.215 for calls completed after the 2,000 minute block of time has been used.

### SECTION 12 – GRANDFATHERED SERVICES

- 12.5 Type 4 Grandfathered Services (continued)
  - 12.5.14 Business Integrated Solutions 5000 Minutes Plan (Grandfathered 9/30/2011)
    - (A) The Business Integrated Solutions 5000 Minutes plan is a direct dialed outbound long distance service offered to business Customers from switched access lines. Service is offered on a flat rate basis, twenty-four hours per day, seven days per week. Customers must dial 1 plus the area code, if applicable, and the terminating number to complete a call utilizing this service. The Business Integrated Solutions 5000 Minutes plan provides the Customer with 5,000 minutes of interexchange (both interstate and intrastate) long distance usage for a flat rate monthly charge. Additional usage over the initial 5,000 minutes will be billed on a flat rate per minute basis. No term discount is applicable to this service. The 5000 plan minutes cannot be used for calls to 900 services, international calling, directory assistance or operator services calls.

Original Page 48

- (B) Customers of this plan must subscribe to one of the following services offered by the Company's Affiliated: BellSouth® Integrated Solutions, BellSouth® Centrex Service, BellSouth® Dedicated Internet Access or BellSouth® Primary ISDN and to a local service term agreement from the Company's Affiliated. Customers who no longer meet the eligibility requirements identified above will be transferred to Business Message Telecommunications Service without further notice until the Customer selects another Company service.
- (C) All calls are billed in one (1) minute increments after an initial period, for billing purposes, of one (1) minute. This service is not offered on an intraLATA only basis.
- (D) The MRC is \$1,114.00, and the per-minute usage rate is \$0.203 for calls completed after the 5,000 minute block of time has been used.

### SECTION 12 – GRANDFATHERED SERVICES

- 12.5 Type 4 Grandfathered Services (continued)
  - 12.5.15 Business Integrated Solutions 3500 Minutes Plan (Grandfathered 9/30/2011)
    - (A) The Business Integrated Solutions 3500 Minutes plan is a direct dialed domestic outbound and domestic inbound long distance service offered to business Customers on switched access lines. Service is offered on a flat rate basis, twenty-four hours per day, seven days per week. For outbound calls Customers must dial 1 plus the area code, if applicable, and the terminating number to complete a call utilizing this service. The Business Integrated Solutions 3500 Minutes plan provides the Customer with 3500 minutes of outbound and inbound interexchange (both interstate and intrastate) long distance usage per billing period for a flat rate monthly charge. Additional usage over the initial 3500 minutes will be billed on a flat rate per minute basis. No term discount is applicable to this service. The 3500 plan minutes cannot be used for calls to 900 services, international calling, directory assistance or operator services calls.

Original Page 49

- (B) Customers of this plan must subscribe to one of the following services offered by the Company's Affiliated ILEC: BellSouth® Integrated Solutions, BellSouth® Centrex Service, BellSouth® Dedicated Internet Access or BellSouth® Primary Rate ISDN and to a local service term agreement from the Company's Affiliated ILEC. Customers who no longer meet the eligibility requirements identified above will be transferred to an alternate plan without further notice.
- (C) All calls are billed in one (1) minute increments after an initial period, for billing purposes, of one (1) minute. This service is not offered on an intraLATA only basis.
- (D) To receive toll-free service (inbound) Customers must subscribe to the AT&T Long Distance Toll Free<sup>SM</sup> Service. The provisions and rates and charges of the AT&T Long Distance Toll Free<sup>SM</sup> Service will apply as described in Section 4.4 of this pricing and service guide except that usage rates are specified (E) below:
- (E) The MRC is \$778.50, and the per-minute usage rate is \$0.207 for calls completed after the 3,500 minute block of time has been used.

#### SECTION 12 – GRANDFATHERED SERVICES

#### 12.5 Type 4 – Grandfathered Services (continued)

- 12.5.16 Business Preferred Country Plan (Grandfathered 12/31/2011)
  - (A) The Business Preferred Country plan is a direct dialed outbound international long distance service offered to Business Customers from switched access lines. Service is offered to each country on a flat rate basis, twenty-four hours per day, seven days per week. Customers must dial 1 plus the country code or area code, if applicable, and the terminating number to complete a call utilizing this service.

Original Page 50

Effective: September 1, 2019

- (B) All calls are billed in six (6) second increments after an initial period, for billing purposes, of thirty (30) seconds. This service is only offered in conjunction with any of the Company's Business Domestic service.
- (C) This plan allows Business Customers to designate one international country for which they will receive a discounted per minute rate. The Customer designates his/her preferred country at the time the plan is selected and may change the preferred country at any time. The Company reserves the right to limit the number of times the preferred country is changed within a given period of time. Calls to non-preferred countries will be billed at the regular rate defined in Section 12.5.16 (E).
- (D) Inbound international toll-free service originated from locations within Canada is available to Customers of this service who also subscribe to AT&T Long Distance Toll Free<sup>SM</sup> in Section 4.4 of this pricing and service guide. The rate table in Section 12.5.16 (E) also contains the per minute rates for AT&T Long Distance Toll Free<sup>SM</sup> Service originating in Canada and terminating throughout the United States.
- (E) Rates and Charges

The MRC is \$3.85, and this charge is waived once a Customer's international usage exceeds \$50.00 for the current billing month. Usage may not be carried to the next billing cycle to meet the waiver requirement.

Original Page 51

Effective: September 1, 2019

### SECTION 12 – GRANDFATHERED SERVICES

- 12.5 Type 4 Grandfathered Services (continued)
  - 12.5.16 Business Preferred Country Plan (continued)
    - (E) Rates and Charges (continued)

The per-minute usage rates are as follows:

	Preferred (	Country Rate	Regular (	Country Rate
	Initial	Each Add'l	Initial	Each Add'l
Country	30 Seconds		30 Seconds	
Afghanistan	\$0.3250	\$0.0650	\$0.3700	\$0.0740
Albania	\$0.4800	\$0.0960	\$0.5500	\$0.1100
Algeria	\$0.3000	\$0.0600	\$0.3450	\$0.0690
Andorra	\$0.1950	\$0.0390	\$0.2250	\$0.0450
Angola	\$0.6450	\$0.1290	\$0.7350	\$0.1470
Anguilla	\$0.2750	\$0.0550	\$0.3150	\$0.0630
Antarctica (Casey Base)	\$0.8100	\$0.1620	\$0.9300	\$0.1860
Antarctica (Scott Base)	\$0.8100	\$0.1620	\$0.9300	\$0.1860
Antigua	\$0.2300	\$0.0460	\$0.2600	\$0.0520
Argentina	\$0.1050	\$0.0210	\$0.1250	\$0.0250
Armenia	\$0.4100	\$0.0820	\$0.4700	\$0.0940
Aruba	\$0.1950	\$0.0390	\$0.2250	\$0.0450
Ascension Island	\$0.4850	\$0.0970	\$0.5550	\$0.1110
Australia	\$0.0750	\$0.0150	\$0.0900	\$0.0180
Austria	\$0.0650	\$0.0130	\$0.0750	\$0.0150
Azerbaijan	\$0.5350	\$0.1070	\$0.6100	\$0.1220
Bahamas	\$0.0900	\$0.0180	\$0.1050	\$0.0210
Bahrain	\$0.3500	\$0.0700	\$0.4000	\$0.0800
Bangladesh	\$0.4650	\$0.0930	\$0.5300	\$0.1060
Barbados	\$0.1550	\$0.0310	\$0.1850	\$0.0370
Belarus	\$0.3150	\$0.0630	\$0.3600	\$0.0720
Belgium	\$0.0650	\$0.0130	\$0.0750	\$0.0150
Belize	\$0.2350	\$0.0470	\$0.2700	\$0.0540
Benin	\$0.2850	\$0.0570	\$0.3300	\$0.0660
Bermuda	\$0.0900	\$0.0180	\$0.1050	\$0.0210
Bhutan	\$0.8000	\$0.1600	\$0.9150	\$0.1830
Bolivia	\$0.2000	\$0.0400	\$0.2300	\$0.0460
Bosnia	\$0.3450	\$0.0690	\$0.3950	\$0.0790
Botswana	\$0.3550	\$0.0710	\$0.4050	\$0.0810
Brazil	\$0.0900	\$0.0180	\$0.1050	\$0.0210
British Virgin Islands	\$0.1950	\$0.0390	\$0.2250	\$0.0450
Brunei	\$0.3650	\$0.0730	\$0.4200	\$0.0840
Bulgaria	\$0.2800	\$0.0560	\$0.3200	\$0.0640
Burkina Faso	\$0.3900	\$0.0780	\$0.4500	\$0.0900
Burundi	\$0.7650	\$0.1530	\$0.8750	\$0.1750
Cambodia	\$0.9800	\$0.1960	\$1.1200	\$0.2240
Cameroon	\$0.4350	\$0.0870	\$0.4950	\$0.0990
Canada Outbound	\$0.0300	\$0.0060	\$0.0350	\$0.0070
Canada Inbound	\$0.0300	\$0.0060	\$0.0350	\$0.0070
Cape Verde Island	\$0.3550	\$0.0710	\$0.4050	\$0.0810
Cayman Islands	\$0.1650	\$0.0330	\$0.1950	\$0.0390
Central African Republic	\$0.6550	\$0.1310	\$0.7500	\$0.1500
Chad	\$1.0700	\$0.2140	\$1.2200	\$0.2440
Chile	\$0.0850	\$0.0170	\$0.1000	\$0.0200
China	\$0.0600	\$0.0120	\$0.0700	\$0.0140

Original Page 52

Effective: September 1, 2019

### SECTION 12 – GRANDFATHERED SERVICES

- 12.5 Type 4 Grandfathered Services (continued)
  - 12.5.16 Business Preferred Country Plan (continued)
    - (E) Rates and Charges (continued)

	Preferred (	Country Rate	Regular C	ountry Rate
	Initial	Each Add'l	Initial	Each Add'l
Country	30 Seconds	6 Seconds	30 Seconds	
Christmas Island	\$0.0750	\$0.0150	\$0.0900	\$0.0180
Cocos Island	\$0.0750	\$0.0150	\$0.0900	\$0.0180
Colombia	\$0.1050	\$0.0210	\$0.1250	\$0.0250
Comoros Islands	\$1.5400	\$0.3080	\$1.7600	\$0.3520
Congo	\$0.4850	\$0.0970	\$0.5550	\$0.1110
Cook Islands	\$0.7200	\$0.1440	\$0.8250	\$0.1650
Costa Rica	\$0.1100	\$0.0220	\$0.1300	\$0.0260
Croatia	\$0.2400	\$0.0480	\$0.2750	\$0.0550
Cuba	\$0.5000	\$0.1000	\$0.5850	\$0.1170
Cyprus	\$0.2550	\$0.0510	\$0.2900	\$0.0580
Czech Republic	\$0.2450	\$0.0490	\$0.2800	\$0.0560
Denmark	\$0.0650	\$0.0130	\$0.0750	\$0.0150
Diego Garcia	\$0.7250	\$0.1450	\$0.8300	\$0.1660
Djibouti	\$0.4250	\$0.0850	\$0.4850	\$0.0970
Dominican Republic	\$0.0700	\$0.0140	\$0.0800	\$0.0160
Dominica	\$0.2350	\$0.0470	\$0.2700	\$0.0540
East Timor	\$0.9650	\$0.1930	\$1.2050	\$0.2410
Easter Island	\$0.0850	\$0.0170	\$0.1000	\$0.0200
Ecuador	\$0.1700	\$0.0340	\$0.2000	\$0.0400
Egypt	\$0.3300	\$0.0660	\$0.3750	\$0.0750
El Salvador	\$0.1100	\$0.0220	\$0.1300	\$0.0260
Equatorial Guinea	\$1.1250	\$0.2250	\$1.2900	\$0.2580
Eritrea	\$0.5450	\$0.1090	\$0.6200	\$0.1240
Estonia	\$0.3950	\$0.0790	\$0.4550	\$0.0910
Ethiopia	\$0.4750	\$0.0950	\$0.5450	\$0.1090
Faeroe Island	\$0.2600	\$0.0520	\$0.3000	\$0.0600
Falkland Island	\$0.6450	\$0.1290	\$0.7350	\$0.1470
Fiji Island	\$0.4450	\$0.0890	\$0.5100	\$0.1020
Finland	\$0.0650	\$0.0130	\$0.0750	\$0.0150
France	\$0.0700	\$0.0140	\$0.0800	\$0.0160
French Antilles	\$0.2800	\$0.0560	\$0.3200	\$0.0640
French Guiana	\$0.2800	\$0.0560	\$0.3200	\$0.0640
French Polynesia	\$0.5750	\$0.1150	\$0.6600	\$0.1320
Gabon	\$0.4750	\$0.0950	\$0.5450	\$0.1090
Gambia	\$0.2950	\$0.0590	\$0.3400	\$0.0680
Georgia	\$0.4500	\$0.0900	\$0.5150	\$0.1030
Germany	\$0.0750	\$0.0150	\$0.0900	\$0.0180
Ghana	\$0.2950	\$0.0590	\$0.3400	\$0.0680
Gibraltar	\$0.4450	\$0.0890	\$0.5100	\$0.1020
Greece	\$0.0650	\$0.0130	\$0.0750	\$0.0150
Greenland	\$0.2800	\$0.0560	\$0.3200	\$0.0640
Grenada	\$0.2450	\$0.0490	\$0.2800	\$0.0560
Guadeloupe	\$0.2750	\$0.0550	\$0.3150	\$0.0630
Guantanamo	\$0.6700	\$0.1340	\$0.7650	\$0.1530
Guatemala	\$0.2250	\$0.0450	\$0.2650	\$0.0530
Guinea	\$0.3550	\$0.0710	\$0.4050	\$0.0810

Original Page 53

Effective: September 1, 2019

### SECTION 12 – GRANDFATHERED SERVICES

- 12.5 Type 4 Grandfathered Services (continued)
  - 12.5.16 Business Preferred Country Plan (continued)
    - (E) Rates and Charges (continued)

	Preferred C	ountry Rate	Regular Country Rate		
	Initial	Each Add'l	Initial	Each Add'l	
Country	30 Seconds	6 Seconds	30 Seconds	6 Seconds	
Guinea-Bissau	\$0.7200	\$0.1440	\$0.8250	\$0.1650	
Guyana	\$0.4300	\$0.0860	\$0.4900	\$0.0980	
Haiti	\$0.1900	\$0.0380	\$0.2250	\$0.0450	
Honduras	\$0.2400	\$0.0480	\$0.2800	\$0.0560	
Hong Kong	\$0.0350	\$0.0070	\$0.0400	\$0.0080	
Hungary	\$0.1800	\$0.0360	\$0.2100	\$0.0420	
Iceland	\$0.2250	\$0.0450	\$0.2550	\$0.0510	
India	\$0.1400	\$0.0280	\$0.1650	\$0.0330	
Indonesia	\$0.1200	\$0.0240	\$0.1350	\$0.0270	
Inmarsat (Atlantic Ocean-East)	\$3.9000	\$0.7800	\$4.4550	\$0.8910	
Inmarsat (Atlantic Ocean-West)	\$3.9000	\$0.7800	\$4.4550	\$0.8910	
Inmarsat (Indian Ocean)	\$3.9000	\$0.7800	\$4.4550	\$0.8910	
Inmarsat (Pacific Ocean)	\$3.9000	\$0.7800	\$4.4550	\$0.8910	
Inmarsat (SNAC)	\$3.9000	\$0.7800	\$4.4550	\$0.8910	
Iran	\$0.5500	\$0.1100	\$0.6300	\$0.1260	
Iraq	\$0.5250	\$0.1050	\$0.6000	\$0.1200	
Ireland	\$0.0650	\$0.0130	\$0.0750	\$0.0150	
Israel	\$0.0750	\$0.0150	\$0.0900	\$0.0180	
Italy	\$0.0750	\$0.0150	\$0.0900	\$0.0180	
Ivory Coast	\$0.5400	\$0.1080	\$0.6150	\$0.1230	
Jamaica	\$0.1900	\$0.0380	\$0.2200	\$0.0440	
Japan	\$0.0350	\$0.0070	\$0.0400	\$0.0080	
Jordan	\$0.3900	\$0.0780	\$0.4500	\$0.0900	
Kazakhstan	\$0.0800	\$0.0160	\$0.0950	\$0.0190	
Kenya	\$0.3700	\$0.0740	\$0.4250	\$0.0850	
Kiribati	\$0.5750	\$0.1150	\$0.6600	\$0.1320	
Korea (North)	\$1.3300	\$0.2660	\$1.5200	\$0.3040	
Korea (South)	\$0.0450	\$0.0090	\$0.0550	\$0.0110	
Kosovo	\$0.0650	\$0.0130	\$0.0750	\$0.0150	
Kuwait	\$0.4050	\$0.0810	\$0.4650	\$0.0930	
Kyrgyzstan	\$0.4800	\$0.0960	\$0.5500	\$0.1100	
Laos	\$1.0600	\$0.2120	\$1.2100	\$0.2420	
Latvia	\$0.4050	\$0.0810	\$0.4650	\$0.0930	
Lebanon	\$0.5250	\$0.1050	\$0.6000	\$0.1200	
Lesotho	\$0.3700	\$0.0740	\$0.4250	\$0.0850	
Liberia	\$0.2800	\$0.0560	\$0.3200	\$0.0640	
Libya	\$0.4200	\$0.0840	\$0.4800	\$0.0960	
Liechtenstein	\$0.0650	\$0.0130	\$0.0750	\$0.0150	
Lithuania	\$0.5250	\$0.1050	\$0.6000	\$0.1200	
Luxembourg	\$0.0650	\$0.1030	\$0.0000	\$0.1200	
Macao	\$0.3350	\$0.0130	\$0.3800	\$0.0130	
Macedonia	\$0.2550	\$0.0070	\$0.2900	\$0.0780	
Madagascar	\$0.2330	\$0.0310	\$1.0550	\$0.0380	
Malawi	\$0.9200	\$0.1640	\$0.3600	\$0.2110	
Malaysia	\$0.3130		\$0.3600		
Maldives	\$0.6300	\$0.0130 \$0.1260	\$0.0730	\$0.0150 \$0.1440	
Manures	φυ.υσυυ	φυ.1200	φυ./200	φυ.1 <del>44</del> 0	

Original Page 54

Effective: September 1, 2019

### SECTION 12 – GRANDFATHERED SERVICES

- 12.5 Type 4 Grandfathered Services (continued)
  - 12.5.16 Business Preferred Country Plan (continued)
    - (E) Rates and Charges (continued)

	Preferred (	Country Rate	Regular Country Rate		
	Initial	Each Add'l	Initial	Each Add'l	
Country	30 Seconds	6 Seconds	30 Seconds	6 Seconds	
Mali Republic	\$0.4550	\$0.0910	\$0.5200	\$0.1040	
Malta Republic	\$0.3150	\$0.0630	\$0.3600	\$0.0720	
Marshall Island	\$0.3200	\$0.0640	\$0.3650	\$0.0730	
Mauritania	\$0.4300	\$0.0860	\$0.4900	\$0.0980	
Mauritius	\$0.5400	\$0.1080	\$0.6150	\$0.1230	
Mayotte Island	\$0.4350	\$0.0870	\$0.4950	\$0.0990	
Mexico 1	\$0.0500	\$0.0100	\$0.0600	\$0.0120	
Mexico 2	\$0.0750	\$0.0150	\$0.0900	\$0.0180	
Micronesia	\$0.3400	\$0.0680	\$0.3900	\$0.0780	
Moldova	\$0.5350	\$0.1070	\$0.6100	\$0.1220	
Monaco	\$0.0650	\$0.0130	\$0.0750	\$0.0150	
Mongolia	\$0.7400	\$0.1480	\$0.8450	\$0.1690	
Montenegro	\$0.2850	\$0.0570	\$0.3300	\$0.0660	
Montserrat	\$0.2350	\$0.0470	\$0.2700	\$0.0540	
Morocco	\$0.3550	\$0.0710	\$0.4050	\$0.0810	
Mozambique	\$0.5500	\$0.1100	\$0.6300	\$0.1260	
Myanmar	\$0.9350	\$0.1870	\$1.0700	\$0.2140	
Namibia	\$0.3450	\$0.0690	\$0.3950	\$0.0790	
Nauru	\$0.5600	\$0.1120	\$0.6400	\$0.1280	
Nepal	\$0.5600	\$0.1120	\$0.6400	\$0.1280	
Netherland Antilles	\$0.1800	\$0.0360	\$0.2100	\$0.0420	
Netherlands	\$0.0700	\$0.0140	\$0.0800	\$0.0160	
Networks	\$3.0000	\$0.6000	\$3.3000	\$0.6600	
New Caledonia	\$0.6300	\$0.1260	\$0.7200	\$0.1440	
New Zealand	\$0.1150	\$0.0230	\$0.1300	\$0.0260	
Nicaragua	\$0.2350	\$0.0470	\$0.2750	\$0.0550	
Niger Republic	\$0.3950	\$0.0790	\$0.4550	\$0.0910	
Nigeria	\$0.3550	\$0.0710	\$0.4200	\$0.0840	
Niue	\$0.8050	\$0.1610	\$0.9200	\$0.1840	
Norfolk Island	\$0.8100	\$0.1620	\$0.9300	\$0.1860	
Norway	\$0.0650	\$0.0130	\$0.0750	\$0.0150	
Oman	\$0.4550	\$0.0910	\$0.5200	\$0.1040	
Pakistan	\$0.2950	\$0.0590	\$0.3500	\$0.0700	
Palau	\$0.5800	\$0.1160	\$0.6650	\$0.1330	
Palestinian Authority	\$0.1550	\$0.0310	\$0.1950	\$0.0390	
Panama	\$0.1600	\$0.0320	\$0.1850	\$0.0370	
Papua New Guinea	\$0.4500	\$0.0900	\$0.5150	\$0.1030	
Paraguay	\$0.2700	\$0.0540	\$0.3100	\$0.0620	
Peru	\$0.1000	\$0.0200	\$0.1200	\$0.0240	
Philippines	\$0.1450	\$0.0290	\$0.1750	\$0.0350	
Poland	\$0.1700	\$0.0340	\$0.1950	\$0.0390	
Portugal	\$0.0650	\$0.0130	\$0.0750	\$0.0150	
Qatar	\$0.4550	\$0.0910	\$0.5200	\$0.1040	
Reunion Island	\$0.4350	\$0.0870	\$0.4950	\$0.0990	
Romania	\$0.3700	\$0.0740	\$0.4250	\$0.0850	

Original Page 55

Effective: September 1, 2019

### SECTION 12 – GRANDFATHERED SERVICES

- 12.5 Type 4 Grandfathered Services (continued)
  - 12.5.16 Business Preferred Country Plan (continued)
    - (E) Rates and Charges (continued)

	<b>Preferred Country Rate</b>		Regular C	Country Rate
	Initial	Each Add'l	Initial	Each Add'l
Country	30 Seconds	6 Seconds	30 Seconds	
Russia	\$0.0800	\$0.0160	\$0.0950	\$0.0190
Rwanda, Republic of	\$0.5100	\$0.1020	\$0.5850	\$0.1170
San Marino	\$0.3350	\$0.0670	\$0.3800	\$0.0760
Sao Tome	\$0.7400	\$0.1480	\$0.8450	\$0.1690
Saudi Arabia	\$0.4000	\$0.0800	\$0.4600	\$0.0920
Senegal	\$0.6200	\$0.1240	\$0.7100	\$0.1420
Serbia	\$0.2850	\$0.0570	\$0.3300	\$0.0660
Seychelles Island	\$0.6500	\$0.1300	\$0.7400	\$0.1480
Sierra Leone	\$0.4350	\$0.0870	\$0.4950	\$0.0990
Singapore	\$0.1300	\$0.0260	\$0.1500	\$0.0300
Slovakia	\$0.2700	\$0.0540	\$0.3100	\$0.0620
Slovenia	\$0.2550	\$0.0510	\$0.2900	\$0.0580
Solomon Island	\$0.6300	\$0.1260	\$0.7200	\$0.1440
Somalia Democratic Rep	\$0.7200	\$0.1440	\$0.8250	\$0.1650
South Africa	\$0.1700	\$0.0340	\$0.2000	\$0.0400
Spain	\$0.0700	\$0.0140	\$0.0800	\$0.0160
Sri Lanka	\$0.4650	\$0.0930	\$0.5300	\$0.1060
St. Helena	\$0.5450	\$0.1090	\$0.6200	\$0.1240
St. Kitts/Nevis	\$0.2300	\$0.0460	\$0.2600	\$0.0520
St. Lucia	\$0.2300	\$0.0460	\$0.2600	\$0.0520
St. Maarten	\$0.1800	\$0.0360	\$0.2100	\$0.0420
St. Pierre Miquelon	\$0.2400	\$0.0480	\$0.2750	\$0.0550
St. Vincent/Grenadines	\$0.2700	\$0.0540	\$0.3100	\$0.0620
South Sudan	\$0.5950	\$0.1190	\$0.6800	\$0.1360
Sudan	\$0.5950	\$0.1190	\$0.6800	\$0.1360
Suriname	\$0.5400	\$0.1080	\$0.6150	\$0.1230
Swaziland	\$0.3300	\$0.0660	\$0.3750	\$0.0750
Sweden	\$0.0650	\$0.0130	\$0.0750	\$0.0150
Switzerland	\$0.0650	\$0.0130	\$0.0750	\$0.0150
Syrian Arab Republic	\$0.5450	\$0.1090	\$0.6200	\$0.1240
Taiwan	\$0.0350	\$0.0070	\$0.0400	\$0.0080
Tajikistan	\$0.1900	\$0.0380	\$0.2200	\$0.0440
Tanzania	\$0.4400	\$0.0880	\$0.5000	\$0.1000
Thailand	\$0.1200	\$0.0240	\$0.1350	\$0.0270
Togo	\$0.4750	\$0.0950	\$0.5450	\$0.1090
Tokelau	\$0.7200	\$0.1440	\$0.8250	\$0.1650
Tonga Island	\$0.6750	\$0.1350	\$0.7700	\$0.1540
Trinidad	\$0.2150	\$0.0430	\$0.2500	\$0.0500
Tunisia	\$0.3400	\$0.0680	\$0.3900	\$0.0780
Turkey	\$0.2050	\$0.0410	\$0.2450	\$0.0490
Turkmenistan	\$0.4650	\$0.0930	\$0.5300	\$0.1060

Original Page 56

Effective: September 1, 2019

### SECTION 12 – GRANDFATHERED SERVICES

### 12.5 Type 4 – Grandfathered Services (continued)

### 12.5.16 Business Preferred Country Plan (continued)

### (E) Rates and Charges (continued)

The per-minute usage rates are as follows: (continued)

	Preferred (	Country Rate	<b>Regular Country Rate</b>		
	Initial	Each Add'l	Initial	Each Add'l	
Country	30 Seconds	6 Seconds	30 Seconds	6 Seconds	
Turks and Caicos Islands	\$0.2550	\$0.0510	\$0.2900	\$0.0580	
Tuvalu	\$0.7950	\$0.1590	\$0.9100	\$0.1820	
Uganda	\$0.3500	\$0.0700	\$0.4000	\$0.0800	
Ukraine	\$0.1250	\$0.0250	\$0.1400	\$0.0280	
United Arab Emirates	\$0.2600	\$0.0520	\$0.3000	\$0.0600	
United Kingdom	\$0.0350	\$0.0070	\$0.0400	\$0.0080	
Uruguay	\$0.2700	\$0.0540	\$0.3100	\$0.0620	
Uzbekistan	\$0.2000	\$0.0400	\$0.2300	\$0.0460	
Vanuatu	\$1.1200	\$0.2240	\$1.2800	\$0.2560	
Vatican City	\$0.0950	\$0.0190	\$0.1100	\$0.0220	
Venezuela	\$0.1100	\$0.0220	\$0.1300	\$0.0260	
Vietnam	\$0.2700	\$0.0540	\$0.3100	\$0.0620	
Wallis Futuna	\$1.2600	\$0.2520	\$1.4400	\$0.2880	
Western Samoa	\$0.4450	\$0.0890	\$0.5100	\$0.1020	
Yemen Arab Republic	\$0.4550	\$0.0910	\$0.5200	\$0.1040	
Zaire	\$0.3400	\$0.0680	\$0.3900	\$0.0780	
Zambia	\$0.3650	\$0.0730	\$0.4200	\$0.0840	
Zimbabwe	\$0.3400	\$0.0680	\$0.3900	\$0.0780	

### (F) International Mobile Termination Charge

See Section 6.4.1 of this pricing and service guide for the applicable International Mobile Termination Charge.

#### SECTION 12 – GRANDFATHERED SERVICES

#### 12.5 Type 4 – Grandfathered Services (continued)

### 12.5.17 AT&T Business International Calling<sup>SM</sup> III (Grandfathered 12/31/2011)

(A) AT&T Business International Calling<sup>SM</sup> III is a direct dialed outbound international long distance service offered to Customers on switched access lines. Service is offered to each country on a flat rate basis, twenty-four hours per day, seven days per week. Customers must dial 1 plus the country code or area code, if applicable, and the terminating number to complete a call utilizing this service.

Original Page 57

Effective: September 1, 2019

- (B) All calls are billed in one (1) minute increments after an initial period, for billing purposes, of one (1) minute. This service is only offered in conjunction with any of the Company's Business Domestic service.
- (C) Inbound international toll-free service originated from locations within Canada is available to Customers of this service who also subscribe to AT&T Long Distance Toll Free<sup>SM</sup> Service in Section 4.4 of this pricing and service guide. The rate table in Section 12.5.17 (E) also contains the per-minute rate for AT&T Long Distance Toll Free<sup>SM</sup> Service originating in Canada and terminating throughout the United States (see Canada Inbound).
- (D) International Mobile Termination Charge
   See Section 6.4.1 of this pricing and service guide for the applicable International Mobile Termination Charge.
- (E) Rates and Charges

The MRC is \$13.00, and the per-minute usage rates are as follows:

Country	Initial Minute	Each Additional Minute	Country	Initial Minute	Each Additional Minute
Afghanistan	\$4.780	\$4.780	Bhutan	\$11.945	\$11.945
Albania	\$7.154	\$7.154	Bolivia	\$3.015	\$3.015
Algeria	\$4.505	\$4.505	Bosnia-Herzegovina	\$5.109	\$5.109
Andorra	\$2.965	\$2.965	Botswana	\$5.293	\$5.293
Angola	\$9.576	\$9.576	Brazil	\$1.668	\$1.668
Anguilla	\$4.130	\$4.130	British Virgin Islands	\$2.965	\$2.965
Antarctica (Casey Base)	\$12.083	\$12.083	Brunei	\$5.482	\$5.482
Antarctica (Scott Base)	\$12.083	\$12.083	Bulgaria	\$4.179	\$4.179
Antigua	\$3.383	\$3.383	Burkina Faso	\$5.852	\$5.852
Argentina	\$1.989	\$1.989	Burundi	\$11.386	\$11.386
Armenia	\$6.128	\$6.128	Cambodia	\$14.547	\$14.547
Aruba	\$2.965	\$2.965	Cameroon	\$6.453	\$6.453
Ascension Island	\$7.249	\$7.249	Canada - Outbound	\$0.549	\$0.549
Australia	\$1.155	\$1.155	Canada - Inbound	\$0.549	\$0.549
Austria	\$1.009	\$1.009	Cape Verde Island	\$5.293	\$5.293
Azerbaijan	\$7.900	\$7.900	Cayman Islands	\$3.015	\$3.015
Bahamas	\$1.668	\$1.668	Central African Republic		\$9.761
Bahrain	\$5.203	\$5.203	Chad	\$15.903	
Bangladesh	\$6.918	\$6.918	Chile	\$1.573	\$1.573
Barbados	\$2.872	\$2.872	China	\$1.061	\$1.061
Belarus	\$4.688	\$4.688	Christmas Island	\$1.155	\$1.155
Belgium	\$1.009	\$1.009	Cocos Island	\$1.155	\$1.155
Belize	\$3.529	\$3.529	Colombia	\$1.989	\$1.989
Benin	\$4.271	\$4.271	Comoros Islands	\$22.925	
Bermuda	\$1.386	\$1.386	Congo	\$7.249	\$7.249

Original Page 58

Effective: September 1, 2019

### SECTION 12 – GRANDFATHERED SERVICES

- 12.5 Type 4 Grandfathered Services (continued)
  - 12.5.17 AT&T Business International Calling<sup>SM</sup> III (continued)
    - (E) Rates and Charges (continued)

Country	Initial Minute	Each Additional Minute	Country	Initial Minute	Each Additional Minute
Cook Islands	\$10.785	\$10.785	Hungary	\$3.290	\$3.290
Costa Rica	\$2.132	\$2.132	Iceland	\$3.339	\$3.339
Croatia	\$3.570	\$3.570	India	\$2.693	\$2.693
Cuba	\$9.389	\$9.389	Indonesia	\$1.759	\$1.759
Cyprus	\$3.803	\$3.803	Inmarsat (Atlantic Ocean-East)	\$58.049	\$58.049
Czech Republic	\$3.619	\$3.619	Inmarsat (Atlantic Ocean-West)		\$58.049
Denmark	\$1.009	\$1.009	Inmarsat (Indian Ocean)	\$58.049	\$58.049
Diego Garcia	\$10.828	\$10.828	Inmarsat (SNAC)	\$58.049	\$58.049
Djibouti	\$6.316	\$6.316	Inmarsat (Pacific Ocean)	\$58.049	\$58.049
Dominica	\$3.529	\$3.529	Iran	\$8.178	\$8.178
Dominican Republic	\$1.334	\$1.334	Iraq	\$7.807	\$7.807
East Timor	\$13.436	\$13.436	Ireland	\$1.009	\$1.009
Easter Island	\$1.573	\$1.573	Israel	\$1.434	\$1.434
Ecuador	\$3.246	\$3.246	Italy	\$1.434	\$1.434
Egypt	\$4.922	\$4.922	Ivory Coast	\$8.037	\$8.037
El Salvador	\$1.989	\$1.989	Jamaica	\$3.570	\$3.570
Equatorial Guinea	\$16.781	\$16.781	Japan	\$0.497	\$0.497
Eritrea	\$8.086	\$8.086	Jordan	\$5.852	\$5.852
			Kazakhstan	\$1.200	
Estonia	\$5.894	\$5.894			\$1.200
Ethiopia	\$7.062	\$7.062	Kenya	\$5.525	\$5.525
Faeroe Island	\$3.899	\$3.899	Kiribati	\$8.593	\$8.593
Falkland Island	\$9.576	\$9.576	Korea (North)	\$19.806	\$19.806
Fiji Island	\$6.638	\$6.638	Korea (South)	\$0.874	\$0.874
Finland	\$1.009	\$1.009	Kosovo	\$1.009	\$1.009
France	\$1.293	\$1.293	Kuwait	\$6.080	\$6.080
French Antilles	\$4.179	\$4.179	Kyrgyzstan	\$7.154	\$7.154
French Guiana	\$4.179	\$4.179	Laos	\$15.718	\$15.718
French Polynesia	\$8.593	\$8.593	Latvia	\$6.080	\$6.080
Gabon	\$7.062	\$7.062	Lebanon	\$7.807	\$7.807
Gambia	\$4.406	\$4.406	Lesotho	\$5.525	\$5.525
Georgia	\$6.693	\$6.693	Liberia	\$4.179	\$4.179
Germany	\$1.434	\$1.434	Libya	\$6.226	\$6.226
Ghana	\$4.406	\$4.406	Liechtenstein	\$1.009 \$7.807	\$1.009 \$7.807
Gibraltar	\$6.638 \$1.009	\$6.638 \$1.009	Lithuania Luxembourg	\$1.009	\$1.009
Greece Greenland	\$4.179	\$4.179	Macao	\$4.968	\$4.968
Grenada	\$3.619	\$3.619	Macedonia	\$3.803	\$3.803
Guadeloupe	\$4.130	\$4.130	Madagascar	\$13.712	\$13.712
Guantanamo Bay	\$9.992	\$9.992	Malawi	\$4.688	\$4.688
Guatemala	\$4.083	\$4.083	Malaysia	\$1.009	\$1.009
Guinea	\$5.293	\$5.293	Maldives	\$9.389	\$9.389
Guinea-Bissau	\$10.785	\$10.785	Mali Republic	\$6.736	\$6.736
Guyana	\$6.361	\$6.361	Malta Republic	\$4.688	\$4.688
Haiti	\$3.529	\$3.529	Marshall Island	\$4.736	\$4.736
Honduras	\$4.456	\$4.456	Mauritania	\$6.361	\$6.361
Hong Kong	\$0.497	\$0.497	Mauritius	\$8.037	\$8.037

Original Page 59

Effective: September 1, 2019

### SECTION 12 – GRANDFATHERED SERVICES

- 12.5 Type 4 Grandfathered Services (continued)
  - 12.5.17 AT&T Business International Calling<sup>SM</sup> III (continued)
    - (E) Rates and Charges (continued)

Country	Initial Minute	Each Additional Minute	Country	Initial Minute	Each Additional Minute
Mayotte Island	\$6.453	\$6.453	Russia	\$1.200	\$1.200
Mexico 1	\$1.009	\$1.009	Rwanda, Republic of	\$7.618	\$7.618
Mexico 2	\$1.434	\$1.434	San Marino	\$4.968	\$4.968
Micronesia	\$5.061	\$5.061	Sao Tome	\$10.973	\$10.973
Moldova	\$7.900	\$7.900	Saudi Arabia	\$5.943	\$5.943
Monaco	\$1.009	\$1.009	Senegal	\$9.246	\$9.246
Mongolia	\$10.973	\$10.973	Serbia	\$4.271	\$4.271
Montenegro	\$4.271	\$4.271	Seychelles Island	\$9.668	\$9.668
Montserrat	\$3.529	\$3.529	Sierra Leone	\$6.453	\$6.453
Morocco	\$5.293	\$5.293	Singapore	\$1.943	\$1.943
Mozambique	\$8.178	\$8.178	Slovakia	\$3.995	\$3.995
Myanmar	\$13.948	\$13.948	Slovenia	\$3.803	\$3.803
Namibia	\$5.109	\$5.109	Solomon Island	\$9.389	\$9.389
Nauru	\$8.316	\$8.316	Somalia Democratic Rep	\$10.785	\$10.785
Nepal	\$8.316	\$8.316	South Africa	\$3.057	\$3.057
Netherland Antilles	\$3.290	\$3.290	South Sudan	\$8.973	\$8.973
Netherlands	\$1.293	\$1.293	Spain	\$1.293	\$1.293
Networks	\$33.490	\$33.490	Sri Lanka	\$6.918	\$6.918
New Caledonia	\$9.389	\$9.389	St. Helena	\$8.086	\$8.086
New Zealand	\$1.668	\$1.668	St. Kitts/Nevis	\$3.383	\$3.383
Nicaragua	\$4.271	\$4.271	St. Lucia	\$3.383	\$3.383
Niger Republic	\$5.894	\$5.894	St. Maarten	\$3.290	\$3.290
Nigeria	\$6.506	\$6.506	St. Pierre Miquelon	\$3.570	\$3.570
Niue	\$11.994	\$11.994	St. Vincent/Grenadines	\$3.995	\$3.995
Norfolk Island	\$12.083	\$12.083	Sudan	\$8.873	\$8.873
Norway	\$1.009	\$1.009	Suriname	\$8.037	\$8.037
Oman	\$6.736	\$6.736	Swaziland	\$4.922	\$4.922
Pakistan	\$5.390	\$5.390	Sweden	\$1.009	\$1.009
Palau	\$8.642	\$8.642	Switzerland	\$1.009	\$1.009
Palestinian Authority	\$2.180	\$2.180	Syrian Arab Republic	\$8.086	\$8.086
Panama	\$3.015	\$3.015	Taiwan	\$0.497	\$0.497
Papua New Guinea	\$6.693	\$6.693	Tajikistan	\$2.824	\$2.824
Paraguay	\$3.995	\$3.995	Tanzania	\$6.506	\$6.506
Peru	\$1.897	\$1.897	Thailand	\$1.759	\$1.759
Philippines	\$2.693	\$2.693	Togo	\$7.062	\$7.062
Poland	\$2.546	\$2.546	Tokelau	\$10.785	\$10.785
Portugal	\$1.009	\$1.009	Tonga Island	\$10.039	\$10.039
Qatar	\$6.736	\$6.736	Trinidad & Tobago	\$3.995	\$3.995
Reunion Island	\$6.453	\$6.453	Tunisia	\$5.061	\$5.061
Romania	\$5.525	\$5.525	Turkey	\$3.803	\$3.803

Original Page 60

Effective: September 1, 2019

#### SECTION 12 – GRANDFATHERED SERVICES

- 12.5 Type 4 Grandfathered Services (continued)
  - 12.5.17 AT&T Business International Calling<sup>SM</sup> III (continued)
    - (E) Rates and Charges (continued)

The per-minute usage rates are as follows: (continued)

Country	Initial Minute	Each Additional Minute	Country	Initial Minute	Each Additiona Minute
Turkmenistan	\$6.918	\$6.918	Vatican City	\$1.434	\$1.434
Turks and Caicos Islands	\$3.803	\$3.803	Venezuela	\$2.132	\$2.132
Tuvalu	\$11.808	\$11.808	Vietnam	\$3.995	\$3.995
Uganda	\$5.203	\$5.203	Wallis Futuna	\$18.740	\$18.740
Ukraine	\$1.854	\$1.854	Western Samoa	\$6.638	\$6.638
United Arab Emirates	\$3.899	\$3.899	Yemen Arab Republic	\$6.736	\$6.736
United Kingdom	\$0.641	\$0.641	Zaire	\$5.061	\$5.061
Uruguay	\$3.995	\$3.995	Zambia	\$5.482	\$5.482
Uzbekistan	\$3.015	\$3.015	Zimbabwe	\$5.061	\$5.061
Vanuatu	\$16.692	\$16.692			

#### 12.5.18 Business Worldwide Plan (Grandfathered 12/31/2011)

- (A) The Business Worldwide plan is a direct dialed outbound international long distance service offered to Customers on switched access lines. Service is offered to each country on a flat rate basis, twenty-four hours per day, seven days per week. Customers must dial 1 plus the country code or area code, if applicable, and the terminating number to complete a call utilizing this service.
- (B) All calls are billed in six (6) second increments after an initial period, for billing purposes, of thirty (30) seconds. This service is only offered in conjunction with any of the Company's Business Domestic service.
- (C) Inbound international toll-free service originated from locations within Canada is available to Customers of this service who also subscribe to AT&T Long Distance Toll Free<sup>SM</sup> Service in Section 4.4 of this pricing and service guide. The rate table in Section 12.5.18 (E) also contains the per-minute rate for AT&T Long Distance Toll Free<sup>SM</sup> Service originating in Canada and terminating throughout the United States (Canada Inbound).
- (D) International Mobile Termination Charge

See Section 6.4.1 of this pricing and service guide for the applicable International Mobile Termination Charge.

### SECTION 12 – GRANDFATHERED SERVICES

### 12.5 Type 4 – Grandfathered Services (continued)

### 12.5.18 Business Worldwide Plan (continued)

### (E) Rates and Charges

### .1 MRC

The MRC is \$46.95 and is waived once a Customer's international usage exceeds \$50.00 for the current billing month. Usage may not be carried to the next billing cycle to meet the waiver requirement.

Original Page 61

Effective: September 1, 2019

.2 The per-minute usage rates are as follows:

Country         Jo Seconds of Seconds of Seconds         Country         Initial of Seconds of Seconds         Additional 30 Seconds of Seconds of Seconds           Afghanistan         \$1.347         \$0.269         Central African Republic seconds         \$2.741         \$0.548           Albania         \$2.020         \$0.404         Chile         \$0.367         \$0.073           Andorra         \$0.834         \$0.167         Chiile         \$0.367         \$0.073           Angolilla         \$1.161         \$0.232         Chiile         \$0.321         \$0.064           Anguilla         \$1.161         \$0.232         Cocos Island         \$0.321         \$0.064           Antarctica (Casey Base)         \$3.414         \$0.683         Compros Island         \$0.321         \$0.064           Antarctica (Casey Base)         \$3.414         \$0.683         Compros Islands         \$6.485         \$1.297           Antarctica (Scott Base)         \$3.414         \$0.082         Cook Islands         \$0.412         \$0.082           Argentina         \$0.1273         \$0.082         Cook Islands         \$3.040         \$0.694           Arustia         \$0.321         \$0.064         Cyroatia         \$0.495         \$0.995           Australia         \$0.321 <th></th> <th>-</th> <th></th> <th></th> <th></th> <th></th>		-				
Country         30 Seconds         6 Seconds           Afghanistan         \$1.347         \$0.269           Albania         \$2.020         \$0.404           Algeria         \$1.273         \$0.255           Andorra         \$0.834         \$0.167           Angola         \$2.695         \$0.539           Angulla         \$1.161         \$0.232           Angulla         \$1.161         \$0.232           Antarctica (Casey Base)         \$3.414         \$0.683           Antarctica (Scott Base)         \$3.414         \$0.683           Antigua         \$0.949         \$0.190           Argentina         \$0.412         \$0.082           Argentina         \$1.739         \$0.348           Accension Island         \$2.042         \$0.408           Ascension Island         \$2.042         \$0.408           Australia         \$0.321         \$0.064           Austria         \$0.275         \$0.055           Aerbaijan         \$2.228         \$0.446           Austria         \$0.275         \$0.055           Bahrain         \$1.458         \$0.292           Barbados         \$0.667         \$0.133           Belgium         <		T *** 1	Each		T 1	Each
Afghanistan         \$1.347         \$0.269         Central African Republic         \$2.741         \$0.548           Albania         \$2.020         \$0.404         Chald         \$4.482         \$0.896           Algeria         \$1.273         \$0.255         Chile         \$0.367         \$0.073           Andorra         \$0.834         \$0.167         China         \$0.249         \$0.050           Angoulla         \$2.695         \$0.539         Christmas Island         \$0.321         \$0.064           Anguilla         \$1.161         \$0.232         Cocos Island         \$0.321         \$0.064           Antarctica (Casey Base)         \$3.414         \$0.683         Colombia         \$0.412         \$0.082           Antarctica (Scott Base)         \$3.414         \$0.683         Comoros Island         \$0.412         \$0.082           Argentina         \$0.412         \$0.082         Cook Islands         \$0.412         \$0.048           Argentina         \$1.739         \$0.342         \$0.408         Cook Islands         \$3.040         \$0.698           Arustalia         \$0.321         \$0.064         Cypus         \$1.066         \$0.213           Australia         \$0.321         \$0.067         \$0.133	C			C		
Albania         \$2.020         \$0.404         Chad         \$4.482         \$0.896           Algeria         \$1.273         \$0.255         Chile         \$0.367         \$0.073           Angola         \$2.695         \$0.539         China         \$0.249         \$0.050           Angolla         \$1.161         \$0.232         Cocos Island         \$0.321         \$0.064           Antarctica (Casey Base)         \$3.414         \$0.683         Colombia         \$0.412         \$0.082           Antarctica (Scott Base)         \$3.414         \$0.683         Comoros Islands         \$6.485         \$1.297           Antigua         \$0.949         \$0.190         Cook Islands         \$0.412         \$0.082           Argentina         \$0.412         \$0.082         Cook Islands         \$0.404         \$0.408           Arruba         \$0.834         \$0.167         Coka Rica         \$0.437         \$0.087           Austriala         \$0.275         \$0.055         Cota Rica         \$0.437         \$0.087           Azerbaijan         \$2.228         \$0.446         Cyprus         \$1.066         \$0.213           Bahamas         \$0.342         \$0.068         Cyprus         \$1.066         \$0.213						
Algeria						
Andorra         \$0.834         \$0.167         China         \$0.249         \$0.050           Angouila         \$2.695         \$0.539         Christmas Island         \$0.321         \$0.064           Anguilla         \$1.161         \$0.232         Cocos Island         \$0.321         \$0.064           Antarctica (Casey Base)         \$3.414         \$0.683         Colombia         \$0.412         \$0.082           Antigua         \$0.949         \$0.190         Colombia         \$0.412         \$0.082           Ariguina         \$0.412         \$0.082         Coors Islands         \$6.485         \$1.297           Ariguina         \$0.412         \$0.082         Coorgo         \$2.042         \$0.408           Arremaia         \$1.739         \$0.348         Cots Islands         \$3.040         \$0.082           Acerbaijan         \$0.231         \$0.064         Covatia         \$0.995         \$0.199           Austriai         \$0.275         \$0.055         Cocach Republic         \$1.020         \$0.204           Austriai         \$1.458         \$0.292         \$0.0667         \$0.133         \$0.667         \$0.133         \$0.667         \$0.133         \$0.667         \$0.133         \$0.667         \$0.133         \$						
Angola         \$2.695         \$0.539         Christmas Island         \$0.321         \$0.064           Anguilla         \$1.161         \$0.232         Cocos Island         \$0.321         \$0.064           Antarctica (Casey Base)         \$3.414         \$0.683         Combia         \$0.412         \$0.082           Antarctica (Scott Base)         \$3.414         \$0.683         Combia         \$0.412         \$0.082           Antigua         \$0.949         \$0.190         Comoros Islands         \$6.485         \$1.297           Argentina         \$0.412         \$0.082         Compoo         \$2.042         \$0.408           Armenia         \$1.739         \$0.348         Costa Rica         \$0.437         \$0.087           Aruba         \$0.834         \$0.167         Costa Rica         \$0.437         \$0.087           Austria         \$0.275         \$0.055         Coceh Republic         \$1.020         \$0.204           Azerbaijan         \$2.228         \$0.044         Cyprus         \$1.066         \$0.213           Bahrain         \$1.458         \$0.292         Dijbouti         \$1.765         \$0.353           Belarus         \$1.320         \$0.054         East Timor         \$5.599         \$1.120						
Anguilla         \$1.161         \$0.232         Cocos Island         \$0.321         \$0.064           Antarctica (Casey Base)         \$3.414         \$0.683         Colombia         \$0.412         \$0.082           Antigua         \$0.949         \$0.190         Compore Islands         \$6.485         \$1.297           Antigua         \$0.949         \$0.190         Compo         \$2.042         \$0.408           Armenia         \$1.739         \$0.348         \$0.167         Congo         \$2.042         \$0.408           Aruba         \$0.834         \$0.167         Croatia         \$0.995         \$0.199           Ascension Island         \$2.042         \$0.408         Costa Rica         \$0.437         \$0.087           Austriai         \$0.321         \$0.064         Cyprus         \$1.066         \$0.213           Austriai         \$0.321         \$0.064         Cyprus         \$1.066         \$0.213           Austriai         \$0.322         \$0.044         Eyprus         \$1.066         \$0.213           Bahrain         \$1.458         \$0.292         Eenmark         \$0.275         \$0.055           Belarius         \$1.320         \$0.390         East Timor         \$5.599         \$1.120						
Antarctica (Casey Base)         \$3.414         \$0.683         Colombia         \$0.412         \$0.082           Antarctica (Scott Base)         \$3.414         \$0.683         Comoros Islands         \$6.485         \$1.297           Antigua         \$0.949         \$0.190         Congo         \$2.042         \$0.408           Argentina         \$1.739         \$0.348         Cook Islands         \$3.040         \$0.608           Armenia         \$1.739         \$0.348         Cook Islands         \$3.040         \$0.608           Armenia         \$1.739         \$0.348         Costa Rica         \$0.437         \$0.087           Aruba         \$0.834         \$0.167         Croatia         \$0.995         \$0.199           Ascension Island         \$2.042         \$0.408         Cuba         \$2.203         \$0.417           Austria         \$0.275         \$0.055         Croatia         \$0.995         \$0.199           Azerbaijan         \$2.228         \$0.446         Denmark         \$0.275         \$0.055           Bahamas         \$0.342         \$0.068         Denmark         \$0.275         \$0.055           Barghados         \$1.950         \$0.390         \$0.790         \$0.275         \$0.055						
Antarctica (Scott Base)         \$3.414         \$0.683         Comoros Islands         \$6.485         \$1.297           Antigua         \$0.949         \$0.190         \$0.082         \$0.402         \$0.408           Argentina         \$1.739         \$0.348         \$0.167         \$0.085         \$0.437         \$0.088           Aruba         \$0.834         \$0.167         \$0.408         \$0.437         \$0.087           Ascension Island         \$2.042         \$0.408         \$0.203         \$0.441           Australia         \$0.321         \$0.064         \$0.203         \$0.441           Austria         \$0.275         \$0.055         \$0.055         \$0.204           Azerbaijan         \$2.228         \$0.446         \$0.275         \$0.055           Bahamas         \$0.342         \$0.068         \$0.275         \$0.055           Bahrain         \$1.458         \$0.292         \$0.667         \$0.339           Barpadosh         \$1.950         \$0.390         \$0.078         \$0.972         \$0.194           Belgium         \$0.275         \$0.055         \$0.455         \$0.455         \$0.667         \$0.133           Benin         \$1.206         \$0.241         \$0.942         \$0.942						
Antigua         \$0.949         \$0.190         Congo         \$2.042         \$0.408           Argentina         \$0.412         \$0.082         Cook Islands         \$3.040         \$0.608           Armenia         \$1.739         \$0.348         Costa Rica         \$0.437         \$0.087           Aruba         \$0.834         \$0.167         Croatia         \$0.995         \$0.199           Ascension Island         \$2.042         \$0.408         Cuba         \$2.203         \$0.441           Australia         \$0.321         \$0.064         Cyprus         \$1.066         \$0.213           Australia         \$0.2275         \$0.055         Czech Republic         \$1.020         \$0.204           Azerbaijan         \$2.228         \$0.446         Denmark         \$0.275         \$0.055           Bahamas         \$0.342         \$0.068         Diego Garcia         \$3.064         \$0.613           Bahrain         \$1.458         \$0.292         Dijbouti         \$1.765         \$0.353           Barbados         \$0.667         \$0.133         Dominican Republic         \$0.275         \$0.055           Belarus         \$1.320         \$0.264         East Timor         \$5.599         \$1.120						
Argentina         \$0.412         \$0.082         Cook Islands         \$3.040         \$0.608           Armenia         \$1.739         \$0.348         Costa Rica         \$0.437         \$0.087           Aruba         \$0.834         \$0.167         Croatia         \$0.995         \$0.199           Ascension Island         \$2.042         \$0.408         Cuba         \$2.203         \$0.441           Australia         \$0.321         \$0.064         Cyprus         \$1.066         \$0.213           Austria         \$0.275         \$0.055         Czech Republic         \$1.020         \$0.204           Azerbaijan         \$2.228         \$0.446         Denmark         \$0.275         \$0.055           Bahamas         \$0.342         \$0.068         Diego Garcia         \$3.064         \$0.613           Bahrain         \$1.458         \$0.292         Dominica         \$0.972         \$0.194           Barbados         \$0.667         \$0.133         Dominica         \$0.972         \$0.194           Belgium         \$0.275         \$0.055         East Timor         \$5.599         \$1.120           Bermuda         \$0.390         \$0.078         El Salvador         \$0.462         \$0.092           Ber						
Armenia         \$1.739         \$0.348         Costa Rica         \$0.437         \$0.087           Aruba         \$0.834         \$0.167         Croatia         \$0.995         \$0.199           Ascension Island         \$2.042         \$0.408         Cuba         \$2.203         \$0.441           Austriia         \$0.321         \$0.064         Cyprus         \$1.066         \$0.213           Austriia         \$0.275         \$0.055         Czech Republic         \$1.020         \$0.204           Azerbaijan         \$2.228         \$0.446         Denmark         \$0.275         \$0.055           Bahrain         \$1.458         \$0.292         Djibouti         \$1.765         \$0.353           Barbados         \$0.667         \$0.390         Dominica         \$0.972         \$0.194           Belarus         \$1.320         \$0.264         East Timor         \$5.599         \$1.120           Belize         \$0.972         \$0.194         Ecuador         \$0.667         \$0.073           Belize         \$0.972         \$0.194         Egypt         \$1.367         \$0.073           Benin         \$1.206         \$0.241         Egypt         \$1.367         \$0.273           Bentian         \$3						
Aruba         \$0.834         \$0.167         Croatia         \$0.995         \$0.199           Ascension Island         \$2.042         \$0.408         \$2.203         \$0.441           Austria         \$0.275         \$0.055         \$1.066         \$0.213           Austria         \$0.275         \$0.055         \$1.020         \$0.204           Azerbaijan         \$2.228         \$0.446         Denmark         \$0.275         \$0.055           Bahamas         \$0.342         \$0.068         Diego Garcia         \$3.064         \$0.613           Bahrain         \$1.458         \$0.292         Dibouti         \$1.765         \$0.353           Bangladesh         \$1.950         \$0.390         Dominica         \$0.972         \$0.194           Belarus         \$1.320         \$0.264         East Timor         \$5.599         \$1.120           Belgium         \$0.275         \$0.055         Easter Island         \$0.367         \$0.073           Belrize         \$0.972         \$0.194         Egypt         \$1.367         \$0.273           Bermuda         \$0.390         \$0.078         El Salvador         \$0.462         \$0.092           Bhutan         \$3.368         \$0.674         Equatorial Guine						
Ascension Island         \$2.042         \$0.408         Cuba         \$2.203         \$0.441           Austriai         \$0.321         \$0.064         Cyprus         \$1.066         \$0.213           Austria         \$0.275         \$0.055         Czech Republic         \$1.020         \$0.204           Azerbaijan         \$2.228         \$0.446         Denmark         \$0.275         \$0.055           Bahamas         \$0.342         \$0.068         Diego Garcia         \$3.064         \$0.613           Bahrain         \$1.458         \$0.292         Dijbouti         \$1.765         \$0.353           Bangladesh         \$1.950         \$0.390         Dominica         \$0.972         \$0.194           Berlarus         \$1.320         \$0.264         Belarus         \$0.275         \$0.055           Belarus         \$1.320         \$0.264         Best Timor         \$5.599         \$1.120           Belgium         \$0.275         \$0.055         East Timor         \$5.599         \$1.120           Belize         \$0.972         \$0.194         Egypt         \$1.367         \$0.273           Bermuda         \$0.390         \$0.078         El Salvador         \$0.462         \$0.092           Bhutan <td></td> <td></td> <td></td> <td></td> <td></td> <td></td>						
Australia         \$0.321         \$0.064         Cyprus         \$1.066         \$0.213           Austria         \$0.275         \$0.055         Czech Republic         \$1.020         \$0.204           Azerbaijan         \$2.228         \$0.446         Denmark         \$0.275         \$0.055           Bahamas         \$0.342         \$0.068         Diego Garcia         \$3.064         \$0.613           Bahrain         \$1.458         \$0.292         Diibouti         \$1.765         \$0.353           Bangladesh         \$1.950         \$0.390         Dominica         \$0.972         \$0.194           Barbados         \$0.667         \$0.133         Dominica         \$0.275         \$0.055           Belarus         \$1.320         \$0.264         East Timor         \$5.599         \$1.120           Belgium         \$0.275         \$0.055         Easter Island         \$0.367         \$0.073           Belize         \$0.972         \$0.194         Ecuador         \$0.667         \$0.133           Bermuda         \$0.390         \$0.078         El Salvador         \$0.462         \$0.092           Bhutan         \$3.368         \$0.674         Equatorial Guinea         \$4.741         \$0.948 <td< td=""><td></td><td></td><td></td><td></td><td></td><td></td></td<>						
Austria         \$0.275         \$0.055           Azerbaijan         \$2.228         \$0.446           Bahamas         \$0.342         \$0.068           Bahrain         \$1.458         \$0.292           Bangladesh         \$1.950         \$0.390           Barbados         \$0.667         \$0.133           Berarus         \$1.320         \$0.264           Belgium         \$0.275         \$0.055           Belgium         \$0.275         \$0.055           Bermuda         \$0.390         \$0.972           Bermuda         \$0.390         \$0.078           Bermuda         \$0.390         \$0.078           Benin         \$1.206         \$0.241           Bermuda         \$0.390         \$0.078           Bermuda         \$0.390         \$0.078           Beritish         \$0.854         \$0.171           Bosnia         \$1.436         \$0.287           Bosnia         \$1.436         \$0.287           Brazil         \$0.342         Extonia         \$1.670         \$0.334           British Virgin Islands         \$0.834         \$0.068         Faeroe Island         \$1.099         \$0.218           Burkina Faso         \$1.645<				Cuba		
Azerbaijan         \$2.228         \$0.446           Bahamas         \$0.342         \$0.068           Bahrain         \$1.458         \$0.292           Bangladesh         \$1.950         \$0.390           Barbados         \$0.667         \$0.133           Belarus         \$1.320         \$0.264           Belgium         \$0.275         \$0.055           Belgium         \$0.275         \$0.055           Belize         \$0.972         \$0.194           Benin         \$1.206         \$0.241           Bermuda         \$0.390         \$0.078           Behutan         \$3.368         \$0.674           Bosnia         \$1.436         \$0.287           Bosnia         \$1.436         \$0.287           Botswana         \$1.483         \$0.297           British Virgin Islands         \$0.834         \$0.167           Brunei         \$1.529         \$0.306           Burkina Faso         \$1.645         \$0.329           Burkina Faso         \$1.645         \$0.329           Burkina Faso         \$1.645         \$0.329           Burundi         \$3.205         \$0.641           Cameroon         \$1.810         \$0.362 <td>Australia</td> <td></td> <td></td> <td></td> <td></td> <td></td>	Australia					
Bahamas         \$0.342         \$0.068           Bahrain         \$1.458         \$0.292           Bangladesh         \$1.950         \$0.390           Barbados         \$0.667         \$0.133           Belarus         \$1.320         \$0.264           Belgium         \$0.275         \$0.055           Belgium         \$0.972         \$0.194           Belize         \$0.972         \$0.194           Benin         \$1.206         \$0.241           Bermuda         \$0.390         \$0.078           Bermuda         \$0.390         \$0.078           Bolivia         \$0.854         \$0.171           Botswana         \$1.483         \$0.287           Borriish Virgin Islands         \$0.834         \$0.167           Brunei         \$1.529         \$0.306           Burkina Faso         \$1.645         \$0.329           Burkina Faso         \$1.645         \$0.329           Burkina Faso         \$1.810         \$0.823           Cameroon         \$1.810         \$0.362           Camada - Outbound         \$0.110         \$0.022           Cape Verde Island         \$1.483         \$0.297           Gabon         \$1.254	Austria			Czech Republic		
Bahrain         \$1.458         \$0.292         Djibouti         \$1.765         \$0.353           Bangladesh         \$1.950         \$0.390         Dominica         \$0.972         \$0.194           Barbados         \$0.667         \$0.133         Dominican Republic         \$0.275         \$0.055           Belarus         \$1.320         \$0.264         East Timor         \$5.599         \$1.120           Belgium         \$0.275         \$0.055         Easter Island         \$0.367         \$0.073           Belize         \$0.972         \$0.194         Ecuador         \$0.667         \$0.133           Benin         \$1.206         \$0.241         Egypt         \$1.367         \$0.273           Bermuda         \$0.390         \$0.078         Egypt         \$1.367         \$0.273           Bermuda         \$0.390         \$0.074         Egypt         \$1.367         \$0.273           Bolivia         \$0.854         \$0.171         Eritrea         \$2.275         \$0.452           Bosnia         \$1.436         \$0.287         Estonia         \$1.670         \$0.334           Boriswana         \$1.483         \$0.297         Estonia         \$1.998         \$0.400           Brunei         \$	Azerbaijan		\$0.446	Denmark	\$0.275	
Bangladesh         \$1.950         \$0.390         Dominica         \$0.972         \$0.194           Barbados         \$0.667         \$0.133         Dominican Republic         \$0.275         \$0.055           Belarus         \$1.320         \$0.264         East Timor         \$5.599         \$1.120           Belgium         \$0.275         \$0.055         Easter Island         \$0.367         \$0.073           Belize         \$0.972         \$0.194         Ecuador         \$0.367         \$0.073           Benin         \$1.206         \$0.241         Egypt         \$1.367         \$0.273           Bermuda         \$0.390         \$0.078         Egypt         \$1.367         \$0.273           Bermuda         \$0.390         \$0.078         Egypt         \$1.367         \$0.273           Bolivia         \$0.354         \$0.171         Eritrea         \$2.275         \$0.452           Bosnia         \$1.436         \$0.287         Estonia         \$1.670         \$0.334           Botswana         \$1.483         \$0.297         Estonia         \$1.670         \$0.334           Brunei         \$1.529         \$0.306         Faeroe Island         \$1.090         \$0.218           Burkin Faso	Bahamas	\$0.342	\$0.068	Diego Garcia	\$3.064	\$0.613
Barbados         \$0.667         \$0.133         Dominican Republic         \$0.275         \$0.055           Belgium         \$0.275         \$0.055         East Timor         \$5.599         \$1.120           Belize         \$0.972         \$0.194         Easter Island         \$0.367         \$0.073           Benin         \$1.206         \$0.241         Egypt         \$1.367         \$0.273           Bermuda         \$0.390         \$0.078         El Salvador         \$0.462         \$0.092           Bhutan         \$3.368         \$0.674         Equatorial Guinea         \$4.741         \$0.948           Bolivia         \$0.854         \$0.171         Eritrea         \$2.275         \$0.455           Botswana         \$1.436         \$0.287         Estonia         \$1.670         \$0.334           British Virgin Islands         \$0.834         \$0.167         Faeroe Island         \$1.090         \$0.218           Brunei         \$1.529         \$0.306         Fiji Island         \$1.857         \$0.539           Burkina Faso         \$1.645         \$0.329         France         \$0.297         \$0.055           Burundi         \$3.205         \$0.641         French Antilles         \$1.179         \$0.236 <td>Bahrain</td> <td>\$1.458</td> <td>\$0.292</td> <td>Djibouti</td> <td></td> <td>\$0.353</td>	Bahrain	\$1.458	\$0.292	Djibouti		\$0.353
Belarus         \$1.320         \$0.264         East Timor         \$5.599         \$1.120           Belgium         \$0.275         \$0.055         Easter Island         \$0.367         \$0.073           Belize         \$0.972         \$0.194         Ecuador         \$0.667         \$0.133           Berinn         \$1.206         \$0.241         Egypt         \$1.367         \$0.273           Bermuda         \$0.390         \$0.078         El Salvador         \$0.462         \$0.092           Bhutan         \$3.368         \$0.674         Equatorial Guinea         \$4.741         \$0.948           Bolivia         \$0.854         \$0.171         Eritrea         \$2.275         \$0.455           Bonsia         \$1.436         \$0.287         Estonia         \$1.670         \$0.334           Botswana         \$1.483         \$0.297         Estonia         \$1.998         \$0.400           Brazil         \$0.342         \$0.068         Faeroe Island         \$1.090         \$0.218           Brunei         \$1.529         \$0.306         Falkland Island         \$2.695         \$0.539           Burkina Faso         \$1.645         \$0.329         France         \$0.297         \$0.055           Burundi </td <td>Bangladesh</td> <td>\$1.950</td> <td>\$0.390</td> <td>Dominica</td> <td>\$0.972</td> <td>\$0.194</td>	Bangladesh	\$1.950	\$0.390	Dominica	\$0.972	\$0.194
Belgium         \$0.275         \$0.055         Easter Island         \$0.367         \$0.073           Belize         \$0.972         \$0.194         Ecuador         \$0.667         \$0.133           Benin         \$1.206         \$0.241         Egypt         \$1.367         \$0.273           Bermuda         \$0.390         \$0.078         El Salvador         \$0.462         \$0.092           Bhutan         \$3.368         \$0.674         Equatorial Guinea         \$4.741         \$0.948           Bolivia         \$0.854         \$0.171         Eritrea         \$2.275         \$0.455           Bosnia         \$1.436         \$0.287         Estonia         \$1.670         \$0.334           Botswana         \$1.483         \$0.297         Ethiopia         \$1.998         \$0.400           Brazil         \$0.342         \$0.068         Faeroe Island         \$1.090         \$0.218           Brunei         \$1.529         \$0.306         Fillsland Island         \$2.695         \$0.539           Burunei         \$1.179         \$0.236         France         \$0.297         \$0.055           Burundi         \$3.205         \$0.641         French Antilles         \$1.179         \$0.236           Cameroon	Barbados	\$0.667	\$0.133	Dominican Republic	\$0.275	\$0.055
Belize         \$0.972         \$0.194         Ecuador         \$0.667         \$0.133           Benin         \$1.206         \$0.241         Egypt         \$1.367         \$0.273           Bermuda         \$0.390         \$0.078         El Salvador         \$0.462         \$0.092           Bhutan         \$3.368         \$0.674         Equatorial Guinea         \$4.741         \$0.948           Bolivia         \$0.854         \$0.171         Eritrea         \$2.275         \$0.455           Bosnia         \$1.436         \$0.287         Estonia         \$1.670         \$0.334           Botswana         \$1.483         \$0.297         Ethiopia         \$1.998         \$0.400           Brazil         \$0.342         \$0.068         Faeroe Island         \$1.090         \$0.218           Brunei         \$1.529         \$0.306         Falkland Island         \$2.695         \$0.539           Burkina Faso         \$1.645         \$0.329         France         \$0.297         \$0.055           Burundi         \$3.205         \$0.641         French Antilles         \$1.179         \$0.236           Cameroon         \$1.810         \$0.362         French Polynesia         \$2.415         \$0.483	Belarus	\$1.320	\$0.264	East Timor	\$5.599	\$1.120
Benin         \$1.206         \$0.241         Egypt         \$1.367         \$0.273           Bermuda         \$0.390         \$0.078         El Salvador         \$0.462         \$0.092           Bhutan         \$3.368         \$0.674         Equatorial Guinea         \$4.741         \$0.948           Bolivia         \$0.854         \$0.171         Eritrea         \$2.275         \$0.455           Bosnia         \$1.436         \$0.287         Estonia         \$1.670         \$0.334           Botswana         \$1.483         \$0.297         Ethiopia         \$1.998         \$0.400           Brazil         \$0.342         \$0.068         Faeroe Island         \$1.090         \$0.218           British Virgin Islands         \$0.834         \$0.167         Falkland Island         \$2.695         \$0.539           Brunei         \$1.529         \$0.306         Fiji Island         \$1.857         \$0.371           Burkina Faso         \$1.645         \$0.329         France         \$0.297         \$0.055           Burundi         \$3.205         \$0.641         French Antilles         \$1.179         \$0.236           Cameroon         \$1.810         \$0.362         French Polynesia         \$2.415         \$0.483	Belgium	\$0.275	\$0.055	Easter Island	\$0.367	\$0.073
Bermuda         \$0.390         \$0.078         El Salvador         \$0.462         \$0.092           Bhutan         \$3.368         \$0.674         Equatorial Guinea         \$4.741         \$0.948           Bolivia         \$0.854         \$0.171         Eritrea         \$2.275         \$0.455           Bosnia         \$1.436         \$0.287         Estonia         \$1.670         \$0.334           Botswana         \$1.483         \$0.297         Ethiopia         \$1.998         \$0.400           Brazil         \$0.342         \$0.068         Faeroe Island         \$1.090         \$0.218           British Virgin Islands         \$0.834         \$0.167         Faeroe Island         \$1.090         \$0.218           Brunei         \$1.529         \$0.306         Fiji Island         \$2.695         \$0.539           Burkina Faso         \$1.179         \$0.236         Finland         \$0.275         \$0.055           Burundi         \$3.205         \$0.641         French Antilles         \$1.179         \$0.236           Cambodia         \$4.113         \$0.823         French Polynesia         \$2.415         \$0.483           Canada - Inbound         \$0.110         \$0.022         Gabon         \$1.998         \$0.400	Belize	\$0.972	\$0.194	Ecuador	\$0.667	\$0.133
Bhutan         \$3.368         \$0.674         Equatorial Guinea         \$4.741         \$0.948           Bolivia         \$0.854         \$0.171         Eritrea         \$2.275         \$0.455           Bosnia         \$1.436         \$0.287         Estonia         \$1.670         \$0.334           Botswana         \$1.483         \$0.297         Ethiopia         \$1.998         \$0.400           Brazil         \$0.342         \$0.068         Faeroe Island         \$1.090         \$0.218           British Virgin Islands         \$0.834         \$0.167         Falkland Island         \$2.695         \$0.539           Brunei         \$1.529         \$0.306         Fiji Island         \$1.857         \$0.371           Bulgaria         \$1.179         \$0.236         Finland         \$0.275         \$0.055           Burundi         \$3.205         \$0.641         French Antilles         \$1.179         \$0.236           Cambodia         \$4.113         \$0.823         French Guiana         \$1.179         \$0.236           Canada - Inbound         \$0.110         \$0.022         Gabon         \$1.998         \$0.400           Canada - Outbound         \$0.110         \$0.022         Gambia         \$1.254         \$0.251	Benin	\$1.206	\$0.241	Egypt	\$1.367	\$0.273
Solivia   Soles   Solivia   Solivi	Bermuda	\$0.390	\$0.078	El Salvador	\$0.462	\$0.092
Bosnia         \$1.436         \$0.287         Estonia         \$1.670         \$0.334           Botswana         \$1.483         \$0.297         Ethiopia         \$1.998         \$0.400           Brazil         \$0.342         \$0.068         Faeroe Island         \$1.090         \$0.218           British Virgin Islands         \$0.834         \$0.167         Faeroe Island         \$1.090         \$0.218           Brunei         \$1.529         \$0.306         Falkland Island         \$2.695         \$0.539           Bulgaria         \$1.179         \$0.236         Fiji Island         \$1.857         \$0.371           Burkina Faso         \$1.645         \$0.329         France         \$0.297         \$0.059           Burundi         \$3.205         \$0.641         French Antilles         \$1.179         \$0.236           Cambodia         \$4.113         \$0.823         French Guiana         \$1.179         \$0.236           Cameroon         \$1.810         \$0.362         French Polynesia         \$2.415         \$0.483           Canada - Inbound         \$0.110         \$0.022         Gabon         \$1.998         \$0.400           Canada - Outbound         \$0.110         \$0.022         Georgia         \$1.879         \$	Bhutan	\$3.368	\$0.674	Equatorial Guinea	\$4.741	\$0.948
Botswana         \$1.483         \$0.297         Ethiopia         \$1.998         \$0.400           Brazil         \$0.342         \$0.068         Faeroe Island         \$1.090         \$0.218           British Virgin Islands         \$0.834         \$0.167         Falkland Island         \$2.695         \$0.539           Brunei         \$1.529         \$0.306         Fiji Island         \$1.857         \$0.371           Bulgaria         \$1.179         \$0.236         Finland         \$0.275         \$0.055           Burkina Faso         \$1.645         \$0.329         France         \$0.297         \$0.059           Burundi         \$3.205         \$0.641         French Antilles         \$1.179         \$0.236           Cambodia         \$4.113         \$0.823         French Guiana         \$1.179         \$0.236           Cameroon         \$1.810         \$0.362         French Polynesia         \$2.415         \$0.483           Canada - Inbound         \$0.110         \$0.022         Gabon         \$1.998         \$0.400           Canada - Outbound         \$0.110         \$0.022         Georgia         \$1.879         \$0.376	Bolivia	\$0.854	\$0.171	Eritrea	\$2.275	\$0.455
Brazil         \$0.342         \$0.068         Faeroe Island         \$1.090         \$0.218           British Virgin Islands         \$0.834         \$0.167         Falkland Island         \$2.695         \$0.539           Brunei         \$1.529         \$0.306         Fiji Island         \$1.857         \$0.371           Bulgaria         \$1.179         \$0.236         Finland         \$0.275         \$0.055           Burkina Faso         \$1.645         \$0.329         France         \$0.297         \$0.059           Burundi         \$3.205         \$0.641         French Antilles         \$1.179         \$0.236           Cambodia         \$4.113         \$0.823         French Guiana         \$1.179         \$0.236           Cameroon         \$1.810         \$0.362         French Polynesia         \$2.415         \$0.483           Canada - Inbound         \$0.110         \$0.022         Gabon         \$1.998         \$0.400           Canada - Outbound         \$0.110         \$0.022         Gambia         \$1.254         \$0.251           Cape Verde Island         \$1.483         \$0.297         Georgia         \$1.879         \$0.376	Bosnia			Estonia		
Brazil         \$0.342         \$0.068         Faeroe Island         \$1.090         \$0.218           British Virgin Islands         \$0.834         \$0.167         Falkland Island         \$2.695         \$0.539           Brunei         \$1.529         \$0.306         Fiji Island         \$1.857         \$0.371           Bulgaria         \$1.179         \$0.236         Finland         \$0.275         \$0.055           Burkina Faso         \$1.645         \$0.329         France         \$0.297         \$0.059           Burundi         \$3.205         \$0.641         French Antilles         \$1.179         \$0.236           Cambodia         \$4.113         \$0.823         French Guiana         \$1.179         \$0.236           Cameroon         \$1.810         \$0.362         French Polynesia         \$2.415         \$0.483           Canada - Inbound         \$0.110         \$0.022         Gabon         \$1.998         \$0.400           Canada - Outbound         \$0.110         \$0.022         Gambia         \$1.254         \$0.251           Cape Verde Island         \$1.483         \$0.297         Georgia         \$1.879         \$0.376	Botswana	\$1.483	\$0.297	Ethiopia	\$1.998	\$0.400
British Virgin Islands         \$0.834         \$0.167         Falkland Island         \$2.695         \$0.539           Brunei         \$1.529         \$0.306         Fiji Island         \$1.857         \$0.371           Bulgaria         \$1.179         \$0.236         Finland         \$0.275         \$0.055           Burkina Faso         \$1.645         \$0.329         France         \$0.297         \$0.059           Burundi         \$3.205         \$0.641         French Antilles         \$1.179         \$0.236           Cambodia         \$4.113         \$0.823         French Guiana         \$1.179         \$0.236           Cameroon         \$1.810         \$0.362         French Polynesia         \$2.415         \$0.483           Canada - Inbound         \$0.110         \$0.022         Gabon         \$1.998         \$0.400           Canada - Outbound         \$0.110         \$0.022         Gambia         \$1.254         \$0.251           Cape Verde Island         \$1.483         \$0.297         Georgia         \$1.879         \$0.376	Brazil		\$0.068	Faeroe Island	\$1.090	\$0.218
Brunei         \$1.529         \$0.306         Fiji Island         \$1.857         \$0.371           Bulgaria         \$1.179         \$0.236         Finland         \$0.275         \$0.055           Burkina Faso         \$1.645         \$0.329         France         \$0.297         \$0.059           Burundi         \$3.205         \$0.641         French Antilles         \$1.179         \$0.236           Cambodia         \$4.113         \$0.823         French Guiana         \$1.179         \$0.236           Cameroon         \$1.810         \$0.362         French Polynesia         \$2.415         \$0.483           Canada - Inbound         \$0.110         \$0.022         Gabon         \$1.998         \$0.400           Canada - Outbound         \$0.110         \$0.022         Gambia         \$1.254         \$0.251           Cape Verde Island         \$1.483         \$0.297         Georgia         \$1.879         \$0.376	British Virgin Islands			Falkland Island		
Bulgaria         \$1.179         \$0.236         Finland         \$0.275         \$0.055           Burkina Faso         \$1.645         \$0.329         France         \$0.297         \$0.059           Burundi         \$3.205         \$0.641         French Antilles         \$1.179         \$0.236           Cambodia         \$4.113         \$0.823         French Guiana         \$1.179         \$0.236           Cameroon         \$1.810         \$0.362         French Polynesia         \$2.415         \$0.483           Canada - Inbound         \$0.110         \$0.022         Gabon         \$1.998         \$0.400           Canada - Outbound         \$0.110         \$0.022         Gambia         \$1.254         \$0.251           Cape Verde Island         \$1.483         \$0.297         Georgia         \$1.879         \$0.376	Brunei					
Burkina Faso         \$1.645         \$0.329         France         \$0.297         \$0.059           Burundi         \$3.205         \$0.641         French Antilles         \$1.179         \$0.236           Cambodia         \$4.113         \$0.823         French Guiana         \$1.179         \$0.236           Cameroon         \$1.810         \$0.362         French Polynesia         \$2.415         \$0.483           Canada - Inbound         \$0.110         \$0.022         Gabon         \$1.998         \$0.400           Canada - Outbound         \$0.110         \$0.022         Gambia         \$1.254         \$0.251           Cape Verde Island         \$1.483         \$0.297         Georgia         \$1.879         \$0.376	Bulgaria					
Burundi         \$3.205         \$0.641         French Antilles         \$1.179         \$0.236           Cambodia         \$4.113         \$0.823         French Guiana         \$1.179         \$0.236           Cameroon         \$1.810         \$0.362         French Polynesia         \$2.415         \$0.483           Canada - Inbound         \$0.110         \$0.022         Gabon         \$1.998         \$0.400           Canada - Outbound         \$0.110         \$0.022         Gambia         \$1.254         \$0.251           Cape Verde Island         \$1.483         \$0.297         Georgia         \$1.879         \$0.376						
Cambodia         \$4.113         \$0.823         French Guiana         \$1.179         \$0.236           Cameroon         \$1.810         \$0.362         French Polynesia         \$2.415         \$0.483           Canada - Inbound         \$0.110         \$0.022         Gabon         \$1.998         \$0.400           Canada - Outbound         \$0.110         \$0.022         Gambia         \$1.254         \$0.251           Cape Verde Island         \$1.483         \$0.297         Georgia         \$1.879         \$0.376	Burundi					
Cameroon         \$1.810         \$0.362         French Polynesia         \$2.415         \$0.483           Canada - Inbound         \$0.110         \$0.022         Gabon         \$1.998         \$0.400           Canada - Outbound         \$0.110         \$0.022         Gambia         \$1.254         \$0.251           Cape Verde Island         \$1.483         \$0.297         Georgia         \$1.879         \$0.376						
Canada - Inbound         \$0.110         \$0.022         Gabon         \$1.998         \$0.400           Canada - Outbound         \$0.110         \$0.022         Gambia         \$1.254         \$0.251           Cape Verde Island         \$1.483         \$0.297         Georgia         \$1.879         \$0.376						
Canada - Outbound         \$0.110         \$0.022         Gambia         \$1.254         \$0.251           Cape Verde Island         \$1.483         \$0.297         Georgia         \$1.879         \$0.376						
Cape Verde Island \$1.483 \$0.297 Georgia \$1.879 \$0.376						
	Cayman Islands	\$0.693	\$0.139	Germany	\$0.297	\$0.059

Original Page 62

Effective: September 1, 2019

### SECTION 12 – GRANDFATHERED SERVICES

- 12.5 Type 4 Grandfathered Services (continued)
  - 12.5.18 Business Worldwide Plan (continued)
    - (E) Rates and Charges (continued)
      - .2 The per-minute usage rates are as follows: (continued)

Country	Initial	Each Additional 6 Seconds	Country	Initial 30 Seconds	Each Additional
Ghana	\$1.254	\$0.251	Country Luxembourg	\$0.275	\$0.055
Gibraltar	\$1.254	\$0.231	Macao	\$1.390	\$0.033
Gibraitar Greece	\$0.275	\$0.371	Macedonia	\$1.066	\$0.278
Greenland	\$1.179				\$0.213
		\$0.236 \$0.204	Madagascar	\$3.880 \$1.320	\$0.776
Grenada	\$1.020		Malawi		
Guadeloupe	\$1.161	\$0.232	Malaysia	\$0.275	\$0.055
Guantanamo	\$2.807	\$0.561	Maldives	\$2.647	\$0.529
Guatemala	\$0.949	\$0.190	Mali Republic	\$1.902	\$0.380
Guinea	\$1.483	\$0.297	Malta Republic	\$1.320	\$0.264
Guinea-Bissau	\$3.040	\$0.608	Marshall Island	\$1.320	\$0.264
Guyana	\$1.785	\$0.357	Mauritania	\$1.785	\$0.357
Haiti	\$0.808	\$0.162	Mauritius	\$2.253	\$0.451
Honduras	\$0.927	\$0.185	Mayotte Island	\$1.810	\$0.362
Hong Kong	\$0.135	\$0.027	Mexico 1	\$0.204	\$0.041
Hungary	\$0.764	\$0.153	Mexico 2	\$0.297	\$0.059
Iceland	\$0.927	\$0.185	Micronesia	\$1.412	\$0.282
India	\$0.554	\$0.111	Moldova	\$2.228	\$0.446
Indonesia	\$0.482	\$0.096	Monaco	\$0.275	\$0.055
Inmarsat (Atlantic Ocean-East)	\$16.396	\$3.279	Mongolia	\$3.088	\$0.618
Inmarsat (Atlantic Ocean-West)	\$16.396	\$3.279	Montenegro	\$1.206	\$0.241
Inmarsat (Indian Ocean)	\$16.396	\$3.279	Montserrat	\$0.972	\$0.194
Inmarsat (Pacific Ocean)	\$16.396	\$3.279	Morocco	\$1.483	\$0.297
Inmarsat (SNAC)	\$16.396	\$3.279	Mozambique	\$2.298	\$0.460
Iran	\$2.298	\$0.460	Myanmar	\$3.950	\$0.790
Iraq	\$2.203	\$0.441	Namibia	\$1.436	\$0.287
Ireland	\$0.275	\$0.055	Nauru	\$2.344	\$0.469
Israel	\$0.297	\$0.059	Nepal	\$2.344	\$0.469
Italy	\$0.297	\$0.059	Netherland Antilles	\$0.764	\$0.153
Ivory Coast	\$2.253	\$0.451	Netherlands	\$0.297	\$0.059
Jamaica	\$0.741	\$0.148	Networks	\$13.953	\$2.791
Japan	\$0.135	\$0.027	New Caledonia	\$2.647	\$0.529
Jordan	\$1.645	\$0.329	New Zealand	\$0.462	\$0.092
Kazakhstan	\$0.342	\$0.068	Nicaragua Nicaragua	\$0.995	\$0.199
Kenya	\$1.553	\$0.311	Niger Republic	\$1.670	\$0.334
Kiribati	\$2.415	\$0.483	Nigeria	\$1.508	\$0.302
Korea (North)	\$5.599	\$1.120	Niue	\$3.389	\$0.678
Korea (South)	\$0.204	\$0.041	Norfolk Island	\$3.414	\$0.683
Kosovo	\$0.204	\$0.041	Norway	\$0.275	\$0.055
Kuwait	\$1.716	\$0.033	Oman	\$1.902	\$0.033
	\$2.020	\$0.343	Pakistan	\$1.902	\$0.380
Kyrgyzstan Laos	\$4.437	\$0.404	Palau	\$1.234	
Latvia					\$0.488
Latvia	\$1.716	\$0.348	Palestinian Authority	\$0.902	\$0.180
Lebanon	\$2.203	\$0.441	Panama	\$0.624	\$0.125
Lesotho	\$1.553	\$0.311	Papua New Guinea	\$1.879	\$0.376
Liberia	\$1.179	\$0.236	Paraguay	\$1.134	\$0.227
Libya	\$1.765	\$0.353	Peru	\$0.437	\$0.087
Liechtenstein	\$0.275	\$0.055	Philippines	\$0.624	\$0.125
Lithuania	\$2.203	\$0.441	Poland	\$0.717	\$0.143

Original Page 63

Effective: September 1, 2019

### SECTION 12 – GRANDFATHERED SERVICES

- 12.5 Type 4 Grandfathered Services (continued)
  - 12.5.18 Business Worldwide Plan (continued)
    - (E) Rates and Charges (continued)
      - .2 The per-minute usage rates are as follows: (continued)

Country	Initial 30 Seconds	Each Additional 6 Seconds	Country	Initial 30 Seconds	Each Additional 6 Seconds
Portugal	\$0.275	\$0.055	Switzerland	\$0.275	\$0.055
Qatar	\$1.902	\$0.380	Syrian Arab Republic	\$2.275	\$0.455
Reunion Island	\$1.810	\$0.362	Taiwan	\$0.135	\$0.027
Romania	\$1.553	\$0.311	Tajikistan	\$0.808	\$0.162
Russia	\$0.342	\$0.068	Tanzania	\$1.833	\$0.367
Rwanda, Republic of	\$2.158	\$0.432	Thailand	\$0.482	\$0.096
San Marino	\$1.390	\$0.278	Togo	\$1.998	\$0.400
Sao Tome	\$3.088	\$0.618	Tokelau	\$3.040	\$0.608
Saudi Arabia	\$1.692	\$0.338	Tonga Island	\$2.832	\$0.566
Senegal	\$2.621	\$0.524	Trinidad & Tobago	\$0.834	\$0.167
Serbia	\$1.206	\$0.241	Tunisia	\$1.412	\$0.282
Seychelles Island	\$2.715	\$0.543	Turkey	\$0.880	\$0.176
Sierra Leone	\$1.810	\$0.362	Turkmenistan	\$1.950	\$0.390
Singapore	\$0.531	\$0.106	Turks and Caicos Islands	\$1.066	\$0.213
Slovakia	\$1.134	\$0.227	Tuvalu	\$3.347	\$0.669
Slovenia	\$1.066	\$0.213	Uganda	\$1.458	\$0.292
Solomon Island	\$2.647	\$0.529	Ukraine	\$0.505	\$0.101
Somalia Democratic Rep	\$3.040	\$0.608	United Arab Emirates	\$1.090	\$0.218
South Africa	\$0.717	\$0.143	United Kingdom	\$0.135	\$0.027
South Sudan	\$2.509	\$0.502	Uruguay	\$1.134	\$0.227
Spain	\$0.297	\$0.059	Uzbekistan	\$0.854	\$0.171
Sri Lanka	\$1.950	\$0.390	Vanuatu	\$4.717	\$0.943
St. Helena	\$2.275	\$0.455	Vatican City	\$0.412	\$0.082
St. Kitts/Nevis	\$0.949	\$0.190	Venezuela	\$0.437	\$0.087
St. Lucia	\$0.949	\$0.190	Vietnam	\$1.134	\$0.227
St. Maarten	\$0.764	\$0.153	Wallis Futuna	\$5.300	\$1.060
St. Pierre Miquelon	\$0.995	\$0.199	Western Samoa	\$1.857	\$0.371
St. Vincent/Grenadines	\$1.134	\$0.227	Yemen Arab Republic	\$1.902	\$0.380
Sudan	\$2.509	\$0.502	Zaire	\$1.412	\$0.282
Suriname	\$2.253	\$0.451	Zambia	\$1.529	\$0.306
Swaziland	\$1.367	\$0.273	Zimbabwe	\$1.412	\$0.282
Sweden	\$0.275	\$0.055			

#### SECTION 12 – GRANDFATHERED SERVICES

#### 12.5 Type 4 – Grandfathered Services (continued)

#### 12.5.19 Business International Savings Plan (Grandfathered 12/31/2011)

(A) The Business International Savings plan is a direct dialed outbound international long distance service offered to Business Customers from switched access lines. Service is offered on a flat rate basis, twenty-four hours per day, seven days per week. Per minute charges apply and will be based on the country called and the duration of the call. Customers must dial 1 plus the country code or area code, if applicable, and the terminating number to complete a call utilizing this service.

Original Page 64

Effective: September 1, 2019

- (B) All calls are billed in six (6) second increments after an initial period, for billing purposes, of thirty (30) seconds. This service is only offered in conjunction with any BellSouth® Business domestic service. MMU charge applies to this service. The Customer will be billed the minimum usage charge if actual usage is below the commitment level in any month. Usage over the minimum usage charge may be carried over to future months to satisfy the commitment.
- (C) International Mobile Termination Charge

See Section 6.4.1 of this pricing and service guide for the applicable International Mobile Termination Charge.

(D) Rates and Charges

The MMU charge is \$24.95, and the per-minute usage rates are as follows:

Country	Initial 30 Seconds	Each Additional 6 Seconds	Country	Initial 30 Seconds	Each Additional 6 Seconds
Afghanistan	\$0.2900	\$0.0580	Benin	\$0.2600	\$0.0520
Albania	\$0.4350	\$0.0870	Bermuda	\$0.0850	\$0.0170
Algeria	\$0.2750	\$0.0550	Bhutan	\$0.7250	\$0.1450
Andorra	\$0.1800	\$0.0360	Bolivia	\$0.1850	\$0.0370
Angola	\$0.5800	\$0.1160	Bosnia	\$0.3100	\$0.0620
Anguilla	\$0.2500	\$0.0500	Botswana	\$0.3200	\$0.0640
Antarctica (Casey Base)	\$0.7350	\$0.1470	Brazil	\$0.0750	\$0.0150
Antarctica (Scott Base)	\$0.7350	\$0.1470	British Virgin Islands	\$0.1800	\$0.0360
Antigua	\$0.2050	\$0.0410	Brunei	\$0.3300	\$0.0660
Argentina	\$0.0900	\$0.0180	Bulgaria	\$0.2550	\$0.0510
Armenia	\$0.3750	\$0.0750	Burkina Faso	\$0.3550	\$0.0710
Aruba	\$0.1800	\$0.0360	Burundi	\$0.6900	\$0.1380
Ascension Island	\$0.4400	\$0.0880	Cambodia	\$0.8850	\$0.1770
Australia	\$0.0700	\$0.0140	Cameroon	\$0.3900	\$0.0780
Austria	\$0.0600	\$0.0120	Canada - Outbound	\$0.0250	\$0.0050
Azerbaijan	\$0.4800	\$0.0960	Canada - Inbound Service	\$0.0250	\$0.0050
Bahamas	\$0.0750	\$0.0150	Cape Verde Island	\$0.3200	\$0.0640
Bahrain	\$0.3150	\$0.0630	Cayman Islands	\$0.1500	\$0.0300
Bangladesh	\$0.4200	\$0.0840	Central African Republic	\$0.5900	\$0.1180
Barbados	\$0.1450	\$0.0290	Chad	\$0.9650	\$0.1930
Belarus	\$0.2850	\$0.0570	Chile	\$0.0800	\$0.0160
Belgium	\$0.0600	\$0.0120	China	\$0.0550	\$0.0110
Belize	\$0.2100	\$0.0420	Christmas & Cocos Islands	\$0.0700	\$0.0140

Original Page 65

Effective: September 1, 2019

### SECTION 12 - GRANDFATHERED SERVICES

- 12.5 Type 4 Grandfathered Services (continued)
  - 12.5.19 Business International Savings Plan (continued)
    - (D) Rates and Charges (continued)

The usage rates are as follows: (continued)

		Each		Initial	Each
	Initial	Additional		30	Additional
Country	30 Seconds	6 Seconds	Country	Seconds	6 Seconds
Colombia	\$0.0900	\$0.0180	Guinea-Bissau	\$0.6550	\$0.1310
Comoros Islands	\$1.3950	\$0.2790	Guyana	\$0.3850	\$0.0770
Congo	\$0.4400	\$0.0880	Haiti	\$0.1750	\$0.0350
Cook Islands	\$0.6550	\$0.1310	Honduras	\$0.2000	\$0.0400
Costa Rica	\$0.0950	\$0.0190	Hong Kong	\$0.0300	\$0.0060
Croatia	\$0.2150	\$0.0430	Hungary	\$0.1650	\$0.0330
Cuba	\$0.4750	\$0.0950	Iceland	\$0.2000	\$0.0400
Cyprus	\$0.2300	\$0.0460	India	\$0.1200	\$0.0240
Czech Republic	\$0.2200	\$0.0440	Indonesia	\$0.1050	\$0.0210
Denmark	\$0.0600	\$0.0120	Inmarsat (SNAC)	\$3.5250	\$0.7050
Diego Garcia	\$0.6600	\$0.1320	Iran	\$0.4950	\$0.0990
Djibouti	\$0.3800	\$0.0760	Iraq	\$0.4750	\$0.0950
Dominican Republic	\$0.0600	\$0.0120	Ireland	\$0.0600	\$0.0120
Dominica	\$0.2100	\$0.0420	Israel	\$0.0650	\$0.0130
East Timor	\$1.2050	\$0.2410	Italy	\$0.0650	\$0.0130
Easter Island	\$0.0800	\$0.0160	Ivory Coast	\$0.4850	\$0.0970
Ecuador	\$0.1450	\$0.0290	Jamaica	\$0.1600	\$0.0320
Egypt	\$0.2950	\$0.0590	Japan	\$0.0300	\$0.0060
El Salvador	\$0.1000	\$0.0200	Jordan	\$0.3550	\$0.0710
Equatorial Guinea	\$1.0200	\$0.2040	Kazakhstan	\$0.0750	\$0.0150
Eritrea	\$0.4900	\$0.0980	Kenya	\$0.3350	\$0.0670
Estonia	\$0.3600	\$0.0720	Kiribati	\$0.5200	\$0.1040
Ethiopia	\$0.4300	\$0.0860	Korea (North)	\$1.2050	\$0.2410
Faeroe Island	\$0.2350	\$0.0470	Korea (South)	\$0.0450	\$0.0090
Falkland Island	\$0.5800	\$0.1160	Kosovo	\$0.0600	\$0.0120
Fiji Island	\$0.4000	\$0.0800	Kuwait	\$0.3700	\$0.0740
Finland	\$0.0600	\$0.0120	Kyrgyzstan	\$0.4350	\$0.0870
France	\$0.0650	\$0.0130	Laos	\$0.9550	\$0.1910
French Antilles	\$0.2550	\$0.0510	Latvia	\$0.3700	\$0.0740
French Guiana	\$0.2550	\$0.0510	Lebanon	\$0.4750	\$0.0950
French Polynesia	\$0.5200	\$0.1040	Lesotho	\$0.3350	\$0.0670
Gabon	\$0.4300	\$0.0860	Liberia	\$0.2550	\$0.0510
Gambia	\$0.2700	\$0.0540	Libya	\$0.3800	\$0.0760
Georgia	\$0.4050	\$0.0810	Liechtenstein	\$0.0600	\$0.0120
Germany	\$0.0650	\$0.0130	Lithuania	\$0.4750	\$0.0950
Ghana	\$0.2700	\$0.0540	Luxembourg	\$0.0600	\$0.0120
Gibraltar	\$0.4000	\$0.0800	Macao	\$0.3000	\$0.0600
Greece	\$0.0600	\$0.0120	Macedonia	\$0.2300	\$0.0460
Greenland	\$0.2550	\$0.0510	Madagascar	\$0.8350	\$0.1670
Grenada	\$0.2200	\$0.0440	Malawi	\$0.2850	\$0.0570
Guadeloupe	\$0.2500	\$0.0500	Malaysia	\$0.0600	\$0.0120
Guantanamo	\$0.6050	\$0.1210	Maldives	\$0.5700	\$0.1140
Guatemala	\$0.2050	\$0.0410	Mali Republic	\$0.4100	\$0.0820
Guinea	\$0.3200	\$0.0640	Malta Republic	\$0.2850	\$0.0570

Original Page 66

Effective: September 1, 2019

### SECTION 12 - GRANDFATHERED SERVICES

- 12.5 Type 4 Grandfathered Services (continued)
  - 12.5.19 Business International Savings Plan (continued)
    - (D) Rates and Charges (continued)

The usage rates are as follows: (continued)

Country	Initial 30 Seconds	Each Additional 6 Seconds	Country	Initial 30 Seconds	Each Additional 6 Seconds
Marshall Island	\$0.2850	\$0.0570	Norfolk Island	\$0.7350	\$0.1470
Mauritania	\$0.3850	\$0.0770	Norway	\$0.0600	\$0.0120
Mauritius	\$0.4850	\$0.0970	Oman	\$0.4100	\$0.0820
Mexico 1	\$0.0450	\$0.0090	Pakistan	\$0.2700	\$0.0540
Mexico 2	\$0.0650	\$0.0130	Palau	\$0.5250	\$0.1050
Micronesia	\$0.3050	\$0.0610	Palestinian Authority	\$0.1950	\$0.0390
Moldova	\$0.4800	\$0.0960	Panama	\$0.1350	\$0.0270
Monaco	\$0.0600	\$0.0120	Papua New Guinea	\$0.4050	\$0.0810
Mongolia	\$0.6650	\$0.1330	Paraguay	\$0.2450	\$0.0490
Montenegro	\$0.2600	\$0.0520	Peru	\$0.0950	\$0.0190
Montserrat	\$0.2100	\$0.0420	Philippines	\$0.1350	\$0.0270
Morocco	\$0.3200	\$0.0640	Poland	\$0.1550	\$0.0310
Mozambique	\$0.4950	\$0.0990	Portugal	\$0.0600	\$0.0120
Myanmar	\$0.8500	\$0.1700	Qatar	\$0.4100	\$0.0820
Namibia	\$0.3100	\$0.0620	Reunion Island (includes Mayotte Island)	\$0.3900	\$0.0780
Nauru	\$0.5050	\$0.1010	Romania	\$0.3350	\$0.0670
Nepal	\$0.5050	\$0.1010	Russia	\$0.0750	\$0.0150
Netherland Antilles	\$0.1650	\$0.0330	Rwanda, Republic of	\$0.4650	\$0.0930
Netherlands	\$0.0650	\$0.0130	San Marino	\$0.3000	\$0.0600
Networks	\$3.0000	\$0.6000	Sao Tome	\$0.6650	\$0.1330
New Caledonia	\$0.5700	\$0.1140	Saudi Arabia	\$0.3650	\$0.0730
New Zealand	\$0.1000	\$0.0200	Senegal	\$0.5650	\$0.1130
Nicaragua	\$0.2150	\$0.0430	Serbia	\$0.2600	\$0.0520
Niger Republic	\$0.3600	\$0.0720	Seychelles Island	\$0.5850	\$0.1170
Nigeria	\$0.3250	\$0.0650	Sierra Leone	\$0.3900	\$0.0780
Niue	\$0.7300	\$0.1460	Singapore	\$0.1150	\$0.0230

Original Page 67

Each Additional 6 Seconds \$0.1310 \$0.1220 \$0.0360 \$0.0610 \$0.0380 \$0.0840 \$0.0460 \$0.1440 \$0.0630 \$0.0220 \$0.0470 \$0.0060 \$0.0490 \$0.0370 \$0.2030 \$0.0180 \$0.0190 \$0.0490 \$0.2280 \$0.0800 \$0.0820 \$0.0610 \$0.0660 \$0.0610

Effective: September 1, 2019

#### SECTION 12 - GRANDFATHERED SERVICES

- 12.5 Type 4 Grandfathered Services (continued)
  - 12.5.19 Business International Savings Plan (continued)
    - (D) Rates and Charges (continued)

      The usage rates are as follows: (continued)

Country	Initial 30 Seconds	Each Additional 6 Seconds	Country	Initial 30 Seconds	
Slovakia	\$0.2450	\$0.0490	Tokelau	\$0.6550	Ī
Slovenia	\$0.2300	\$0.0460	Tonga Island	\$0.6100	I
Solomon Island	\$0.5700	\$0.1140	Trinidad	\$0.1800	I
Somalia Democratic Rep	\$0.6550	\$0.1310	Tunisia	\$0.3050	l
South Africa	\$0.1550	\$0.0310	Turkey	\$0.1900	I
South Sudan	\$0.5400	\$0.1080	Turkmenistan	\$0.4200	I
Spain	\$0.0650	\$0.0130	Turks and Caicos Islands	\$0.2300	I
Sri Lanka	\$0.4200	\$0.0840	Tuvalu	\$0.7200	Ī
St. Helena	\$0.4900	\$0.0980	Uganda	\$0.3150	Ī
St. Kitts/Nevis	\$0.2050	\$0.0410	Ukraine	\$0.1100	Ī
St. Lucia	\$0.2050	\$0.0410	United Arab Emirates	\$0.2350	Ī
St. Maarten	\$0.1650	\$0.0330	United Kingdom	\$0.0300	Ī
St. Pierre Miquelon	\$0.2150	\$0.0430	Uruguay	\$0.2450	Ī
St. Vincent/Grenadines	\$0.2450	\$0.0490	Uzbekistan	\$0.1850	Ī
Sudan	\$0.5400	\$0.1080	Vanuatu	\$1.0150	Ī
Suriname	\$0.4850	\$0.0970	Vatican City	\$0.0900	Ī
Swaziland	\$0.2950	\$0.0590	Venezuela	\$0.0950	Ī
Sweden	\$0.0600	\$0.0120	Vietnam	\$0.2450	Ī
Switzerland	\$0.0600	\$0.0120	Wallis Futuna	\$1.1400	Ī
Syrian Arab Republic	\$0.4900	\$0.0980	Western Samoa	\$0.4000	I
Taiwan	\$0.0300	\$0.0060	Yemen Arab Republic	\$0.4100	I
Tajikistan	\$0.1750	\$0.0350	Zaire	\$0.3050	Ī
Tanzania	\$0.3950	\$0.0790	Zambia	\$0.3300	I
Thailand	\$0.1050	\$0.0210	Zimbabwe	\$0.3050	I
Togo	\$0.4300	\$0.0860			ĺ

#### SECTION 12 - GRANDFATHERED SERVICES

- 12.5 Type 4 Grandfathered Services (continued)
  - 12.5.20 Business Preferred Rate Plan (Grandfathered 12/31/2011)
    - (A) This plan is a direct dialed outbound long distance service offered to business Customers from switched access lines. Service is offered on a flat rate basis, twenty-four hours per day, seven days per week. Customers must dial 1 plus the area code, if applicable, and the terminating number to complete a call utilizing this service.
    - (B) All calls are billed in six (6) second increments after an initial period, for billing purposes, of thirty (30) seconds. This service is not offered on an intraLATA only basis.
    - (C) Customers of this plan must subscribe to local exchange service provided by the Company's Affiliated ILEC.
    - (D) Rates and Charges
      - .1 MRC

The MRC is \$60.00.

The MRC will be waived in instances where the Customer's combined monthly billing for intrastate and interstate usage under the plan is \$25.00 or more. Surcharges, taxes and other similar fees shall not be included in the calculation for the waiving of the MRC.

Original Page 68

Effective: September 1, 2019

.2 Per Minute Usage Rates – Direct Dial

Initial 30 Seconds	Each Add'l 6 Seconds		
\$0.3100	\$0.0620		

### .3 Discounts

Customers will be given a discount off of total monthly usage charges billed based upon the combined actual monthly intrastate and interstate usage charges billed under the plan. Surcharges, taxes and other similar fees shall not be included in the calculation for determination of the monthly usage.

Discount Percentages

<b>Monthly Usage Charge</b>	Discount
\$0.00 - \$49.99	0.00%
\$50.00 - \$199.99	2.89%
\$200.00 - \$200.01 +	5.80%

#### **SECTION 12 - GRANDFATHERED SERVICES**

- 12.5 Type 4 Grandfathered Services (continued)
  - 12.5.21 Business Appreciation Plan (Grandfathered 12/31/2011)
    - (A) The Business Appreciation plan is a direct dialed outbound long distance service offered to business Customers from switched access lines. Service is offered on a flat rate basis, twenty-four hours per day, seven days per week. Customers must dial 1 plus the area code, if applicable, and the terminating number to complete a call utilizing this service.

Original Page 69

- (B) Customers to this plan must, at the same time as they subscribe to this service, also place a new service order for a 36-month local service term agreement or a Complete Choice for Business<sup>®</sup> Elite Promotion Election agreement, for at least 1-3 business lines, from the Company's Affiliated ILEC.
- (C) All calls are billed in six (6) second increments after an initial period, for billing purposes, of thirty (30) seconds. This service is not offered on an intraLATA only basis.
- (D) The MRC is \$1.00 and the per minute usage rate is \$0.0600.
- 12.5.22 Business Appreciation II Plan (Grandfathered 12/31/2011)
  - (A) The Business Appreciation II plan is a direct dialed outbound long distance service offered to business Customers from switched access lines. Service is offered on a flat rate basis, twenty-four hours per day, seven days per week. Customers must dial 1 plus the area code, if applicable, and the terminating number to complete a call utilizing this service.
  - (B) Customers to this plan must, at the same time as they subscribe to this service, also place a new service order for a 36-month local service term agreement or a Complete Choice for Business® Elite Promotion Election agreement, for 4 or more business lines, from the Company's Affiliated ILEC.
  - (C) All calls are billed in six (6) second increments after an initial period, for billing purposes, of thirty (30) seconds. This service is not offered on an intraLATA only basis.
  - (D) The MRC is \$1.00 and the per minute usage rate is \$0.0500.

#### SECTION 12 - GRANFATHERED SERVICES

- 12.5 Type 4 Grandfathered Services (continued)
  - 12.5.23 Business Platinum Preferred Rate Plan (Grandfathered 12/31/2011)
    - (A) The Business Platinum Preferred Rate plan is a direct dialed outbound long distance service offered to business Customers from switched access lines. Service is offered on a flat rate basis, twenty-four hours per day, seven days per week. Customers must dial 1 plus the area code, if applicable, and the terminating number to complete a call utilizing this service.

Original Page 70

Effective: September 1, 2019

- (B) All calls are billed in six (6) second increments after an initial period, for billing purposes, of thirty (30) seconds. This service is not offered on an intraLATA only basis.
- (C) Customers to this service are required to commit to a minimum monthly commitment (MMC). The Customer will be billed the MMC level if actual usage is below the commitment level in any month. Usage over the MMC level may not be carried over to future months to satisfy any commitment level in those months.
- (D) Rates and Charges
  - .1 MMC

The MMC is \$250.00.

Customers will be given a discount off of total monthly charges billed based upon the actual monthly usage billed for each month. Monthly usage is calculated utilizing a combination of intrastate usage, interstate usage, Directory Assistance charges or Operator Services usage and surcharges. Other surcharges, taxes and similar fees shall not be included in the calculation for determination of the monthly usage.

- .2 The per minute usage rate is \$0.0630.
- .3 Discounts

Customers will be given a discount off of total monthly charges billed based upon the actual monthly usage charges billed. Monthly usage is calculated utilizing a combination of intrastate or interstate usage. Surcharges, taxes and other similar fees shall not be included in the calculation for determination of the monthly usage. Usage over the minimum monthly commitment level may not be carried over to future months to satisfy any commitment level in those months.

Discount Percentages

<b>Monthly Usage Charge</b>	Discount
\$0.00 - \$499.99	0.00%
\$500.00 - \$999.99	3.18%
\$1,000.00 - \$1,999.99	6.35%
\$2,000.00 - \$2,000.01 +	7.94%

#### SECTION 12 - GRANDFATHERED SERVICES

- 12.5 Type 4 Grandfathered Services (continued)
  - 12.5.24 Business Monthly Saver Plan II (Grandfathered 12/31/2011)
    - (A) The Business Monthly Saver plan II is a direct dialed outbound long distance service offered to business Customers from switched access lines. Service is offered on a flat rate basis, twenty-four hours per day, seven days per week. Customers must dial 1 plus the area code and the terminating number to complete a call utilizing this service.

Original Page 71

Effective: September 1, 2019

- (B) All calls are billed in six (6) second increments after an initial period, for billing purposes, of thirty (30) seconds. This service is not offered on an intraLATA only basis.
- (C) Customers to this service are required to commit to a BellSouth Long Distance term plan of one year or longer and an MMC charge. The Customer will be billed the MMC amount if actual usage is below the commitment amount in any month. Usage over the MMC amount may not be carried over to future months to satisfy any commitment level in those months. The commitment amount can be reached by any combination of intrastate usage, interstate usage, international usage, Directory Assistance charges or Operator Services usage and surcharges. Other surcharges, taxes and similar fees shall not be included in the calculation for the determination of whether the minimum monthly commitment amount has been met. Term discounts apply to actual intrastate and interstate usage only.

### (D) MMC Rates

<b>Minimum Monthly Commitment</b>	Per Minute Usage Rate
\$500.00	\$0.0600
\$1,000.00	\$0.0580
\$2,000.00	\$0.0570

#### SECTION 12 - GRANDFATHERED SERVICES

- 12.5 Type 4 Grandfathered Services (continued)
  - 12.5.25 Business Value Plan (Grandfathered 12/31/2011)
    - (A) The Business Value Plan is a direct dialed outbound long distance service offered to Business Customers on switched access lines. Service is offered on a flat rate basis, twenty-four hours per day, seven days per week. Customers must dial 1 plus the area code, if applicable, and the terminating number to complete a call utilizing this service.

Original Page 72

- (B) The monthly recurring charge for this plan will be waived for Customers who subscribe to Complete Choice® for Business Plan provided by the Company's Affiliated ILEC, a local term agreement provided by the Company's Affiliated ILEC, or a Company term agreement.
- (C) All calls are billed in one minute (1) increments after an initial period, for billing purposes, of one (1) minute. This service is not offered on an intraLATA only basis.
- (D) The MRC is \$49.00, and the per-minute usage rate is \$2.45.

#### SECTION 12 - GRANDFATHERED SERVICES

- 12.5 Type 4 Grandfathered Services (continued)
  - 12.5.26 AT&T Business Unlimited Calling<sup>SM</sup> III (Grandfathered 5/31/2012)
    - (A) AT&T Business Unlimited Calling<sup>SM</sup> III is an outbound calling and optional inbound long distance calling plan. This plan is established at the BTN level; multiple BTN aggregation is not available with this plan. If the Customer or Applicant selects a different long distance calling plan for specific WTNs, the Customer is required to establish a separate BTN for each plan. This Plan is available for the following Business Customers who:

#### OPTION 1 - NEW OR EXISTING

.1 utilize Switched Access to reach the long distance network for outbound calling and to receive calls from the long distance network for AT&T Long Distance Toll Free<sup>SM</sup> Service;

Original Page 73

- .2 subscribe to the Company for the provision of interstate and intrastate (interLATA and intraLATA) service for outbound long distance calling on each AT&T business access line per service location, and may also subscribe to the Company for the provision of toll-free service for inbound long distance calling;
- .3 newly subscribe to or currently subscribe to AT&T Business Local Calling Assurance<sup>SM</sup> from Affiliated ILEC of the Company on at least one (1), but no more than four (4), business access lines (*i.e.*, WTNs) on a single BTN account;
- .4 newly subscribe or currently subscribe to one of the following services from an Affiliate of the Company: (a) AT&T U-verse High Speed Internet-Business Edition; (b) FastAccess® Business DSL Lite; (c) FastAccess® Business DSL; (d) FastAccess® Business DSL Plus; or (e) FastAccess® Business DSL 6.0;
- newly subscribe or currently subscribe to one of the following services from an Affiliate of the Company with a minimum of two (2) years term commitment:
  (a) wireless voice; (b) wireless data; (c) wireless voice and data; or (d) wireless laptop card;
- .6 agree the Company may combine its wireless Affiliate's billing with the Company's wireless billing, except when Customers are participating in the wireless Affiliate's SBS Wireless Deposit Waiver Program; and,
- .7 commit to subscribe to this plan under a 2-year (24 Month) AT&T All for Less Subscriber Agreement.

#### SECTION 12 - GRANDFATHERED SERVICES

- 12.5 Type 4 Grandfathered Services (continued)
  - 12.5.26 AT&T Business Unlimited Calling<sup>SM</sup> III (continued)
    - (A) This plan is available for the following Business Customers who: (continued)

#### OPTION 2 - WINBACK

.1 utilize Switched Access to reach the long distance network for outbound calling and to receive calls from the long distance network for AT&T Long Distance Toll Free<sup>SM</sup> Service;

Original Page 74

- .2 subscribe to the Company for the provision of interstate and intrastate (interLATA and intraLATA) service for outbound long distance calling on each AT&T business access line per service location, and may also subscribe to the Company for the provision of toll-free service for inbound long distance calling;
- .3 Meet a minimum of one (1) of the following conditions:
  - .a previously subscribed to local dial tone service of an Affiliated ILEC of the Company and have cancelled that Service; or
  - be a current local telephone Customer in the Affiliated ILEC local territory who is now moving dial tone services from a competitor of the Company to the Affiliated ILEC;
- .4 newly subscribe to or currently subscribe to AT&T Business Local Calling Assurance<sup>SM</sup> from Affiliated ILEC of the Company on at least one (1), but no more than four (4), business access lines (*i.e.*, WTNs) on a single BTN account;
- .5 newly subscribe or currently subscribe to one of the following services from an Affiliate of the Company: (a) AT&T U-verse High Speed Internet-Business Edition; (b) FastAccess® Business DSL Lite; (c) FastAccess® Business DSL; (d) FastAccess® Business DSL Plus; or (e) FastAccess® Business DSL 6.0;
- .6 newly subscribe or currently subscribe to one of the following services from an Affiliate of the Company with a minimum of two (2) years commitment:
  - (a) wireless voice; (b) wireless data; (c) wireless voice and data; or (d) wireless laptop card; or,
  - .b (a) AT&T Tech Support 360<sup>SM</sup> Advanced Service Package and AT&T Web Hosting<sup>SM</sup> Shared Hosting; or (b) AT&T Tech Support 360<sup>SM</sup> Premium Service Package and AT&T Tech Support 360<sup>SM</sup> Back-up and Go
- .7 agree the Company may combine its wireless Affiliate's billing with the Company's wireless billing, except when Customers are participating in the wireless Affiliate's SBS Wireless Deposit Waiver Program; and,
- .8 commit to subscribe to this plan under a 2-year (24 Month) AT&T All for Less Subscriber Agreement.

#### SECTION 12 - GRANDFATHERED SERVICES

12.5 Type 4 – Grandfathered Services (continued)

12.5.26 AT&T Business Unlimited Calling<sup>SM</sup> III (continued)

(B) A single business entity with more than one BTN at the business entity's physical service location is eligible for this plan on one (1) BTN only, provided the total number of AT&T business access lines do not exceed ten (10) at that location. Additionally, service under this plan is limited to one BTN with one (1) to four (4) business access lines per legal business entity.

Original Page 75

- (C) The Customer may subscribe to this plan for outbound Service only, or for both outbound Service and inbound toll-free calling. Customers subscribing to the plan receive unlimited domestic 1+ direct-dialed outbound calling. Where a Customer subscribes to AT&T Long Distance Toll Free<sup>SM</sup> Service, inbound toll-free calls may originate on any type of access and are terminated via Switched Access to the Customer's location. (See Section 4.4 and Section 3 of this pricing and service guide for optional feature, rules and regulations, and general information regarding AT&T Long Distance Toll Free<sup>SM</sup> Service).
- (D) Inbound switched toll-free calls and fully automated operator assisted, and operator dialed calls billed to the calling card are not included in the unlimited MOUs. Toll free calls are billed in increments of six (6) seconds subject to a minimum connect time (initial period) of thirty (30) seconds. Calling card calls are billed in one (1) minute increments subject to a minimum connect time (initial period) of one (1) minute. Toll free calls and calling card per minute rates and per call charges associated with this plan are subject to change with prior notification to the Customer.
- (E) Call Detail is available as an optional feature to this plan. For an additional MRC, Customers who select this option will receive itemized call detail of their zero-rated calls. Activation and deactivation of the feature will begin within the next billing cycle following a Customer's request. Customers may add or remove the Call Detail optional feature from this plan at no charge.
- (F) AT&T All for Less Subscriber 2-Year (24 Month) Agreement
  - .1 If during the 2-Year term the Customer adds or removes access lines under this plan, the 2-Year team will restart for a new 2-Year term. Customers who remain on this plan at the expiration of the 2-Year term will be billed on a month-to-month basis at the rates defined in Section 12.5.26 (H).
  - .2 If prior to the expiration date of the 2-Year agreement the Customer chooses to terminate long distance service with the Company, the Customer may be charged the following Early Termination Fee (ETF) based on Option 1, Option 2 or Option 3 defined above:
    - Option 1 \$5.00 multiplied by the number of months remaining in the term
    - Option 2 \$2.50 multiplied by the number of months remaining in the term
    - Option 3 \$2.50 multiplied by the number of months remaining in the term
  - .3 If the Company no longer offers this plan and if the Customer: (a) moves to a new location; or, (b) changes the number of access lines under the BTN(s) to which this plan applies, then Customer's service under this plan will terminate, and Customer will not be subject to termination charges.

#### SECTION 12 - GRANDFATHERED SERVICES

12.5 Type 4 – Grandfathered Services (continued)

12.5.26 AT&T Business Unlimited Calling<sup>SM</sup> III (continued)

#### (G) Restrictions

.1 All business access lines and/or working lines under the participating BTN must be provisioned on this plan and term discounts do not apply to this plan.

Original Page 76

- .2 This plan is provided for standard voice calling involving live dialog between individuals. This plan is not intended for use as a substitute for dedicated or open circuits, or similar applications. This plan may not be used for data connections, including but not limited to, modem to modem calls, remote access applications, Internet access, or Intranet access, including access to corporate LANs. Additionally, this plan may not be used for dedicated point-to-point connections between equipment that leaves a circuit connected without a contemporary, continuous voice communication. This plan may not be used for any of the following: auto dialers; PBX trunks; ground start line or trunks; ISDN service, including PRI; foreign exchange services; Remote Call Forwarding/Telebranch Service; public telephone service; public access smartpay phones; analog to digital conversion; digital PBX service, including local access provisioned via T-1 facilities; WATS service; PBX/PABX/EABX services; non-square electronic key telephone systems; hybrid key telephone systems; predictive calling/dialing systems; automatic outbound dialing systems; any type of automatic call distribution system; or the functional equivalent of any such systems listed above. This plan may be used for fax transmissions, excluding broadcast fax applications. If the Company determines that the Customer is in violation of the above listed restrictions, the Customer shall forfeit eligibility for rates under this plan and will be moved to Business MTS unless an alternative plan is selected by the Customer. If, during a billing cycle, the Customer has more than 10 calls to any single number of a duration of more than 3 hours each or any single call of greater than 10 hours of duration, the Customer will be presumed to be in violation of these restrictions. In such case, Customer may be asked to provide reasonable proof to the Company that it is not using the Service for a prohibited purpose. Failure to provide such proof to the Company shall be treated as a violation of the terms and conditions of this Plan.
- .3 If the Company determines that the Customer is in violation of any of the restrictions listed above and/or the total number of Access Lines exceeds the maximum of ten (10), the Customer shall forfeit eligibility for rates under this plan and will be moved to Business MTS unless an alternative plan is selected by the Customer. If moved to Business MTS, the rates associated with Business MTS will apply in lieu of the rates specified in (H) below.

#### SECTION 12 - GRANDFATHERED SERVICES

12.5 Type 4 – Grandfathered Services (continued)

12.5.26 AT&T Business Unlimited Calling<sup>SM</sup> III (continued)

(H) Rates and Charges

.1 The outbound long distance calling MRC is as follows:

Option 1

AT&T All for Less 2-Year Agreement: \$10.00 per access line Month-to-Month: \$20.00 per access line

Original Page 77

Effective: September 1, 2019

Option 2

AT&T All for Less 2-Year Agreement: \$5.00 per access line Month-to-Month: \$20.00 per access line

Option 3

AT&T All for Less 2-Year Agreement: \$5.00 per access line Month-to-Month: \$20.00 per access line

.2 The inbound toll-free service per minute usage rate is \$0.055. See Section 4.4 of this pricing and service guide for the AT&T Long Distance Toll Free<sup>SM</sup> Service MRC.

.3 Call Detail

The MRC is \$5.00/line up to four (4) lines.

#### SECTION 12 - GRANDFATHERED SERVICES

#### 12.5 Type 4 – Grandfathered Services (continued)

12.5.27 AT&T Business Block of Time<sup>SM</sup> 500 II (Grandfathered 12/12/2014)

AT&T Business Block of TimeSm 500 II plan is a bundled intrastate/intersate outbound calling and inbound long distance calling plan.

(A) This plan is available to new or existing Business Customers who:

#### **NEW OR EXISTING**

.1 utilize Switched Access to reach the long distance network for outbound calling and to receive calls from the long distance network for toll free calling;

Original Page 78

Effective: September 1, 2019

- .2 subscribes to the Company for interLATA (interstate and intrastate) and intraLATA (local toll) Service for outbound long distance calling on each access line, and may also subscribe to the Company's TFS for inbound long distance calling;
- .3 subscribe to AT&T Business Local Calling<sup>SM</sup> or Complete Choice<sup>®</sup> for Business package provided by an Affiliated ILEC of the Company with a new, or restart an existing, 1-Year term agreement; and,
- .4 meets one of the following:

#### OPTION 1

newly subscribe to one of the following services from an Affiliate of the Company for a 1 or 2-Year term commitment: (a) AT&T U-verse High Speed Internet-Business Edition; (b) FastAccess® Business DSL Lite; (c) FastAccess® Business DSL; (d) FastAccess® Business DSL Plus; or (e) FastAccess® Business DSL 6.0; or

#### **OPTION 2**

newly subscribes to one of the following for a two (2) year term: (a) Wireless Voice; (b) Wireless Data (c) Wireless Voice and Data; or (d) Wireless Laptop card, and agree the Company may combine the Affiliate of the Company's wireless billing with the Company's wireline billing except when participating in the Affiliate of the Company's SBS Wireless Deposit Waiver Program; or,

#### **OPTION 3**

Currently subscribe to Internet Service from an Affiliate of the Company and agree to update current Internet Service to AT&T U-verse High Speed Internet-Business Edition Elite (6Mbps) or higher.

- .5 specify at the time of ordering if the MOUs are to be used for outbound calling, or both outbound and inbound toll-free calling; and,
- .6 request to be provisioned under this plan.

#### SECTION 12 - GRANDFATHERED SERVICES

- 12.5 Type 4 Grandfathered Services (continued)
  - 9.5.27 AT&T Business Block of Time<sup>SM</sup> 500 II (continued)
    - (A) This plan is available to new or existing Business Customers who: (continued)

### RETENTION

.1 utilize Switched Access to reach the long distance network for outbound calling and optionally, to receive calls from the long distance network for TFS;

Original Page 79

Effective: September 1, 2019

- .2 subscribes to the Company for interLATA (interstate and intrastate) and intraLATA (local toll) Service for outbound long distance calling on each access line and may also subscribe to the Company's TFS for inbound long distance calling;
- .3 currently subscribe to local dial tone service from an Affiliated ILEC of the Company (including where that Affiliate is operating as a CLEC) and state an intention to change local carriers for economic reasons and/or in response to a competitive offer;
- .4 subscribe to AT&T Business Local Calling<sup>SM</sup> or Complete Choice<sup>®</sup> for Business package provided by an Affiliated ILEC of the Company with a new, or restart an existing, 1-Year term agreement;
- .5 currently subscribes to Internet Service or Wireless Service as defined and offered by an Affiliate of the Company and/or Affiliated Wireless provider of the Company; and,
- .6 request to be provisioned under this plan.

#### (B) Rates and Charges

This plan provides the Customer with a 500 minute block of time for placing (1+) Direct-Dialed domestic outbound long distance calling or for both (1+) Direct-Dialed domestic outbound calling and inbound toll free service calls. The MRC is \$25.00, and the outbound intrastate/interstate and/or inbound toll-free calling per-minute usage rate is \$0.057 for calls completed or received after the 500 block of time has been used.

#### SECTION 12 - GRANDFATHERED SERVICES

- 12.5 Type 4 Grandfathered Services (continued)
  - 12.5.28 AT&T Business Unlimited Calling<sup>SM</sup> V (Grandfathered 12/12/2014)
    - (A) AT&T Business Unlimited Calling SM V is an outbound calling and optional inbound long distance calling plan. This plan is established at the BTN level; multiple BTN aggregation is not available with this plan. If the Customer selects a different long distance calling plan for specific WTNs, those WTNs cannot be included under the BTN account(s) used for this plan. This plan is available to Business Customers in Alabama, Florida, Georgia, Kentucky, Louisiana, Mississippi, North Carolina, South Carolina and Tennessee who:
      - .1 NEW OR EXISTING ONE (1) TO TEN (10) ACCESS LINES
        - utilize Switched Access to reach the long distance network for outbound calling and optionally, to receive calls from the long distance network for TFS;

Original Page 80

Effective: September 1, 2019

- .b subscribe to the Company for interLATA (interstate and intrastate) and intraLATA (local toll) Service for outbound long distance calling on each AT&T business access line per service location for a one (1) or two (2) year term agreement, and may also subscribe to the Company's TFS for inbound long distance calling;
- .c currently or newly subscribe to and maintain at least one (1) but no more than ten (10) business access lines (*i.e.*, BTNs and WTNs) from an Affiliated ILEC of the Company on a single BTN account per service location;
- .d subscribe to this plan only on business access lines subscribing to AT&T Business Local Calling SM or Complete Choice® for Business provided by an Affiliated ILEC of the Company with a new, or restart an existing, 1-Year term agreement; and,
- .e meet one of the following:

#### OPTION 1

newly subscribe to one of the following services from an Affiliate of the Company for a 1 or 2-Year term commitment: (a) AT&T U-verse High Speed Internet-Business Edition; (b) FastAccess® Business DSL Lite; (c) FastAccess® Business DSL; (d) FastAccess® Business DSL Plus; or (e) FastAccess® Business DSL 6.0; or,

#### SECTION 12 - GRANDFATHERED SERVICES

- 12.5 Type 4 Grandfathered Services (continued)
  - 12.5.28 AT&T Business Unlimited Calling<sup>SM</sup> V (continued)
    - (A) This plan is available to Business Customers in Alabama, Florida, Georgia, Kentucky, Louisiana, Mississippi, North Carolina, South Carolina and Tennessee who: (continued)
      - .1 NEW OR EXISTING ONE (1) TO TEN (10) ACCESS LINES (continued)
        - .e meets one of the following: (continued)

#### **OPTION 2**

.a newly subscribe to one of the following services from an Affiliate of the Company with a minimum of a 2-Year term commitment: (a) wireless voice; (b) wireless data; (c) wireless voice and data; or (d) wireless laptop card; and,

Original Page 81

Effective: September 1, 2019

.b agree the Company may combine its wireless Affiliate billing with the Affiliated ILEC billing except when Customers are participating in the wireless Affiliate's SBS Wireless Deposit Waiver Program; or,

#### **OPTION 3**

Currently subscribe to Internet Service from an Affiliate of the Company and agree to upgrade current Internet service to AT&T Uverse High Speed Internet-Business Edition Elite (6Mbps) or higher; or,

#### **OPTION 4**

- .a currently subscribe to: (a) AT&T Business Unlimited Calling<sup>SM</sup>, Business Unlimited Plan, AT&T Business Unlimited Calling<sup>SM</sup> III, or AT&T Business Unlimited Calling<sup>SM</sup> IV;
- .b currently subscribe to Complete Choice® for Business or AT&T Business Local Calling Assurance<sup>SM</sup> from an Affiliated ILEC of the Company;
- .c currently subscribe to Internet service as defined and offered by an Affiliate of the Company and/or wireless service provided by an Affiliate Wireless provider of the Company; and,
- .d request to be provisioned under this plan.

#### SECTION 12 - GRANDFATHERED SERVICES

- 12.5 Type 4 Grandfathered Services (continued)
  - 12.5.28 AT&T Business Unlimited Calling<sup>SM</sup> V (continued)
    - (A) This plan is available to Business Customers in Alabama, Florida, Georgia, Kentucky, Louisiana, Mississippi, North Carolina, South Carolina and Tennessee who: (continued)
      - .1 NEW OR EXISTING ONE (1) TO TEN (10) ACCESS LINES (continued)
        - .e meets one of the following: (continued)

#### **OPTION 5**

a gree to re-subscribe to this plan for a new one (1) or two (2) year term agreement at the end of any term or re-subscription term; and

Original Page 82

- .b continues to meet all other requirements in Section A.1 .a-.d;
   and
- currently or newly subscribes to the Services in Section A.1.e Option 1 or Option 2 from an Affiliate of the Company and currently or newly subscribes to one of the following service packages from Affiliate(s) of the Company:
  - .i AT&T Tech Support 360SM Backup and Go with AT&T Website Solutions  $^{\rm SM}$ , or AT&T Tech Support  $360^{\rm SM}$  Advanced; or
  - .ii AT&T Tech Support 360<sup>SM</sup> Premium Service with AT&T Tech Support 360<sup>SM</sup> Backup and Go Premium and AT&T Website Solutions<sup>SM</sup>

#### **SECTION 12 - GRANDFATHERED SERVICES**

- 12.5 Type 4 Grandfathered Services (continued)
  - 12.5.28 AT&T Business Unlimited Calling<sup>SM</sup> V (continued)
    - (A) This plan is available to Business Customers in Alabama, Florida, Georgia, Kentucky, Louisiana, Mississippi, North Carolina, South Carolina and Tennessee who: (continued)
      - .2 RETENTION ONE (1) to FOUR (4) ACCESS LINES
        - utilize Switched Access to reach the long distance network for outbound calling and optionally, to receive calls from the long distance network for TFS;

Original Page 83

- .b subscribe to the Company for interLATA (interstate and intrastate) and intraLATA (local toll) Service for outbound long distance calling on each access line per service for a one (1) or two (2) year term, and may also subscribe to the Company's TFS for inbound long distance calling;
- .c currently subscribe to local dial tone service from an Affiliated ILEC of the Company (including where that Affiliate is operating as a CLEC) and state an intention to change local carriers for economic reasons and/or in response to a competitive offer;
- .d currently subscribe to and maintain at least one (1) but no more than four (4) business access lines (*i.e.*, BTNs and WTNs) from an Affiliated ILEC of the Company on a single BTN account per service location;
- .e subscribe to this plan on business access lines subscribing to AT&T Business Local Calling<sup>SM</sup> or Complete Choice® for Business package from an Affiliated ILEC of the Company with a new, or restart an existing, 1-Year term agreement;
- f. currently subscribed to Internet Service or Wireless Service as defined and offered by an Affiliate of the Company and/or Affiliated Wireless provider of the Company: and,
- .g request to be provisioned under this plan.
- (B) A single business entity with more than one BTN at the business entity's physical service location is eligible for this plan on one (1) BTN only, provided the total number of AT&T business access lines do not exceed ten (10) at that location. Additionally, service under this plan is limited to one BTN with one (1) to ten (10) business access lines per legal business entity.
- (C) The Customer may subscribe to this plan for outbound Service only, or for both outbound Service and inbound toll-free calling. Customers subscribing to the plan receive unlimited domestic 1+ direct-dialed outbound calling. Where a Customer subscribes to TFS, inbound toll-free calls may originate on any type of access and are terminated via Switched Access to the Customer's location. (See Section 3 and Section 4.4 of this pricing and service guide for optional feature, rules and regulations, and general information regarding TFS).

#### SECTION 12 - GRANDFATHERED SERVICES

12.5 Type 4 – Grandfathered Services (continued)

12.5.28 AT&T Business Unlimited Calling<sup>SM</sup> V (continued)

(D) TFS calls, multi-party conference calls (except those calls placed by using the Three-Way Calling feature are included), calls to 900, 976, 700 numbers or other calls to access information services, directory assistance, operator services and international calling are not included in the unlimited minute of use. TFS calls are billed in increments of six (6) seconds subject to a minimum connect time (initial period) of thirty (30) seconds. TFS calls per minute rates associated with this plan are subject to change with prior notification to the Customer.

Original Page 84

- (E) Call Detail is available as an optional feature to this plan. For an additional MRC, Customers who select this option will receive itemized call detail of their zero-rated calls. Activation and deactivation of the feature will begin within the next billing cycle following a Customer's request. Customers may add or remove the Call Detail optional feature from this plan at no charge.
- (F) Term Agreement
  - .1 1-Year term agreements may be oral agreements that do not require a signed Agreement. 2-year term agreements require a signed Agreement.
  - .2 Written agreements must be signed under any method accepted by the Company, including electronically, and must be received by the Company within thirty (30) calendar days from the order date. If the Company does not receive the Customer's signed agreement within the thirty (30) calendar days, and the Service has been provisioned in reliance on the Customer's order for Service, the Service will be provided on a month-to-month basis and the Customer will be billed the month-to-month rates specified in Section 4.6.25(I).
  - .3 The 1-Year agreement includes an option to re-subscribe for up to two additional 1-Year terms. AT&T will provide Customer a confirmation letter outlining the details of the agreement. In addition, AT&T will provide subsequent notifications regarding the upcoming re-subscription option at least 60 days prior to the expiration date of the initial and first renewal term (if applicable); and, unless Customer requests otherwise before the expiration date, a new 1-Year renewal term will commence under the same terms and conditions. Should Customer decide to disconnect the Service prior to the expiration of any 1-Year term, Customer may be liable for early termination charges. At the end of the second renewal, the Customer will be billed the month-to-month rates specified in Section 12.5.28 (I).
  - .4 If the Customer selects the 2-Year term, at the expiration of the 2-Year term the Customer will be billed the month-to-month rates specified in Section 12.5.28 (I).
  - .5 If the Company no longer offers this Plan, and if the Customer moves to a new location or changes the number of access lines under the BTN(s) to which this Plan applies then the Customer's Service under the Plan will terminate, and the Customer will not be subject to termination charges.

#### SECTION 12 - GRANDFATHERED SERVICES

12.5 Type 4 – Grandfathered Services (continued)

12.5.28 AT&T Business Unlimited Calling<sup>SM</sup> V (continued)

- (G) Early Termination Fee
  - .1 If prior to the expiration date of the initial or any renewal term, Customer chooses to either:
    - .a terminate long distance service with the Company: or,
    - .b change its calling plan from the AT&T Business Unlimited Calling<sup>SM</sup>
      V prior to the expiration of the Term Agreement, the Customer may be charged an early termination fee ("ETF"). The ETF shall be \$5.00 multiplied by the number of months remaining in the term. Customer may change its calling plan to the AT&T Business Block of Time II Plan (rate option 500 minutes or greater) as specified in Section 12.X of this pricing and service guide; in such case, ETF shall not apply.

Original Page 85

- .2 The Company will not charge an ETF if Customer cancels this Plan if at the same time the Customer agrees to replace some or all their existing service with Internet Protocol (IP) service, Wireless, or any functionally equivalent service from an Affiliate of the Company for the purpose of placing outbound and/or inbound live voice communications outside of the Customer's local calling area.
- (H) This plan is provided for standard voice calling involving live dialog between two individuals. This plan is not intended for use as a substitute for dedicated or open circuits, or similar applications. This plan is not available for resale. This plan may not be used for data connections, including but not limited to, modem to modem calls, remote access applications, Internet access, or Intranet access, including access to corporate LANs. Additionally, this plan may not be used for dedicated point-to-point connections between equipment that leaves a circuit connected without a contemporary, continuous voice communication. This plan may not be used for any of the following: auto dialers; PBX trunks; ground start line or trunks; ISDN service, including PRI; foreign exchange services; Remote Call Forwarding/Telebranch Service; public telephone service; public access smart-pay phones; analog to digital conversion; digital PBX service, including local access provisioned via T-1 facilities; WATS service; PBX/PABX/EABX services; non-square electronic key telephone systems; hybrid key telephone systems; predictive calling/dialing systems; automatic outbound dialing systems; any type of automatic call distribution system; or the functional equivalent of any such systems listed above. This plan may be used for fax transmissions, excluding broadcast fax applications. If the Company determines that the Customer is in violation of the above listed restrictions, the Customer shall forfeit eligibility for rates under this plan and will be moved to Business MTS unless an alternative Plan is selected by the Customer. If, during a billing cycle, the Customer has more than 10 calls to any single number of a duration of more than 3 hours each or any single call of greater than 10 hours of duration, the Customer will be presumed to be in violation of these restrictions. In such case, Customer may be asked to provide reasonable proof to the Company that it is not using the Service for a prohibited purpose. Failure to provide such proof to the Company shall be treated as a violation of the terms and conditions of this plan.

### SECTION 12 - GRANDFATHERED SERVICES

- 12.5 Type 4 Grandfathered Services (continued)
  - 12.5.28 AT&T Business Unlimited Calling<sup>SM</sup> V (continued)
    - (I) Rates and Charges
      - .1 The outbound long distance calling MRC is as follows:

1-Year Agreement \$10.00 per access line 2-Year Agreement: \$10.00 per access line Month-to-Month \$10.00 per access line

.2 The inbound toll-free service per minute usage rate is \$0.055. See Section 4.4 of this pricing and service guide for the AT&T Long Distance Toll Free<sup>SM</sup> Service MRC.

Original Page 86

Effective: September 1, 2019

- .3 For operator services and directory assistance see Section 8 for rates and charges.
- .4 Call Detail

The MRC is \$5.00/line up to ten (10) lines.

### SECTION 12 - GRANDFATHERED SERVICES

- 12.5 Type 4 Grandfathered Services (continued)
  - 12.5.29 AT&T<sup>®</sup> United Kingdom and Germany Value Plus Plan (Grandfathered 12/01/2005)
    - (A) The AT&T® United Kingdom and Germany Value Plus Plan is a direct dialed outbound international long distance service offered to residential Customers on switched access lines. Service is offered on a flat rate basis, twenty-four hours per day, seven days per week. Customers must dial 1 plus the country code or area code, if applicable, and the terminating number to complete a call utilizing this service.

Original Page 87

- (B) For a MRC, the Customer receives a block of 3,000 minutes for placing (1) 1+ Direct-Dialed International calls originating in the United States from a Customer's presubscribed line and terminating to Germany and United Kingdom. All usage in excess of the block of time will be billed at a fixed rate per minute. Calls are billed in one (1) minute increments after an initial period, for billing purposes, of one (1) minute. This plan does not provide call detail information on the Customer's monthly bill for calls to Germany, except for applicable international mobile termination charges. Call detail information for international calls to other countries will be provided. This plan is only offered in conjunction with the Company's Domestic Service. Customers may not subscribe to more than one international calling plan on the same account.
- (C) Per minute charges apply to calls to other international countries based on the country called and the duration of the call in addition to the MRC. An international wireless termination charge applies to all applicable calls.
- (D) In order to be eligible for this plan, the Company must be able to verify that the Customer meets the following eligibility requirements. Customers who no longer meet these eligibility requirements will not be eligible for this plan and may be offered an alternative plan or have their service suspended, restricted or canceled without notice.
  - .1 This plan is available to Customers with one to three lines at a location.
  - .2 Customers must subscribe to the Company's Affiliated ILEC.
  - .3 This plan is not available on an account that is the recipient of charges billed from another location.
  - .4 Customer lines associated with educational institutions (colleges, universities, etc.) are not eligible for this plan.
  - .5 The block of time minutes cannot be used for calls that include multi-party conference calls (except those calls placed by using the Three-Way Calling features are included), calls to 900 numbers or other calls to access information services, directory assistance, operator services, international calling to countries other than Germany, domestic calls, and toll-free calling services. This plan is not available for resale.
  - .6 If usage under this plan is not consistent with typical residential Customer usage, at the Company's sole discretion, the Company may offer the Customer an alternative plan or suspend, restrict or cancel the Customer's service without prior notice. Calls that are not consistent with typical residential voice usage include but are not limited to: use for general business purposes, commercial facsimile, auto-dialing, resale, international call forwarding, call centers, long distance access to internet service providers and telemarketing.

BUSINESS AND RESIDENTIAL SERVICES

Original Page 88

Effective: September 1, 2019

# PRICING AND SERVICE GUIDE SECTION 12 - GRANDFATHERED SERVICES

## 12.5 Type 4 – Grandfathered Services (continued)

12.5.29 AT&T® United Kingdom and Germany Value Plus Plan (continued)

### (E) Rates and Charges

The MRC is \$42.95 and the per minute usage rate in \$0.05 for calls completed after the 3,000 block of time has been used. Per-minute rates to countries not included in the block of time are as follows:

Country	Initial Minute	Each Additional Minute	Country	Initial Minute	Each Additional Minute
Afghanistan	\$1.51	\$1.51	Bulgaria	\$0.35	\$0.35
Albania	\$1.00	\$1.00	Burkina Faso	\$0.94	\$0.94
Algeria	\$0.49	\$0.49	Burundi	\$1.68	\$1.68
Andorra	\$0.43	\$0.43	Cambodia	\$2.07	\$2.07
Angola	\$1.31	\$1.31	Cameroon	\$0.49	\$0.49
Anguilla	\$0.53	\$0.53	Canada	\$0.05	\$0.05
Antarctica (Casey Base)	\$1.62	\$1.62	Cape Verde Island	\$0.75	\$0.75
Antarctica (Scott Base)	\$1.62	\$1.62	Cayman Islands	\$0.25	\$0.25
Antigua	\$0.44	\$0.44	Central African Republic	\$1.34	\$1.34
Argentina	\$0.18	\$0.18	Chad	\$2.20	\$2.20
Armenia	\$0.53	\$0.53	Chile	\$0.19	\$0.19
Aruba	\$0.42	\$0.42	China	\$0.10	\$0.10
Ascension Island	\$1.02	\$1.02	Christmas Island	\$0.12	\$0.12
Australia	\$0.12	\$0.12	Cocos Island	\$0.12	\$0.12
Austria	\$0.12	\$0.12	Colombia	\$0.15	\$0.15
Azerbaijan	\$0.77	\$0.77	Comoros Islands	\$3.18	\$3.18
Bahamas	\$0.24	\$0.24	Congo	\$0.68	\$0.68
Bahrain	\$0.59	\$0.59	Cook Islands	\$1.47	\$1.47
Bangladesh	\$0.50	\$0.50	Costa Rica	\$0.26	\$0.26
Barbados	\$0.48	\$0.48	Croatia	\$0.32	\$0.32
Belarus	\$0.32	\$0.32	Cuba	\$0.91	\$0.91
Belgium	\$0.12	\$0.12	Cyprus	\$0.52	\$0.52
Belize	\$0.50	\$0.50	Czech Republic	\$0.26	\$0.26
Benin	\$0.60	\$0.60	Denmark	\$0.12	\$0.12
Bermuda	\$0.19	\$0.19	Diego Garcia	\$1.31	\$1.31
Bhutan	\$1.64	\$1.64	Djibouti	\$0.87	\$0.87
Bolivia	\$0.34	\$0.34	Dominican Republic	\$0.18	\$0.18
Bosnia	\$0.35	\$0.35	Dominica	\$0.35	\$0.35
Botswana	\$0.77	\$0.77	East Timor	\$1.69	\$1.69
Brazil	\$0.14	\$0.14	Easter Island	\$0.19	\$0.19
British Virgin Islands	\$0.43	\$0.43	Ecuador	\$0.32	\$0.32
Brunei	\$0.77	\$0.77	Egypt	\$0.37	\$0.37

## SECTION 12 - GRANDFATHERED SERVICES

## 12.5 Type 4 – Grandfathered Services (continued)

## 12.5.29 AT&T® United Kingdom and Germany Value Plus Plan (continued)

(E) Rates and Charges (continued)

Per-minute rates to countries not included in the block of time are as follows: (continued)

Original Page 89

Country	Initial Minute	Each Additional Minute	Country	Initial Minute	Each Additional Minute
El Salvador	\$0.26	\$0.26	Kiribati	\$1.10	\$1.10
Equatorial Guinea	\$2.31	\$2.31	Korea (South)	\$0.08	\$0.08
Eritrea	\$0.80	\$0.80	Korea (North)	\$1.01	\$1.01
Estonia	\$0.60	\$0.60	Kosovo	\$0.12	\$0.12
Ethiopia	\$0.95	\$0.95	Kuwait	\$0.47	\$0.47
Faeroe Island	\$0.49	\$0.49	Kyrgyzstan	\$0.45	\$0.45
Falkland Island	\$1.08	\$1.08	Laos	\$2.17	\$2.17
Fiji Island	\$0.94	\$0.94	Latvia	\$0.35	\$0.35
Finland	\$0.12	\$0.12	Lebanon	\$0.55	\$0.55
France	\$0.12	\$0.12	Lesotho	\$0.77	\$0.77
French Antilles	\$0.60	\$0.60	Liberia	\$0.59	\$0.59
French Guiana	\$0.60	\$0.60	Libya	\$0.87	\$0.87
French Polynesia	\$1.20	\$1.20	Liechtenstein	\$0.12	\$0.12
Gabon	\$1.04	\$1.04	Lithuania	\$0.35	\$0.35
Gambia	\$0.53	\$0.53	Luxembourg	\$0.12	\$0.12
Georgia	\$0.72	\$0.72	Macao	\$0.70	\$0.70
Germany	IN BLOCK	IN BLOCK	Macedonia	\$0.52	\$0.52
Ghana	\$0.44	\$0.44	Madagascar	\$2.00	\$2.00
Gibraltar	\$0.94	\$0.94	Malawi	\$0.49	\$0.49
Greece	\$0.12	\$0.12	Malaysia	\$0.18	\$0.18
Greenland	\$0.54	\$0.54	Maldives	\$1.29	\$1.29
Grenada	\$0.42	\$0.42	Mali Republic	\$0.95	\$0.95
Guadeloupe	\$0.57	\$0.57	Malta Republic	\$0.66	\$0.66
Guatemala	\$0.26	\$0.26	Marshall Island	\$0.68	\$0.68
Guinea	\$0.75	\$0.75	Mauritania	\$0.90	\$0.90
Guinea-Bissau	\$1.47	\$1.47	Mauritius	\$1.12	\$1.12
Guyana	\$0.76	\$0.76	Mayotte Island (see R	eunion Is	sland)
Haiti	\$0.46	\$0.46	Mexico 1	\$0.10	\$0.10
Honduras	\$0.40	\$0.40	Mexico 2	\$0.10	\$0.10
Hong Kong	\$0.09	\$0.09	Micronesia	\$0.70	\$0.70
Hungary	\$0.26	\$0.26	Moldova	\$0.46	\$0.46
Iceland	\$0.46	\$0.46	Monaco	\$0.12	\$0.12
India	\$0.28	\$0.28	Mongolia	\$1.51	\$1.51
Indonesia	\$0.32	\$0.32	Montenegro	\$0.41	\$0.41
Inmarsat (SNAC)	\$8.10	\$8.10	Montserrat	\$0.50	\$0.50
Iran	\$0.55	\$0.55	Morocco	\$0.53	\$0.53
Iraq	\$0.69	\$0.69	Mozambique	\$1.13	\$1.13
Ireland	\$0.12	\$0.12	Myanmar	\$2.05	\$2.05
Israel	\$0.12	\$0.12	Namibia	\$0.73	\$0.73
Italy	\$0.12	\$0.12	Nauru	\$1.04	\$1.04
Ivory Coast	\$0.68	\$0.68	Nepal	\$1.15	\$1.15
Jamaica	\$0.37	\$0.37	Netherlands	\$0.12	\$0.12
Japan	\$0.09	\$0.09	Netherland Antilles	\$0.40	\$0.40
Jordan	\$0.55	\$0.55	Networks	\$3.95	\$3.95
Kazakhstan	\$0.15	\$0.15	New Caledonia	\$1.29	\$1.29
Kenya	\$0.62	\$0.62	New Zealand	\$0.12	\$0.12

## SECTION 12 - GRANDFATHERED SERVICES

- 12.5 Type 4 Grandfathered Services (continued)
  - 12.5.29 AT&T® United Kingdom and Germany Value Plus Plan (continued)
    - (E) Rates and Charges (continued)

Per-minute rates to countries not included in the block of time are as follows: (continued)

Original Page 90

		Each			Fools
	Initial	Each Additional		Initial	Each Additional
Country	Minute	Minute	Country	Minute	Minute
Nicaragua	\$0.36	\$0.36	St. Maarten	\$0.40	\$0.40
Niger Republic	\$0.84	\$0.84	St. Pierre Miquelon	\$0.45	\$0.45
Nigeria Nigeria	\$0.53	\$0.53	St. Vincent/Grenadines	\$0.51	\$0.51
Niue	\$1.61	\$1.61	South Sudan	\$0.88	\$0.88
Norfolk Island	\$1.62	\$1.62	Sudan	\$0.88	\$0.88
Norway	\$0.12	\$0.12	Suriname	\$1.12	\$1.12
Oman	\$0.95	\$0.95	Swaziland	\$0.69	\$0.69
Pakistan	\$0.32	\$0.32	Sweden	\$0.12	\$0.12
Palau	\$1.15	\$1.15	Switzerland	\$0.12	\$0.12
Palestinian Authority	\$0.33	\$0.33	Syrian Arab Republic	\$0.88	\$0.88
Panama	\$0.33	\$0.33	Taiwan	\$0.07	\$0.07
Papua New Guinea	\$0.95	\$0.95	Tajikistan	\$0.28	\$0.28
Paraguay	\$0.36	\$0.36	Tanzania	\$0.49	\$0.49
Peru	\$0.20	\$0.20	Thailand	\$0.24	\$0.24
Philippines	\$0.17	\$0.17	Togo	\$0.99	\$0.99
Poland	\$0.15	\$0.15	Tokelau	\$2.82	\$2.82
Portugal	\$0.12	\$0.12	Tonga Island	\$1.40	\$1.40
Qatar	\$0.86	\$0.86	Trinidad and Tobago	\$0.33	\$0.33
Reunion Island (includes	\$0.92	\$0.92		¢0.40	¢0.40
Mayotte Island)			Tunisia	\$0.49	\$0.49
Romania	\$0.32	\$0.32	Turkey	\$0.39	\$0.39
Russia	\$0.15	\$0.15	Turkmenistan	\$0.41	\$0.41
Rwanda, Republic of	\$1.06	\$1.06	Turks and Caicos Islands	\$0.49	\$0.49
San Marino	\$0.95	\$0.95	Tuvalu	\$1.63	\$1.63
Sao Tome	\$1.41	\$1.41	Uganda	\$0.49	\$0.49
Saudi Arabia	\$0.46	\$0.46	Ukraine	\$0.22	\$0.22
Senegal	\$0.68	\$0.68	United Arab Emirates	\$0.37	\$0.37
Serbia	\$0.41	\$0.41	United Kingdom	IN BLOCK	IN BLOCK
Seychelles Island	\$1.34	\$1.34	Uruguay	\$0.36	\$0.36
Sierra Leone	\$0.53	\$0.53	Uzbekistan	\$0.41	\$0.41
Singapore	\$0.13	\$0.13	Vanuatu	\$2.12	\$2.12
Slovakia	\$0.25	\$0.25	Vatican City	\$0.12	\$0.12
Slovenia	\$0.52	\$0.52	Venezuela	\$0.18	\$0.18
Solomon Island	\$1.07	\$1.07	Vietnam	\$0.54	\$0.54
Somalia Democratic Rep	\$1.46	\$1.46	Wallis Futuna	\$2.08	\$2.08
South Africa	\$0.35	\$0.35	Western Samoa	\$0.90	\$0.90
Spain	\$0.12	\$0.12	Yemen Arab Republic	\$0.87	\$0.87
Sri Lanka	\$0.94	\$0.94	Zaire	\$0.59	\$0.59
St. Helena	\$1.03	\$1.03	Zambia	\$0.59	\$0.59
St. Kitts/Nevis	\$0.46	\$0.46	Zimbabwe	\$0.49	\$0.49
St. Lucia	\$0.41	\$0.41			

#### SECTION 12 - GRANDFATHERED SERVICES

- 12.5 Type 4 Grandfathered Services (continued)
  - 12.5.30 AT&T Unlimited Nationwide Calling Basic Preferred Plan (Grandfathered 1/27/2009)
    - (A) The AT&T Unlimited Nationwide Calling Basic Preferred plan is a direct dialed outbound long distance service offered to single line or multi-line residential Customers on switched access lines. Service is offered on a flat rate basis, twenty-four hours per day, seven days per week. Customers must dial 1 plus the area code, if applicable, and the terminating number to complete a call utilizing this service. The AT&T Unlimited Nationwide Calling Basic Preferred plan provides the Customer with unlimited minutes of interexchange (both interstate and intrastate) long distance usage for a flat rate monthly charge (certain restrictions, as outlined below, do apply). This plan does not provide call detail information on the Customer's monthly bill. This service is not offered on an intraLATA or intrastate only basis.

Original Page 91

- (B) In order to be eligible for this plan, the Company must be able to verify that the Customer meets the following eligibility requirements. Customers who no longer meet these eligibility requirements will not be eligible for this plan and may be offered an alternative plan or have their service suspended, restricted or canceled without prior notice. Customers with one to three lines at a location are eligible for this service if they meet the following requirements:
  - Single line Customers must subscribe to BellSouth Long Distance and must also subscribe to and maintain one of the residential services as defined and offered by the Affiliated ILEC of the Company. Multi-line Customers must subscribe to BellSouth Long Distance and must also subscribe to and maintain one of the residential services as defined and offered by the Affiliated ILEC of the Company on each line under this plan. This plan may also be combined with one (1) or more products or services offered by the Affiliate of the Company as defined and offered by the Affiliate of the Company in conjunction with a subscription to this plan. Customers required to subscribe to one (1) or more products offered by an Affiliate of the Company must continue to subscribe to those products.

### SECTION 12 - GRANDFATHERED SERVICES

- 12.5 Type 4 Grandfathered Services (continued)
  - 12.5.30 AT&T Unlimited Nationwide Calling Basic Preferred Plan (continued)
    - (B) (continued)
      - .2 Customers must presubscribe to the Company for both intraLATA and interLATA long distance service on all lines subscribed to under this plan.
      - .3 Customer lines associated with educational institutions, (colleges, universities, etc.) are not eligible for this plan.

Original Page 92

- .4 This plan cannot be used for any use inconsistent with residential service.
- .5 This plan is not available for resale.
- .6 This plan is not available on an account that is the recipient of charges billed from another location.
- .7 Unlimited plan usage does not include multi-party conference calls (except those calls placed by using the Three-Way Calling feature are included), calls to 900, 976, 700 numbers and other calls to access information services, directory assistance, operator services, international calling and toll-free calling services.
- .8 If usage under this plan is not consistent with typical residential Customer usage, at the Company's sole discretion, the Company may offer the Customer an alternative plan or suspend, restrict or cancel the Customer's service without prior notice. Calls that are not consistent with typical residential voice usage include but are not limited to: use for general business purposes, commercial facsimile, auto-dialing, resale, call centers, long distance access to internet service providers and telemarketing.
- .9 The Company reserves the right to exclude certain terminating telephone numbers from this plan. Calls to these numbers will be billed at applicable MTS rates.
- (C) The MRC is \$26.99.

#### SECTION 12 - GRANDFATHERED SERVICES

- 12.5 Type 4 Grandfathered Services (continued)
  - 12.5.31 AT&T Unlimited Nationwide Calling Advantage 3 (Grandfathered 3/31/2012)
    - (A) AT&T Unlimited Nationwide Calling Advantage 3 is a bundled outbound only interstate and intrastate long distance usage calling plan that is offered to Residential Customers on switched access lines. Customers must dial 1 plus the area code, if applicable, and the called telephone number from their presubscribed telephone line. This plan provides the Customer with unlimited minutes of interexchange long distance usage for a single monthly recurring charge (certain terms and conditions, as described below, apply). This plan does not provide call detail information on the Customer's bill.

Original Page 93

- (B) In order to be eligible for this plan, the Company must be able to verify that the Customer meets the following terms and conditions. Customers who no longer meet these terms and conditions will not be eligible for this plan and may be offered an alternative plan or have their service suspended, restricted or canceled without prior notice.
  - .1 Customers must subscribe to and maintain the following services from the Company's Affiliated ILEC:
    - .a access line, and;
    - verticals feature package as defined and offered by the Company's Affiliated ILEC.
  - .2 Customers must subscribe to and maintain one (1) qualifying product or service from a qualified affiliate of the Company as defined and offered by the affiliate of the Company.
  - .3 Customers must presubscribe to the Company for the provision of interstate, intrastate IntraLATA and intrastate InterLATA service.
  - .4 This plan is only available for a maximum of three (3) lines at the same location.
  - .5 This plan is not available on an account that is the recipient of charges billed from another location.
  - .6 Unlimited plan usage does not include multi-party conference calls (except those calls placed by using Three-Way Calling feature are included), calls to 900, 976, 700 numbers or other calls to access information services, directory assistance, operator services, international calling and toll-free calling services.
  - .7 This plan cannot be used for any use inconsistent with typical residential voice service. If usage under this plan is not consistent with typical residential voice Customer usage, at the Company's sole discretion, the Company may offer the Customer an alternative plan or suspend, restrict or cancel the Customer's service without prior notice, and the Customer may not be eligible to resubscribe to this plan. Calls that are not consistent with typical residential voice usage include but are not limited to: use for general business purposes, commercial facsimile, auto-dialing, resale, call centers, long distance access to internet service providers and telemarketing.
  - .8 The monthly recurring charge is \$13.00.

#### SECTION 12 - GRANDFATHERED SERVICES

### 12.6 Type 5 – Grandfathered Services

- 12.6.1 Business Unlimited Plan<sup>/1/</sup> (Grandfathered 1/31/2012)
  - (A) The Business Unlimited plan is a direct dialed domestic outbound long distance voice service offered to business Customers with one to twelve switched access lines. Service is offered on a flat rate basis, twenty-four hours per day, seven days per week. Customers must dial 1 plus the area code, if applicable, and the terminating number to complete a call utilizing this service. The Business Unlimited plan provides the Customer with unlimited minutes of interexchange (both interstate and intrastate) long distance usage for a flat rate monthly charge (certain restrictions, as outlined below, do apply). This plan does not provide call detail information on the Customer's monthly bill for usage eligible for unlimited calling but does provide call detail information for usage not included in the unlimited calling. Customers may choose to receive this call detail information for an additional monthly recurring charge. This service is not offered on an intraLATA only basis.

Original Page 94

- (B) Toll-free service is available with this plan for Customers who also subscribe to the AT&T Long Distance Toll Free<sup>SM</sup> Service. The provisions and rates and charges of the AT&T Long Distance Toll Free<sup>SM</sup> Service will apply as described in Section 4.4 of this pricing and service guide except that usage rates are specified below. Toll-free usage is not included in the unlimited usage.
- (C) In order to be eligible for this plan, the Company must be able to verify that the Customer meets the following eligibility requirements. If the Company determines that the Customer's use of service violates any of these restrictions the Company, at its sole option, may move the Customer to another plan or suspend, restrict or cancel the Customer's service without prior notice. Customers are eligible for this service if they meet the following requirements:
  - .1 Customers must be new or existing subscribers to the Company and must also be new or existing subscribers to the Company's affiliated incumbent local exchange carrier. Customers must subscribe to Complete Choice® For Business plan offered by the Company's Affiliated ILEC on each line at each location under this plan. The Customer must also receive billing for each of these services from the Company's Affiliated ILEC.
  - .2 At the time of subscription to this plan, Customers must subscribe to a local service term agreement with the Company's Affiliated ILEC.
  - .3 This plan is available to business Customers with one to twelve business lines at a single location and on a single billing account. All lines at the location must subscribe to this plan.
- /1/ Effective January 31, 2012, the Business Unlimited Plan is no longer available to new Customers or existing Customers who move locations or transfer service to another location. Customers may add or remove lines in accordance with the plan terms and conditions as long as the plan service remains at the same Customer Premises.

#### SECTION 12 - GRANDFATHERED SERVICES

### 12.6 Type 5 – Grandfathered Services (continued)

### 12.6.1 Business Unlimited Plan (continued)

(D) Unlimited plan usage does not include multi-party conference calls (except those calls placed by using the Three-Way Calling feature are included), calls to 900, 976, 700 numbers and other calls to access information services, directory assistance, calling card, operator services, international calling and toll-free calling services.

### (E) Restrictions

.1 Unlimited plan usage does not include multi-party conference calls (except those calls

Original Page 95

- .2 This plan is not available for resale.
- .3 This plan is not available to Customers with an account that bills to another number or is the recipient of charges billed from another number.
- .4 The Company reserves the right to exclude certain terminating telephone numbers from this plan. Calls to these numbers will be billed at applicable MTS rates.
- .5 Service under this plan cannot be used with the following applications or services: call center environment or in connection with any similar such application, auto-dialers or any similar type of device, PBX or PBX-like equipment, data transmission equipment, Centrex service, foreign exchange service, public telephone services, ISDN services, broadcast facsimile services or the equivalents of any such services.
- .6 Customers on this plan may only place as many concurrent calls as they have individual lines subscribed to under this plan.
- .7 Term discounts are not applicable to this service.
- .8 Customer lines associated with education institutions (colleges, Universities, etc.) or other businesses that aggregate end user traffic are not eligible for this plan.

BUSINESS AND RESIDENTIAL SERVICES

Original Page 96

Effective: September 1, 2019

# PRICING AND SERVICE GUIDE SECTION 12 - GRANDFATHERED SERVICES

## 12.6 Type 5 – Grandfathered Services (continued)

## 12.6.1 Business Unlimited Plan (continued)

## (F) Rates and Charges

### .1 Direct Dialed Outbound Service MRC

<b>Number of Lines</b>	MRC
1	\$86.00
2	\$157.00
3	\$228.00
4	\$299.00
5	\$370.00
6	\$441.00
7	\$512.00
8	\$583.00
9	\$654.00
10	\$725.00
11	\$796.00
12	\$867.00

### .2 Toll-Free Service

The per-minute usage rate is \$0.0600. All toll-free inbound calls are billed in six (6) second increments after an initial period, for billing purposes, of thirty (30) seconds. See Section 4.4 of this pricing and service guide for the toll-free service MRC.

### .3 Call Detail MRC

<b>Number of Lines</b>	MRC
1	\$5.00
2	\$10.00
3	\$15.00
4	\$20.00
5	\$25.00
6	\$30.00
7	\$35.00
8	\$40.00
9	\$45.00
10	\$50.00
11	\$55.00
12	\$60.00

#### SECTION 12 - GRANDFATHERED SERVICES

- 12.6 Type 5 Grandfathered Services (continued)
  - 12.6.2 AT&T Business Unlimited Calling<sup>SM</sup> IV (Grandfathered 8/12/2015)
    - (A) AT&T Business Unlimited Calling SM IV is an outbound calling and optional inbound Switched TFS long distance calling plan. This plan is established at the BTN level; multiple BTN aggregation is not available with this plan. If the Customer or Applicant selects a different long distance calling plan for specific WTNs, the Customer or Applicant is required to establish a separate BTN for each plan. This plan is available to Business Customers in Alabama, Florida, Georgia, Kentucky, Louisiana, Mississippi, North Carolina, South Carolina and Tennessee who:

### WIN/WINBACK

- .1 Option 1 One (1) to four (4) access lines
  - utilize Switched Access to reach the long distance network for outbound calling and to receive calls from the long distance network for TFS;

Original Page 97

- .b subscribe to the Company for interLATA (interstate and intrastate) and intraLATA (local toll) Service for outbound long distance calling on each access line, and may also subscribe to the Company's TFS for inbound long distance calling;
- .c meet a minimum of one (1) of the following conditions:
  - i have previously subscribed to local dial tone Service of an Affiliated ILEC of the Company (including where that Affiliate is operating as a CLEC) and have cancelled that Service; or,
  - .ii are currently a local telephone customer of a competitor in the local serving territory of one of the Affiliated ILECs of the Company (including where that Affiliated is operating as a CLEC) and are now moving dial tone services from the competitor to the Affiliated ILEC;
- .d newly subscribe to a 1-Year term agreement to AT&T Business Local Calling Assurance SM from an Affiliated ILEC of the Company (including where that Affiliate is operating as a CLEC) on at least one (1), but no more than four (4) business access lines (*i.e.*, WTNs) on a single BTN;
- currently subscribe to one of the following wireless Services from an Affiliate of the Company: a) Wireless Voice; (b) Wireless Data; (c) Wireless Voice and Data; or (d) Wireless Laptop card; or currently subscribe to one of the following services from an Affiliate of the Company for a 1-Year term: (a) AT&T U-Verse High Speed Internet-Business Edition; (b) FastAccess® Business DSL; (c) FastAccess® Business DSL Lite; (d) FastAccess® Business DSL Plus; or (e) FastAccess® Business DSL 6.0; or

#### SECTION 12 - GRANDFATHERED SERVICES

- 12.6 Type 5 Grandfathered Services (continued)
  - 12.6.1 AT&T Business Unlimited Calling<sup>SM</sup> IV (continued)
    - (A) This plan is available to Business Customers in Alabama, Florida, Georgia, Kentucky, Louisiana, Mississippi, North Carolina, South Carolina and Tennessee who: (continued)

#### WIN/WINBACK (continued)

- .1 Option 1 One (1) to Four (4) access lines (continued)
  - .f newly subscribe to one of the following services from an Affiliate of the Company for a minimum 1- Year term: (a) AT&T U-Verse High Speed Internet-Business Edition; (b) FastAccess® Business DSL; (c) FastAccess® Business DSL Lite; (d) FastAccess® Business DSL Plus; or (e) FastAccess® Business DSL 6.0 or newly subscribe (for a minimum 2-Year term) to one of the following wireless services from an affiliate of the Company: (a wireless Voice; (b) Wireless Data; (c) Wireless Voice and Data; or (d) Wireless Laptop card;

Original Page 98

- .g agree the Company may combine the Affiliate of the Company's wireless billing with the Company's wireline billing except when participating in the Affiliate of the Company's SBS Wireless Deposit Waiver Program; and,
- .h commit to subscribe to this plan for a one (1) year term.
- .2 Option 2 One (1) to Ten (10) access lines
  - utilize Switched Access to reach the long distance network for outbound calling and to receive calls from the long distance network for TFS;
  - .b subscribe to the Company for interLATA (interstate and intrastate) and intraLATA (local toll) Service for outbound long distance calling on each access line, and may also subscribe to the Company's TFS for inbound long distance calling;
  - .c meet a minimum of one (1) of the following conditions:
    - have previously subscribed to local dial tone Service of an Affiliates ILEC of the Company (including where that Affiliate is operating as a CLEC) and have cancelled that Service; or,
    - are currently a local telephone customer of a competitor in the local serving territory of one of the Affiliates ILECs of the Company (including where that Affiliated is operating as a CLEC) and are now moving dial tone services from the competitor to the Affiliated ILEC;
  - .d newly subscribe to a 1-Year term agreement to AT&T Business Local Calling SM or Complete Choice® for Business from an Affiliate ILEC of the Company (including where that Affiliate is operating as a CLEC) on at least one (1), but no more than ten (10) business access lines (*i.e.*, WTNs) on a single BTN;

#### SECTION 12 - GRANDFATHERED SERVICES

- 12.6 Type 5 Grandfathered Services (continued)
  - 12.6.2 AT&T Business Unlimited Calling<sup>SM</sup> IV (continued)
    - (A) This plan is available to Business Customers in Alabama, Florida, Georgia, Kentucky, Louisiana, Mississippi, North Carolina, South Carolina and Tennessee who: (continued)

#### WIN/WINBACK (continued)

- .2 Option 2 One (1) to ten (10) access lines (continued)
  - .e currently subscribe to one of the following wireless Services from an Affiliate of the Company: a) Wireless Voice; (b) Wireless Data; (c) Wireless Voice and Data; or (d) Wireless Laptop card; or currently subscribe to one of the following services from an Affiliate of the Company for a 1-Year term: (a) AT&T U-Verse High Speed Internet-Business Edition; (b) FastAccess® Business DSL; (c) FastAccess® Business DSL Lite; (d) FastAccess® Business DSL Plus; or (e) FastAccess® Business DSL 6.0; or,

Original Page 99

- .f newly subscribe to one of the following services from an Affiliate of the Company for a minimum 1-Year term: (a) AT&T U-Verse High Speed Internet-Business Edition; (b) FastAccess® Business DSL; (c) FastAccess® Business DSL Lite; (d) FastAccess® Business DSL Plus; or (e) FastAccess® Business DSL 6.0 or newly subscribe (for a minimum 2-Year term) to one of the following wireless services from an affiliate of the Company: (a wireless Voice; (b) Wireless Data; (c) Wireless Voice and Data; or (d) Wireless Laptop card;
- g agree the Company may combine the Affiliate of the Company's wireless billing with the Company's wireline billing except when participating in the Affiliate of the Company's SBS Wireless Deposit Waiver Program; and,
- .h commit to subscribe to this plan for a one (1) year term.

#### SECTION 12 - GRANDFATHERED SERVICES

- 12.6 Type 5 Grandfathered Services (continued)
  - 12.6.2 AT&T Business Unlimited Calling<sup>SM</sup> IV (continued)
    - (A) This plan is available to Business Customers in Alabama, Florida, Georgia, Kentucky, Louisiana, Mississippi, North Carolina, South Carolina and Tennessee who: (continued)

#### RETENTION

- .1 Option 1 One (1) to four (4) access lines
  - utilize Switched Access to reach the long distance network for outbound calling and to receive calls from the long distance network for TFS:

Original Page 100

- .b subscribe to the Company for interLATA (interstate and intrastate) and intraLATA (local toll) Service for outbound long distance calling on each access line, and may also subscribe to the Company's TFS for inbound long distance calling;
- .c currently subscribe to local dial tone service from an Affiliated ILEC of the Company (including where that Affiliate is operating as a CLEC) and state an intention to change local carriers for economic reasons and/or in response to a competitive offer;
- .d newly subscribe to or restart a 1-Year term agreement to AT&T Business Local Calling Assurance SM from an Affiliated ILEC of the Company (including where that Affiliate is operating as a CLEC) on at least one (1), but no more than four (4) business access lines (i.e., WTNs) on a single BTN; or
- .e currently subscribe to one of the following wireless Services from an Affiliate of the Company: a) Wireless Voice; (b) Wireless Data; (c) Wireless Voice and Data; or (d) Wireless Laptop card; or currently subscribe to one of the following services from an Affiliate of the Company for a 1-Year term: (a) AT&T U-Verse High Speed Internet Business Edition; (b) FastAccess® Business DSL; (c) FastAccess® Business DSL Lite; (d) FastAccess® Business DSL Plus; or (e) FastAccess® Business DSL 6.0;
- .f newly subscribe to one of the following services from an Affiliate of the Company for minimum 1-Year term: (a) AT&T U-Verse High Speed Internet-Business Edition; (b) FastAccess® Business DSL; (c) FastAccess® Business DSL Lite; (d) FastAccess® Business DSL Plus; or (e) FastAccess® Business DSL 6.0 or newly subscribe for a minimum 2-Year term to one of the following wireless services from an affiliate of the Company: (a) wireless Voice; (b) Wireless Data; (c) Wireless Voice and Data; or (d) Wireless Laptop card;
- .g agree the Company may combine the Affiliate of the Company's wireless billing with the Company's wireline billing except when participating in the Affiliate of the Company's SBS Wireless Deposit Waiver Program; and,
- .h commit to subscribe to this plan for a one (1) year term.

#### SECTION 12 - GRANDFATHERED SERVICES

- 12.6 Type 5 Grandfathered Services (continued)
  - 12.6.2 AT&T Business Unlimited Calling<sup>SM</sup> IV (continued)
    - (A) This plan is available to Business Customers in Alabama, Florida, Georgia, Kentucky, Louisiana, Mississippi, North Carolina, South Carolina and Tennessee who: (continued)

#### **RETENTION** (continued)

- .2 Option 2 Five (5) to ten (10) access lines
  - utilize Switched Access to reach the long distance network for outbound calling and to receive calls from the long distance network for TFS:

Original Page 101

- .b subscribe to the Company for interLATA (interstate and intrastate) and intraLATA (local toll) Service for outbound long distance calling on each access line, and may also subscribe to the Company's TFS for inbound long distance calling;
- .c currently subscribe to local dial tone service from an Affiliated ILEC of the Company (including where that Affiliate is operating as a CLEC) and state an intention to change local carriers for economic reasons and/or in response to a competitive offer;
- .d newly subscribe to or restart a 1-Year term agreement to AT&T Business Local Calling<sup>SM</sup> or AT&T Complete Choice<sup>®</sup> for Business from an Affiliated ILEC of the Company (including where that Affiliate is operating as a CLEC) on at least five (5), but no more than ten (10) business access lines (*i.e.*, WTNs) on a single BTN; or,
- .e currently subscribe to one of the following wireless Services from an Affiliate of the Company: a) Wireless Voice; (b) Wireless Data; (c) Wireless Voice and Data; or (d) Wireless Laptop card; or currently subscribe to one of the following services from an Affiliate of the Company for a 1-Year term: (a) AT&T U-Verse High Speed Internet-Business Edition; (b) FastAccess® Business DSL; (c) FastAccess® Business DSL Lite; (d) FastAccess® Business DSL Plus; or (e) FastAccess® Business DSL 6.0; or,
- .f newly subscribe to one of the following services from an Affiliate of the Company for a minimum 1-Year term: (a) AT&T U-Verse High Speed Internet-Business Edition; (b) FastAccess® Business DSL; (c) FastAccess® Business DSL Lite; (d) FastAccess® Business DSL Plus; or (e) FastAccess® Business DSL 6.0 or newly subscribe (for a minimum 2-Year term) to one of the following wireless services from an affiliate of the Company: (a wireless Voice; (b) Wireless Data; (c) Wireless Voice and Data; or (d) Wireless Laptop card;
- .g agree the Company may combine the Affiliate of the Company's wireless billing with the Company's wireline billing except when participating in the Affiliate of the Company's SBS Wireless Deposit Waiver Program; and,
- .h commit to subscribe to this plan for a one (1) year term.

### **SECTION 12 - GRANDFATHERED SERVICES**

- 12.6 Type 5 Grandfathered Services (continued)
  - 12.6.2 AT&T Business Unlimited Calling<sup>SM</sup> IV (continued)
    - (A) This plan is available to Business Customers in Alabama, Florida, Georgia, Kentucky, Louisiana, Mississippi, North Carolina, South Carolina and Tennessee who: (continued)
    - (B) Term Agreement
      - .1 Customers must subscribe to an AT&T All for Less Subscriber 1-Year (12 Month) Agreement ("Term Agreement").
      - .2 Customers who remain on this plan at the expiration of the 1-Year term will be billed on a month-to-month basis at the rates defined below in Section 12.6.2 (I).

Original Page 102

Effective: September 1, 2019

.3 If the Company no longer offers this plan, and if the Customer moves to a new location or changes the number of access lines under the BTN(s) to which this plan applies then the Customer's Service under the Plan will terminate, and Customer will not be subject termination charges.

#### SECTION 12 - GRANDFATHERED SERVICES

### 12.6 Type 5 – Grandfathered Services (continued)

### 12.6.2 AT&T Business Unlimited Calling<sup>SM</sup> IV (continued)

(C) A single legal business entity with more than one BTN at that legal business entity's physical service location is eligible for this plan on one (1) BTN only, and only if the total number of business access lines does not exceed ten (10) at that location. Additionally, service under this plan is limited to one BTN with one (1) to ten (10) business access lines per legal business entity.

Original Page 103

Effective: September 1, 2019

- (D) The Customer may subscribe to this plan solely for outbound long distance calling, or for both outbound long distance calling and inbound toll free calling. Customers subscribing to the plan receive unlimited domestic 1+ direct-dialed outbound calling. Where a Customer subscribes to TFS, inbound toll-free calls may originate on any type of access and are terminated via Switched Access to the Customer's location. See Section 3 and Section 4.4 of this pricing and service guide for optional features, rules and regulations, and general information regarding TFS.
- (E) TFS calls, calls to 900. 976, 700 numbers or other calls to access information services, directory assistance, operator services and international calling are not included in the unlimited minutes of use. Toll-free calls are billed in increments of six (6) seconds subject to a minimum connect time (initial period) of thirty (30) seconds. TFS calls per minute rates associated with this plan are subject to change with prior notification to the customer.

### (F) Restrictions

This plan is provided for standard voice calling involving live dialog between individuals. This plan is not intended for use as a substitute for dedicated or open circuits, or similar applications. This plan may not be used for data connections, including but not limited to, modem to modem calls, remote access applications, Internet access, or Intranet access (including access to corporate LANs). Additionally, this plan may not be used for dedicated point-to-point connections between equipment that leaves a circuit connected without a contemporary, continuous voice communication. The plan may not be used for auto dialers; PBX trunks; ground start line or trunks; ISDN service, including PRI; foreign exchange services; Remote Call Forwarding/Telebranch Service; public telephone service; public access smart-pay phones; analog to digital conversion digital PBX service, including local access provisioned via T-1 facilities; WATS service; PBX/PABX/EABX services; nonsquare electronic key telephone systems; hybrid key telephone systems; predictive calling/dialing systems; automatic outbound dialing systems; any type of automatic call distribution system; or the functional equivalent of any such systems listed above. This plan may be used for fax transmissions, excluding broadcast fax applications (which are prohibited). Where customer has more than 10 calls to any single number of a duration of more than 3 hours each during a billing cycle or any single call of greater than 10 hours of duration, Customer will be presumed to be in violation of these restrictions. In such case, Customer may be asked to provide reasonable proof to the Company that it is not using the Service for a prohibited purpose. Failure to provide such proof to the Company shall be treated as a violation of the terms and conditions of this plan.

#### SECTION 12 - GRANDFATHERED SERVICES

### 12.6 Type 5 – Grandfathered Services (continued)

## 12.6.2 AT&T Business Unlimited Calling<sup>SM</sup> IV (continued)

### (G) Call Detail

Call Detail is available as an optional feature to this Plan. For an additional MRC Customers who select this optional feature will receive the itemized call detail of their zero-rated calls. Activation and deactivation of the feature will begin within the next billing cycle following Customer's request. Customers may add or remove the Call Detail optional feature from this Plan at no charge.

Original Page 104

Effective: September 1, 2019

## (H) Early Termination Fee (ETF)

- .1 If prior to the expiration date of the Term Agreement, the Customer chooses to either: (a) terminate long distance Service with the Company; or (b) change its calling Plan from AT&T Business Unlimited Calling M IV prior to the expiration of the Term Agreement, the Customer may be charged an Early Termination Fee ("ETF"). The ETF shall be \$2.50 multiplied by the number of months remaining in the term. Customer may change its calling Plan to the AT&T Business Block of Time II Plan (rate option 700 minutes or greater) as specified in Section 4 of this pricing and service guide; in such case ETF shall not apply.
- .2 The Company will not charge an ETF if Customer cancels this Plan if at the same time the Customer agrees to replace some or all their existing service with Internet Protocol (IP) service, Wireless, or any functionally equivalent service from an Affiliate of the Company for the purpose of placing outbound and/or inbound live voice communications outside of the Customer's local calling area.

### (I) Rates and Charges:

(A) The outbound long distance MRC is as follows:

1-Year Term: \$10.00 per access line Month-to-Month: \$10.00 per access line

Customers will receive a monthly bill credit reward amount equal to \$5.00 per line for the duration of the 1-Year Term Agreement. During the 1-Year Term, Federal Universal Service Fee, Federal Regulatory Fee, taxes and other similar fees and surcharges will be calculated on the MRC prior to the calculation of the monthly bill credit reward amount.

- (B) The AT&T Long Distance Toll Free<sup>SM</sup> Service per-minute usage rate is \$0.055. For the AT&T Long Distance Toll Free<sup>SM</sup> Service MRC, see Section 4.4 of this pricing and service guide.
- (C) Call Detail

The MRC is \$5.00/line up to ten (10) lines.